

Bio-Based Plant Oil Polymers and Composites

Edited by: Samy A. Madbouly Chaoqun Zhang Michael R. Kessler



BIO-BASED PLANT OIL POLYMERS AND COMPOSITES



PLASTICS DESIGN LIBRARY (PDL) PDL HANDBOOK SERIES Series Editor: Sina Ebnesajjad, PhD (<u>sina@FluoroConsultants.com</u>) President, FluoroConsultants Group, LLC Chadds Ford, PA, USA www.FluoroConsultants.com

The **PDL Handbook Series** is aimed at a wide range of engineers and other professionals working in the plastics industry, and related sectors using plastics and adhesives.

PDL is a series of data books, reference works and practical guides covering plastics engineering, applications, processing, and manufacturing, and applied aspects of polymer science, elastomers and adhesives.

Recent titles in the series

Biopolymers: Processing and Products, Michael Niaounakis (ISBN: 9780323266987) Biopolymers: Reuse, Recycling, and Disposal, Michael Niaounakis (ISBN: 9781455731459) Carbon Nanotube Reinforced Composites, Marcio Loos (ISBN: 9781455731954) Extrusion, 2e, John Wagner & Eldridge Mount (ISBN: 9781437734812) Fluoroplastics, Volume 1, 2e, Sina Ebnesajjad (ISBN: 9781455731992) Handbook of Biopolymers and Biodegradable Plastics, Sina Ebnesajjad (ISBN: 9781455728343) Handbook of Molded Part Shrinkage and Warpage, Jerry Fischer (ISBN: 9781455725977) Handbook of Polymer Applications in Medicine and Medical Devices, Kayvon Modjarrad & Sina Ebnesaiiad (ISBN: 9780323228053) Handbook of Thermoplastic Elastomers, Jiri G Drobny (ISBN: 9780323221368) Handbook of Thermoset Plastics, 2e, Hanna Dodiuk & Sidney Goodman (ISBN: 9781455731077) High Performance Polymers, 2e, Johannes Karl Fink (ISBN: 9780323312226) Introduction to Fluoropolymers, Sina Ebnesajjad (ISBN: 9781455774425) Ionizing Radiation and Polymers, Jiri G Drobny (ISBN: 9781455778812) Manufacturing Flexible Packaging, Thomas Dunn (ISBN: 9780323264365) Plastic Films in Food Packaging, Sina Ebnesajjad (ISBN: 9781455731121) Plastics in Medical Devices, 2e, Vinny Sastri (ISBN: 9781455732012) Polylactic Acid, Rahmat et al. (ISBN: 9781437744590) Polyvinyl Fluoride, Sina Ebnesajjad (ISBN: 9781455778850) Reactive Polymers, 2e, Johannes Karl Fink (ISBN: 9781455731497) The Effect of Creep and Other Time Related Factors on Plastics and Elastomers, 3e, Laurence McKeen (ISBN: 9780323353137) The Effect of Long Term Thermal Exposure on Plastics and Elastomers, Laurence McKeen (ISBN: 9780323221085) The Effect of Sterilization on Plastics and Elastomers, 3e, Laurence McKeen (ISBN: 9781455725984) The Effect of Temperature and Other Factors on Plastics and Elastomers, 3e, Laurence McKeen (ISBN: 9780323310161) The Effect of UV Light and Weather on Plastics and Elastomers, 3e, Laurence McKeen (ISBN: 9781455728510) Thermoforming of Single and Multilayer Laminates, Ali Ashter (ISBN: 9781455731725) Thermoplastics and Thermoplastic Composites, 2e, Michel Biron (ISBN: 9781455778980) Thermosets and Composites, 2e, Michel Biron (ISBN: 9781455731244)

To submit a new book proposal for the series, or place an order, please contact David Jackson, Acquisitions Editor

david.jackson@elsevier.com

BIO-BASED PLANT OIL POLYMERS AND COMPOSITES

Samy A. Madbouly Chaoqun Zhang Michael R. Kessler







William Andrew is an imprint of Elsevier The Boulevard, Langford Lane, Kidlington, Oxford, OX5 1GB, UK 225 Wyman Street, Waltham, MA 02451, USA

Copyright © 2016 Elsevier Inc. All rights reserved.

No part of this publication may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or any information storage and retrieval system, without permission in writing from the publisher. Details on how to seek permission, further information about the Publisher's permissions policies and our arrangements with organizations such as the Copyright Clearance Center and the Copyright Licensing Agency, can be found at our website: www.elsevier.com/permissions.

This book and the individual contributions contained in it are protected under copyright by the Publisher (other than as may be noted herein).

Notices

Knowledge and best practice in this field are constantly changing. As new research and experience broaden our understanding, changes in research methods, professional practices, or medical treatment may become necessary.

Practitioners and researchers must always rely on their own experience and knowledge in evaluating and using any information, methods, compounds, or experiments described herein. In using such information or methods they should be mindful of their own safety and the safety of others, including parties for whom they have a professional responsibility.

To the fullest extent of the law, neither the Publisher nor the authors, contributors, or editors, assume any liability for any injury and/or damage to persons or property as a matter of products liability, negligence or otherwise, or from any use or operation of any methods, products, instructions, or ideas contained in the material herein.

British Library Cataloguing-in-Publication Data

A catalogue record for this book is available from the British Library

Library of Congress Cataloging-in-Publication Data

A catalog record for this book is available from the Library of Congress

ISBN: 978-0-323-35833-0

For information on all William Andrew publications visit our website at http://store.elsevier.com/



Publisher: Matthew Deans Acquisition Editor: David Jackson Editorial Project Manager: Peter Gane Production Project Manager: Nicky Carter Designer: Maria Ines Cruz

Typeset by Thomson Digital Printed and bound in USA

Contents

Co Pro	ntrib eface.	utors	ix xi
1	Int Jia	t roduction to Plant Oils hao Chen, Stephanie Ovola-Revnoso, and Martin Thuo	1
	1 1	Introduction	1
	1.1	Chemistry and Biochemistry of Plant Oils	3
	1.2	1.2.1 Biosynthesis of Ternenoids	3
		1.2.2 Biosynthesis of Fatty Acids	6
		1.2.3 Chemical Reactivity of Plant Oils.	6
		1.2.4 Chemical Characterization	10
	1.3	Major Sources of Plant Oils	11
	1.4	Extraction and Processing of Plant Oils	11
		1.4.1 Small-Scale Extraction and Processing	11
		1.4.2 Large-Scale Extraction and Processing	12
		1.4.2.1 Palm Oil	12
		1.4.3 Postextraction Processing of Fatty Acids	13
	1.5	Outlook	13
	Ref	ferences	
2	Pla	ant Oil-Based Derivatives	
	Cha	aoqun Zhang and Samy A. Madbouly	
	2.1	Introduction	
	2.2	Plant Oil-Based Derivatives	
		2.2.1 Fatty Acids	21
		2.2.2 Fatty Amides/Nitriles/Amines	
		2.2.3 Alcohols	
		2.2.4 Ester Derivatives	25
		2.2.5 Epoxy Derivatives	
		2.2.6 Conjugates	
		2.2.7 Other Derivatives	
	2.3	Conclusions	
	Ref	ferences	
3	Pla	ant Oil-Based Polyurethanes	
	The	omas F. Garrison and Michael R. Kessler	
	3.1	Polyurethane Chemistry	
	3.2	Plant Oil-Based Polyurethanes	
	3.3	Developing New Sources of Vegetable Oils	40
	3.4	Polyol Methods	
		3.4.1 Epoxidation/Ring-Opening	44
		3.4.2 Ozonolysis	44
		3.4.3 Amidation	44
	3.5	Flame Retardant Polyols	45
	3.6	Alternative Routes for Making Polyurethanes	45

	3.7 Applications of Plant Oil-Based Polyurethanes	
	3.8 Foams	
	3.8.1 Flexible Foams	
	3.8.2 Rigid Foams	
	3.9 Coatings	
	3.9.1 Waterborne Dispersions	
	3.9.2 Antimicrobial Coatings	
	3.10 Shape Memory Polymers	
	3.11 Conclusions	
	References	
4	4 Plant Oil-Based Polyhydroxyurethanes	
	Ashley Johns, Luke Gibbons, Madeline Smith, Kyle Edwards, and Rafa	el L. Quirino
	4.1 Introduction	
	4.2 Petroleum-Based Polyurethanes	
	4.3 Polyurethanes from Bio-Based Polyols	
	4.4 Polyhydroxyurethanes (PHUs)	
	4.4.1 Introduction to PHUs	
	4.4.2 Petroleum-Based PHUs	
	4.4.3 Bio-Based PHUs	61
	4.4.3.1 Triglycerides	61
	4.4.3.2 Glycerol	
	4.4.3.3 Terpenes	
	4.4.3.4 Hybrid PHUs and Composites	
	4.5 Alternative Systems	
	4.6 Conclusions	
	Keterences	
5	5 Plant Oil-Based Polvester	
	Larissa R. Fonseca, José L. Silva Sá, and Benedito S. Lima-Neto	
	5.1 Introduction	73
	5.2 Processes and Monomers	74
	5.3 Thermoplastic Polvesters	
	5.4 Biodegradable Polyesters	
	5.5 Unsaturated Polyester Resin (UPR)	
	5.6 Other Applications	
	5.7 Applications of Plant Oil-Based Polyester as an Alternative for Pet	troleum-Based Polyester
	References	
_		07
0	6 Plant Oll-Based Polyether	
	1 Declare ed	07
	6.1 Background	
	0.1.1 Introduction	
	6.1.2 Motheda	
	0.2 Michilous	89
	6.2.2 Durifications ofter Extractions	۷۶ مم
	6.2.3 Modification from Oil to Polyather Monomers	48 ۱۵
	6.2.4 Polymerization from Oil Source to Polyether	
	0.2.4 rorymenzation from On Source to roryether	

	6.3 Properties of Plant Oil-Based Polyethers	91
	6.4 Applications	93
	6.5 Challenges and Trends	94
	References	95
7	Plant Oil-Based Epoxy Intermediates for Polymers	
	7.1 Introduction	
	7.2 Epoxidation of Plant Oils	
	7.3 Derivatives of Epoxidized Plant Oils	
	7.3.1 Epoxy Ring Opening by Alcohols	
	7.3.2 Epoxy Ring Opening by Water	
	7.3.3 Amine-to-Epoxide Reactions	
	7.3.4 Epoxy Ring Opening with Halogen Reagents	
	7.3.5 Epoxy King Opening by Lactic Acid	108
	7.3.6 Hydrogenation of Epoxidized Oils	
	7.3.7 Acrylated Epoxidized Plant Oils	
	7.3.8 Carbonated Plant Oil and its Derivatives	
	7.3.9 Epoxy Ring-Opening Reactions by Anhydrides	110 111
	7.3.10 Epoxy King-Opening Polymerization	111
	7.4. Dichard Even Decim	112
	7.4 Biobased Epoxy Resins	113 114
	7.4.1 Annyande-Curea Epoxiaized Ons	114
	7.4.2 Resins of Activitied-Epoxidized Plant Oils	114 116
	7.5. Conclusions	110 117
	Pafarancas	/ 110 110
	Further Reading	110
	Turtier Reading	123
8	Enzyme-Assisted Synthesis of Plant Oil-Based Polymers	127
	Yi-Shu Tai and Kechun Zhang	
	8.1 Introduction	
	8.2 Enzyme-Assisted Synthesis of Plant Oil-Based Polyesters	
	8.2.1 Lipase-Catalyzed Synthesis of Functional Polyesters	
	8.2.1.1 Biodegradable and Cross-Linkable Polyesters	
	8.2.1.2 Epoxide-Containing Polyesters	
	8.2.1.3 Oleic Acid-Based Polyesters	
	8.2.2 Lipase-Catalyzed Polycondensation of Unsaturated or Epoxidized	120
	α, ω -Dicarboxylic Acids and Diols	
	8.2.3 Lipase-Catalyzed Polycondensation of Hydroxy Fatty Acids and Derivatives	
	8.2.3.1 Poly(Ricinoleaic Acid) and its Derivatives	133
	8.2.3.2 Poly(Ambrettolide Epoxide) and Poly(Alcuritic Acid)	
	8.2.5.5 Polyesters from the Monomer <i>cis</i> -9,10-Epoxy-18-Hydroxyoctadecanoic Acid	138
	8.5 Enzyme-Assisted Plant Ull Transformations.	
	0.5.1 Sugar Fally Acid Esters	138
	6.5.2 Iransesternication	
	0.3.5 Elizyille-Catalyzed Epoxidation.	141 142
	0.5.4 Faity Actus Hyutoxytation.	143 144
	0.4 CONCIUSION	144 115 1
		14)

9	Plant Oil-Based Nanocomposites	
	Samy A. Madbouly, Chaoqun Zhang, and Jing Zhang	
	9.1 Introduction	
	9.2 Plant Oil-Based/Natural Filler Composites	
	9.3 Plant Oil-Based/Carbon Nanotubes Composites	
	9.4 Plant Oil-Based PU/Graphene Oxide Composites	
	9.4.1 Surface Modification of Graphene Oxide	
	9.4.2 Synthesis and Characterization of Bio-Based PU/GO Nanocomposites	
	9.5 Plant Oil-Based/Clay and Silica Nanocomposites	
	References	
10	Fiber Reinforced Plant Oil-Based Composites	
	Kunwei Liu, Chaoqun Zhang, and Samy A. Madbouly	
	10.1 Introduction	167
	10.2 Fiber-Reinforced Plant Oil-Based Vinyl Polymer Composites	168
	10.3 Fiber-Reinforced Plant Oil-Based Epoxy Composites	176
	10.4 Fiber-Reinforced Plant Oil-Based Polyurethane Composites	
	10.5 Fiber-Reinforced Plant Oil-Based Form Composites	184
	10.6 Conclusion	186
	References	
11	Application of Plant Oil-Based Products in Structural Health Monitoring.	
		101
	11.1 Introduction	
	11.2 Background	
	11.2.1 Materials	
	11.2.2 Sensor Fabrication	
	11.2.3 Sensing Principle	
	11.5 Materials Properties	
	11.2.2 Dispersion of Nanoffler	
	11.3.2 Dielectric Properties	
	11.5.5 Inermogravimetric Analysis (IGA)	
	11.4 Laboratory Experiments	
	11.4.1 Experimental Setup	
	11.4.2 Results and Discussion	
	References	
12	The Future of Essential Oils as a Pest Biocontrol Method	
	Onvier Sparagano, James I Michard, and David George	
	12.1 Introduction	
	12.2 Range of Use	
	12.3 Mechanisms	
	12.4 Future Research	
	12.5 Conclusion	
	Keterences	
	Further Keading	
Ind	lex	

Jiahao Chen

Department of Materials Science and Engineering; Micro-Electronic Research Center, Iowa State University, Ames, IA, USA

Eric W. Cochran

Department of Chemical and Biological Engineering, Iowa State University, Ames, IA, USA

Jonathan M. Curtis

Lipid Chemistry Group (LCG), Department of Agricultural, Food, and Nutritional Science (AFNS), University of Alberta, Edmonton, AB, Canada

Kyle Edwards

Chemistry Department, Georgia Southern University, Statesboro, GA, USA

Larissa R. Fonseca

Instituto de Química de São Carlos, Universidade de São Paulo, SP, Brazil

Michael J. Forrester

Department of Chemical and Biological Engineering, Iowa State University, Ames, IA, USA

Thomas F. Garrison

Department of Mechanical and Materials Engineering, Washington State University, Pullman, WA, USA

David George

Northumbria University, Newcastle upon Tyne, UK; Head of Entomology, Stockbridge Technology Centre, North Yorkshire, UK

Luke Gibbons

Chemistry Department, Georgia Southern University, Statesboro, GA, USA

Ashley Johns

Chemistry Department, Georgia Southern University, Statesboro, GA, USA

Michael R. Kessler

Department of Mechanical and Materials Engineering, Washington State University, Pullman, WA, USA

Simon Laflamme

Department of Civil, Construction, and Environmental Engineering; Department of Electrical and Computer Engineering, Iowa State University, Ames, IA, USA

Benedito S. Lima-Neto

Instituto de Química de São Carlos, Universidade de São Paulo, SP, Brazil

Kunwei Liu

Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA

Samy A. Madbouly

Department of Chemistry, Faculty of Science, Cairo University, Orman-Giza, Egypt; Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA

Tolibjon S. Omonov

Lipid Chemistry Group (LCG), Department of Agricultural, Food, and Nutritional Science (AFNS), University of Alberta, Edmonton, AB, Canada

Stephanie Oyola-Reynoso

Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA

James Pritchard

Royal Veterinary College, Hatfield, Hertfordshire, UK

Rafael L. Quirino

Chemistry Department, Georgia Southern University, Statesboro, GA, USA

José L. Silva Sá

Centro de Ciências da Natureza, Universidade Estadual do Piauí, PI, Brazil

Madeline Smith

Chemistry Department, Georgia Southern University, Statesboro, GA, USA

Olivier Sparagano

Vice-Chancellor Office, Coventry University, Coventry, UK

Yi-Shu Tai

Department of Chemical Engineering and Materials Science, University of Minnesota, Minneapolis, MN, USA

Fana Teffera

Department of Chemical and Biological Engineering, Iowa State University, Ames, IA, USA

Martin Thuo

Department of Materials Science and Engineering; Micro-Electronic Research Center; Biopolymer and Biocomposites Research Team, Center for Bioplastics and Biocomposites, Iowa State University, Ames, IA, USA

Chaoqun Zhang

Department of Mechanical Engineering, Texas A&M University, College Station, TX, USA

Kechun Zhang

Department of Chemical Engineering and Materials Science, University of Minnesota, Minneapolis, MN, USA

Jing Zhang

Department of Corrosion Prevention and Polymer Materials, College of Materials Science and Engineering, Jiangsu University of Science and Technology, Zhenjiang, China

Most plastics are produced entirely from petroleum. However, the volatility of oil markets and diminishing petroleum reserves have motivated a desire to reduce dependence on petroleum products. Together with an increase in environmental awareness, this has promoted the development and use of alternative, biorenewable, environmentallyfriendly plastics from biomass. The growing interest in replacing petroleum-based products by inexpensive, renewable, natural materials is important for sustainable development into the future and is already having a significant impact on the polymer industry and the environment. Plant oils represent one of the most promising renewable resources that can be used to create and develop a new class of bioplastics and composites with tailored multifunctional properties due to their unique chemical structure. Recently, polymers and composites derived from plant oil-based monomers have shown thermomechanical properties that rival those of their counterparts from petroleum because of the intrinsic structure and unique properties of plant oils. Additionally, new novel technology has improved the physical and thermomechanical properties of biobased polymers and enabled multifunctionality, such as antibacterial behavior, which further promotes their application in new areas, including sensors and medical devices.

In *Bio-Based Plant Oil Polymers and Composites*, we describe some of the latest technologies in the development of polymeric materials and composites from plant oils (thermoplastics, thermosets, nanocomposites, and fiber-reinforced composites) and summarize their applications. This book is likely to be of immediate practical interest to academic researchers, industrial scientists, engineers, and students in the fields of materials science and engineering and polymer science. The timeliness of the text owes to the increasing transformation from a fossil resource dependent society into a society dependent on renewable resources.

This book covers details of the preparation, characterization, and applications of vegetable oil-based polymers and composites. Chapter 1 describes sources, types, properties, structures, traditional usage, worldwide production (quantity/distribution), quality, and purity of plant oils. Chapter 2 describes important industry chemicals derived from plant oils, such as biofuels, lubricants, and adhesives, and examines their properties relative to their nonrenewable counterparts. Chapter 3 discusses the development of plant oil-based polyols for polyurethanes via different methods. This chapter also reports some critical applications of polyurethanes and waterborne polyurethane dispersions. Chapter 4 presents the development of plant oil-based polyhydroxyurethane and its advantages compared with traditional polyurethane products. Chapter 5 discusses the development of unsaturated polyester resins and its application as biomaterials and damping materials. Chapter 6 reports the fundamental aspects of plant oil-based polyether synthesis, processing, and characterization. Chapter 7 details the epoxidation method of plant oil, the ring-opening reaction of epoxidized vegetable oil by different initiators, and the polymerization of the epoxidized plant oils using different hardeners. Chapter 8 discusses polymer synthesis by

various enzymes as well as modification of plant oil as monomers for polymeric materials by catalyzing oxidation, reduction, and transesterification reactions. Chapter 9 highlights substantial enhancement in mechanical, thermal, and dielectric properties of plant oil-based polyurethane by three different types of surface-modified and functionalized nanofillers including lignin, graphene, and carbon nanotubes. Chapter 10 presents plant oil-based composites reinforced with fibers, such as long glass fibers and banana fibers. The reinforcement mechanism and the treatment of fiber surfaces for improved interfacial strength are also discussed. Chapter 11 discusses the applications of plant oil-based products in structural health monitoring. Finally, Chapter 12 covers the current application of essential oils as pest control products and its prospection and future.

Jiahao Chen^{1,2}, Stephanie Oyola-Reynoso¹, and Martin Thuo^{1,2,3}

¹Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA ²Micro-Electronic Research Center, Iowa State University, Ames, IA, USA ³Biopolymer and Biocomposites Research Team, Center for Bioplastics and Biocomposites, Iowa State University, Ames, IA, USA

		Ο U T L I N E			
1.1	Introd	luction	1		
1.2	Chem	istry and Biochemistry of Plant Oils	3		
	1.2.1	Biosynthesis of Terpenoids	3		
	1.2.2	Biosynthesis of Fatty Acids	6		
	1.2.3	Chemical Reactivity of Plant Oils	6		
	1.2.4	Chemical Characterization	10		
1.3	Major	r Sources of Plant Oils	11		
1.4	Extra	ction and Processing of Plant Oils	11		
	1.4.1	Small-Scale Extraction and Processing	11		
	1.4.2	Large-Scale Extraction and Processing	12		
	1.4.3	Postextraction Processing of Fatty Acids	13		
1.5	Outlo	ok	13		
Ref	eferences 14				

1.1 Introduction

Plant oils is a class of lipids that have low melting points, are primarily long chain hydrocarbons, and have varying degree of unsaturation. Plant oils are a potentially renewable feedstock to a variety of industrial or consumer products that has not been fully exploited [1–3]. Plant oils can be classified into two main classes based on their biosynthetic origin namely, (i) fatty acids and their glycerol condensation products [4], and (ii) iso-

prenoid derived hydrocarbons and their condensation products. In most literature related to biorenewable feedstock, especially those focused on polymerization [5–22], fatty acidderived oils bear the most significance since they can be extracted in significant quantities. Composition of fatty acids, and therefore their condensed triglycerides, show little variation within a species and has become a good target for chemotaxonomy [2]. Plant biosynthesis has evolved a mechanism to keep the melting points of fatty acids as low as possible, hence their oily nature. This evolutionary strategy has led to these fatty acids having, preferentially, functional groups that are associated with low melting points [2]. The isoprenoidderived oils (commonly known as essential oils), though less explored, offer an opportunity to increase complexity of products derived from plant oils and as such are reviewed in this chapter alongside the triglyceride-based plant oils. Plant oils are chemically heterogeneous.

A majority of plant oils, however, has high quantities of triglycerides (>80%), with the rest of the oil being primarily di- and monoacylglycerols, phospholipids, terpenes, sterols, and fat-soluble vitamins. Plant oils are therefore named according to their biological origin – mostly based on the plant name; for example, soy oil and castor oil – even though every oil has unique physical (e.g., melting temperatures, density, and refractive index) and chemical properties. To assess the chemical quality of the oil, several metrics have been adopted. These metrics include (i) hydroxyl value (number of free -OH groups), (ii) acid value (free -CO₂H groups), (iii) olefin or iodine value (number of free C=C

bonds), and (iv) peroxide value (measure of oxidative degradation) (Figure 1.1).

Plant oils have been used in many forms throughout the history of mankind. From lighting [4], lubricants [4], paints in ancient art [23], perfumes, fragrances, to polymer synthesis in recent years [14,24–27], humans have adopted plant-derived oils for a myriad of uses. More recently plant oils have come to prominence as an alternative source for products, like polymers, that have primarily been derived from petroleum. The challenge of sustainability in the petrochemical industry has pushed the need for other sources of industrial feedstock and/or more environmentally benign sources. Better agricultural practices have also helped promote biorenewables as a feasible source of industrial feedstock without jeopardizing global food security.

World production of oil producing crop has been on the rise since the 1980s with current production rising to billions of pounds per year (Table 1.1) [3,28]. Brazil, for example, is projected to produce 95.5 million tons of soybeans in 2015 (17% increase from 2014, and >500% increase since 1980) [28]. The ability



Figure 1.1 General chemical structure of triglycerides, the primary structure on which bulk plant oils are based. The structure is derived from condensation of a carbonyl moiety with a glyceride unit to give mono-, di-, or triglycerides. In some cases, the hydroxyl moiety of the parent glyceride unit contains a phosphate to give phospholipids.

Year	Wheat	Coarse Grains	Total Grains	Oilseed	Cotton
2013	658	1136	2266	475	124
2014	715	1278	2471	504	120
2015	723	1272	2472	532	119

Table 1.1 Production of Grains and Oil-Producing Crops Between 2001 and 2015 Shows SignificantIncrease in the Global Production (in Million Metric Tons) [28]

to optimize agricultural output, either through mechanization and better farming methods or science and technological advancements, has allowed previous low producing countries to emerge as leaders in cultivation of oil producing plants. For example, palm oil production in Malaysia has gained prominence due to the adaptation of the African palm tree in Southeast Asia while soybean farming in Brazil has benefited from mechanized farming [29,30]. It is projected that agricultural output for grains and oilseed will increase, especially with the developing and emerging economies adopting better farming practices.

Plant oils are often obtained as a mixture of compounds making their use in polymer synthesis, in their crude form, a challenge. As a caution, however, the chemical composition of any plant oil can only be roughly defined since, like all natural product blends, it depends on (i) the plant cultivar, (ii) growing conditions, (iii) part of the plant from which the oil is extracted, and (iv) the health of the plant. Soy oil, for example, is rich in triglycerides (99%) but also contains about eight major fatty acids [31]. Besides variation in chemical identity, fatty acid chains in soy oil have different number of carbons and different degrees of saturation. Postextraction chemical modification, however, allows for the oils' complexity to be narrowed down for applications that require isotropic composition. In some cases, however, the diversity in chemical functionalities is desired. Paints and coating, for example, have been a historically unique application of plant oils since unsaturation in the oils leads to cross-linking and/or polymerization on exposure to ambient conditions.

This chapter reviews aspect of plant oils that are relevant to a multidisciplinary audience with a bias toward polymer synthesis and engineering application. Plant oils cover a breadth of compounds that would be difficult to review in a single chapter. The most pertinent information regarding biosynthesis, reactivity, sources, production, and processing is given.

1.2 Chemistry and Biochemistry of Plant Oils

With advances in genetic engineering, and a growing demand for plant-derived industrial feedstock, understanding the biosynthetic origin of a desired product can motivate genetic engineering of different organisms to maximize bioproduction. A brief overview of both the isoprenoid-derived oils (essential oils) and the fatty acid precursors to glycerides is given below. This is not an exhaustive review but rather points at the differences between the two mechanisms, with a special emphasis on key steps or unique intermediates.

1.2.1 Biosynthesis of Terpenoids

There are two major classes of plant oils, glycerides (major) and terpenoids (minor) [32,33]. Most low molecular weight plant

oils belong to the terpene family, one of the most structurally diverse class of compounds [33]. Terpenoids are derived from an isoprene unit (Figure 1.2a). Terpenoids are structurally diverse and contain C = C, carboxyl and hydroxyl groups among others (Figure 1.2). The smaller terpenes (Figure 1.2b) can either be linear or cycloalkyls with various degrees of oxidation and stereochemical compositions. The ability of elongated isoprenes to cyclize can lead to aromatization to give elaborate structures that can be used in polymer synthesis (Figure 1.2c). Although less common, isoprenoids can condense with glycols through ether linkages (e.g., phytonyl glycerol, Figure 1.2b). These alkylated glycerols are structurally analogous to glycerides that are major components of commodity plant oils. There are also other terpenoid derivatives, like sterols, that are not of interest to the polymer community but have other uses as secondary metabolites (Figure 1.2d).

Polyisoprenoids, molecules derived from multiple IPP fragments, are biosynthesized through subsequent elongation of a dimethylallyl pyrophosphate (DMAPP) starting unit by IPP (Figure 1.2a). This elongation reaction is catalyzed by prenyltransferases like polyisoprenoid diphosphate synthases. The elongation reaction is a nucleophilic elimination of the allylic diphosphate on the growing chain with subsequent α -H elimination on the added IPP (Figure 1.2a). The subsequent α -H elimination on the IPP can lead to cis- or trans-olefins in the polyisoprenoid. The presence of electronic rich olefins on terpenoids opens the possibilities, at a first level, for cyclization, condensation, reduction, and/or oxidation (Figure 1.2b). Any other subsequent structural modification bestows this class of compound with complexity that would make them a rich entry point to biopolymer synthesis. Figure 1.2a and b

shows some examples of structurally diverse molecules with potential for use as monomers. With increased structural diversity, however, the compounds are no longer oils but span a plethora of physical forms from waxy compounds to crystalline compounds like the steroids (Figure 1.2d).

Terpenes are, biosynthetically, derived from two isomeric compounds, isopentyl diphosphate (IPP) and DMAPP through successfully elongation [32,33], that is often accompanied by structural modifications through oxidation, reduction, or, cyclization (Figure 1.2c and d). There are two main biochemical pathways for the synthesis of the IPP precursor in plants, namely, the mavelonate (MVA) pathway and the methylerythritol (MEP) pathway - also referred as deoxyxylulose phosphate (DOX) pathway (Figure 1.3). The MVA pathway starts with Claisen condensation of acetyl coenzyme A (AcCoA) to an acetylacetyl CoA (AcAcCoA) (Figure 1.3a). This condensation reaction is then followed by an aldol condensation of the AcAcCoA with a third AcCoA to give 3-hydroxy-3-methylglutaryl-CoA (HMG-CoA), which is subsequently reduced to give mavelonic acid. Eventual reductive decarboxylation leads to the terpene-starting unit, IPP (Figure 1.3a). The MEP (DOX) pathway (Figure 1.3b) starts with condensation of pyruvic acid with glyceraldehyde-3-phosphate as catalyzed by 1-deoxy-D-xylulose-5-phosphate synthase (DXS) with a thiamin diphosphate cofactor (Figure 1.3b). Demethvlation and reduction of the methylketone to an aldehyde followed by cytidinylation 4-diphosphocytidyl-2-C-methyl-Dgives erythritol. Subsequent C-2 phosphorylation followed by cyclization of the diphosphate leads to a rare cyclic diphosphate that is then dephosphorylated and reduced to give either IPP or DMAPP.



Figure 1.2 Overview of terpenes, a key class of molecules of essential oils. (a) Biosynthesis of terpenes from DMAPP and IPP to give monoterpenes or more elaborate structures. (b) Examples of chemical structures of commonly found terpenes, illustrating the structural diversity even in the low-molecular weight molecules. (c) Examples of polyolefin containing terpenes and/or other polymerizable groups. (d) Examples of structurally complex terpenes, a class dominated by steroids and related structures.



Figure 1.3 Biosynthesis of terpene precursors, dimethylallyl pyrophosphate (DMAPP) and IPP. (a) The MVA pathway, and (b) the DOX pathway also referred to as the MEP pathway.

1.2.2 Biosynthesis of Fatty Acids

Biosynthesis of fatty acid-derived oils is less elaborate compared to terpenoid-based oils. Glycerides (tri-, di-, and monoacylglycerides) are derived from condensation of fatty acids (see examples in Figure 1.4a) with a glycerol to give esters linkages. Since fatty acid are derived from acetyl CoA, they are limited in structural complexity - in comparison to terpenoids, but presence of olefins and hydrolyzable esters allows for simple transformations (Figure 1.4b). Fatty acids are derived from malonyl-Coenzyme A (malonyl CoA), which is synthesized by carboxylation of acetyl CoA by a biotinylated enzyme (Figure 1.4c). The malonyl CoA is sequentially elongated, 2 carbons at a time, by coupling with acetyl CoA. This elongation process (Figure 1.4c) is mediated by an acyl carrier protein (ACP) that is similar to a CoA with a difference that the phosphopantetheine group is attached to a serine rather than a nucleotide.

1.2.3 Chemical Reactivity of Plant Oils

Plant oils, fatty acids, or isoprenoidderived, often contain C=C bonds, carboxylic acids, or hydrolyzable esters (Figure 1.4b). Presence of alkene moieties allows for, potentially, many reactions to be carried out on plant oils (Figure 1.5b). The greatest challenge is heterogeneity in composition of the oil, hence the need to assess functional group compatibility to allow for successful, or selective, transformation(s). Figure 1.5 gives a summary of potential reactions of alkenes (Figure 1.5b) and carboxylic acids/esters (Figure 1.5a). We only highlight key chemical transformation to illustrate the potential of these two functional groups, carboxylic acids and olefins, in increasing diversity



Figure 1.4 An overview of fatty acid derived oils. (a) Examples of common plant-derived fatty acids. (b) Triglycerides with examples of a variety of polymerizable or modifiable moieties to illustrate potential complexity and opportunities for further chemical elaboration or polymerization. (c) An abbreviated schematic of the biosynthesis of various fatty acids from malonyl-Coenzyme A (malonyl-CoA) that is derived from carboxylation of acetyl-CoA.



Figure 1.5 An overview of some reactions of carboxylic acids (a) and alkenes (b). The type of reactions range from simple carboxylate condensation to cycloadditions like Diels–Alder reaction and cyclopropyration in the alkenes.

of the reactive sites in plant oils. Detailed chemical transformations have been given in the literature [1,9,13,14,25–27,34–43]. For a discussion and practical guide on performing the chemical transformations listed below, the reader is referred to the Vogel's text on practical organic chemistry [44].

The chemistry of carboxylic acid and esters is dominated by the presence of an electrophilic carbon and the presence of a cleavable C—O single bond (for esters) or the acid proton (or analogous carboxylate anion) on the carboylic acid. From the carboxylic acid, direct or indirect condensation of the carboxvlate with nucleophilic moieties can lead to amines, amides, esters, anhydrides, or acyl halides [45–49]. Reduction of the carboxylate leads to a terminal hydroxyl while activation of the -OH into good leaving groups, for example, a tosylate, allows for monoalkylation of the carboxylate using mild reagents to give ketones. These reactions can be used to create intermediate moieties that can be used to further increase structural diversity of the oils and/or alter the oils' properties and/or aid in processing.

Olefin chemistry is dictated by the presence of an electron-rich C=C bond. This allows electrophilic and/or nucleophilic addition across the π -bond. The addition can be symmetric or asymmetric depending on the functional groups adjacent to the olefin that may influence the sharing of the π -electron cloud. Common symmetric additions include bromination, hydrogentation, dihydroxylations, epoxidation, and cyclopropanation. Asymmetric additions are dominated by Markovnikov and anti-Markovnikov's additions, oxymercuration-demercurations, halohydrin formation, hydroborationoxidations, and related reactions. Besides additions, olefins can undergo cycloadditions depending on other associated functional groups in the molecule. The most common cycloaddition reaction for olefins is the 4 + 2Diels–Alder reaction in which an olefin reacts with a diene to give a cyclohexene [47] or the 2 + 2 reaction to give a cyclobutane [50]. Formation of three-membered rings [51] with either a carbon (cyclopropanation), oxygen (epoxidation), or sulfur (thiirane formation) can also be achieved either as a terminal product or as an intermediate structure to further chemical transformations.

Cleavage of the C=C bond can also be achieved through processes like ozonolysis to either give aldehydes or carboxylic acids. This cleavage has the advantage that, for plant oils, it can be used to reduce the size of the fatty acid chain while introducing reactive groups on the shortened oils or to alter their thermal properties (products of ozonolysis are more polar than the C=C bond). Olefin metathesis [45,46,51–55] can also be used to either cleave or couple olefins in plant oils. Through olefin cross-metathesis with a smaller olefin, the size of the fatty acid can be shortened; however, this is likely to occur alongside polymerization or cross-metathesis between the fatty acid chains. Olefin metathesis can also be used to alter the physical properties of an oil through controlled cross-linking to create large, oligomeric, macromonomers or to prepare polymers. A combination of the above reactions can be used to develop functional group selective chemistry based on the physical properties of the resultant compounds [52,54]. Where stereochemistry is desired, a plethora of chemical transformation exists to transform either one or both of the olefinic carbons into chiral centers [53]. Examples of chiral additions across olefins include asymmetric hydrogenation (Knowles or Noyori's catalysts), Sharpless bis-dihydroxylation using osmium catalysts, Jacobsen's epoxidation,

asymmetric hydroboration–oxidation using chiral auxiliaries (e.g., isopinocamphenylboranes). Introduction of chiral centers requires the use of chiral auxiliaries and exploitation of sterics around the olefin.

1.2.4 Chemical Characterization

The development of plant oils as commercially viable chemicals, which was dominated for a long time by essential oils, has significantly evolved in line with development of analytical and separation techniques [56]. Due to their relatively low molecular weights and compositional diversity, gas chromatography (GC), mass spectrometry (MS), and liquid chromatography (LC) have been widely used to characterize their composition [56]. GC has a sufficiently high sensitivity that allows for even minor components of the oil to be analyzed. One disadvantage is the need for standards to ascertain the identity of the peak, although retention indices (RI) can be used to predict a possible structure assuming homology to known components. The use of RI is favorable to plant oils since structure similarities, due to a series of oils emanating from the same biosynthetic pathway, is very likely. A similar approach can also be used with LC for components that are not readily vaporized on a GC, for example, triglycerides and carboxylic acids. Since plant oils are isolated as mixtures, MS on its own cannot be used to characterize an oil blend without prior purification or isolation of individual components.

Supercritical fluids, with their high density, high diffusivity, and low viscosity, have been used as the mobile phase in chromatographic separation of plant oils. Capillary supercritical fluid chromatography (SFC) has shown reliable component quantifications of oxygenated compounds than GC [57,58]. Counter current chromatography (CCC), a form of liquid–liquid partition chromatography, has also been used to characterize plant oils.

Development in coupled (hyphenated) techniques, like GC-MS and LC-MS, has allowed for separation with concomitant identification of all components in an oil blend or other complex biological samples [56,59–61]. For low molecular weight compounds, GC-MS is used to generate a profile and identity for each of the components in the oil. For polar compounds, derivatization with nonpolar moieties is often used to allow for vaporization, promote uniform flow (sharp peaks), and prevent significant damage to the stationary phase of the GC column. Other coupled techniques that can be used to analyze plant oils are often technically involved, albeit with higher certainty in chemical structures in most cases. The recently reported LC-MS-NMR technique [59,60], for example, has great potential in plant oil analysis even though it is technically involving.

Chemical characterization of plant oils is often challenging and mischaracterization is possible due to structural homology and/or presence of difficult-to-resolve minor components. With techniques like GC and LC, there is potential for peaks to overlap leading to misidentification, although preparative techniques can be used to isolate pure components for full chemical characterization. Isolation of small pure quantities of plant oil components can be achieved through preparative thin-layer chromatography, preparative LC, distillation or chemical derivatization.

Other chemical characterizations that are used to identify chemical constituents of oils are ultraviolet (UV) spectroscopy, infrared (IR) spectroscopy, Raman spectroscopy and nuclear magnetic resonance (NMR).

1.3 Major Sources of Plant oils

The ISTA Mielke (Hamburg, Germany) recognizes 17 commodity oils and fats, 13 of which are vegetable oils. Vegetable oils can be sourced from (i) annual crops that need to be planted each year or season (e.g., canola, sunflower, soybeans, castor, and linseed), (ii) tree crops that take long time to mature but can be harvested for many years (e.g., coconut and palm tree), or (iii) as secondary products from plants grown for other purposes (e.g., cotton and soybeans). The focus of this section is not to review where the oils are derived from but to highlight the parts of the plants from which the oils can be derived. We highlight both glycerides- and isoprene-based oils.

Plant oils can be derived from many parts of the plant and a full review of the sources of plant oils is beyond the scope of this chapter. To give a few examples, leaves of Eucalyptus are used to produce significant amount of oils that are rich in citronellal. Southeast Asia and western Africa has also led the effort in production of palm oil, which is extracted from the fruit of the palm tree. Common plant oils, predominantly used in the fragrance and perfume industry, are derived from barks/wood (camphor, sandal wood), flowers (rose oil, chamomile, and jasmine), roots and rhizomes (valerian and ginger), fruits or fruit peels (palm or citrus oil), seeds and berries (almond oil, Juniper, anise oil, nutmeg oil), and resins (Myrrh, benzoin, and frankincense). Although there is a large diversity in the part of the plant from which the oil can be extracted, most oils have similar compounds, since they are derived from a limited number of biosynthetic pathways, albeit in varying concentrations. The part of the plant in which the oil accumulates depends on the evolutional role

of these oils to the plants, with most of them having evolved as semiochemicals for *intra*and *inter*-species chemical communication.

Understanding of the biosynthetic pathways and the role these oils play in the plant can potentially be used to develop genetic and/or external methods to increase yields. Several genetically engineered plants and/or enzymes have been developed to increase the production of the oils but the technology is not yet at an industrial scale.

1.4 Extraction and Processing of Plant Oils

1.4.1 Small-Scale Extraction and Processing

Extraction technique employed in obtaining plant oils depends on the required amount, desired purity, the part of the plant from which the oil is sourced and the use. For qualitative analysis, oils can be extracted directly from the oil glands or the stoma using glass capillaries, or headspace analysis for volatiles exudates. For semiguantitative amounts adsorbents can be used to obtain small quantities of the oil, especially exudates, sometimes without significantly affecting the plant. For laboratory scale oil extractions, soxhlet extraction or a Deanand-Stark apparatus can be used to obtain gram quantities of oil. The soxhlet extraction is desired where the oil has a higher boiling point than the solvent, the source has high fiber/particulate content, or plant contains thermally degradable compounds that can contaminate the oil. The Dean-and-Stark apparatus is used to coevaporate the oil with a solvent in which the oil is immiscible, often hydrodistillation, with the oil accumulating in Dean-and-Stark apparatus. Immiscibility of oil and the solvent leads to a two-phase distillate that is readily separated.

More recently, the use of polymeric membranes to separate organic compounds either for the sake of purification [62–65] or during a chemical transformation [45,46,51– 54,66–71] has been reported. The Bowden's approach to purification of fatty acids [62,63,65] relies on the selective transport across a thin polymeric membrane that can be tuned, through degree of cross-linking and/or swelling, to allow for efficient purification of plant oils.

1.4.2 Large-Scale Extraction and Processing

For large quantities, or processes where the oil is a starting material in a production process, distillation or cold press techniques are widely used [4]. Due to the wide variation in techniques that can be employed, detailed analysis of the processing of palm is used to illustrate the process.

1.4.2.1 Palm Oil

Palm oil producing trees are natives of West Africa but they were imported to Southeast Asia where, in some places, have become a major revenue stream. Extraction of the palm oil from the fruits involves cleaning and mechanical (screw) pressing of the palm fruits. The oil is then filtered, clarified, and purified to give the crude palm oil (Figure 1.6a). To obtain the major components, however, further chemical or physical refining is required (Figure 1.6b). The chemical process involves alkali neutralization followed by earth bleaching and deodorizing to give the refined neutral bleached palm oil (NBD PO). The physical refining process is analogous to chemical refining with the exception that the neutralization step is replaced with degumming. The degumming process is used to remove



Figure 1.6 Processing of plant oils, palm oil as an example. (a) A general schematic of how palm fruits are processed after harvesting. The oil is extracted by cold pressing then purified to give industrial grade product. (b) The oil can then be refined through chemical or physical processes to give stearin and olein after fractionation.



Figure 1.7 A general schema showing the processing of oils and fats into various chemical or reactive moieties using simple chemical reactions like esterifications of hydrogenation. The diversity of reactive moieties indicates the potential of oils and fats.

phospholipids analogous to the alkali neutralization (hydrolysis). Upon fractionation of the refined palm oil, stearin, and olein are obtained as the major products (Figure 1.6).

1.4.3 Postextraction Processing of Fatty Acids

Based on functional group diversity in the purified plant oil, several chemical processes can be used to improve the quality of the oil. Postextraction modifications can be used to introduce polymerizable chemical moieties in the plant oils. For polymer synthesis, although most oils contain polymerizable groups like epoxide, olefins, and hydroxyls, it is often desired to introduce moieties like norbornenes that can be polymerized using organometallic catalysts. Larock and coworkers, for example, have shown that by introducing norbornenes in castor oil and soybean oil, they can use the introduced norbornene to synthesize thermosets through ring-opening metathesis polymerization (ROMP) [13,14,25-27,43].

At an industrial scale, oils and fats can be saponified into the glycerol and fatty acids, hydrogenated to give saturated oil, or chemically modified to give derivatives of the triglycerides (Figure 1.7) [72]. Further processing can lead to alcohols, esters, ethers or other desired compounds. The postextraction modification depends on the use and possible chemical transformations based on available chemical moieties (Figure 1.7).

1.5 Outlook

The use of plant-derived oils is on the rise. With an increase in the need for biorenewable and sustainable engineering, plant-derived polymers are more desired. A challenge to the scientists and engineer in this field is to find means to use plant-derived materials without compromising food security. Genetic engineering has the potential to maximize plant oil production. This book addresses future prospects of plant oils in detail in later chapters.

References

- [1] U. Biermann, W. Friedt, S. Lang, W. Luhs, G. Machmuller, J.O. Metzger, M.R. Klaas, H.J. Schafer, M.P. Schneider, New syntheses with oils and fats as renewable raw materials for the chemical industry, Angew. Chem. Int. Ed. 39 (13) (2000) 2206–2224.
- [2] H. Baumann, M. Buhler, H. Fochem, F. Hirsinder, H. Zoebelein, J. Falbe, Natural fats and oils – renewable raw materials for the chemical industry, Angew. Chem. Int. Ed. 27 (1) (1988) 41–62.
- [3] D.B. Turley, The chemical value of biomass, in: J. Clark, F. Deswarte (Eds.), Introduction to Chemicals From Biomass, John Wiley & Sons, New York, 2008.
- [4] R. Verhe, M. Mittelbach, S. Mateo, V. Eychenne, P. De Caro, Z. Mouloungui, C.V. Stevens, Industrial products from lipids and proteins, in: C.V. Stevens, R. Verhe (Eds.), Renewable Bioresources: Scope and Modification for Non-Food Applications, John Wiley & Sons, New York, 2004, pp. 208–250.
- [5] P.A. Fokou, M.A.R. Meier, Use of a renewable and degradable monomer to study the temperature-dependent olefin isomerization during ADMET polymerizations, J. Am. Chem. Soc. 131 (5) (2009) 1664–1665.
- [6] L. Montero de Espinosa, M.A.R. Meier, Plant oils: the perfect renewable resource for polymer science?, Eur. Polym. J. 47 (5) (2011) 837–852.
- [7] J.C. Ronda, G. Lligadas, M. Galia, V. Cadiz, A renewable approach to thermosetting resins, React. Funct. Polym. 73 (2) (2013) 381–395.
- [8] S. Warwel, F. Bruse, C. Demes, M. Kunz, M.R. Klaas, Polymers and

surfactants on the basis of renewable resources, Chemosphere 43 (1) (2001) 39–48.

- [9] U. Biermann, J.O. Metzger, M.A.R. Meier, Acyclic triene metathesis oligoand polymerization of high oleic sun flower oil, Macromol. Chem. Physic. 211 (8) (2010) 854–862.
- [10] J.V. Crivello, R. Narayan, Epoxidized triglycerides as renewable monomers in photoinitiated cationic polymerization, Chem. Mater. 4 (3) (1992) 692–699.
- [11] J.P. Dwan'Isa, A.K. Mohanty, M. Misra, L.T. Drzal, Bio-based polyurethanes and their composites: present status and future perspective, in: A.K. Mohanty, M. Misra, L.T. Drzal (Eds.), Natural Fibers, Biopolymers, and Biocomposites, CRC Press, Boca Raton, FL, 2005, pp. 775–805.
- [12] A. Guo, D. Demydov, W. Zhang, Z.S. Petrovic, Polyols and polyurethanes from hydroformylation of soybean oil, J. Polym. Environ. 10 (1/2) (2002) 49–52.
- [13] P. Henna, R.C. Larock, Novel thermosets obtained by the ring-opening metathesis polymerization of a functionalized vegetable oil and dicyclopentadiene, J. Appl. Polym. Sci. 112 (3) (2009) 1788–1797.
- [14] P.H. Henna, M.R. Kessler, R.C. Larock, Fabrication and properties of vegetable-oil-based glass fiber composites by ring-opening metathesis polymerization, Macromol. Mater. Eng. 293 (12) (2008) 979–990.
- [15] W. Kaminsky, M. Fernandez, New polymers by copolymerization of ole-fins with bio oil components, Eur. J. Lipid Sci. Technol. 110 (9) (2008) 841–845.
- [16] G. Lligadas, J.C. Ronda, M. Galia, V. Cadiz, Plant oils as platform chemi-

cals for polyurethane synthesis: current state-of-the-art, Biomacromolecules 11 (11) (2010) 2825–2835.

- [17] Z.S. Petrovic, I. Cvetkovic, D.P. Hong, X. Wan, W. Zhang, T.W. Abraham, J. Malsam, Vegetable oil-based triols from hydroformylated fatty acids and polyurethane elastomers, Eur. J. Lipid Sci. Technol. 112 (1) (2009) 97–102.
- [18] D. Quinzler, S. Mecking, Renewable resource-based poly(dodecyloate) by carbonylation polymerization, Chem. Commun. (36) (2009) 5400–5402.
- [19] D. Quinzler, S. Mecking, Linear semicrystalline polyesters from fatty acids by complete feedstock molecule utilization, Angew. Chem. Int. Ed. 49 (25) (2010) 4306–4308 S4306/1–S4306/14.
- [20] F. Stempfle, P. Roesle, S. Mecking, Long-chain polyesters via chemical catalytic conversions of fatty acid esters, ACS Symp. Ser. 1105 (2012) 151–164.
- [21] O. Tueruenc, M.A.R. Meier, The thiolene (click) reaction for the synthesis of plant oil derived polymers, Eur. J. Lipid Sci. Technol. 115 (1) (2013) 41–54.
- [22] S. Warwel, F. Bruese, M. Kunz, New polymers from plant oils as raw material, Fresen. Environ. Bull. 12 (6) (2003) 534–539.
- [23] M.R. Van De Mark, K. Sandefur, Vegetables oils in paints and coatings, in: S.Z. Erhan (Ed.), Industrial Uses of Vegetable Oil, AOCS Publishing, Peoria, 2009, pp. 143–162.
- [24] U. Poth, Drying oils and related products, Ullman's Encyclopedia of Industrial Chemistry, Wiley-VCH Verlag GmbH & Co, Weinheim, 2001.
- [25] T.C. Mauldin, K. Haman, X. Sheng, P. Henna, R.C. Larock, M.R. Kessler, Ring-opening metathesis polymerization of a modified linseed oil with vary-

ing levels of crosslinking, J. Polym. Sci. Part A Polym. Chem. 46 (20) (2008) 6851–6860.

- [26] Q. Tian, R.C. Larock, Model studies and the ADMET polymerization of soybean oil, J. Am. Oil Chemists Soc. 79 (5) (2002) 479–488.
- [27] Y. Xia, Y.S. Lu, R.C. Larock, Ringopening metathesis polymerization (ROMP) of norbornenyl-functionalized fatty alcohols, Polymer 51 (1) (2010) 53–61.
- [28] USDA, World Agricultural Production. USDA, Washington, DC, USA, 2015.
- [29] Y. Basiron, Palm oil production through sustainable plantations, Eur. J. Lipid Sci. Technol. 109 (4) (2007) 289–295.
- [30] W. Rupilius, S. Ahmad, Palm oil and palm kernel oil as raw materials for basic oleochemicals and biodiesel, Eur. J. Lipid Sci. Technol. 109 (4) (2007) 433–439.
- [31] K. Liu, Soybeans: Chemistry, Technology, and Utilization, International Thompson Publishing, New York, (1997).
- [32] C.A. Citron, J.S. Dickschat, Microbial terpenes: a perspective on their chemistry and biosynthesis, Synlett 25 (6) (2014) 766–782.
- [33] J.S. Dickschat, Isoprenoids in threedimensional space: the stereochemistry of terpene biosynthesis, Natural Prod. Rep. 28 (12) (2011) 1917–1936.
- [34] U. Biermann, U. Bornscheuer, M.A.R. Meier, J.O. Metzger, H.J. Schaefer, Oils and fats as renewable raw materials in chemistry, Angew. Chem. Int. Ed. 50 (17) (2011) 3854–3871 S3954/1– S3954/36.
- [35] U. Biermann, W. Butte, T. Eren, D. Haase, J.O. Metzger, Regio- and stereoselective Diels–Alder additions of ma-

leic anhydride to conjugated triene fatty acid methyl esters, Eur. J. Org. Chem. (23) (2007) 3859–3862.

- [36] U. Biermann, W. Friedt, S. Lang, W. Luehs, G. Machmueller, J.O. Metzger, M.R.g. Klaas, H.J. Schaefer, M.P. Schneider, New syntheses with oils and fats as renewable raw materials for the chemical industry, Biorefineries Ind. Processes Prod. 2 (2006) 253–289.
- [37] M. Eissen, J.O. Metzger, E. Schmidt, U. Schneidewind, 10 years after Rio – concepts on the contribution of chemistry to a sustainable development, Angew. Chem. Int. Ed. 41 (3) (2002) 414–436.
- [38] M.A.R. Meier, J.O. Metzger, U.S. Schubert, Plant oil renewable resources as green alternatives in polymer science, Chem. Soc. Rev. 36 (11) (2007) 1788–1802.
- [39] J.O. Metzger, Fats and oils as renewable feedstock for chemistry, Eur. J. Lipid Sci. Technol. 111 (9) (2009) 865–876.
- [40] J.O. Metzger, M. Eissen, Concepts on the contribution of chemistry to a sustainable development. Renewable raw materials, C. R. Chim. 7 (6–7) (2004) 569–581.
- [41] J.O. Metzger, S. Fuermeier, New type of skipped oligoaziridines. Synthesis of new fatty acid derivatives containing aziridine functions, Eur. J. Org. Chem. (3) (1999) 661–664.
- [42] J.O. Metzger, U. Linker, Free radical additions to unsaturated fatty compounds, Fett Wissenschaft Technologie 93 (7) (1991) 244–249.
- [43] P.H. Henna, R.C. Larock, Rubbery thermosets by ring-opening metathesis polymerization of a functionalized castor oil and cyclooctene, Macromol. Mater. Eng. 292 (12) (2007) 1201–1209.

- [44] A.I. Vogel, Vogel's Textbook of Practical Organic Chemistry, fifth ed., Pearson Education Limited, Essex, England, 1989.
- [45] M.B. Runge, M.T. Mwangi, A.L. Miller II, M. Perring, N.B. Bowden, Cascade reactions using LiAlH4 and Grignard reagents in the presence of water, Angew. Chem. Int. Ed. 47 (5) (2008) 935–939.
- [46] M.B.J. Atkinson, S. Oyola-Reynoso, R.E. Luna, D.K. Bwambok, M.M. Thuo, Pot-in-pot reactions: a simple and green approach to efficient organic synthesis, RSC Adv. 5 (1) (2015) 597–607.
- [47] E. Plettner, A. Mohle, M.T. Mwangi, J. Griscti, B.O. Patrick, R. Nair, R.J. Batchelor, F. Einstein, 2-Chlorobicyclo [2.2.1]hept-5-ene-2-carboxamide and 2-chlorobicyclo[2.2.1]heptane-2-carboxamide as precursors of bicyclo[2.2.1] hept-5-en-2-one and bicyclo [2.2.1] heptan-2-one: resolution, absolute configuration and hydrogen-bonding properties, Tetrahedron Asym. 16 (16) (2005) 2754–2763.
- [48] M.M. Thuo, W.F. Reus, F.C. Simeone, C. Kim, M.D. Schulz, H.J. Yoon, G.M. Whitesides, Replacing —CH₂CH₂ with —CONH— does not significantly change rates of charge transport through AgTS-SAM//Ga2O3/EGaIn junctions, J. Am. Chem. Soc. 134 (26) (2012) 10876–10884.
- [49] H.J. Yoon, N.D. Shapiro, K.M. Park, M.M. Thuo, S. Soh, G.M. Whitesides, The rate of charge tunneling through self-assembled monolayers is insensitive to many functional group substitutions, Angew. Chem. Int. Ed. 51 (19) (2012) 4658–4661 S4658/1– S4658/23.

- [50] M.B.J. Atkinson, A.N. Sokolov, D.-K. Bucar, S.V.S. Mariappan, M.T. Mwangi, M.C. Tiedman, L.R. MacGillivray, Applications of hydrogen-bond-acceptor templates to direct in-phase reactivity of a diene diacid in the solid state, Photochem. Photobiol. Sci. 10 (9) (2011) 1384–1386.
- [51] M.T. Mwangi, M.B. Runge, K.M. Hoak, M.D. Schulz, N.B. Bowden, A materials approach to site-isolation of Grubbs catalysts from incompatible solvents and *m*-chloroperoxybenzoic acid, Chem. Eur. J. 14 (22) (2008) 6780–6788.
- [52] M.T. Mwangi, M.B. Runge, N.B. Bowden, Occlusion of Grubbs' catalysts in active membranes of polydimethylsiloxane: catalysis in water and new functional group selectivities, J. Am. Chem. Soc. 128 (45) (2006) 14434–14435.
- [53] M.T. Mwangi, M.D. Schulz, N.B. Bowden, Sequential reactions with Grubb's catalyst and AD-mix- α/β using PDMS thimbles, Org. Lett. 11 (1) (2009) 33–36.
- [54] M.B. Runge, M.T. Mwangi, N.B. Bowden, New selectivities from old catalysts. Occlusion of Grubbs' catalysts in PDMS to change their reactions, J. Organometal. Chem. 691 (24–25) (2006) 5278–5288.
- [55] M.D. Schulz, M.B.J. Atkinson, R.J. Elsey, M.M. Thuo, Copper(I) halides inhibit olefin isomerized by-products from phosphine-based Grubbs' metathesis catalysts in polar protic solvents, Transition Metal Chem. 39 (7) (2014) 763–767.
- [56] K.H. Kubeczka, History and sources of essential oils, in: K.H.C. Baser, G. Buchbauer (Eds.), Handbook of Essential Oils Science, Technology, and

Applications, CRC Press, Boca Raton, 2009, pp. 3–35.

- [57] P. Manninen, H. Kallio, Supercritical fluid chromatography–gaschromatography of volatiles in cloudberry (*Rubus chamaemorus*) oil extracted with supercritical carbon dioxide, J. Chromatogr. A 787 (1997) 276–282.
- [58] P. Manninen, M.L. Riekkola, Y. Holm, R. Hiltunen, SFC in analysis of aromatic plants, J. High Resolut. Chromatogr. 13 (1990) 167–169.
- [59] R.M. Gathungu, S.S. Bird, D.P. Sheldon, R. Kautz, P. Vouros, W.R. Matson, B.S. Kristal, Identification of metabolites from liquid chromatography– coulometric array detection profiling: gas chromatography–mass spectrometry and refractionation provide essential information orthogonal to LC– MS/microNMR, Anal. Biochem. 454 (2014) 23–32.
- [60] R.M. Gathungu, J.T. Oldham, S.S. Bird, C.W.T. Lee-Parsons, P. Vouros, R. Kautz, Application of an integrated LC–UV–MS–NMR platform to the identification of secondary metabolites from cell cultures: benzophenanthridine alkaloids from elicited *Eschscholzia californica* (california poppy) cell cultures, Anal. Methods 4 (5) (2012) 1315–1325.
- [61] S.S. Bird, D.P. Sheldon, R.M. Gathungu, P. Vouros, R. Kautz, W.R. Matson, B.S. Kristal, Structural characterization of plasma metabolites detected via LC-electrochemical coulometric array using LC-UV fractionation, MS, and NMR, Anal. Chem. 84 (22) (2012) 9889–9898.
- [62] N.B. Bowden, A. Gupta, Methods for separating mixtures of fatty acids. 2013-US20339 2014088607, 20130104, 2014.

- [63] N.B. Bowden, A. Gupta, Methods for separating mixtures of compounds. 2014-US14477749 20150065660, 20140904, 2015.
- [64] N.B. Bowden, A. Gupta, T.R. Long, High-density polydicyclopentadiene. 2012-US13546252 20130020262, 20120711, 2013.
- [65] A. Gupta, N.B. Bowden, Separation of *cis*-fatty acids from saturated and *trans*fatty acids by nanoporous polydicyclopentadiene membranes, ACS Appl. Mater. Interf. 5 (3) (2013) 924–933.
- [66] M.B. Runge, M.T. Mwangi, A.L. Miller II, M. Perring, K.M. Hoak, M.D. Schulz, N.B. Bowden, PDMS thimbles for the development of cascade reactions: a materials approach to organic chemistry, PMSE Preprints 100 (2009) 690–691.
- [67] T.R. Long, N.B. Bowden, PDMS, in: P.L. Fuchs (Ed.), Handbook of Reagents for Organic Synthesis: Reagents for Silicon-Mediated Organic Synthesis, John Wiley & Sons, West Sussex, UK, 2011, pp. 403–408.
- [68] A. Gupta, T.R. Long, D.G. Rethwisch, N.B. Bowden, Retention of palladium and phosphine ligands using

nanoporous polydicyclopentadiene thimbles, Chem. Commun. 47 (37) (2011) 10236–10238.

- [69] A.L. Miller, N.B. Bowden, Siteisolation and recycling of PdCl(2) using PDMS thimbles, J. Org. Chem. 74 (13) (2009) 4834–4840.
- [70] N. Bowden, M.B. Runge, A.L. Miller III, Apparatus and method for carrying out multiple reactions for preparing alcohols, pyrrolidines, epoxides, biaryls and other related compounds. 2009-476978 20090299102, 20090602, 2009.
- [71] A.L. Miller II, N.B. Bowden, A materials approach to the dual-site isolation of catalysts bonded to linear polymers and small, ionic molecules for use in one-pot cascade reactions, Adv. Mater. 20 (21) (2008) 4195–4199.
- [72] K. Hill, Industrial development and application of biobased oleochemicals, in: G. Centri, R.A. Van Santen (Eds.), Catalysis for Renewables: From Feedstock to Energy Production, Wiley-VCH Verlag GmbH & Co, Weinheim, 2007, pp. 75–99.

Chaoqun Zhang¹ and Samy A. Madbouly^{2,3}

¹Department of Mechanical Engineering, Texas A&M University, College Station, TX, USA ²Department of Chemistry, Faculty of Science, Cairo University, Orman-Giza, Egypt ³Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA

		Ο U T L I N E			
2.1	Introd	luction	19		
2.2	Plant	Oil-Based Derivatives	21		
	2.2.1	Fatty Acids	21		
	2.2.2	Fatty Amides/Nitriles/Amines	21		
	2.2.3	Alcohols	22		
	2.2.4	Ester Derivatives	25		
	2.2.5	Epoxy Derivatives	28		
	2.2.6	Conjugates	30		
	2.2.7	Other Derivatives	31		
2.3	Concl	usions	32		
Refe	References 33				

2.1 Introduction

The depleting of fossil resources, the everincreasing emission of greenhouse gases and toxic waste, and stringent environmental regulation have triggered an increasing interest in developing industrially compounds and polymers from natural resources as an alternative of petroleum-based counterparts [1,2]. Various renewable resources are available, including cellulose, starch, natural oils, and sugars. Among them, plant oils are among the most promising options; they offer excellent properties, including world-wide availability, inherent sustainability nature, and relatively low price [3,4]. A large number of plant oils, such as soybean oil, rapseed oil, and palm oil, are considered to be the most

important renewable feedstock processed in the chemical industry to decrease the dependence on depleting petroleum resources.

Vegetable oils are esters formed between glycerin and different fatty acids containing from 8 to 24 carbons and from 0 to 7 carbon– carbon double bonds (see Table 2.1), depending on the plant type and climatic conditions of harvest. Typical structure of vegetable oils is shown in Figure 2.1. Naturally existing carbon double bonds in most vegetable oils are located in the 9–16th carbon, which makes them less reactivity due to radical trapping by allyl hydrogens from methylene group between double bonds. However, in some oils, such as Tung oil, conjugated double bonds are available, which demonstrate high polymerization activity. Ester group is another

		Fatty Acid Composition (%)				
	Double Bonds	Palmitc	Stearic	Oleic	Linoleic	Linolenic
Castor	3.0	1.5	0.5	5.0	4.0	0.5
Corn	4.5	10.9	2.0	25.4	59.6	1.2
Linseed	6.6	5.5	3.5	19.1	15.3	56.6
Olive	2.8	13.7	2.5	71.1	10.0	0.6
Palm	1.7	42.8	4.2	40.5	10.1	-
Soybean	4.6	11.0	4.0	23.4	53.3	7.8
Canola	3.9	4.1	1.8	60.9	21.0	8.8

 Table 2.1
 Degree of Unsaturation, Composition of Common Vegetable Oils



Figure 2.1 Typical structure of vegetable oils $(R_1, R_2, R_3$ represent fatty acid chains).

reactive site for most of vegetable oils, and some oils contain other reactive groups such as hydroxyl or epoxy as shown in Table 2.2. For example, castor oil contains 90% ricinoleic acid, which contains ne hydroxyl group on 12th carbon. Vernonia oil has a functionality of 2.8 epoxy rings per triglyceride. All these inherent functionalities offer the possibility of these renewable resources being transformed into industrially significant compounds (fatty acid, fatty amides/nitriles/amines, alcohols, ester derivatives, epoxy, conjugates) via several efficient chemical modifications. The basic oleochemicals are fatty acids (ca. 52%), the respective methyl esters (ca. 11%), amines (ca. 9%), and alcohols (ca. 25%) [1].

The world production of major vegetable oils has risen from 95 million tons in 2002/2003 to 154 million tons in 2012/2013 at an average rate just over 5 million tons a year [1]. Soybean oil, palm oil, and rapeseed oil range highest world production of vegetable oils. Most of the global soybean oil production is located in North America and South America.

Table 2.2	Six Common	Fatty A	cids Cor	nposition	in Vegetable	e Oils
-----------	------------	---------	----------	-----------	--------------	--------

-		7
Fatty Acids	Formula	Structure
Caprylic	$C_8 H_{16} O_2$	~~~~Соон
Palmitc	C ₁₆ H ₃₂ O ₂	~~~~~СООН
Stearic	C ₁₈ H ₃₆ O ₂	~~~~~ ^{Соон}
Oleic	C ₁₈ H ₃₄ O ₂	Соон
Linoleic	C ₁₈ H ₃₂ O ₂	Соон
Linolenic	C ₁₈ H ₃₀ O ₂	Соон
Ricinoleic	C ₁₈ H ₃₄ O ₃	ОН
α -Eleostearic	C ₁₈ H ₃₀ O ₂	~~~~Соон
Vernolic	C ₁₈ H ₃₂ O ₂	о Соон



Figure 2.2 Hydrolysis of vegetable oils.

For example, in the United States, the three largest crops planted are corn, soybean, and wheat with planted acreage of 30%, 28%, and 23%, respectively. However, rapeseed production is the major crop in Europe, while palm oil dominates in Asia. These three oils are the most attractive for large-scale industrial products due to low price. From 2001 to 2005, 15% of soybean oil was used for industry application [5]. Although the majority of vegetable oils are primarily produced for food and feed purposes, minor vegetable oils such as castor and linseed oil are almost solely used for industrial applications.

Recently, there has been a steady growth in the use of plant oils in the painting and coating industries, such as shampoos lubricants, emulsifiers, cosmetics, plasticizers, biodiesels and pharmaceuticals, as well as various polymeric materials.

2.2 Plant Oil-Based Derivatives

2.2.1 Fatty Acids

The hydrolysis or saponification of different vegetable oils under the influence of water, temperature, pressure, and catalyst produces a mixture of C_6 up to C_{20} fatty acids. Partial hydrolysis of triglycerides yields mono- and di-glycerides and fatty acids, while complete hydrolysis generates glycerol and fatty acids as shown in Figure 2.2. Generally, the reaction between plant oils and water is pretty slow due to the hydrophobic nature of plant oils. Increased temperature or pressures improve the solubility of plant oils in water, leading to rapid hydrolysis rates [6]. Also, organic acid (such as HCl, H_2SO_4) and basic (NaOH) can be used to accelerate the reaction. When basic is used, the final product is fatty acid sodium salts, which are used to make soaps, skin products, candles, and perfumes.

Also, fatty acids provide useful chemicals for polymers development. For example, castor oil fatty acids were used as ring opener for preparation of polyols for polyurethane coating and foam [4].

2.2.2 Fatty Amides/Nitriles/ Amines

Fatty amines can be obtained by hydrogenation of fatty nitriles, which are produced



Figure 2.3 Preparation of fatty amides/nitriles/amines.

by the reaction between triglycerides, or fatty acids, or fatty esters and ammonia with elimination of two molecules of water at high temperature (423-723 K) in the presence of dehydrating catalyst (such as silica gel, alumina, or iron-based catalysts, and so on) [6,7]. Actually, the products from the hydrogenation of nitriles are mixtures of primary, secondary, and tertiary amines as shown in Figure 2.3. Excess of ammonia used in the process would prohibit the release of the ammonia, resulting in suppressing formation of secondary and tertiary amines. The choice of the hydrogenation catalyst also affects the selectivity of the process. For example, nickel or cobalt catalyst favored the formation of primary amines, while copper chromite catalysts promoted the formation of secondary amines. Primary amines and their salts find applications as flotation agents, anticaking and water-repellent agent, corrosion inhibitors, lubricating additives, bactericides, or additives for fuels and gasoline [8,9]. They are also intermediates for the production of quaternary ammonium salts. The secondary amines are used mainly as intermediates for the preparation of difatty dimethyl quaternary ammonium compounds.

2.2.3 Alcohols

Alcohols are obtained by conversation of plant oils directly, called polyols, or by catalytic hydrogenolysis of fatty acids or fatty esters derived from them, called fatty alcohols. The former are usually used to prepare polyurethane coating, foam, and dispersions. The hydroxyl groups in polyols are usually introduced in the react sites of double– double bonds located in the chain backbone of plant oils. The common alcoholysis methods that are used to prepare these polyols are as shown below.

 Epoxidation/ring opening: Most vegetable oil-based polyols are produced by the epoxidation of vegetable oils, followed by oxirane ring opening with proton donors. Several proton donors can be used to ring open the epoxide, such as alcohol, acid.

- 2. Transesterification/amidation: Two common routes to synthesize vegetable oil-based polyols involve transesterification with various polyols, or amidation, typically with diethanolamine, of vegetable oils.
- 3. Hydroformylation/reduction: The hydroformylation/reduction of vegetable oils involves the addition of hydroxymethyl groups to the carbon–carbon double bonds in the fatty acid chains using a two-step procedure. In the first step, the vegetable oil is treated with syngas (a mixture of CO and H₂) in the presence of a Rh or Co catalyst to give aldehydes, which are then reduced with H₂ in the second step to yield hydroxyl groups.
- 4. Ozonolysis/reduction: This method involves the efficient cleavage of the carbon–carbon double bonds present in the fatty acid chains to give, under the right conditions, shorter chain alcohols. In the first step, an ozonide is formed by treating a vegetable oil with ozone, which is then reduced by zinc to give an aldehyde, and is further reduced with Raney nickel to the corresponding primary alcohol. The ozonide can also be treated with sodium borohydride to afford the primary alcohol directly.

Based on these routes, several research groups are dedicated to combining the advantages offered by the modification of carbon–carbon double bonds and those offered by the modification of esters to develop high functionality polyols. For example, Kong used 1, 3-propanediol for ring-opening of epoxidized vegetable oils and transesterification with the triglycerides [10]. Lligadas developed oligomeric polyether polyols by ring-opening polymerization of epoxidized methyl oleate followed by reduction of the ester groups [11]. Epoxidation of vegetable oils, followed by reduction of epoxy groups and esters, is one method developed by Zhang to prepare high functional polyols from castor oil, soybean oil, and linseed oil for polyurethane coating [3].

Fatty alcohols are produced from hydrogenolysis of fatty acids or fatty esters catalyzed by copper chromite, palladiumrhenium, or rhodium-tin systems with large excess of hydrogen due to its low solubility in the reaction mixture, which introduces mass transport resistance and limits the concentration of hydrogen at the surface of the catalyst [12]. However, these catalysts are not selective toward carboxyl group and carbon-carbon double bonds. It is reported that zinc chromite catalyst, Zn-Cr catalyst, zincaluminum, iron-zinc, and zinc chromite show chemoselective reduction toward carboxylic group, preserving the double bonds [6,13,14]. One example of chemoselective reduction of fatty esters is production of oleyl alcohols, which is classified by refined, technical, and industrial according to their purity as shown in the figure. The refined oleyl alcohols are used primarily in cosmetic and pharmaceutical preparations. The technical grade of oleyl alcohol is used as a chemical intermediate for the preparation of surfaceactive agents such as ethoxylates, sulphates, and esters. The industrial grades of oleyl alcohol and their derivatives are widely used in lubricants and as petroleum additives [15]. Fatty alcohols by hydrogenation of fatty acids and their methyl esters are also used for the production of surfactants [1]. With their hydrophilic head and hydrophobic tail, fatty alcohols interpose themselves between water and water-insoluble substances. Properties of surfactants are additionally governed by the alkyl chain length of their hydrophobic
Class	Hydroxyl Value	Iodine Value	Cloud Point
Refinded	205–215	85–95	<10
Technical	205–220	80–90	<15
Industrial	205–225	65–75	<28

Table 2.3 Commercial Class of Oleyl Alcohol [15]

part: wetting agents (C8–C10), detergents (C12–C16), emulsifiers, and softeners (C18–C22) [16] (see Table 2.3).

The rapid emergence of biodiesel industry causes mass production of glycerol as an organic byproduct (a 10% byproduct of biodiesel production via the transesterification of vegetable oils), which attracts attention from both academia and industry to find application for this cheap renewable alcohol. With its three OH groups, several chemical pathways, such as oxidation, reduction, dehydration, esterification, etherification, were developed to transform it into a large number of chemicals as shown in Figure 2.4. For example, oxidation toward primary and secondary OH in glycerin using stoichiometric mineral acid or fermentation routes produce glyceric acid, dihydroxyacetone, and hydroxypyruvic acid, which find application in medicine, cosmetics [6]. The



Figure 2.4 Possible application and transformation of glycerol [6].



Figure 2.5 (a) Mo-catalyzed reduction of sulfoxides (1 mmol) with crude glycerol (2.4 g, 5–8% purity), (b) recycling study of the Mo-catalyzed reduction of bis(p-tolyl) sulf-oxide with glycerol [18].

transesterification of dimethyl carbonate with glycerol in the presence of tetra-*n*-butylammonium bromide at 120°C leads to glyceryl carbonate, which is employed as a solvent, additive, and chemical intermediate in the synthesis of polyurethane and glycidol [17].

Recently, a new application for glycerol has been found that can be used as a reducing agent and solvent in Mo-catalyzed deoxy-genation of sulfoxides, which conventionally usually used phosphorous(III) reagents or silanes [18]. This green reaction medium allows high yields, easy product isolation, and catalyst recycling at least five times with negligible change in its activity (see Figure 2.5).

Glycerol has also been investigated to be a solvent for separation, a solvent for biocatal-

ysis, a solvent for enhancing reaction selectivity, and a promoting solvent for organic synthesis water due to its low toxicity, low price, large availability, renewability high boiling point, and low vapor pressure. The author claims that glycerol is an available green solvent that affords innovative solutions to the substitution of the conventionally used volatile organic solvents [19].

2.2.4 Ester Derivatives

Vegetable oils, esters by itself formed between glycerin and different fatty acids, are one of the most important renewable raw materials for a sustainable chemistry. To meet

Table 2.4 Viscosity of Various Vegetable Oils Compared to Diesel Fuel [25]



Figure 2.6 Transesterification of triglycerides to produce fatty acid methyl esters.

need for specific application, several ester derivatives have been obtained from vegetable oils. For example, fatty acid monoglycerides and diglycerides are obtained by glycerinolysis of triglycerides with an excess of glycerol at temperatures of 200-250°C in the presence of base catalysts, such as KOH or Ca(OH), under an inert gas atmosphere [20]. The final products are usually mixture of 35-60% of monoglycerides and 35-50% of diglycerides. The separation can be achieved by distillation due to different emulsifying ability, but it is very expensive. Thus, these mixtures of monoglyceride and diglyceride are used as emulsifiers in the foodstuffs industry and cosmetics [21,22].

With the depleting of petroleum resources, there is increased interest in developing vegetable oil-based alternative fuel for diesel engines. However, due to the higher viscosity (see Table 2.4) and tendency to polymerize of vegetable oil, it does not atomize properly for combustion during injection into the cylinder, leading to failure of the lubricating oil, and may ruin the engine [25]. Transesterfication of vegetable oils into methyl and ethyl esters (Figure 2.6) or blending with diesel fuels reduces the problems encountered to produce biodiesel with several outstanding advantages among other new renewable and clean engine fuel alternatives. The structure of vegetable oilbased ester can be tailored by transesterification variables, such as ratio of glycerides to alcohol, the catalyst-type concentration, reaction temperature and pressure, the reaction time and the free fatty acids, and purity of the reactants [24,25]. For example, Clark has employed starbons-based catalysts (S400-SO₂H) to convert waste oils containing high free fatty acid into fatty acid methyl esters, a biodiesel-like fuel under both conventional and microwave heating (Figure 2.7) [23]. It is found that the Starbons acids initiate both esterification of fatty acids with methanol and triglyceride transesterification. The yield of fatty acid methyl esters was reported >95% (Figure 2.7a–c). Reactions rates could be increased by employing microwave, reducing reaction times from >12 h (using conventional heating) to <45 min, without a reduction in



Figure 2.7 Activity of starbon-400SO₃H in the esterification/transesterification of waste oils rapeseed waste frying oil (10 wt% free fatty acids) (left); sunflower + olive waste frying oil (17 wt% free fatty acids) (right). (a) Under conventional heating at 80°C. (b) Under 300 W microwave irradiation at 80–85°C. (c) Activity comparison of different solid acids in the production of fatty acid methyl esters. (d) Comparison of recycle uses of starbons acid under conventional heating and MW [23].

activity. The Starbons-based catalyst maintained high fatty acid methyl esters conversions even after five recycle runs (>70 mol%), with the use of MW heating proving to be beneficial in terms of prolonging the high activity of the catalyst over a high number of catalytic runs (Figure 2.7) [25]. The transesterification of fatty acid esters with sucrose produces fatty acid derivative as surfactants that are used in applications such as cosmetics, cleaning, and washing [26]. Similar surfactants can be prepared by the condensation of glycosides with a fatty alcohol and esterification of sorbitol with fatty acid [27]. The transesterification of the vegetable oil with other polyols, such as trimethylolpropane, neopentyl glycol, or pentaerythritol polyols, produces a new generation of hydraulic fluids with many advantages and provides longevity comparable to that of petroleum-based fuel at moderate temperature [28,29].

2.2.5 Epoxy Derivatives

Epoxidation can be considered a common and useful method to transform the nature feedstock to significant industry compounds. The carbon–carbon double bonds on the fatty acid chains provide the reactive sites for epoxidation by chemical routine and enzymatic routine [30,31].

The most preferred epoxidation of vegetable oils is Prileschajew reaction, which is conducted in the presence of peracids (peracetic or performic acid) due to ready availability and low price [30,32]. The degree of epoxidation can be controlled by reaction ratios of starting materials, reaction time, and reaction temperatures. Acetic/peracetic acid is largely used because it leads to lower ringopening than formic/performic acid. However, this method is not selective and the acid used would cause corrosion of the equipment as well as residue issue.

Thus, the heterogenization of such processes using solid catalysts is of great interest from industrial and environmental points of view (see Figure 2.8). Transition metal complexes such as Mo (VI), rhenium, oxomolybdenum are among the options. Maritana investigated the epoxidation of soybean oil with bis(acetyl-acetonato)dioxomolybdenum (VI) [MoO₂(acac)₂] as a catalyst in the presence of *tert*-butyl hydroperoxide as an oxidizing agent. The highest yield of epoxidation was $54.1 \pm 0.9\%$ at 110° C for 2 h. It also found that the unsaturation in the middle fatty acids chains was less susceptible to epoxidation in comparison with terminal double bonds of short chain alkenes [33,34].

Chemo-enzymatic epoxidation is another interesting route for the synthesis of epoxidized fatty acid derivatives because it avoids the ring-opening of epoxide [31]. Warwel reported that an immobilized lipase from *Candida antarctica* (Novozym 435) first converts unsaturated carboxylic acids to percarboxylic acids, followed by epoxidation themselves by intermolecular oxygen transfer (see Figure 2.9)



Figure 2.8 Mechanism of epoxidation of carbon double bonds by hydroperoxide in the presence of a transition metal catalyst.



Figure 2.9 Epoxidation of fatty acids by lipase.



Figure 2.10 Perhydrolysis of plant oils.



Figure 2.11 Three interconnected equilibria in epoxidation of plant oils by lipase [36].

with high yields of 72–91% after 16 h. Furthermore, the lipase is recovered and reused 15 times to produce epoxy-stearic acid on a multigram scale [35].

When a triglyceride is treated with hydrogen peroxide in the presence of Novozym 435, perfatty acids are formed by the perhydrolysis of plant oils (see Figure 2.10) and three interconnected equilibria (see Figure 2.11) existed in reaction mixture, resulting in epoxidized triglycerides containing small amounts of epoxidized free fatty acids and mono- and diglycerides. The formation of these side products could be avoided by simply adding a small amount of free fatty acids before reaction, and high yields can be obtained (rapeseed oil: 91%, sunflower oil: 88%, linseed oil: 80%) [36].

Epoxidized vegetable oils and epoxidized fatty acid are largely used as plasticizers, nonmetallic stabilizer, and antioxidants for PVC as they offer ease of processing and improve flexibility [37]. The epoxides react with hydrogen chloride released from the PVC resins under the prolonged action of light and heat, restoring the labile chlorine atoms back into the polymer chains (see Figure 2.12) [38]. Also, the epoxides might act as a polyene blocker, opening by reaction at a labile hydrogen site in the PVC to form an ether linkage with the chlorinated epoxy [39]. These reactions prevent PVC from further dehydrochlorination, preserving its color and limiting loss in plasticized PVC properties at high temperatures. Bouchareb found that the tensile strength of PVC decreases gradually with the content of epoxidized sunflower oil, whereas the elongation at break increases. However, increasing the degree of epoxidation of epoxidized sunflower oil will increase in tensile strength and decrease in elongation at break. The authors also pointed out that the plasticizing effect of epoxidized sunflower oil was higher than that of epoxidized soybean oil because of its higher compatibility with PVC [39]. Bueno-Ferrer evaluated the use of epoxidized soybean oil for PVC stabilization in commercial lids, and found that the thermal stability of PVC was highly dependent on the plasticizer concentration and a significant increase in thermal degradation temperature with increased content of epoxidized soybean oil, permitting their use in food processing at high temperatures [38].

Epoxy vegetable oils are also preferred lubricants, not only because they acquire most of the properties required for lubricants (such as high index viscosity, low volatility, and good lubricity), but also because they overcome the disadvantage of poor



Figure 2.12 (a) Scheme of the reaction of the oxirane ring in epoxidized soybean oil with HCI (PVC) inhibiting the thermal degradation process of PVC. (b) Plasticizers found in the gaskets of commercial lids, DBS-dibutyl sebacate, DOA-dioctyl adipate, ATBC-acetyl tributyl citrate. (c) Initial degradation temperatures (95%) of raw materials, cured plastiso, and gaskets for commercial lids [38].

oxidative and thermal stability of raw vegetable oils. For example, Wu reported that the epoxidized rapeseed oil has good friction-reducing and extreme pressure abilities, because the three-member ring of oxirane could form polyester or polyether material due to tribo-polymerization, which is tribologically effective to reduce friction [40].

2.2.6 Conjugates

Vegetable oils that contain fatty acids with conjugated double bonds are valuable drying agents in paints, varnishes, and inks. In conjugated fatty acids, cross-linking reactions occur by the action of atmospheric oxygen in room temperature when oxygen directly attacks the conjugated double bonds and forms free radicals. Although the fatty acid chains in triglyceride molecules contain a varied number of carbon-carbon double bonds, only few vegetable oils naturally contain conjugated carbon double bonds, such as Tung oil, which were of the prime interest to fine wood finishes. Most of these double bonds in fatty acid chains are not reactive enough to produce viable materials by free radical or cationic polymerization, in which double bonds are separated by methylene group, commonly referred to as being "methylene interrupted." Castor oil has only one carbon-carbon double bond in each fatty acid chain, and thus is cataloged as nondrying oil. However, it can be dehydrated at about 250°C and in the presence of catalysts (e.g., concentrated sulphuric acid, activated earth) and under an inert atmosphere or vacuum, forming chemicals containing conjugated double for making semidrying or drying oil resembling Tung oil, which is used extensively in paints and varnishes [41].

Isomerization of nonconjugated double bonds in fatty acid into conjugated counterparts has been regarded as a useful method to render vegetable oils new application for coating and paints application. Krompiec investigated the effect of different ruthenium complexes ([RuHCl(CO)(PPh₃)₃) as catalyst on the isomerization of rapeseed, soybean, sunflower, and linseed oils. The conversions of conjugation were 30, 54, 54, and 50 wt% for rapeseed, soybean, sunflower, and linseed oils. However, high isomerization temperature (212–226°C) leads to 10 wt% polymerization of triglyceride oils [42]. Sleeter reduced isomerization temperature to 180°C for linseed and soybean oils with ruthenium catalysts in the presence of formic acid [43]. Larock found that the yields of homogeneous isomerization of vegetable oils can be increased to high yields (95%) using [RhCl(C₈H₁₄)₂]₂ as catalyst in 60°C for 24 h [44].

2.2.7 Other Derivatives

Vegetable oil is a preferred carbon precursor with a low H content for the production of high-quality CNT production. Spray pyrolysis-assisted CVD is often used for the synthesis of CTN for these liquid bio-resources, in which vegetable oils are vaporized at low temperature on a furnace, and then transported by a carrier gas into a second furnace for pyrolysis and deposition at higher temperature with Fe/Co zeolite catalysts. The typical spray pyrolysis setup is shown in Figure 2.13.



Figure 2.13 Spray pyrolysis setup for the synthesis of aligned CNTs [46].



Figure 2.14 SEM images of CNT from palm oil on a Si substrate: (a) peeled off bundle of CNT, (b) side view of the aligned CNTs [49].

Turpentine oil [45], neem oil [46], coconut oil [47], eucalyptus oil [48], and palm oil [49] have been used as the carbon source for CNTs production with variable amorphous and diameters. In a spray pyrolysis process with carefully designed Fe zeolite catalysts and palm oil as carbon source, Suriani synthesized a bundle of mixed multiwalled and single-walled carbon nanotubes (CNTs) of about 90% purity, in which diameters of the single-walled carbon nanotubes were estimated to be 0.6–1.2 nm as shown in Figure 2.14 [49]. By controlling the growth temperature and concentration of catalyst, Paul prepared multiwalled carbon nanotubes with diameters in the range of 80–90 nm [47]. Tiwari reported a scalable synthesis of aligned multiwalled carbon nanotubes bundles with length in the range of 20–50 μ m from neem oils by spray pyrolysis at 825°C. The diameters of multiwalled CNTs were found to be 15–30 nm [46].

2.3 Conclusions

With the advance of science and technology in the last decade, many important chemical compounds have been developed by modern synthetic methods, enzymatic and microbiological methods. Some of them have been employed as substitution of petroleumbased counterparts to benefit the whole society. And others are still in the laboratory due to the higher price or lower properties. However, with the breeding of new oil plants including the use of gene technology, more and new oil plants containing fatty acids with interesting desired properties and high yields will be introduced and cultivated for chemical utilization. Also, academia and industry are struggling to develop novel technology to improve conversion, and especially selectivity, low the coast to make possible a commercial application of those processes. In the coming years, the utilization of plant oils as renewable raw materials in the chemical industry is continued to be a challenge and direction for research and development.

References

- U. Biermann, U. Bornscheuer, M.A.R. Meier, J.O. Metzger, H.J. Schafer, Oils and fats as renewable raw materials in chemistry, Angew. Chem. Int. Ed. 50 (17) (2011) 3854–3871.
- [2] J.O. Metzger, U. Bornscheuer, Lipids as renewable resources: current state of chemical and biotechnological conversion and diversification, Appl. Microbiol. Biot. 71 (1) (2006) 13–22.
- [3] C.Q. Zhang, R. Ding, M.R. Kessler, Reduction of epoxidized vegetable oils: a novel method to prepare bio-based polyols for polyurethanes, Macromol. Rapid. Commun. 35 (11) (2014) 1068–1074.
- [4] C.Q. Zhang, Y. Xia, R.Q. Chen, S. Huh, P.A. Johnston, M.R. Kessler, Soy-castor oil based polyols prepared using a solvent-free and catalyst-free method and polyurethanes therefrom, Green Chem. 15 (6) (2013) 1477–1484.

- [5] M.N. Belgacem, A. Gandini, Monomers, polymers and composites from renewable resources foreword, Monomers Polymers Composites Renewable Resources (2008) V–V10.
- [6] A. Corma, S. Iborra, A. Velty, Chemical routes for the transformation of biomass into chemicals, Chem. Rev. 107 (6) (2007) 2411–2502.
- [7] A. Biswas, B.K. Sharma, J.L. Willett, S.Z. Erhan, H.N. Cheng, Soybean oil as a renewable feedstock for nitrogencontaining derivatives, Energ. Environ. Sci. 1 (6) (2008) 639–644.
- [8] S.H. Feairheller, R.G. Bistline, A. Bilyk, R.L. Dudley, M.F. Kozempel, M.J. Haas, A novel technique for the preparation of secondary fatty amides. 3. Alkanolamides, diamides and aralkylamides, J. Am. Oil Chem. Soc. 71 (8) (1994) 863–866.
- [9] L.I. Kopylov, Y.N. Shekhter, A.A. Gureev, M.B. Bakaleinikov, Physicochemical and functional-properties of mixtures of Akor-1 inhibitor and alkanolamides, Chem. Tech. Fuels Oil+ 16 (1–2) (1980) 62–64.
- [10] X.H. Kong, G.G. Liu, H. Qi, J.M. Curtis, Preparation and characterization of high-solid polyurethane coating systems based on vegetable oil derived polyols, Prog. Org. Coat. 76 (9) (2013) 1151–1160.
- [11] G. Lligadas, J.C. Ronda, M. Galia, U. Biermann, J.O. Metzger, Synthesis and characterization of polyurethanes from epoxidized methyl oleate based polyether polyols as renewable resources, J. Polym. Sci. Pol. Chem. 44 (1) (2006) 634–645.
- [12] T. Voeste, H. Buchold, Production of fatty alcohols from fatty-acids, J. Am. Oil. Chem. Soc. 61 (2) (1984) 350–352.

- [13] Y. Pouilloux, F. Autin, A. Piccirilli, C. Guimon, J. Barrault, Preparation of oleyl alcohol from the hydrogenation of methyl oleate in the presence of cobalt-tin catalysts, Appl. Catal. A Gen. 169 (1) (1998) 65–75.
- [14] K. De Oliveira, Y. Pouilloux, J. Barrault, Selective hydrogenation of methyl oleate into unsaturated alcohols in the presence of cobalt-tin supported over zinc oxide catalysts, J. Catal. 204 (1) (2001) 230–237.
- [15] R.R. Egan, G.W. Earl, J. Ackerman, Properties and uses of some unsaturated fatty alcohols and their derivatives, J. Am. Oil Chem. Soc. 61 (2) (1984) 324–329.
- [16] J.R. Falbe, Surfactants in Consumer Products: Theory, Technology, and Application, Springer-Verlag, Berlin, New York, (1987) p xxiii, 547 p.
- [17] R.A. Grey, Preparation of cyclic carbonates using alkylammonium and tertiary amine catalysts, US5091543, (1992).
- [18] N. Garcia, P. Garcia-Garcia, M.A. Fernandez-Rodriguez, D. Garcia, M.R. Pedrosa, F.J. Arnaiz, R. Sanz, An unprecedented use for glycerol: chemoselective reducing agent for sulfoxides, Green Chem. 15 (4) (2013) 999–1005.
- [19] Y.L. Gu, F. Jerome, Glycerol as a sustainable solvent for green chemistry, Green Chem. 12 (7) (2010) 1127–1138.
- [20] A.E. Bailey, D. Swern, Bailey's Industrial Oil and Fat Products, 3d ed., Interscience Publishers, New York, (1964) p xiii, 1103 p.
- [21] J.L. Vanhaften, Fat-based food emulsifiers, J. Am. Oil Chem. Soc. 56 (1979) A831–A835.
- [22] H. Maag, Fatty-acid derivatives important surfactants for household, cosmetic and industrial purposes, J.

Am. Oil Chem. Soc. 61 (2) (1984) 259–267.

- [23] R. Luque, J.H. Clark, Biodiesel-likebiofuels from simultaneous transesterification/ esterification of waste oils with a biomass-derived solid acid catalyst, Chemcatchem 3 (3) (2011) 594–597.
- [24] A. Srivastava, R. Prasad, Triglyceridesbased diesel fuels, Renew. Sust. Energ. Rev. 4 (2) (2000) 111–133.
- [25] H.J. Harwood, Oleochemicals as a fuel: mechanical and economic-feasibility, J. Am. Oil Chem. Soc. 61 (2) (1984) 315–324.
- [26] D.B. Sarney, M.J. Barnard, DA Mac-Manus, E.N. Vulfson, Application of lipases to the regioselective synthesis of sucrose fatty acid monoesters, J. Am. Oil Chem. Soc. 73 (11) (1996) 1481–1487.
- [27] R. Gulati, P. Arya, B. Malhotra, A.K. Prasad, R.K. Saxena, J. Kumar, A.C. Watterson, V.S. Parmar, Novel biocatalytic esterification reactions on fatty acids: synthesis of sorbitol 1(6)-monostearate, Arkivoc (2003) 159–170.
- [28] T. Mang, W. Dresel, Lubricants and Lubrications, Wiley-VCH, Weinheim, New York, Chichester, (2001) p xxxix, 759 p.
- [29] T. Mang, K. Bobzin, T. Bartels, Industrial Tribology: Tribosystems, Friction, Wear and Surface Engineering, Lubrication, Wiley-VCH, Weinheim, (2011) p xxvii, 644 p.
- [30] A. Kockritz, A. Martin, Oxidation of unsaturated fatty acid derivatives and vegetable oils, Eur. J. Lipid Sci. Tech. 110 (9) (2008) 812–824.
- [31] H. Uyama, M. Kuwabara, T. Tsujimoto, S. Kobayashi, Enzymatic synthesis and curing of biodegradable epoxidecontaining polyesters from renewable resources, Biomacromolecules 4 (2) (2003) 211–215.

- [32] A.S.A Hazmi, M.M. Aung, L.C. Abdullah, M.Z. Salleh, M.H. Mahmood, Producing Jatropha oil-based polyol via epoxidation and ring opening, Ind. Crop. Prod. 50 (2013) 563–567.
- [33] M. Farias, M. Martinelli, D.P. Bottega, Epoxidation of soybean oil using a homogeneous catalytic system based on a molybdenum (VI) complex, Appl. Catal. A Gen. 384 (1–2) (2010) 213–219.
- [34] M. Farias, M. Martinelli, G.K. Rolim, Immobilized molybdenum acetylacetonate complex on montmorillonite K-10 as catalyst for epoxidation of vegetable oils, Appl. Catal. A Gen. 403 (1–2) (2011) 119–127.
- [35] S. Warwel, M.R.G. Klaas, Chemoenzymatic epoxidation of unsaturated carboxylic acids, J. Mol. Catal. B Enzym. 1 (1) (1995) 29–35.
- [36] M.R.G. Klaas, S. Warwel, Chemoenzymatic epoxidation of unsaturated fatty acid esters and plant oils, J. Am. Oil Chem. Soc. 73 (11) (1996) 1453–1457.
- [37] Z.S. Petrovic, A. Zlatanic, C.C. Lava, S. Sinadinovic-Fiser, Epoxidation of soybean oil in toluene with peroxoacetic and peroxoformic acids: kinetics and side reactions, Eur. J. Lipid Sci. Tech. 104 (5) (2002) 293–299.
- [38] C. Bueno-Ferrer, M.C. Garrigos, A. Jimenez, Characterization and thermal stability of poly(vinyl chloride) plasticized with epoxidized soybean oil for food packaging, Polym. Degrad. Stabil. 95 (11) (2010) 2207–2212.
- [39] B. Bouchareb, M.T. Benaniba, Effects of epoxidized sunflower oil on the mechanical and dynamical analysis of the plasticized poly(vinyl chloride), J. Appl. Polym. Sci. 107 (6) (2008) 3442–3450.
- [40] X.D. Wu, X.G. Zhang, S.R. Yang, H.G. Chen, D.P. Wang, The study of epoxi-

dized rapeseed oil used as a potential biodegradable lubricant, J. Am. Oil Chem. Soc. 77 (5) (2000) 561–563.

- [41] D.S. Ogunniyi, Castor oil: vital industrial raw material, Bioresource Technol. 97 (9) (2006) 1086–1091.
- [42] S. Krompiec, R. Penczek, M. Krompiec, T. Pluta, H. Ignasiak, A. Kita, S. Michalik, M. Matlengiewicz, M. Filapek, Transition metals for conjugation of polyunsaturated acids and their esters, Curr. Org. Chem. 13 (9) (2009) 896–913.
- [43] R.T. Sleeter, Method of conjugating double bonds in drying oils, 1998.
- [44] R.C. Larock, X.Y. Dong, S. Chung, C.K. Reddy, L.E. Ehlers, Preparation of conjugated soybean oil and other natural oils and fatty acids by homogeneous transition metal catalysis, J. Am. Oil Chem. Soc. 78 (5) (2001) 447–453.
- [45] P. Ghosh, T. Soga, R.A. Afre, T. Jimbo, Simplified synthesis of single-walled carbon nanotubes from a botanical hydrocarbon: turpentine oil, J. Alloy Compd. 462 (1–2) (2008) 289–293.
- [46] R. Kumar, R.S. Tiwari, O.N. Srivastava, Scalable synthesis of aligned carbon nanotubes bundles using green natural precursor: neem oil, Nanoscale Res. Lett. 6 (2011).
- [47] S. Paul, S.K. Samdarshi, A green precursor for carbon nanotube synthesis, New Carbon Mater. 26 (2) (2011) 85–88.
- [48] P. Ghosh, R.A. Afre, T. Soga, T. Jimbo, A simple method of producing singlewalled carbon nanotubes from a natural precursor: eucalyptus oil, Mater. Lett. 61 (17) (2007) 3768–3770.
- [49] A.B. Suriani, A.A. Azira, S.F. Nik, R.M. Nor, M. Rusop, Synthesis of vertically aligned carbon nanotubes using natural palm oil as carbon precursor, Mater. Lett. 63 (30) (2009) 2704–2706.

Page left intentionally blank

Thomas F. Garrison and Michael R. Kessler

Department of Mechanical and Materials Engineering, Washington State University, Pullman, WA, USA

Ο U T L I N E

3.1	Polyurethane Chemistry						
3.2	Plant Oil-Based Polyurethanes						
3.3	Developing New Sources of Vegetable Oils						
3.4	Polyol Methods 4						
	3.4.1	Epoxidation/Ring-Opening	44				
	3.4.2	Ozonolysis	44				
	3.4.3	Amidation	44				
3.5	Flame	e Retardant Polyols	45				
3.6	Alterr	native Routes for Making Polyurethanes	45				
3.7	Applications of Plant Oil-Based Polyurethanes						
3.8	Foam	s	46				
	3.8.1	Flexible Foams	47				
	3.8.2	Rigid Foams	47				
3.9	Coati	ngs	47				
	3.9.1	Waterborne Dispersions	48				
	3.9.2	Antimicrobial Coatings	48				
3.10	.10 Shape Memory Polymers						
3.11	11 Conclusions 49						
Ref	eferences 49						

This chapter provides a brief overview of polyurethane chemistry and plant oil-based polyols. For some readers, this may be their first introduction to plant oil-based polyurethanes, while for others, this will be a review of the current literature. The primary objective of this chapter is to highlight the newest developments on plant oil-based polyurethanes rather than provide a comprehensive account of the field. In the past several years, a number of excellent reviews focused on this subject have been published. In 2008, Petrovic wrote a comprehensive review on vegetable oil-based polyurethanes [1]. Three years later, Pfister et al. reviewed recent progress in the field [2]. More recently, Desroches et al. wrote a comprehensive review on industrial products made from vegetable oil-based polyurethanes [3]. Many of the references in this chapter will highlight current research trends in plant oil-based polyurethanes since these reviews were published.

3.1 Polyurethane Chemistry

Polyurethanes are polymers which are typically made by reacting isocyanate groups from polyisocyanates with hydroxyl groups from polyols to form urethane bonds (see Scheme 3.1). Polymer chain extension occurs when polyols and isocyanates with a functionality of two (e.g., diols and diisocyanates) are reacted together. It should be noted that functionalities greater than two generate polymer crosslinking. Typically, polyurethanes are formulated with a slight excess of isocyanates with water [4].

Isocyanates are highly reactive functional groups, which can react with a number of other functional groups including amines, carboxylic acids, ureas, and other isocyanate groups. Reactions of isocyanates with water and amines are discussed later in the section on polyurethane foams, while the other reactions are outside the scope of this chapter. Delebecq et al. wrote an excellent review on isocyanate chemistry, which also covers the development of blocked isocyanates [5].

Polyurethanes are versatile materials because the chemical structure of the isocyanate and polyol starting materials can be readily changed. Isocyanates can either be aromatic or aliphatic. Aromatic isocyanates have better thermal and flame retardant properties due to the stability of the aromatic rings and are used extensively in polyurethane foams. Common aromatic isocyanates used industrially include toluene diisocyanate (TDI), methylene diphenyl diisocyanate (MDI), and polymeric MDI. In contrast, aliphatic isocyanates are often used in coatings applications because they are more resistant to yellowing under UV exposure [6]. Examples of aliphatic diisocyanates include isophorone diisocyanate (IPDI), hexamethylene diisocyanate (HDI), and bis(4-isocyanatocyclohexyl)methane ($H_{12}MDI$) (Figure 3.1). Pentamethylene diisocyanate (PDI), a new bio-based aliphatic isocyanate designed for polyurethane coatings, is expected to be made commercially available in 2015 [7].

3.2 Plant Oil-Based Polyurethanes

In recent years, considerable research has focused on the replacement of petroleumbased starting materials for polyurethanes





Figure 3.1 Structures of common diisocyanates.

with plant oil-based polyols [8]. Most plant oil-based polyols are made from triglycerides, although fatty acids and fatty acid derivatives can be used to make polyols. Triglycerides consist of three fatty acid chains connected to a glycerin backbone via ester linkages (see Figure 3.2). The most common fatty acids range in length from 16 to 20 carbons and vary in unsaturation from 0 to 3 double bonds (see Figure 3.3) [9]. Examples of saturated fatty acids include palmitic acid $(C16:0)^1$, and stearic acid (C18:0). Common unsaturated acids include oleic acid (C18:1), linoleic acid (C18:2), and linolenic acid (C18:3). Overall unsaturation of triglycerides is determined by the relative composition of fatty acids and is typically measured by iodine values². Table 3.1 lists typical compositions for common plant oils.

¹The convention for naming fatty acids is to list the number of carbons in the fatty acid chain followed by the number of double bonds. For example, palmitic acid (C16:0) has 16 carbon atoms and 0 double bonds.

²Iodine value (IV) or iodine number is a measure of the degree of unsaturation and is determined by the amount of iodine absorbed per gram of sample. Iodine values are typically expressed in terms of centigrams of iodine per gram of sample (weight percent of absorbed iodine) [10]. Higher iodine values indicate a greater amount of unsaturation.



Figure 3.3 Most common fatty acids found in plant oils.

In fatty acids found in most plant oils, carboxylic acid groups and unsaturated carboncarbon double bonds are the only naturally occurring functional groups. Three notable exceptions are ricinoleic acid, lesquerolic acid, and vernolic acid (see Figure 3.4). Ricinoleic acid, which makes up about 87% of castor oil, contains a hydroxyl group on the C12 position and a carbon-carbon double bond between C9 and C10. Lesquerolic acid, a component of lesquerella oil, is similar to castor oil because it has a hydroxyl group (C14) and a carbon-carbon double bond (between C11 and C12). With 20 carbons, lesquerella acid is slightly longer than the 18 carbon ricinoleic acid. Vernolic acid, the major component of vernonia oil, contains an oxirane (epoxy group) between the C12 and C13 carbon atoms and a double bond between C9 and C10.

3.3 Developing New Sources of Vegetable Oils

Abundant vegetable oils such as soybean oil, castor oil, and corn oil have already found widespread use in plant oil-based polyurethanes. Palm oil, despite is abundance, has found relatively limited use in making polyols

	Fatty Acid Composition (%)						lodine	
Oil	C12:0	C14:0	C16:0	C18:0	C18:1	C18:2	C18:3	Value
Castor ^b	-	-	1–2	0.9–2	2.9–6	3–5	0–0.5	81–91
Corn	0–0.3	0–0.3	9.2–16.5	0–3.3	20–42.2	39.4–65.6	0.5–1.5	107–135
Linseed	-	-	5.7–7	3–4	20–20.3	17–17.3	52–54	170–203
Olive	-	0–0.1	7.5–20	0.5–5.0	55–83	3.5–21	0–1.5	75–94
Palm ^c	0-0.4	0.5–2.0	40–48	3.5–6.5	36–44	6.5 – 12.0	0–0.5	49–55
Palm Kernal ^c	40–55	14–18	6.5–0.3	1.3–3	12–21	1–3.5	0–0.7	14–21
Peanut	0-0.1	0-0.1	8.3–4.0	1.9–4.4	36.4–67.1	14.0–43.0	0–0.1	83–107
Rapeseed	0.1	0.2	1.5–6	0.5–3.1	8–60	11–23	5–13	94–120
Soybean	0–0.1	0-0.2	9.7–13.3	3.0–5.4	17.7–28.5	49.8–57.1	5.5–9.5	118–139

Table 3.1 Fatty Acid Composition and Iodine Values of Common Plant Oilsª

^aCompiled information from Physical and Chemical Characteristics of Oils, Fats, and Waxes.

^bCastor oil is composed of 88% ricinoleic acid.

°Elaeis guineensis.



Figure 3.4 Structures of naturally occurring hydroxylated and epoxidized fatty acids.

for polyurethanes due to low unsaturation [2]. Seed oil from flowering plants such as rapeseed [11], sunflower (*Helianthus annuus* L.) [12], and *Camelina sativa* (a member of the mustard family) [13] is all being explored as sustainable sources of plant oil-based polyols. Algal oils are also a potential source of plant oil-based polyols, although there have been relatively few studies on using algal oil to make polymers [14]. There is significant interest in using a wide variety of underutilized plant oils from tree nuts and fruit seeds found in developing regions of the world as sustainable local or regional sources of polyols. For example, *Mesua ferrea* L., commonly known as nahar, is a tree found in India and other parts of Asia that has seeds with triglyceride contents of up to 70%. Nahar seed oil has been used to make polyols via transesterification

with glycerol [15]. Jatropha (Jatropha curcas L.), which originated in the Americas, is now being grown globally for its seed oil as a source of biodiesel [16]. Jatropha oil-based polyols have been prepared from crude jatropha oil using conventional epoxidation and ring opening [17]. There was also a recent study that used seed oil from Prosopis juliflora, a shrub or small tree also known as mesquite, to make polyols for polyurethanes [18]. Yellow passion fruit (Passiflora edulis Sims f. flavicarpa Degener), a native plant in South America, is widely cultivated for its juice that generates large amounts of seeds as waste by-products. Passion fruit seed oil contains high amounts of unsaturated linoleic acid (73%) that can be used to make plant oil-based polyols [19]. Recent studies have investigated the synthesis of passion fruit oil-based polyols for making polyurethane foams [20].

3.4 Polyol Methods

Castor oil can be used to make polyurethanes without additional processing because it has an average hydroxyl functionality of 2.7 per triglyceride due to its high content of ricinoleic acid. Lesquerella oil naturally contains hydroxyl groups, but its potential for commercial applications is hampered by limited cultivation. All other plant oils require conversion of either carbon–carbon double bonds or the ester groups into hydroxyls. Polyols can be formed from double bonds by epoxidation/oxirane ring opening, ozonolysis/reduction, and hydroformylation/ reduction (see Scheme 3.2). Examples of ester group modifications include transesterification with a polyol and amidation with an amine polyol (see Scheme 3.2). Recent developments for several of these methods are discussed further below.

Polyol structures formed by the different methods can play a significant role in determining reactivity of the polyols, as well as the material properties of the resulting polyurethanes. For example, primary hydroxyl groups are more reactive with isocyanates than secondary hydroxyls. Ozonolyis/reduction, hydroformylation/reduction, transesterification, and amidation are routes that form primary hydroxyls, whereas epoxidation/oxirane ring opening generates secondary hydroxyls. Hydroformylation inserts an additional carbon atom into the polyol structure, while ozonolysis yields polyols with decreased molecular weights. With the exception of polyols formed by ozonolysis, most plant oil-based polyols contain dangling chains that can act as internal plasticizers. The hydroxyl functionality of plant oil-based polyols - the average number of hydroxyl groups per molecule – can vary based on the route of polyol formation, the unsaturation of the plant oil, and the extent of reaction.

Molecular weight and hydroxyl functionality are two of the most important factors that control the overall properties of polyurethanes. In practice, hydroxyl (OH) numbers are more commonly used by formulators than functionalities because hydroxyl numbers correlate the hydroxyl content per mass of polyol sample. Hydroxyl numbers are easily determined by wet analytical techniques³. Equivalent weight of polyol,

³Hydroxyl numbers are determined by following a standard analytical procedure, such as ASTM D1957. In these procedures, the hydroxyl groups are first acetylated and then titrated with potassium hydroxide. The units are given by milligrams KOH per gram polyol.



Scheme 3.2 Common routes to vegetable oil-based polyols.4

another important term, is the average molecular weight of the polyol divided by the average hydroxyl functionality. Relationships between molecular weight, functionality, hydroxyl number, and equivalent weights⁵ are shown in the following equations [21]:

$$Functionality = \frac{Molecular Weight}{Equivalent Weight}$$

⁴Adapted from a reaction scheme by Pfister et al. [2].

⁵In these equations, 56.1 is the molecular weight of potassium hydroxide, while 1000 is the conversion between milligrams and grams.

Equivalent Weight = $\frac{56.1*1000}{\text{OH Number}}$

 $Functionality = \frac{Molecular Weight*OH Number}{56.1*1000}$

3.4.1 Epoxidation/ Ring-Opening

Most widely available plant oil polyols are produced by some variation of epoxidation of double bonds, followed by nucleophilic ring opening. The most common method of epoxidizing plant oils uses peracids, which are often formed *in situ* with hydrogen peroxide and an acid, such as formic acid, acetic acid, or propionic acid [22]. Recyclable heterogeneous catalysts for ring opening oxiranes, such as sulfamic acid-functionalized iron nanoparticles, are being developed as sustainable alternatives to homogenous acid catalysts [23].

Ring opening of oxiranes is typically performed with alcohols, water, acids, or by hydrogenation [17]. In some cases, ring opening can be used to introduce additional functionality to plant oil-based polyols. For example, methacrylic acid has been used to ring open epoxidized plant oils to produce methacrylated plant oil-based polyols that can be further functionalized [13]. Zhang et al. recently reported a novel two-step approach for making plant oil-based polyols consisting of epoxidation of plant oils, followed by reduction of the epoxides and esters. In this study, epoxidized soybean oil and epoxidized linseed oil were reduced with lithium aluminum hydride (Li-AlH₄) to form fatty acid-based polyols [24].

3.4.2 Ozonolysis

During ozonolysis, ozone reacts with double bonds to form an unstable ozonide ring.

Aldehydes are formed upon decomposition of the ozonide rings. These carbonyls are subsequently reduced via hydrogenation to form primary hydroxyl groups [25]. Alternatively, polyols can be formed by reacting the unstable ozonides with diols (e.g., ethylene glycol) to form polyols with ester linkages [26]. A drawback of the ozonolysis/hydrogenation route is that the cleavage of double bonds during the ozonolysis step causes a reduction in molecular weight [27]. Furthermore, ozonolysis/hydrogenation of plant oils generates a mixture of products (alcohols, diols, and triols). Solvent selection is an important factor in determining the products and yields of ozonolysis. Mixtures of protic and aprotic solvents have been reported to maximize the yields of aldehydes, while minimizing the production of carboxylic acids [28].

3.4.3 Amidation

The amidation of fatty acid esters using diethanolamine to make amide polyols is an interesting approach for making plant oil-based polyols (see Scheme 3.3). This approach is particularly useful for feedstock with high free fatty acid contents. Tall oil, a by-product of Kraft pulping, is known for its high amounts of free fatty acids. Recently, tall oil fatty acids (TOFA) were used to make fatty acid-based amide polyols [29].

Amidation is also useful for nondrying oils, such as palm oil, which have high amounts of saturated fatty acids. Epoxidation and ring opening of nondrying oils tends to result in polyols with low hydroxyl functionalities. Amide polyols made from palm tree (*Elaeis guineensis*) oil have been used to make polyurethane coatings [30]. Likewise, seed oil from neem (*Azadiriachta indica juss*), a tree found in the Indian subcontinent, has been reacted with diethanolamine to make amide diols [31].



Scheme 3.3 Reaction scheme for making fatty acid-based amide polyols.

The composition of neem seed oil contains approximately 45% saturated fatty acids (palmitic, stearic, and arachidic). Other plant oils that have been used to make amide polyols include cottonseed oil [32] and olive oil [33].

Amide diols can be further functionalized into polyetheramides. Neem oil-based amide diols were further functionalized by reacting with bisphenol-A to make polyetheramides [31]. The thermal and mechanical properties of polyurethanes made from neem oil-based polyetheramides indicate that these materials have strong potential as resins in surface coatings. Polyetheramides have been synthesized from olive oil-based amide diols [33]. One drawback of amide diols is the high temperature necessary for making the reaction proceed. Recently, it has been shown that microwave assisted techniques are a more sustainable way to make amide diols and polyetheramides [33].

3.5 Flame Retardant Polyols

Polyurethanes, especially those made from plant oil-based polyols, require the use of flame retardants in their formulations. Flame retardants often contain phosphorus, halogen, boron, metal hydroxides, or nitrogen-based functional groups [34]. However, the use of halogenated (i.e., brominated or chlorinated) flame retardants is declining due to the potential release of toxic and corrosive combustion products [35]. As a result, many efforts to make bio-based flame retardants have focused on modifying plant oils with phosphorus-containing functional groups [36]. Boron-modified plant oil polyols have been prepared from castor oil [37] and linseed oil [38], but the flame retardant properties of these polyols were not reported.

Phosphorylated polyols with flame retardant properties have been synthesized from epoxidized soybean oil and phosphoric acid [39]. Rigid polyurethane foams made with phosphorylated polyols having phosphorus contents ranging from 0.4 wt% to 0.8 wt% displayed comparable flame retardant effects as commercial flame retardants based on limiting oxygen indexes (LOI) [40]. In another recent study, castor oil-based flame retardant polyols were prepared with diethyl phosphate [24]. The synthesis of these castor oil-based polyols followed a three-step process consisting of alcoholysis with glycerin, epoxidation with formic acid/hydrogen peroxide, and ring opening with diethyl phosphate. Resulting foams with 3 wt% phosphorus made with the flame retardant polyols showed a 24.3% improvement in LOI compared to foams made with gylcerolysis castor oil.

3.6 Alternative Routes for Making Polyurethanes

While the vast majority of research in bio-based polyurethanes has focused on using vegetable oil-based polyols, other routes for making renewable polyurethanes are being explored. Aliphatic vegetable oil-based isocyanates have been investigated [2]. The price of aliphatic amines has been a major factor limiting the development of renewable isocyanates [5].

Extensive research efforts have been made to find safer routes for making polyurethanes because isocyanates are toxic and can cause significant respiratory problems [41]. Furthermore, phosgene precursors of isocyanates are even more dangerous to the environment and to human health [42]. Phosgene-free methods of making isocyanates from fatty acids are a potential path for making more environmentally friendly isocyanates [43]. Polyurethane that avoids the use of isocyanate groups altogether has been prepared by reacting cyclo-carbonate functionalized soybean oil with diamines [41].

3.7 Applications of Plant Oil-Based Polyurethanes

Polyurethanes make up one of the most widely used and most versatile classes of polymers. However, the vast majority of polyurethanes are made from petrochemicalbased feedstock. Applications of bio-based polyurethanes from plant oils include flexible foams [44], rigid foams [45], coatings [21], elastomers [46], and adhesives [47]. A few applications of plant oil-based polyurethanes, including foams, coatings, and shape memory polymers, will be discussed in greater detail below.

3.8 Foams

Foam materials, including polyurethane foams, are lightweight, resilient, energy absorbent materials with a range of thermal and mechanical properties that can be readily tailored to meet the application requirements [48]. The polyurethane foam industry in the United States generates \$9.5 billion dollars a year and is expected to continue to grow at an annual growth rate of 2.6% over the next 5 years [49]. Downstream demand from new construction, furniture manufacturing, and automobile manufacturing are all expected to grow over the next few years as the economy recovers from the recession.

As described previously, urethane bonds are formed when isocyanate groups react with hydroxyl groups (see Scheme 3.1). The urethane reaction is sometimes referred to as the "gel" reaction. In polyurethane foams, a second chemical reaction, the urea reaction, occurs when an isocyanate reacts with water (see Scheme 3.4). The urea reaction is a twostep process that begins with an isocyanate



Scheme 3.4 Urea formation in polyurethane foams.

group reacting with water to yield an amine and a carbon dioxide molecule. In the second step, the amine subsequently reacts with a second isocyanate group to form a urea linkage. The urea reaction is sometimes called the "blow" reaction because CO_2 gas is generated, which causes the formation of the cell structures in polyurethane foams.

Foams that are made with CO_2 generated by water reacting with isocyanates are called "water blown." In certain applications, additional blowing agents are added to improve the thermal insulation properties or performance at extreme temperatures. In recent years, many of the conventional blowing agents that caused ozone depletion have been replaced with more environmentally friendly blowing agents [50]. Examples of environmentally friendly blowing agents include HFC-365mc, HFC-245fa, and pentane isomers [51].

3.8.1 Flexible Foams

Flexible foams are used in automotive, furniture, packaging, and bedding applications.

In the United States, polyurethane flexible foams make up 37.6% of the revenues in the foam industry [49]. TDI is the most common isocyanate used in manufacturing flexible foams [52]. Although overexposure to TDI can cause respiratory sensitization, industry sponsored studies have shown that PU flexible foam mattresses do not pose a health risk to consumers [53]. Flexible foams require low crosslink densities, which can be achieved by using polyols with either low functionalities or high molecular weights [54].

Palm oil is a good source of plant oil-based polyols for replacing conventional polyether

polyols in polyurethane flexible foams because it has low levels of unsaturation. Palm oil-based polyols with low hydroxyl functionalities and minimal amounts of residual double bonds have been synthesized by epoxidizing and ring opening palm triglycerides [55]. However, the amount of polyether polyols that can be replaced with significant changes to the formulation is limited. In one reported study, significant shrinkage was observed when more than 15 wt% palm oilbased polyols were used [55].

3.8.2 Rigid Foams

Rigid foams are used in a wider range of applications including construction, appliances, packaging, industrial, and transportation. Rigid foams typically have heavier densities than flexible foams. MDI and polymeric MDI are the most common isocyanates used in rigid foams. Rigid foam products make up 23.6% of the revenues generated by foams domestically [49]. Soybean oil-based polyols have been investigated extensively in recent years for developing rigid polyurethane foams [56].

3.9 Coatings

Polyurethane coatings are widely used in applications such as construction, transportation, furniture, textile and fibers, and electronics. There has been a move in recent years to develop sustainable polymeric coatings that reduce the use of petrochemical-based polymers, as well as the emission of volatile organic compounds (VOCs). Spray-applied VOC-free polyurethane coatings with high solid contents have been made with rapeseed oil-based polyols [57]. Hyperbranched plant oil-based polyurethanes have potential for use as thin film materials in packaging applications [12]. Water repellant polyurethane coatings have been made using alkoxysilane castor oil synthesized from thiol-ene coupling of 3-mercaptopropyl trimethoxysilane [58].

3.9.1 Waterborne Dispersions

Waterborne polyurethane dispersions (PUDs) are widely used polymers with a diverse range of applications, including paints, inks, coatings, adhesives, paper, and textiles [59]. PUDs offer various advantages, including low viscosity, high molecular weights, and good film forming properties [60]. PUDs are excellent choice of materials for reducing the emission of volatile organic compounds (VOC) generated by organic solvents [61]. Recently, a plant oil-based ionic segment for dispersing polyurethanes was synthesized as a substitute for dimethylol propionic acid (DMPA) by ring-opening epoxidized linseed oil with glycol followed by saponification [62]. A fully bio-based PUD was synthesized from castor oil and castor oil-based derivatives [63]. In this study, a linear castor oil-based diisocyanate was made using a Curtiss rearrangement, while the castor oil-based hydrophilic chain extender was prepared though a thiol-ene coupling of 3-mercaptopropionic acid.

3.9.2 Antimicrobial Coatings

There has been growing interest in developing plant oil-based cationic polyurethane coatings with antimicrobial properties in biomedical applications and food packaging. Quaternary ammonium compounds are well known for their antimicrobial properties [64]. Xia et al. examined the effects of different amine polyols on the antimicrobial, mechanical, and thermal properties of soybean oilbased cationic polyurethane coatings [65]. This study reported that N-methyl diethanolamine provided a good balance of properties, including the best thermal stability compared to the other amine polyols tested. In a related study, polyurethanes with higher relative ratios of N-methyl diethanolamine and lower crosslink densities showed increased antimicrobial activity [66]. In a different study. polyurethanes made from soybean oil-based polyols functionalized with quaternary ammonium salts used alkylating agents (methyl iodide and benzyl chloride) as an alternative to using an acid [67]. Boron-modified linseed oil polyurethanes have shown moderate antimicrobial activity against E. coli and high antimicrobial activity against S. aureus [38]. Metallohybrid plant oil-based polyols containing copper have been used to make polyurethane coatings with antimicrobial activity against S. aureus and E. coli [68].

3.10 Shape Memory Polymers

Shape-memory polymers are an important class of polymers that are being used in a wide range of applications. Shape memory polymers retain a shape upon deformation, but will respond to a stimulus, such as temperature above a certain threshold, and return to its original predeformed shape [69]. Hyperbranched polyurethanes with shape memory properties have been prepared using monoglyceride polyols synthesized from transesterification of plant oils with glycerin [15]. Castor oil-based monoglyceride polyols demonstrated better shape memory properties due to higher crosslinking, while monoglyceride polyols with lower levels of unsaturation showed better thermal stability [70]. Another recent study using different soybean oil-based polyols confirmed the importance of glycerol crosslinking in shape recovery [71].

3.11 Conclusions

Plant oils and fatty acids are an important industrial feedstock for making sustainable polyurethanes. In recent years, there has been much work on developing regional sources of polyols from currently underutilized sources of plant oils. The goals of these efforts are to reduce dependence on petrochemicals, minimize environmental impacts, and improve local economies. There is also considerable interest in synthesizing multifunctional plant oil-based polyols to improve their performance in a wide range of applications. For example, phosphorus-modified plant oilbased polyols can impart fire resistance in polyurethane foams, while methacrylic acidmodified polyols provide graft sites for acrylic chains in hybrid polymers. Polyurethane coatings with antimicrobial properties can be synthesized using modified polyols. Although most of the attention has focused on developing plant oil-based polyols, new advances in bio-based isocyanates are expected to make an impact. Plant oil-based polyurethanes are already being used in major applications such as foams and coatings.

References

 Z.S. Petrovic, Polyurethanes from vegetable oils, Polym. Rev. 48 (2008) 109– 155, doi: 10.1080/15583720701834224.

- [2] D.P. Pfister, Y. Xia, R.C. Larock, Recent advances in vegetable oilbased polyurethanes, ChemSusChem 4 (2011) 703–717, doi: 10.1002/ cssc.201000378.
- [3] M. Desroches, M. Escouvois, R. Auvergne, S. Caillol, B. Boutevin, From vegetable oils to polyurethanes: synthetic routes to polyols and main industrial products, Polym. Rev. 52 (2012) 38–79, doi: 10.1080/15583724.2011.640443.
- [4] G.N. Manvi, R.N. Jagtap, Effect of DMPA content of polyurethane dispersion on coating properties, J. Dispers. Sci. Technol. 31 (2010) 1376–1382, doi: 10.1080/01932690903269776.
- [5] E. Delebecq, J.P. Pascault, B. Boutevin, F. Ganachaud, On the versatility of urethane/urea bonds: reversibility, blocked isocyanate, and non-isocyanate polyurethane, Chem. Rev. 113 (2012) 80– 118, doi: 10.1021/cr300195n.
- [6] N. Karak, Vegetable oil-based polyurethanes, in: Vegetable Oil-Based Polymers: Properties, Processing and Applications, Woodhead Publishing, Cambridge, GBR, 2012 p. 146–79.
- [7] http://www.business-standard.com/ content/b2b-plastics-polymers/ bayer-materialscience-developseco-friendly-isocyanate-for-puindustry-115012100758_1.html.
- [8] Y. Xia, R.C. Larock, Vegetable oilbased polymeric materials: synthesis, properties, and applications, Green Chem. 12 (2010) 1893–1909, doi: 10.1039/C0GC00264J.
- [9] D. Firestone (Ed.), Physical and Chemical Characteristics of Oils, Fats, and Waxes, AOCS Press, Champaign, II, 1999.
- [10] ASTM InternationalASTM Standard D5554, 1995 (2011), Test Method for Determination of the Iodine Value of

Fats and Oils, ASTM International, West Conshohocken, PA, (2011).

- [11] U. Stirna, A. Fridrihsone, B. Lazdiņa, M. Misāne, D. Vilsone, Biobased polyure-thanes from rapeseed oil polyols: structure, mechanical and thermal properties, J. Polym. Environ. 21 (2013) 952–962, doi: 10.1007/s10924-012-0560-0.
- [12] B. Das, U. Konwar, M. Mandal, N. Karak, Sunflower oil based biodegradable hyperbranched polyurethane as a thin film material, Ind. Crops Prod. 44 (2013) 396–404, doi: 10.1016/j.indcrop.2012.11.028.
- [13] B. Balanuca, A. Lungu, A.M. Hanganu, L.R. Stan, E. Vasile, H. Iovu, Hybrid nanocomposites based on POSS and networks of methacrylated camelina oil and various PEG derivatives, Eur. J. Lipid Sci. Technol. 116 (2014) 458– 469, doi: 10.1002/ejlt.201300370.
- [14] Z.S. Petrović, X. Wan, O. Bilić, A. Zlatanić, J. Hong, I. Javni, M. Ionescu, J. Milić, D. Degruson, Polyols and poly-urethanes from crude algal oil, J. Am. Oil Chem. Soc. 90 (2013) 1073–1078, doi: 10.1007/s11746-013-2245-9.
- [15] H. Kalita, N. Karak, Mesua ferrea L. seed oil-based hyperbranched shape memory polyurethanes: effect of multifunctional component, Polym. Eng. Sci. 52 (2012) 2454–2461, doi: 10.1002/ pen.23200.
- [16] S.A. Edrisi, R.K. Dubey, V. Tripathi, M. Bakshi, P. Srivastava, S. Jamil, H.B. Singh, N. Singh, P.C. Abhilash, *Jatropha curcas* L.: a crucified plant waiting for resurgence, Renew. Sustain. Energy Rev. 41 (2015) 855–862, doi: 10.1016/j.rser.2014.08.082.
- [17] A.S.A. Hazmi, M.M. Aung, L.C. Abdullah, M.Z. Salleh, M.H. Mahmood, Producing Jatropha oil-based polyol via

epoxidation and ring opening, Ind. Crops Prod. 50 (2013) 563–567, doi: 10.1016/j.indcrop.2013.08.003.

- [18] D.S. Tathe, R.N. Jagtap, Synthesis of bio-based polyurethane from modified *Prosopis juliflora* oil, J. Am. Oil Chem. Soc. 90 (2013) 1405–1413, doi: 10.1007/s11746-013-2275-3.
- [19] C.R. Malacrida, N. Jorge, Yellow passion fruit seed oil (Passiflora edulis f. flavicarpa): physical and chemical characteristics, Braz. Arch. Biol. Technol. 55 (2012) 127–134.
- [20] R. Vasconcelos Vieira Lopes, N. Loureiro, A. Pezzin, A. Gomes, I. Resck, M. Sales, Synthesis of polyols and polyurethanes from vegetable oils-kinetic and characterization, J. Polym. Res. 20 (2013) 1–9, doi: 10.1007/s10965-013-0238-x.
- [21] T.F. Garrison, M.R. Kessler, R.C. Larock, Effects of unsaturation and different ring-opening methods on the properties of vegetable oil-based polyurethane coatings, Polymer 55 (2014) 1004–1011, doi: 10.1016/j.polymer.2014.01.014.
- [22] E. Milchert, A. Smagowicz, G. Lewandowski, Optimization of the epoxidation of rapeseed oil with peracetic acid, Org. Process. Res. Dev. 14 (2010) 1094–1101, doi: 10.1021/op900240p.
- [23] B. Kollbe Ahn, H. Wang, S. Robinson, T.B. Shrestha, D.L. Troyer, S.H. Bossmann, X.S. Sun, Ring opening of epoxidized methyl oleate using a novel acid-functionalized iron nanoparticle catalyst, Green Chem. 14 (2012) 136, doi: 10.1039/c1gc16043e.
- [24] L. Zhang, M. Zhang, L. Hu, Y. Zhou, Synthesis of rigid polyurethane foams with castor oil-based fl ame retardant polyols, Ind. Crops Prod. 52 (2014) 380–388 doi:10.1016/j.indcrop.2013.10.043.

- [25] V.H.R. de Souza, S.A. Silva, L.P. Ramos, S.F. Zawadzki, Synthesis and characterization of polyols derived from corn oil by epoxidation and ozonolysis, J. Am. Oil Chem. Soc. 89 (2012) 1723–1731, doi: 10.1007/s11746-012-2063-5.
- [26] P. Tran, D. Graiver, R. Narayan, Ozone-mediated polyol synthesis from soybean oil, J. Am. Oil Chem. Soc. 82 (2005) 653–659, doi: 10.1007/s11746-005-1124-z.
- [27] Z.S. Petrović, W. Zhang, I. Javni, Structure and properties of polyurethanes prepared from triglyceride polyols by ozonolysis, Biomacromolecules 6 (2005) 713–719, doi: 10.1021/bm049451s.
- [28] T.S. Omonov, E. Kharraz, J.M. Curtis, Ozonolysis of canola oil: a study of product yields and ozonolysis kinetics in different solvent systems, J. Am. Oil Chem. Soc. 88 (2011) 689–705, doi: 10.1007/s11746-010-1717-4.
- [29] V. Yakushin, U. Stirna, O. Bikovens, M. Misane, I. Sevastyanova, D. Vilsone, Synthesis and characterization of novel polyurethanes based on tall oil, Mater. Sci. 19 (2013)doi: 10.5755/j01. ms.19.4.2666.
- [30] S.D. Rajput, D.G. Hundiwale, P.P. Mahulikar, V.V. Gite, Fatty acids based transparent polyurethane films and coatings, Prog. Org. Coat. 77 (2014) 1360–1368, doi: 10.1016/j.porgcoat.2014.04.030.
- [31] A.B. Chaudhari, A. Anand, S.D. Rajput, R.D. Kulkarni, V.V. Gite, Synthesis, characterization and application of *Azadirachta indica juss* (neem oil) fatty amides (AIJFA) based polyurethanes coatings: a renewable novel approach, Prog. Org. Coat. 76 (2013) 1779–1785 doi: 10.1016/j.porgcoat.2013.05.016.

- [32] P.D. Meshram, R.G. Puri, A.L. Patil, V.V. Gite, Synthesis and characterization of modified cottonseed oil based polyesteramide for coating applications, Prog. Org. Coat. 76 (2013) 1144–1150, doi: 10.1016/j.porgcoat.2013.03.014.
- [33] M. Alam, N.M. Alandis, Microwave assisted synthesis and characterization of olive oil based polyetheramide as anticorrosive polymeric coatings, Prog. Org. Coat. 75 (2012) 527–536, doi: 10.1016/j.porgcoat.2012.06.001.
- [34] A. Dasari, Z.Z. Yu, G.P. Cai, Y.W. Mai, Recent developments in the fire retardancy of polymeric materials, Prog. Polym. Sci. 38 (2013) 1357–1387, doi: 10.1016/j.progpolymsci.2013.06.006.
- [35] L.M. De Espinosa, M.A.R Meier, J.C. Ronda, M. Galià, V. Cádiz, Phosphorus-containing renewable polyesterpolyols via ADMET polymerization: synthesis, functionalization, and radical crosslinking, J. Polym. Sci. Part Polym. Chem. 48 (2010) 1649–1660, doi: 10.1002/pola.23887.
- [36] M. Moreno, G. Lligadas, J.C. Ronda, M. Galià, V. Cádiz, Phospha–Michael addition to enone-containing triglyceride derivatives as an efficient route to flame retardant renewable thermosets, J. Polym. Sci. Part Polym. Chem. 50 (2012) 3206–3213, doi: 10.1002/pola.26106.
- [37] D. Akram, E. Sharmin, S. Ahmad, Synthesis, characterization and corrosion protective properties of boron-modified polyurethane from natural polyol, Prog. Org. Coat. 63 (2008) 25–32, doi: 10.1016/j.porgcoat.2008.04.003.
- [38] D. Akram, E. Sharmin, S. Ahmad, Development and characterization of boron incorporated linseed oil polyurethanes, J. Appl. Polym. Sci. 116 (2010) 499–508, doi: 10.1002/app.31579.

- [39] Y. Guo, J.H. Hardesty, V.M. Mannari, J.L. Massingill Jr., Hydrolysis of epoxidized soybean oil in the presence of phosphoric acid, J. Am. Oil Chem. Soc. 84 (2007) 929–935, doi: 10.1007/ s11746-007-1126-5.
- [40] M. Heinen, A.E. Gerbase, C.L. Petzhold, Vegetable oil-based rigid polyurethanes and phosphorylated flame-retardants derived from epoxydized soybean oil, Polym. Degrad. Stab. 108 (2014) 76–86, doi: 10.1016/j. polymdegradstab.2014.05.024.
- [41] J. Guan, Y. Song, Y. Lin, X. Yin, M. Zuo, Y. Zhao, X. Tao, Q. Zheng, Progress in study of non-isocyanate polyurethane, Ind. Eng. Chem. Res. 50 (2011) 6517–6527, doi: 10.1021/ie101995j.
- [42] M.S. Kathalewar, P.B. Joshi, A.S. Sabnis, V.C. Malshe, Non-isocyanate polyurethanes: from chemistry to applications, RSC Adv. 3 (2013) 4110, doi: 10.1039/c2ra21938g.
- [43] O. Kreye, H. Mutlu, M.A.R. Meier, Sustainable routes to polyurethane precursors, Green Chem. 15 (2013) 1431– 1455, doi: 10.1039/C3GC40440D.
- [44] A. Prociak, P. Rojek, H. Pawlik, Flexible polyurethane foams modified with natural oil based polyols, J. Cell Plast. 48 (2012) 489–499, doi: 10.1177/0021955X12446210.
- [45] S. Tan, T. Abraham, D. Ference, C.W. Macosko, Rigid polyurethane foams from a soybean oil-based polyol, Polymer 52 (2011) 2840–2846, doi: 10.1016/j.polymer.2011.04.040.
- [46] M.F. Sonnenschein, V.V. Ginzburg, A.L. Grzesiak, K. Schiller, B.L. Wendt, Design, polymerization, and properties of polyurethane elastomers from miscible, immiscible, and hybridized seed-oil derived soft segment blends, J. Polym.

Sci. Part Polym. Chem. 53 (2015) 93– 102, doi: 10.1002/pola.27450.

- [47] M.M. Aung, Z. Yaakob, S. Kamarudin, L.C. Abdullah, Synthesis and characterization of Jatropha (*Jatropha curcas* L.) oil-based polyurethane wood adhesive, Ind. Crops Prod. 60 (2014) 177–185, doi: 10.1016/j.indcrop.2014.05.038.
- [48] V. Srivastava, R. Srivastava, On the polymeric foams: modeling and properties, J. Mater. Sci. 49 (2014) 2681–2692, doi: 10.1007/s10853-013-7974-5.
- [49] W. McKitterick, Urethane Foam Manufacturing in the US (IBISWorld Industry Report No. 32615), IBISWorld Services, Melbourne, Australia, (2015).
- [50] H. Lim, E.Y. Kim, B.K. Kim, Polyurethane foams blown with various types of environmentally friendly blowing agents, Plast. Rubber Compos. 39 (2010) 364–369, doi: 10.1179/1743289 10X12691245469835.
- [51] L. Zipfel, P. Barthtlemy, P. Dournel, The next generation blowing agents: from one single product to a product range, J. Cell Plast. 35 (1999) 345–364, doi: 10.1177/0021955X9903500405.
- [52] E. Vangronsveld, S. Berckmans, M. Spence, Toluene diisocyanate emission to air and migration to a surface from a flexible polyurethane foam, Ann. Occup. Hyg. 57 (2013) 650–661, doi: 10.1093/annhyg/mes105.
- [53] S.M. Arnold, M.A. Collins, C. Graham, A.T. Jolly, R.J. Parod, A. Poole, T. Schupp, R.N. Shiotsuka, M.R. Woolhiser, Risk assessment for consumer exposure to toluene diisocyanate (TDI) derived from polyurethane flexible foam, Regul. Toxicol. Pharmacol. 64 (2012) 504–515, doi: 10.1016/j. yrtph.2012.07.006.

- [54] P. Rojek, A. Prociak, Effect of different rapeseed-oil-based polyols on mechanical properties of flexible polyurethane foams, J. Appl. Polym. Sci. 125 (2012) 2936–2945, doi: 10.1002/app.36500.
- [55] H. Pawlik, A. Prociak, Influence of palm oil-based polyol on the properties of flexible polyurethane foams, J. Polym. Environ. 20 (2011) 438–445, doi: 10.1007/s10924-011-0393-2.
- [56] H. Fan, A. Tekeei, G.J. Suppes, F.-H. Hsieh, Rigid polyurethane foams made from high viscosity soy-polyols, J. Appl. Polym. Sci. 127 (2013) 1623– 1629, doi: 10.1002/app.37508.
- [57] U. Stirna, A. Fridrihsone-Girone, V. Yakushin, D. Vilsone, Processing and properties of spray-applied, 100% solids polyurethane coatings from rapeseed oil polyols, J. Coat Technol. Res. 11 (2014) 409–420, doi: 10.1007/ s11998-013-9545-8.
- [58] C. Fu, Z. Yang, Z. Zheng, L. Shen, Properties of alkoxysilane castor oil synthesized via thiol-ene and its polyurethane/siloxane hybrid coating films, Prog. Org. Coat. 77 (2014) 1241–1248, doi: 10.1016/j.porgcoat.2014.03.020.
- [59] B.K. Kim, Aqueous polyurethane dispersions, Colloid Polym. Sci. 274 (1996) 599–611, doi: 10.1007/ BF00653056.
- [60] Y. Lu, R.C. Larock, Soybean-oil-based waterborne polyurethane dispersions: effects of polyol functionality and hard segment content on properties, Biomacromolecules 9 (2008) 3332–3340, doi: 10.1021/bm801030g.
- [61] Y. Lu, R.C. Larock, Soybean oilbased, aqueous cationic polyurethane dispersions: synthesis and properties, Prog. Org. Coat. 69 (2010) 31–37, doi: 10.1016/j.porgcoat.2010.04.024.

- [62] R. Chen, C. Zhang, M.R. Kessler, Anionic waterborne polyurethane dispersion from a bio-based ionic segment, RSC Adv. 4 (2014) 35476, doi: 10.1039/C4RA07519F.
- [63] C. Fu, Z. Zheng, Z. Yang, Y. Chen, L. Shen, A fully bio-based waterborne polyurethane dispersion from vegetable oils: from synthesis of precursors by thiol-ene reaction to study of final material, Prog. Org. Coat. 77 (2014) 53–60, doi: 10.1016/j.porgcoat.2013.08.002.
- [64] Y. Gao, I.L. Kyratzis, Antimicrobial finishing of wool using an oxidative pretreatment to enhance the exhaustion of quaternary ammonium compounds, J. Appl. Polym. Sci. 125 (2012) E71– E78, doi: 10.1002/app.36412.
- [65] Y. Xia, Z. Zhang, M.R. Kessler, B. Brehm-Stecher, R.C. Larock, Antibacterial soybean-oil-based cationic polyurethane coatings prepared from different amino polyols, ChemSusChem 5 (2012) 2221–2227, doi: 10.1002/ cssc.201200352.
- [66] T.F. Garrison, Z. Zhang, H.J. Kim, D. Mitra, Y. Xia, D.P. Pfister, B.F. Brehm-Stecher, R.C. Larock, M.R. Kessler, Thermo-mechanical and antibacterial properties of soybean oil-based cationic polyurethane coatings: effects of amine ratio and degree of crosslinking, Macromol. Mater. Eng. 299 (2014) 1042– 1051, doi: 10.1002/mame.201300423.
- [67] H. Bakhshi, H. Yeganeh, S. Mehdipour-Ataei, M.A. Shokrgozar, A. Yari, S.N. Saeedi-Eslami, Synthesis and characterization of antibacterial polyurethane coatings from quaternary ammonium salts functionalized soybean oil based polyols, Mater. Sci. Eng. C 33 (2013) 153–164, doi: 10.1016/j. msec.2012.08.023.

- [68] E. Sharmin, F. Zafar, D. Akram, S. Ahmad, Plant oil polyol nanocomposite for antibacterial polyurethane coating, Prog. Org. Coat 76 (2013) 541–547, doi: 10.1016/j.porgcoat.2012.10.027.
- [69] E.D. Rio, G. Lligadas, J.C. Ronda, M. Galià, M.a.R. Meier, V. Cádiz, Polyurethanes from polyols obtained by ADMET polymerization of a castor oilbased diene: characterization and shape memory properties, J. Polym. Sci. Part Polym. Chem. 49 (2011) 518–525, doi: 10.1002/pola.24466.
- [70] H. Kalita, N. Karak, Biobased hyperbranched shape-memory polyurethanes: effect of different vegetable oils, J. Appl. Polym. Sci. 131 (2014) 39579, doi: 10.1002/app.39579.
- [71] S. Miao, N. Callow, P. Wang, Y. Liu, Z. Su, S. Zhang, Soybean oil-based polyurethane networks: shape-memory effects and surface morphologies, J. Am. Oil Chem. Soc. 90 (2013) 1415–1421, doi: 10.1007/s11746-013-2273-5.

4 Plant Oil-Based Polyhydroxyurethanes

Ashley Johns, Luke Gibbons, Madeline Smith, Kyle Edwards, and Rafael L. Quirino

Chemistry Department, Georgia Southern University, Statesboro, GA, USA

	Ο U T L I N E				
4.1	Introduction				
4.2	Petroleum-Based Polyurethanes				
4.3	Polyurethanes from Bio-Based Polyols 5				
4.4	Polyhydroxyurethanes (PHUs)5				
	4.4.1 Introduction to PHUs	59			
	4.4.2 Petroleum-Based PHUs	60			
	4.4.3 Bio-Based PHUs	61			
4.5	Alternative Systems67				
4.6	Conclusions 68				
Ref	eferences 69				

4.1 Introduction

The development of novel bio-based polymers from inexpensive and renewable materials has the potential to greatly impact our current economy, the environment, and our energy matrix. Due to the commercial importance of the plastics and coatings industries, and the uncertainty of the oil market, the replacement of petroleum-based materials with bio-based counterparts can be greatly advantageous. The advantages of bio-based materials are the ready availability of large quantities of renewable starting materials at insignificant costs, the potential of producing more bio-degradable materials than the virtually indestructible petroleum-based polymers, the possibility of obtaining properties not currently available in commercial petroleumbased products, and their overall intrinsic low toxicity. The current legislation and the frequent dramatic fluctuations in the price of oil also create an urgent need, from an industrial point of view, for renewable starting materials and bio-based products that have the potential to replace petroleum-based materials.

Vegetable oils are amongst the most widely used renewable raw materials [1]. They find use in paints [2], biocoatings [3–7], biofuels [8], and as building blocks for bio-based polymers [9–11]. Traditional polyurethanes with promising properties for use as coatings and adhesives



Figure 4.1 Generic chemical structure of polyurethanes.

have been obtained from vegetable oil-based polyols, and have been successfully used in the preparation of anionic [4], and cationic dispersions [5]. Hybrid emulsion systems have also been proposed [6,7]. The technology involved in these vegetable oil-based polyurethane systems is remarkably simple, and although it represents a step in the right direction, it does not address one of the major points of concern on the polyurethane production, which is the use of toxic chemicals such as diisocyanates.

This chapter presents the reader with an overview of the major accomplishments in bio-based polyurethane research, with a special focus on polyhydroxyurethanes (PHUs). The discussion starts with a brief description of the chemistry involved in polyurethanes (PUs), and then transitions into the advancements reported by select research groups on plant oilbased PUs and related systems. The last section of this manuscript addresses the chemistry involved in PHUs and the various proposed biobased starting materials investigated to date, with an entire subsection dedicated to plant oil-based PHUs. Alternative PU systems from nonisocyanate sources are briefly described in this chapter, and concluding remarks give a perspective on the future of bio-based PHUs.

4.2 Petroleum-Based Polyurethanes

Polyurethanes are a class of polymers composed of repeating subunits linked together by urethane groups (R'-O-CO-NH-R) (Figure 4.1). Polyurethanes can be employed in a variety of applications due to the variety of subunits that can be used in their preparation. The subunits' structure determines bulk properties, such as flexibility, thermal stability, and mechanical strength. The range of applications for polyurethanes includes insulating foams, hard plastic parts, sealants, coatings, and durable flexible lining or tubes for the construction Industry.

The starting materials for most polyurethanes are diisocyanates and polyols. Typically, a diol is used, and occasionally a catalyst. Each monomer has at least two reactive functional groups, allowing the reaction to continue at the termini of the chains as the polymer grows [12]. Regardless of the exact starting material, the reaction proceeds to form polymer chains with repeating urethane groups as depicted in Figure 4.1. In that transformation, the hydroxyl group of the polyol acts as a nucleophile attacking the carbon of the isocyanate group as the pi electrons between C and O are pushed onto the oxygen. The re-establishment of the carbonoxygen double bond pushes the electron density onto the nitrogen atom. A final proton transfer completes the mechanism, resulting in a urethane functional group. The reaction can be catalyzed by the addition of a base, which enhances the nucleophilicity of the polyol through deprotonation. The catalyst is regenerated during the final proton transfer step. This mechanism is widely accepted and therefore will not be explicitly presented in this chapter.

The structure of the polyol used in the preparation of a polyurethane significantly affects the properties of the resulting polymer [13]. The longer the carbon chain of the polyol is, the more ductile and flexible the resulting polymer. Reversely, shorter carbon chains in the polyol result in more rigid and durable polymers. The functionality of the polyol influences the crosslink density of the final material and therefore has a direct impact on the properties of the resulting polyurethane [14].

The structure of the diisocyanate used in the synthesis of polyurethanes is also crucial for the final properties of the polymer. Aromatic diisocyanates, such as toluene diisocyanate (TDI), or methylene diphenyl diisocyanate (MDI), are traditionally used for their high reactivity and stiffness conferred to the final material, most likely associated with pi-stacking interactions of the polymer chains. Polyurethanes made with TDI are usually more flexible than those made with MDI [15].

Recent research efforts have been targeted at replacing traditionally used polyols and diisocyanates with eco-friendly derivatives in the synthesis of polyurethanes. The use of diisocyanates represents a potential toxicity hazard in the manufacturing of polyurethanes as any unreacted isocyanate group left in the final product constitutes a risk that limits the applications of the final polymer. Additionally, bio-based polyols have been investigated as substitutes for petroleumderived polyols commonly used in the synthesis of polyurethanes [16]. Another approach consists of the preparation of polymers containing repeating urethane linkages formed by alternative functional groups. One such example is the preparation of hydroxyurethanes from cyclic carbonates and amines. Such approach leads to polymers with properties very similar to traditional polyurethanes, with the added advantage of avoiding the use of diisocyanates and helping to mitigate toxicity and environmental concerns [17].

4.3 Polyurethanes from Bio-Based Polyols

Plant oils consist of triglycerides with fatty acid chains of varying length and degree of unsaturation depending on the plant from which the oil is extracted. These structural variations directly affect the physical and chemical properties of the oil, such as its reactivity toward other reagents. Thus, different plant oils can be classified as drying, semidrying, and nondrying oils, depending on their ability to auto-oxidize in the presence of O₂ from air. Some plant oils bear functional groups, like castor oil, which contains a hydroxyl (-OH) group attached to Carbon 12 (C_{12}) of the fatty acid chain. For illustration purposes, the accepted chemical structure of soybean oil is provided in Figure 4.2.

The carbon-carbon double bonds in plant oils can be reacted in many different ways to form bio-based polymers. The direct polymerization of carbon-carbon double bonds in triglycerides through thermal, cationic, and/or free radical processes, as well as ROMP and ADMET, is discussed in detail elsewhere [18]. Alternatively, the carbon-carbon double bonds in the fatty acid chains of vegetable oils can be epoxidized to form epoxidized triglycerides with various degrees of epoxidation [19]. Epoxidized oils can then be reacted with a variety of nucleophiles to append different functional groups to the triglyceride and therefore increase its reactivity. Acrylated epoxidized soybean oil (AESO), the product of the reaction of acrylic acid with epoxidized soybean oil (ESO), has been extensively studied



Figure 4.2 Accepted chemical structure of soybean oil.



Figure 4.3 Steps involved in the conversion of plant oils into polyols for the preparation of polyurethanes. (a) epoxidation, (b) ring opening, (c) condensation.

and is currently commercially available [20]. Recently, AESO has been used to produce semirigid thermosetting foams with a high bio-based content and desirable mechanical properties for industrial uses [21]. Besides soybean oil, epoxidized methyl oleate has also been acrylated and polymerized to afford pressure-sensitive adhesives [22].

The ring opening reaction of the epoxide groups in epoxidized vegetable oils results in bio-based polyols. Such polyols can react with diisocyanates to give vegetable oil-based polyurethanes (Figure 4.3). Castor oil and ricinoleic acid, both of which are natural polyols, have been directly used in the preparation of polyurethanes [23–25]. Polyols based on a variety of epoxidized oils (sunflower, canola, soybean, corn, linseed [26], and passion fruit oils) (Rodrigues et al., unpublished work) have been polymerized with methylene diphenyl diisocyanate (MDI) to give polyurethanes.

It has been found that the differences in the properties of these polyurethane networks result primarily from different crosslink densities, and are independent of the position of the reactive sites in the fatty acids.

Anionic waterborne polyurethane dispersions (PUDs) with no volatile organic compounds (VOCs) and uniform particle sizes have been prepared with 50-60 wt% methoxylated soybean oil polyols (MSOLs) [4]. These soybean oil-based polyurethanes have been obtained by the reaction of MSOLs with isophorone diisocyanate (IPDI) and dimethylolpropionic acid (DMPA), followed by neutralization with triethylamine and dispersion in water [4]. It was found that, besides the expected crosslink density dependence, the structure and the properties of the resulting materials greatly depend on the hard segment content of the PUs [4]. Along the same lines, vegetable oil-based cationic PUDs with uniform particle sizes have been prepared from castor oil, MSOLs, and AESObased polyols [5]. The cocondensation of these plant oil-based polyols with *N*-methyl diethanol amine (MDEA) and MDI resulted in polyurethanes baring tertiary amines that were finally protonated to yield the desired waterborne, plant oil-based, cationic PUDs with improved adhesion to anionic substrates, such as leather and glass [5].

Polyurethane-acrylic hybrid systems have also been designed from soybean oil-based polyols [27]. In these systems, AESO has been condensed with toluene 2,4-diisocyanate, and DMPA, resulting in anionic PUDs with unreacted carboncarbon double bonds. The subsequent free radical emulsion copolymerization of such PUDs with acrylic comonomers, such as butyl acrylate or methyl methacrylate, resulted in hybrid latexes with 15-60 wt% bio-based content [6,27]. The resulting plant oil-based hybrid latexes exhibit properties comparable to those of known polypropylene glycol-based polyurethane-acrylic hybrid latex films. Improved thermo-mechanical properties have been obtained by performing a seeded emulsion polymerization of 10–60 wt% of styrene or butyl acrylate in the presence of an anionic MSOL-based PUD as seed particles [7]. In that core-shell hybrid system, the PUD forms the latex shell, and serves as a high molecular weight emulsifier, while the vinyl polymers form the core [7].

4.4 Polyhydroxyurethanes (PHUs)

4.4.1 Introduction to PHUs

With a chemical structure very similar to that of regular polyurethanes, polyhydroxyurethanes have monomers connected through urethane linkages, with additional hydroxyl functional groups present on the polymer backbone (Figure 4.4). The presence of additional hydroxyl groups on PHUs allows for more possibilities of functionalization



Figure 4.4 Simplified mechanism showing the synthesis of a hydroxyurethane from a cyclic carbonate and a primary amine.
and crosslinking in comparison to regular PUs. This leads to materials that have a wide range of applications, including tissue engineering [28], and polymer coatings [29]. In comparison to PUs, PHUs exhibit similar bulk properties with the added advantage of being more environmentally friendly. Indeed, in most cases, the production of PHUs does not involve the use of hazardous compounds, such as the diisocyanates frequently used in the synthesis of PUs. Due to the absence of isocyanates in their formulation, PHUs are also referred to as nonisocyanate polyurethanes (NIPU). The use of more environmentally friendly, and more cost-effective starting materials, including byproducts from other industrial processes has lately made PHUs an appealing alternative to regular PUs.

Several different starting materials can be used for the synthesis of PHUs. Petroleumbased, as well as biorenewable molecules, have been successfully used for the preparation of nonisocyanate polyurethanes. The most notable sources for the synthesis of biobased PHUs include vegetable oils, glycerol carbonates, and terpenes [17]. Each one of these topics will be covered in more detail in the upcoming subsections. One of the most promising bio-based PHU systems is obtained by the nucleophilic polyaddition of polyamines to polycyclic carbonates prepared from vegetable oils or their derivatives [17]. PHUs can also be prepared from glycerol carbonate-based intermediates using a step growth polyaddition synthesis [30]. A simplified mechanism for the addition of amines and cyclic carbonates is presented in Figure 4.4 [31]. As detailed later in the text, depending on the structure of the starting cyclic carbonate used in the synthesis (5-membered versus 6-membered), PHUs containing either a primary or a secondary alcohol group may result.

4.4.2 Petroleum-Based PHUs

Polyhydroxyurethanes can be synthesized from petroleum-based molecules, as investigated, for example, by Endo et al. with the use of a five-membered cyclic carbonate, resulting in PHUs with both primary and secondary hydroxyl groups, with molecular weights of approximately 15,000 g/mol [32]. In the synthesis of PHUs from seven-membered cyclic carbonates, two different diamines were used, 4,9-dioxa-1,12-dodecanediamine (DODDA), and *p*-xylylenediamine (*p*-XDA) [34]. Endo found that using DODDA with a 6-h reaction time yielded complete conversion of the cyclic carbonates at temperatures ranging from 30°C to 70°C [33]. Increasing the reaction temperature resulted in an increase of the final polymer molecular weights, which ranged from 32,800 g/mol to 35,700 g/mol [34]. p-XDA was found to be less reactive than DODDA, causing both lower reaction yields and lower molecular weights [33].

In a similar system, the preparation of PHUs from the reaction between terephthaloyl dicyclocarbonate (DCter) and 1,10-diaminodecane (DA10) was optimized using rheometry [34]. The ideal temperature leading to high molecular weight chains was found to be 90°C [35]. Recently, Endo et al. synthesized PHUs with a silicone backbone [35]. Polyaddition synthesis was used substituting the diamine with silicone diamine [35]. This change resulted in a lower glass transition temperature, as well as an increased hydrophobicity [35]. Possible applications of this silicone backbone PHU include antifouling, anticorrosion, and waterrepellent coatings [35]. Typically, PHUs are prepared from bio-based resources in an effort to move away from dependence on petroleum products and into materials with a more limited carbon footprint.

4.4.3 Bio-Based PHUs

Nonisocyanate polyurethanes can be successfully prepared from bio-based resources leading to environmentally friendly products with great appeal due to their intrinsic low toxicity. Among the different bio-based PHUs investigated to date, the ones prepared from cyclic carbonates derived from triglycerides, glycerol, and terpenes stand out due to the abundance and low cost of the starting materials. Following a different approach, organic/ inorganic hybrid PHUs have been prepared from the reaction of cyclic carbonates with amino-modified silica nanoparticles, leading to materials with very interesting mechanical properties [36]. Each one of these topics is covered in more detail next.

4.4.3.1 Triglycerides

Triglycerides can be easily converted into reactive monomers by modification of their chemical structure. Functional group transformations can potentially convert unsaturated triglycerides or fatty acids into polymerizable moieties. Polyurethanes from triglyceride-based polyols are one of the most common bio-based polymers for coating applications [37]. As discussed earlier, the preparation of nonisocyanate polyurethanes from triglycerides can be successfully achieved, and such materials have proven to be very promising. Recently, soybean, sunflower, and palm oils have served as starting materials for the preparation of biobased PHUs [38]. Triglyceride-based cyclic carbonates can be obtained by epoxidation of unsaturated vegetable oils with hydrogen peroxide in acetic or formic acid [39], followed by carbonation, carried out in the presence of carbon dioxide and a catalyst [40]. The polyaddition reaction is then carried

out through ring-opening polymerization of the di- and polyfunctional cyclic carbonates with di- or polyamines [38,41,42].

When supercritical carbon dioxide is used in the preparation of cyclic carbonates from vegetable oils, decreases of approximately 66% in reaction time have been detected in comparison to reactions carried out under regular conditions. In general, supercritical conditions yield preferentially five-membered ring carbonates over their six-membered ring counterparts [17]. Catalysts such as quaternary ammonium halides, polystyrene-bound quaternary ammonium salts, and alkali metal halides have been shown to work most effectively in these reactions [38].

Cyclic carbonates have been shown to produce hydroxyurethanes upon aminolysis with a primary amine. Multiple procedures have been reported for the synthesis of fiveand six-membered cyclic carbonates at the terminus of aliphatic chains [17]. Overall, when five-membered cyclic carbonates are attacked by a primary amine, two possible isomers can be obtained, with selectivity favoring the product containing a secondary hydroxyl group. The more symmetrical sixmembered cyclic carbonates can only produce a single hydroxyurethane [17,38,43]. For the synthesis of polyhydroxyurethanes, a polyaddition reaction must occur between bifunctional cyclic carbonates and aliphatic diamines [38].

The synthesis of plant oil-based polyhydroxyurethanes has been carried out under various conditions. Wilkes reported the preparation of PHUs with tensile strength of 0.2–1.5 MPa and 70–170% elongation at break through the heating of carbonated soybean oil with various amines [44]. Such mixtures were heated at 70°C for 10h, followed by a 3 h stage at 100°C [44]. In a related study, a cyclic carbonate was developed from cashew nut shell liquid and reacted with hexamethylene diamine and isophorone diamine at 120 and 150°C for various time intervals. The resulting materials held potential application in coatings formulations [45].

Tamami et al. reported the synthesis of novel triglyceride-based PHUs through the use of soybean oil. The reaction of epoxidized soybean oil with carbon dioxide was catalyzed by tetrabutylammonium bromide (TBAB) at 110°C. The carbonated soybean oil was then reacted with primary di- and triamines, resulting in products with mechanical and thermal properties that vary according to the type of amine used [39]. Carbonated soybean oil was also shown to react with *n*butylamine to produce β -hydroxyurethane systems [39].

In another study, Bahr et al. prepared PHUs from soybean and linseed oils [42]. Both TBAB and silica-supported 4-pyrrolidinopyridinium iodide were used as catalysts in the conversion of epoxidized oils into their carbonated equivalents [42]. The conversion rates were monitored, and TBAB was determined to be the most efficient catalyst [42]. The cyclic carbonates were cured with 1,2-ethane diamine (EDA), 1,4-butane diamine (BDA), and isophorone diamine (IPDA) [42]. Compared with the previous literature, Bahr reported increased glass transition temperatures as well as improved stiffness for the systems prepared [42].

Hybrid PHUs, as opposed to their linear counterparts derived from the reaction of bifunctional cyclic carbonates with diamines [17], have been prepared by a four-step method consisting in: (i) carbonation of epoxidized unsaturated fatty acid triglycerides with conversion rates ranging from 35% to 85%, (ii) stoichiometric ring opening of the carbonate groups with primary mono-amines, (iii) followed by subsequent ring-opening of the remaining epoxide groups with primary diamines, and finally (iv) crosslinking this intermediate with difunctional comonomers bearing amino-reactive groups [40]. The usage of this method can lessen the time required for synthesis and potentially increase the properties of the resulting products [40].

Numerous other polymeric networks have been reported, with soybean and linseed oilbased systems being the most extensively studied. Bio-based PHUs have also been prepared from other nontriglyceride-based, biorenewable derivatives, such as cashew nut shell liquid [40]. Overall, the use of triglycerides holds much potential for the future as a safe and environmentally friendly route for the production of polyhydroxyurethanes [17].

4.4.3.2 Glycerol

Glycerol is a widely available bio-based resource that can be used in the synthesis of polyhydroxyurethanes without an intermediate isocyanate. Often obtained as the byproduct of industrial processes involving vegetable oils, such as soap manufacturing or biodiesel production, glycerol holds potential for PHU preparation due to its easy conversion to glycerol carbonate. A variety of cyclic carbonates can be synthesized from glycerol and subsequently reacted with a range of amines to produce polyhydroxyurethanes through ring-opening polymerization [17].

Several reaction conditions have been studied for the conversion of glycerol to glycerol carbonate, as well as the reaction of the resulting carbonate with amines to form PHU networks. The reaction of glycerol with compounds such as carbon dioxide, urea, dimethyl carbonate, ethylene carbonate, and propylene carbonate leads to the formation



Figure 4.5 Example of the step-wise synthesis of hydroxyurethanes from glycerol.

of glycerol carbonate [46–49]. The synthetic approach is illustrated in Figure 4.5 for dimethyl carbonate. Glycerol carbonate may be further modified to form larger carbonate compounds, altering the structure of the final PHU product. Likewise, the structure and substitution pattern of the amine used in the ring-opening polymerization step has an impact on the final product.

Nohra et al. reported a PHU yield of 87% when the ring opening was carried out using amines with up to 12 carbon atoms in the alkyl chain [46]. A decrease in the yield was observed for amines with alkyl chains bearing more than 12 carbon atoms [46]. It was also observed that the yield increases with temperature and amine/carbonate ratio [46]. For an amine/carbonate ratio of 1:1.4, a 98% PHU yield is obtained when the reaction mixture is heated at 90°C for 4 h [46]. The use of additional catalysts during the synthesis of PHUs has also been investigated. Addition of catalytic amounts of 1,4-diazabicyclo[2.2.2]octane (DABCO) has been shown to allow curing to occur at room temperature [50]. Along the same

lines, it has been observed that addition of 5 mol% of LiBr enhances the ring opening polymerization reaction [51].

In 2012, Raether et al. found that both linear and branched polyhydroxyurethanes can be prepared from glycerol carbonate [52]. Raether stated that the polymerization of glycerol carbonate, an amine, and an alkylene oxide could result in either branched or linear PHUs [52]. As expected, the structure of the networks was a direct result of the reaction conditions, the temperature employed, and the reagents used [52]. Furthermore, control over the degree of branching was achieved by controlling the number of resulting hydroxyl groups on the PHU, as each additional hydroxyl group increases crosslinking [52].

PHU systems have been successfully formed through the use of 4-[(prop-2-en-1-yloxy) methyl]-1,3-dioxolan-2-one (AGC), a derivative of glycerol carbonate that can be easily obtained from the reaction of glycerol with allylic acid chlorides under basic conditions [30]. Without the use of a photoinitiator or solvent, AGC can be coupled



Figure 4.6 Synthesis of a difunctional cyclic carbonate from glycerol.

with 2,20-oxidiethanethiol, leading to the formation of difunctional cyclic carbonates [30]. This process is illustrated in Figure 4.6. The polyaddition of these difunctional cyclic carbonates with 1,10-diaminodecane results in polymers with molecular weights between 7000 g/mol and 9000 g/mol, and glass transition temperatures ranging from 14°C to 31°C [30].

In another report, phenoxycarbonyloxymethyl ethylene carbonate was prepared from glycerol carbonate and phenyl chloroformate [53]. The synthesis of this dicarbonate is shown in Figure 4.7. Both carbonate groups on the ethylene carbonate undergo polycondensation with diamines to produce amorphous PHUs with primary and secondary pendant hydroxyl groups [53]. It was observed that the glass transition temperatures of these systems decrease with increasing alkyl chain lengths of the diamines [53].

Cyclic carbonates derived from triglycerides often exhibit unclear functionality due to the complexity and variability of the triglyceride structure in vegetable oils. When cyclic carbonates are prepared from glycerol, a definite structure is obtained, and the preparation of monomers with known functionality becomes possible. Three different types of difunctional cyclic carbonate monomers have been recently prepared from glycerol carbonate [54]. In the synthesis of five-membered difunctional cyclic glyceryl carbonate diesters, fatty acids or diesters are added to glycerol carbonate through esterification or transesterification [54]. Since all materials involved in the synthesis of such difunctional carbonates are obtained from



Figure 4.7 Preparation of phenoxycarbonyloxymethyl ethylene carbonate from glycerol carbonate.

renewable resources, this process holds much potential as an alternative in the production of PHUs.

Ultimately, the importance of glycerol and glycerol carbonate may increase in the future due to the range of possible applications, availability, and low toxicity. The incorporation of glycerol into polyhydroxyurethanes presents a pathway for isocyanate-free synthesis of PHUs.

4.4.3.3 Terpenes

Terpenes consist of a wide variety of naturally occurring organic compounds biosynthesized primarily by plants from isoprene. These highly odorous compounds are well known for giving different plants their distinct scents and have been utilized as agents for aromatherapy. Overall, terpenes are characterized by a variety of differing carbon skeletons with little or no functional groups (Figure 4.8) [55]. These carbon skeletons typically include one or more carbon–carbon double bonds that have been shown to be highly reactive toward free-radical initiators for the synthesis of bio-based polymers and copolymers [56]. Terpenes also show potential as starting materials in the production of cyclic carbonates [41]. Because of the wide abundance and low cost of this environmentally friendly, carbon-rich biomass [41,56], terpenes are promising candidates for the isocyanate-free production of polyhydroxyurethanes.

The use of terpenes for the preparation of polyurethanes has been reported recently [57]. In that instance, an anionic polyol dispersion has been prepared by modifying a terpene-based epoxy resin with *p*aminobenzoic acid [57]. The dispersion was then crosslinked with a hexamethylene diisocyanate tripolymer to create a waterborne polyurethane/epoxy resin hybrid coating [57].

Unsaturated terpenes, along with unsaturated plant oils and natural polyols, have also



Figure 4.8 Isoprene-derived terpenes.



Figure 4.9 Step-wise synthesis of a PHU from limonene.

played a role in carbon dioxide fixation in the synthesis of nonisocyanate polyurethanes [41]. They have been used as intermediates along with bio-based amine curing agents in the production of PHUs from cyclic carbonates [41]. Along the same lines, linear and crosslinked terpene-based PHUs have been prepared by curing carbonated terpenes, such as limonene dicarbonate, with different diamines [58]. The step-wise synthesis is illustrated in Figure 4.9. It has been shown that terpene-based cyclic carbonates afford much higher CO₂ fixation and do not contain ester groups, which makes them stand out from conventional plant oil-based cyclic carbonates. The absence of ester groups prevents problems, such as ester cleavage and amide formation, during the amine curing step. Such problems are notorious for impairing network formation, as well as leading to low molecular weight polyol emissions, which can plastify nonisocyanate polyurethanes, and therefore compromise mechanical performance [58].

4.4.3.4 Hybrid PHUs and Composites

Organosiliconic polymers incorporate silicon-containing monomers into organic polymers and utilize silicon's unique properties to yield materials that combine high temperature stability and excellent low temperature elastomeric properties [59]. In that context, polysiloxane-modified polyhydroxyurethanes have been prepared from the reaction of five-membered cyclic carbonate polysiloxane compounds with amines, as illustrated in Figure 4.10 [60,61]. These hybrid polymers find potential applications in the production of thermal recording media, imitation leather, thermoplastic polyolefin coatings, and weather strips [60].

One issue associated with the use of polyhydroxyurethanes involves their poor water resistance due to their extended hydroxyl group functionality. To address this problem, some formulations of PHUs have been developed to confer materials with waterresistant properties. Along these lines, cyclic carbonates have been recently prepared from acrylic epoxy oligomers for the subsequent preparation of PHUs with high water and weather stabilities [58].

Bifunctional cyclic carbonates were reacted with amines to quantitatively yield PHUs that were then reacted with isophorone diisocyanate and end-capped with acrylate to form prepolymers [62]. These prepolymers were mixed with reactive diluents and liquid crystals (LCs), and subjected to UV cure



Figure 4.10 Step-wise synthesis of a PHU–polysiloxane hybrid material.

to form transparent hybrid PHU-acrylic/LC composite films [62].

PHU/attapulgite nanocomposites (PHU/ ATP) have been prepared by the in-situ surface-initiated polymerization of a fivemembered cyclic carbonate, 2,2-bis[p-(1,3dioxolan-2-one-4-yl-methoxy)phenyl] propane (B5CC), and hexamethylene diamine [63]. In this case, amino-modified attapulgite nanoparticles served as the reactive surface for the synthesis of the composite [63]. A similar technology has been employed in the preparation of PHUs reinforced with silica nanoparticles [36]. It has been proven that grafting significantly enhances the stability of the nanoparticles in the polymer matrix by increasing the affinity of its surface with organic substrates [36].

4.5 Alternative Systems

Alternative systems with a chemical structure very similar to PHUs can be obtained in an environmentally benign way from sources other than cyclic carbonates. For example, polyurethanes have been efficiently synthesized in aqueous solution by a polymerization process that is free of isocyanates and catalysts [64]. This method has been shown to be effective and environmentally friendly [64]. In the procedure, *bis*(pentafluorophenyl) carbonate was first activated by reaction with 1,6-hexanediol and polyethylene glycol at various ratios [64]. Subsequent condensation polymerization of the activated carbonates with polyetheramines under aqueous conditions resulted in polyurethanes [64]. Despite exhibiting very similar properties in comparison to PUs and PHUs, this system differs from the traditional PHUs described earlier in the text by the absence of hydroxyl groups along the final polymer backbone.

Likewise, polyurethanes with no additional hydroxyl groups can be generated from the reaction of vegetable oil-derived carbamates and diols [65]. In fact, it was recently reported that dimethyl dicarbamates can be prepared from dicarboxylic acids obtained from plant oils [65]. The carbamate precursors prepared from vegetable oils already contain urethane functional groups that are modified to yield the desired PUs. These two approaches represent interesting and sustainable alternatives for the preparation of polyurethanes from nonisocyanate sources.

4.6 Conclusions

As mentioned initially, polyurethanes are very versatile polymers that are currently used in a wide variety of industrial and large-scale applications. Extensive research efforts surrounding the chemistry of polyurethanes have resulted, over the years, in various materials with different properties, designed for specific applications. Despite its widespread use and solid chemistry background, PUs are typically synthesized from diisocyanates, which exhibit a strong bioincompatibility. For that reason, the production of PUs has raised environmental and health concerns that limit its applications to some extent.

In an effort to make PUs more environmentally friendly, an intense effort has been put into the preparation and use of bio-based polyols. Among the many options, polyols obtained from plant oils became very popular due to the availability of the starting materials and the simple chemistry involved. Although appealing, the use of bio-based polyols for the preparation of PUs does not address the main concern, i.e., the use of hazardous diisocyanates.

More recently, a new class of polymers (polyhydroxyurethanes) consisting of monomers connected by urethane linkages and additional hydroxyl groups sparked the interest of many researchers. Due to their structural similarities, such polymers exhibit properties comparable to those of regular PUs, with the added advantage of not requiring the use of diisocyanates for their synthesis. Indeed, PHUs are typically prepared by the ringopening polymerization of multifunctional cyclic carbonates and amines. Over the past few years, PHU-related research has focused mainly on the investigation of possible biobased cyclic carbonate components. Initially, the most obvious options were cyclic carbonates obtained from plant oils, due to the same reasons plant oil-based polyols are so popular in PU production.

A growing interest in alternative biobased cyclic carbonates currently arises from the need to prevent undesirable ester cleavages and low molecular weight products that are normally formed when plant oil-based cyclic carbonates are used in the synthesis of PHUs. The most current trends in alternative sources of bio-based cyclic carbonates are glycerol, which is also, to some extent, derived from plant oils, and terpenes. Current trends also point to an increased interest in hybrid PHU systems, in which PHU networks are simultaneously formed along other polymers, such as acrylics, polyolefins, epoxy, and/or polysiloxanes. It is expected that research in this area will continue to grow as novel materials with improved properties may potentially address the issues associated with the currently known systems.

References

- [1] G.W. Huber, S. Iborra, A. Corma, Chem. Rev. 106 (2006) 4044.
- [2] M.R.V.D. Mark, K. Sandefur, in: S.Z. Erhan (Ed.), Industrial Uses of Vegetable Oil, AOCS Press, Peoria, IL, 2005, pp. 143.
- [3] P.P. Kundu, R.C. Larock, Prog. Org. Coat. 65 (2009) 10.
- [4] Y. Lu, R.C. Larock, Biomacromolecules 9 (2008) 3332.
- [5] Y. Lu, R.C. Larock, ChemSusChem 3 (2010) 329.
- [6] Y. Lu, R.C. Larock, Biomacromolecules 8 (2007) 3108.
- [7] Y. Lu, Y. Xia, R.C. Larock, Prog. Org. Coat. 71 (2011) 336.
- [8] D.G. Lima, V.C.D. Soares, E.B. Ribeiro, D.A. Carvalho, E.C.V. Cardoso, F.C. Rassi, K.C. Mundim, J.C. Rubim, P.A.Z. Suarez, J. Anal. Appl. Pyrolysis 71 (2004) 987.
- [9] Y. Lu, R.C. Larock, ChemSusChem 2 (2009) 136.
- [10] F. Li, R.C. Larock, Biomacromolecules 4 (2003) 1018.
- [11] P.P. Kundu, R.C. Larock, Biomacromolecules 6 (2005) 797.
- [12] E. Delebecq, J.P. Pascault, B. Boutevin,F. Ganachaud, Chem. Rev. 113 (2013) 80.
- [13] J. Zhang, W. Tu, Z. Dai, J. Coat. Technol. Res. 10 (2013) 887.
- [14] H.H. Wang, J. Mou, Y.H. Ni, G.Q. Fei, C.L. Si, J. Zou, Express Polym. Lett. 7 (2013) 443.
- [15] J. Blackwell, M. Nagarajan, T. Hoitink, in ACS Books, Symposium Series, vol. 172, Urethane Chemistry and Applications, 1981, K.N. Edwards, W.F. Gum, J.E. Johnson, F.E. Bailey Jr, R.S. Graff, W.G. Glasser, D. Klempner, K.C.

Frisch (Eds.), ACS Publications, Washington, DC, p. 179.

- [16] R. Gu, S. Konar, M. Sain, J. Am. Oil Chem. Soc. 89 (2012) 2103.
- [17] B. Nohra, L. Candy, J. Blanco, C. Guerin, Y. Raoul, Z. Mouloungui, Macromolecules 46 (2013) 3771.
- [18] R.L. Quirino, R.C. Larock, in ACS Books, Symposium Series, vol. 1063, Renewable and Sustainable Polymers, 2011, G.F. Payne, P.B. Smith (Eds.), ACS Publications, Washington, DC, p. 37.
- [19] S. Park, F. Jin, J. Lee, Macromol. Rapid Commun. 25 (2004) 724.
- [20] S.N. Khot, J.J. Lascala, E. Can, S.S. Morye, G.I. Williams, G.R. Palmese, S.H. Kusefoglu, R.P. Wool, J. Appl. Polym. Sci. 82 (2001) 703.
- [21] L.M. Bonnaillie, R.P. Wool, J. Appl. Polym. Sci. 105 (2007) 1042.
- [22] S.P. Bunker, R.P. Wool, J. Polym. Sci., Part A: Polym. Chem. 40 (2002) 451.
- [23] H. Yeganeh, M.R. Mehdizadeh, Eur. Polym. J. 40 (2004) 1233.
- [24] Z.S. Petrovic, I. Cvetkovic, D. Hong, X. Wan, W. Zhang, T. Abraham, J. Malsam, J. Appl. Polym. Sci. 108 (2008) 1184.
- [25] K.P. Somani, S.S. Kansara, N.K. Patel, AK. Rakshit, Int. J. Adhesion Adhesives 23 (2003) 269.
- [26] A. Zlatanic, C. Lava, W. Zhang, Z.S. Petrovic, J. Polym. Sci., Part B: Polym. Phys. 42 (2004) 809.
- [27] Y. Lu, R.C. Larock, J. Appl. Polym. Sci. 119 (2011) 3305.
- [28] N. Kayaman-Apohan, E. Çakmakçı, Ç. Çiçek, M.V. Kahraman, M. Arslan, S.E. Kuruca, Chem. Eng. Trans. 32 (2013) 1087.
- [29] C. Nechifor, C.L. Ciobanu, D.O. Dorohoi, C. Ciobanu, U. P. B. Sci. Bull. Series A 71 (2009) 97.

- [30] S. Benyahya, M. Desroches, R. Auvergne, S. Carlotti, S. Caillol, B. Boutevin, Polym. Chem. 2 (2011) 2611.
- [31] J. Bernard, US patent 0312, 502 A1, 2009.
- [32] H. Tomita, F. Sanda, T. Endo, J. Polym. Sci. 39 (2001) 851.
- [33] H. Tomita, F. Sanda, T. Endo, J. Polym. Sci. 39 (2001) 4091.
- [34] S. Benyahya, B. Boutevin, S. Caillol, V. Lapinte, J. Habas, Soc. Chem. Ind. 61 (2012) 918.
- [35] B. Ochiai, H. Kojima, T. Endo, J. Polym. Sci. Part A: Polym. Chem. 52 (2014) 1113.
- [36] Z. Tang, P. Liu, J. Guo, Z. Su, Polym. Comp. 30 (2009) 445.
- [37] A.S. More, L. Maisonneuve, T. Lebarbe,B. Gadenne, C. Alfos, H. Cramail, Eur.J. Lipid Sci. Technol. 115 (2013) 61.
- [38] G.L. Wilkes, S. Seungman, B. Tamami, US patent 7,045,577 B2, 2006.
- [39] B. Tamami, S. Sohn, G.L. Wilkes, J. Appl. Polym. Sci. 92 (2004) 883.
- [40] O. Figovsky, L. Shapovalov, A. Leykin,O. Birukova, R. Potashnikova, Int. Lett.Chem. Phys. Astro. 3 (2012) 59.
- [41] H. Blattmann, M. Fleischer, M. Bahr, R. Mulhaupt, Macromol. Rapid Commun. 35 (2014) 1238.
- [42] M. Bahr, R. Mulhaupt, Green Chem. 14 (2012) 483.
- [43] M. Desroches, S. Benyahya, V. Besse, R. Auvergne, B. Boutevin, S. Caillol, Lipid Technol. 26 (2014) 35.
- [44] O. Birukov, O. Figovsky, A. Leykin, R. Potashnikov, L. Shapovalov, US patent 20,120,208,967A1, 2012.
- [45] M. Kathalewar, A. Sabnis, D. D'Mello, Eur. Polym. J. 57 (2014) 99.
- [46] B. Nohra, L. Candy, J. Blanco, Y. Raoul, Z. Mouloungui, J. Am. Oil Chem. Soc. 89 (2012) 1125.

- [47] M. Aresta, A. Dibenedetto, F. Nocito, C. Pastore, J. Mol. Catal. A Chem. 257 (2006) 149.
- [48] A. Takagaki, K. Iwatani, S. Nishimura, E. Kohki, Green Chem. 12 (2010) 578.
- [49] J. Stewart, B. Weckhuysen, P. Bruijnincx, Catal. Today 2014, in press 10.1016/j.cattod.2014.06.035.
- [50] M. Fleischer, H. Blattmann, R. Mulhaupt, Green Chem. 15 (2013) 934.
- [51] L. Annunziata, A. Diallo, S. Fouguay, G. Michaud, F. Simon, J. Brusson, J. Carpentier, S. Guillaume, Green Chem. 16 (2014) 1947.
- [52] R. Raether, J. Brand, US patent 20,120,215,020A1, 2012.
- [53] L. Ubaghs, N. Fricke, H. Keul, H. Hocker, Macromol. Rapid Commun. 25 (2004) 517.
- [54] H. Cramail, A. Boyer, E. Cloutet,B. Gadenne, A. Carine, US patent 0259087A1, 2012.
- [55] M. Bähr, R. Mühlhaupt, B.S. Ritter, Int patent WO/2012/171659, 2012.
- [56] P.A. Wilbon, F. Chu, C. Tang, Macromol. Rapid Commun. 34 (2013) 8.
- [57] G. Wu, Z. Kong, J. Chen, S. Huo, G. Liu, Prog. Org. Coat. 77 (2014) 315.
- [58] O. Figovsky, L. Shapovalov, A. Leykin,O. Birukova, R. Potashnikova, Sci. Israel Technol. Adv. 14 (2012) 95.
- [59] A. Rahimi, P. Shokrolahi, Int. J. Inorg. Mat. 3 (2001) 843.
- [60] K. Hanada, K. Kimura, K. Takahashi, O. Kawakami, M. Uruno, US patents 20,120,231,184, 20,120,232,289, and 20,120,237,701, 2012.
- [61] K. Hanada, K. Kimura, K. Takahashi,O. Kawakami, M. Uruno, US patent 20,130,171,896A1, 2013.
- [62] Y.H. Cho, B.K. Kim, J. Appl. Polym. Sci. 81 (2001) 2744.

- [63] L. Huo, T. Li, J. Guo, J. Macromol. Sci. Part A Pure Appl. Chem. 47 (2010) 457.
- [64] H. Sardon, A. Engler, J. Chan, D. Coady, J. O'Brien, D. Mecerreyes, Y. Yang, J. Hedrick, Green Chem. 15 (2013) 1121.
- [65] M. Unverferth, O. Kreye, A. Prohammer, M. Meier, Macromol. Rapid Commun. 34 (2013) 1569.

Page left intentionally blank

Larissa R. Fonseca¹, José L. Silva Sá², and Benedito S. Lima-Neto¹

¹Instituto de Química de São Carlos, Universidade de São Paulo, SP, Brazil ²Centro de Ciências da Natureza, Universidade Estadual do Piauí, PI, Brazil

	Ο U Τ L Ι Ν Ε	
5.1	Introduction	73
5.2	Processes and Monomers	74
5.3	Thermoplastic Polyesters	76
5.4	Biodegradable Polyesters	77
5.5	Unsaturated Polyester Resin (UPR)	78
5.6	Other Applications	81
5.7	Applications of Plant Oil-Based Polyester as an Alternative	
	for Petroleum-Based Polyester	82
References		

5.1 Introduction

Alternatives to replace chemicals from petroleum, silicate minerals, pure metals, wood, etc., have been searched to attend the necessities of Earth's growing population. By 2040, we will have to feed 9–10 billion people and provide them with energy and materials [1]. Important features in this search involve aspects associated with the primary source, processing and final product such as, price, shipping, weight, recycling, biocompatibility, and chemical, physical-chemical, mechanical, and thermal properties [2].

Plant oil is an environmentally friendly resource [3]. Diverse new materials derived from plant oil have been invented and inno-

vated to attend the worldwide demands with perspective to avoid environmental problems.

The success of using plant oil in various fields of science is mainly associated with the type of reaction methodology used to obtain an addressed material. In particular, reactions from plant oil are currently recognized and largely employed in polymer science [4].

The diversity of application and advantages of plant oil compared with the usual chemicals from petroleum in classical polymerization methods have led many researchers to initiate developments to obtain new polymeric materials from plant oil in the last three decades. Furthermore, many polymerization procedures have been reformulated considering that the use of plant oil



Figure 5.1 Polycondensation reaction to obtain polyester.

follows the Green Chemical Principles, with a valuable gain over the use of harmful reagents [5].

Plant oil is an important source that can be used as a precursor for different industrial applications. In the field of polymeric material, plant oils are interesting because the functional groups present in triglyceride molecules, such as double bond, hydroxyl, epoxy, ester, acid, etc., can provide a great variety of polymers, with different structures and functionalities, via a desired chemical reaction. Considering their availability and/or possible biodegradability, there is particular interest of the industry in biobased polymers, and scientific research on this type of carbon source has also increased [6,7]. This advance in biobased polymers may use this valuable natural source from oils that are not used as food or without impacts in this purpose.

Plant oil-based polymers can generate:

- polyolefins,
- polyamides,
- polyethers,
- polyesters, among others.

As it is well known, esters are produced from the condensation of a carboxylic acid with an alcohol. Thus, esters are functional groups derived from carboxylic acid, where the hydroxyl (OH) group is replaced with an alkoxy (OR) group. Natural triglycerides from plant oils present this functional group in their molecular compositions, as well as mono and diglycerides, and fatty acid methyl ester (FAME) from transesterification reactions of plant oils. The presence of the ester group in these molecules could already induce the production of a type of macromolecule containing the functional ester group via modifications of other functional groups present in the chains to be polymerized. In this type of structure, the ester group is part of the pendant unit in the main chain. This does not characterize a polyester. Differently, polyesters are defined as polymers containing at least one ester group per repeating unit (R-COO-R') in the main chain that was involved in the polymerization reaction. Thus, generally, polyesters are formed by a condensation reaction between a diacid and a dialcohol, with elimination of one water molecule (Figure 5.1). The polycondensation reaction occurs through step growth polymerization. The *mers* units R^1 and R^2 are the repeating units in the polyester main chain, and they can drive the applications depending on the used monomers.

Polyesters are one of the most important classes of polymers in the world and the demand for biobased polyester has increased in the past years aiming a more sustainable chemistry. In addition, the versatility of the ester linkage has increased the development of novel polyesters and copolyesters from plant oil, as well as their applications [8,9].

5.2 Processes and Monomers

There are many procedures for the syntheses of plant oil-modified polyesters. For example:

- monoglyceride process,
- fatty acid process,
- fatty acid-oil process.

The monoglyceride process obtained from monopolyol can occur with or without a catalyst. Without a catalyst, a temperature of at least 280°C is necessary [10]. Catalysts can be acids or bases, but acid catalysts are more efficient. Therefore, plant oil-based polyester resins can be obtained using enzyme as a catalyst. In the fatty acid process, polyesters are obtained at ca. 260°C when reacting with polyol, dibasic acid, and fatty acid, simultaneously [10]. This process is faster, more controlled, and more reproducible than the monoglyceride acid process [11,12]. The fatty acid-oil process involves fatty acid, oil, polyol, and dibasic acid, producing polyester with high viscosity, which is an advantage compared with the monoglyceride process [10].

The type of monomer, aliphatic (saturated/ unsaturated) or aromatic, is essential to define the polyester obtained from plant oils. Fatty acids can be utilized as monomers in polycondensation reactions because the carboxyl group at the end of the chain, the hydroxyl groups, occasionally present in the chain and functionalized double bonds. The characteristics of the fatty acids, such as degree of unsaturation, number of carbons, and functional group, provide different properties to the biopolymer synthesized.

If the monomer is bifunctional (f = 2) (diacid and dialcohol), a linear polyester is obtained. If the monomer has functionality greater than 2 (f > 2), in other words, more than two reactive sites on a monomer, branched polymers are obtained, and there is the possibility of forming a polymer network.

Soybean, linseed, tung, and castor oil are interesting triglycerides applied in polymer chemistry, because of the occurrence of many reactions by using the double bond or the hydroxyl group present in the fatty acids (Figure 5.2) from these oils. Thus, plant oil can also be an effective approach to reduce petrochemical compounds.

Castor oil has been a feedstock of great demand by the pharmaceutical and chemical



Figure 5.2 Structure of fatty acids.

	Oleic Acid (C18:1)	Linoleic Acid (C18:2)	Linolenic Acid (C18:3)	Any Special Fatty Acid
Castor oil	7.0	5.0	-	86–90 (ricinoleic acid)
Linseed oil	12.0–34.0	17–24	35–60	-
Soybean oil	22–34	43–56	5–11	-
Tung oil	8.0	4.0	3.0	80 (α-eleostearic acid)

Table 5.1 Fatty Acid Composition of Vegetable Oils

industry [13,14]. The high content of ricinoleic acid (Table 5.1) is the reason for the versatile value of castor oil in technology because of functionalization at either the double bond or the hydroxyl group. Soybean, linseed, and tung oil present highly unsaturated chains (Table 5.1) that are reactive in free radical or cationic polymerization.

5.3 Thermoplastic Polyesters

Different combinations of aliphatic and aromatic monomers in the principal polymeric chain can determine different types of thermoplastic polyesters for different purposes [15]. The units R^1 and R^2 (Figure 5.1) can be combined in three different ways:

• When R¹ and R² are aliphatic groups, the polyester can be used in the medical field as biomaterials for implants, drug delivery, prosthesis, etc. Further, the obtained polyester can present features such as biodegradability.

- When R¹ and R² are aromatic groups, the polyester presents superior mechanical properties and heat resistance, which are applied in high-performance thermoplastics.
- When R¹ is aliphatic and R² is aromatic, the polyesters are considered engineering thermoplastics.

Amorphous aliphatic polyesters usually present low T_g values, while amorphous aromatic polyesters present high T_g values [15]. The thermal stability is influenced by the aromatic group content, because the rings bring rigidity to the polymer chain. Hence, molecular mobility is affected by the aromatic ring sits in a plane [15].

Linear polyesters without or with pendant groups are typical thermoplastics (Figure 5.3), which are widely used because of their mechanical strength, low cost, easy



Figure 5.3 Illustration of linear polymer without (a) or with (b) pendant groups.

processability, and resistance to chemical and biological attack.

The chain length in linear polyesters can affect their properties [16–20]. One common way to obtain linear long-chain polyesters is by polycondensation of monomers synthesized from plant oil via three steps: selfmetathesis, hydrogenation, and reduction. Aliphatic polyesters are usually semicrystalline materials with low melting point owing to the absence of cross-linking.

Chain length shows a tendency to increase the melting point, as example, in the polyester ($-OC-R^1-COO-R^2-$), in which R¹ and R² are CH₂ groups [18]. When the number of CH₂ groups for R¹ is 21, the increasing number of CH₂ groups for R² changes the T_m values in the following order describing a saturation profile: $R^2 = 6$ ($T_m = 92^{\circ}C$); $R^2 = 12$ ($T_m = 100^{\circ}C$); $R^2 = 19$ ($T_m = 106^{\circ}C$); $R^2 = 23$ ($T_m = 108^{\circ}C$); $R^2 = 38$ ($T_m = 109^{\circ}C$); $R^2 = 44$ ($T_m = 111^{\circ}C$). These values are comparable with the T_m value of low-density polyethylene.

Random and block copolyesters with pendant chains were synthesized from lauric and stearic acid to obtain bifunctional compounds selectively and to promote the syntheses of linear polymers [21]. Longe side-chains increased the crystallinity and raised the melting temperature because of the side-chain crystallization. In all the cases, block copolyesters showed higher melting temperature than random copolyesters.

5.4 Biodegradable Polyesters

Since plastic waste has accumulated in the environment, leading to long-term environmental and waste management problems,



Figure 5.4 Structure of PHAs.

much concern has been placed on pollution. As biobased polymers are possible candidates to reduce pollution problems caused by plastic wastes, their biodegradability has been extensively investigated.

In response to problems with plastic waste and harmful effects to the environment, there has been considerable interest in the development of materials that could be more biodegradable than petroleumbased plastics. As a result, polymers such as polylactides, polysaccharides, and polyhydroxyalkanoates have been considered substitutes for conventional plastics, beside aliphatic polyesters.

Polyhydroxyalkanoates (PHAs) are linear polyesters of hydroxyalkanoates (HAs) produced in nature by bacterial fermentation of sugar or lipids (Figure 5.4). They can behave either as thermoplastic or as elastomeric materials, with melting points ranging from 40°C to 180°C [22]. They present useful mechanical properties and excellent biodegradability under various conditions by a multiplicity of microorganisms within a period of 1 year [22,23].

So far, PHAs have been recognized as good candidates for biodegradable plastics; unfortunately, their high price compared with conventional plastics has limited their use in a wide range of applications [22]. This occurs because of the high synthesis costs of PHAs, which limits their production on a large scale [22]. Therefore, there is a need for the development of novel processes using inexpensive carbon substrates as plant oils. The use of waste plant oils generated from the food industry and food



Figure 5.5 Structure of poly(sebacic acid-co-ricinoleic acid).

service could be reasonable from the economic and environmental point of view. The conversion of these residues into new biomaterials is an eco-innovative approach. Soybean oil, palm oil, and corn oil are desirable carbon sources for PHA production as they are relatively cheaper than most sugars [22].

Various plant oils have been tested and proved to be effective carbon sources to PHA biosynthesis with interesting material properties. The physical and material properties of PHAs depend on their monomer composition and chemical structure, such as the length of the pendent groups in the polymer backbone, the chemical nature, or the distance between the ester linkages [24]. PHAs synthesized from fatty acids depend on the synthesis procedure, as well as the degradation pathway [25]. The microorganism used and the culture conditions depend on the monomer incorporated in the polymer chain.

It has been reported that the theoretical yield of PHA production from fatty acids is 0.65 g/g, whereas the theoretical yield of PHA production from glucose ranges from 0.30 g/g to 0.40 g/g [26]. This is an advantage for plant oil.

Methyl ester of 9-hydoxynonanoic acid from castor oil produced high molecular weight linear polyester (62,200 g/mol) by self-transesterification [27]. Polyester presented higher melting point (70°C) and glass transition (-31°C) than polycaprolactone. The new polymer is potentially biodegradable and more thermally and hydrolytically stable than PHAs, with shorter methylene chains between ester groups. This may have some interesting applications in industry and medicine in substitution to polycaprolactone (PCL).

Poly(sebacic acid-*co*-ricinoleic acid) is a linear polyester with pendant aliphatic chains (Figure 5.5). It has been studied as a biodegradable polymer to act as drug delivery systems by insertion of ricinoleic acid into poly(sebatic acid). Ester bonds are more stable toward hydrolysis than anhydride bonds [28,29].

5.5 Unsaturated Polyester Resin (UPR)

Thermoset plastics present cross linked chains by covalent bonds produced during the curing process, resulting in a stable material because of the chemical bond type involved (Figure 5.6) [30].

Thermoset polyesters have been denominated as unsaturated polyester resins (UPRs), which are obtained from a typical free-radical polymerization. The chain



Figure 5.6 Illustration of cross linked polymer.

growth polymerization occurs via an unsaturated monomer and a vinyl monomer. The vinyl monomer acts as a solvent for the unsaturated polyester, decreasing the viscosity, and as an agent of cure.

UPRs are currently the most largely utilized thermosetting polymers because of their low cost, easy handling, and good balance of mechanical, electrical, and chemical properties [31]. Their room temperature cure capability, good mechanical properties, and transparency make these materials popular thermosets. Curing of unsaturated polyesters occurs through a polymerization reaction that causes cross-linking among individual linear polymer chains. In the past decade, there has been a growing trend to incorporate plant oils and/or their derivatives into UPR to produce novel biobased materials to reduce petroleum consumption [32]. The reinforced composite materials and nonreinforced materials of UPRs have been broadly employed in the aerospace, automotive, maritime, infra-structure, military, sports, and industrial fields [31].

Maleates alcoholyzed triglyceride resins from castor and soybean oils were synthesized to obtain partial biobased UPRs (Figure 5.7) [32–34]. The alcoholysis reaction can be performed with aliphatic alcohols, such as pentaerythritol and glycerol, and an aromatic alcohol, such as bisphenol A propoxylate. The resulting alcoholysis product can be maleated and cured in the presence of styrene to obtain thermoset polymers.

There is an extra hydroxyl group on each fatty acid chain of castor oil that is expected to have a maleate C = C functionality of 3–4. It was copolymerized with styrene to produce rigid plastics with high performance. This polymeric matrix showed a high cross-link density (4400 mol/m³) which was comparable with that of petroleum-derived UPRs [33,35,36]. The high functionality of the castor oil did not reduce the crosslink density of the resulting material.

Maleates are unreactive in homopolymerization, but are very reactive in copolymerization with styrene. Styrene is the typical vinyl monomer utilized in UPR synthesis because of its low viscosity, low cost, and reactivity with the unsaturated sites of polyesters [34]. Maleates alcoholyzed triglycerides resins were solids at room temperature. Thus, styrene is used in the formulations to make the resins more processable and ensure the polymerization of maleate reactive groups.

Maleates castor oil (MACO) was used as a biomodifier in UPR at different proportions (5, 10, and 15 wt%), and fly ash was used as a filler [37]. With the incorporation of 5 wt% MACO, the impact strength of the UPR matrix increased by 52% without any loss in modulus [37]. It was evident that the storage modulus (E') value gradually decreased with the addition of MACO [37]. These results indicate that brittleness of UPR-based composites can be improved by introducing MACO [37].

Epoxidized plant oils can be used in substitution to polyester resins. Methyl ester of soybean oil (EMS) and epoxidized methyl linseedate (EML) were employed to toughen UPR and biocomposites prepared from the modified UPRs [35]. Biobased resins were



Figure 5.7 Structures of maleates alcoholyzed triglycerides resins from castor and soybean oil.

prepared by a partial substitution of UPR by EMS or EML, and reinforced with natural fibers, nano clays, or layered silicates [35]. Toughness of the biobased resins was improved by the incorporation of EMS or EML. The biobased UPRs containing more than 20 wt% EML were not transparent because of the creation of EML-rich rubbery phases. The behavior of neat UPRs with 0–10 wt% EML was elastic [35].

Novel biomaterials were the blending of an unsaturated polyester resin/styrene mix with tung oil, which offered improved impact strength at a very low content of oil [38]. The principal compound of tung oil is a glyceride based on α -elaeostearic acid, a highly unsaturated conjugated system. Dicyclopentadiene was inserted in UPR, and then it was modified with tung oil via the intermolecular Diels-Alder reaction. The oil was incorporated into UPR via the intermolecular Diels-Alder reaction between double (C = C) bonds of the polyester chain and conjugated triene groups of the oil [31]. The thermal stability $(T_{10} \text{ and } T_{50})$ and toughness were improved when the UPR content tung oil was equal to or greater than 10%. The mechanical and thermal properties of the polymer matrices were influenced by the synergistic effects of phase separation and crosslink density. The mechanical properties are benefit to modify the petroleum-based materials with soft bioresources. These biomaterials, including renewable resources, are very promising in the application of fiber-reinforced composite materials.

5.6 Other Applications

The production of polyester from ricinoleic acid with and without *in situ* insertion of magnetic Fe_3O_4 nanoparticles with their surface modified by ricinoleic acid has been obtained [39]. It was observed that the magnetic Fe_3O_4 nanoparticles act as Lewis acid catalysts during the polymerization. Using the Fe_3O_4 nanoparticles modified by ricinoleic acid, the final polymeric material had an increase in the reaction rate, exhibiting good thermal stability and behaving as superparamagnetic material.

There is an enormous demand for lightweight materials for use in the transportation and construction sectors [40]. Natural fibers exhibit many advantageous properties. They are low-density materials yielding relatively lightweight composites with high specific properties [41]. Oil palm fiber is a good reinforcement in the polyester matrix. It is hard and tough, and its porous surface morphology is useful for better mechanical interlocking with matrix resin for composite production [40,42]. Thermal stability, strength, dielectric constant, and electrical conductivity were improved in the presence of oil palm fiber.

Another application related to oil palm was the development of polymer biocomposites from unsaturated isophethalic polyester resin containing powdered oil palm shell as a function of powder particle size, increasing thermal stability, and improving mechanical properties [43].

Polymeric biomaterials are currently dominated by thermoplastic polyesters such as poly(lactic acid) (PLA), poly(glycolic acid) (PGA), polycaprolactone (PCL), and their blends or copolymers [44]. Many esters, such as fatty acids, are endogenous to human metabolism, and therefore biocompatible. A biomaterial is designed to be used in intimate contact with living tissue; then, it is essential that the implanted material not causes any harmful effects to host tissues and organs [45]. In spite of the fact that many polyesters can break down to natural metabolic products by simple hydrolysis, they did not present adverse effects [44]. Although these biomaterials have been well characterized and fabricated to match the biochemical properties of soft tissues, there is generally a lack of mechanical compatibility between thermoplastic polymer implants and living tissues [45].

Some biopolymers obtained from natural oils are flexible and rubbery [46]. Generally, they are prepared as cross-linked copolymers [46].

The cross-linking process for unsaturated polyester from soybean oil fatty acid was performed by thermal and UV irradiation with 365 nm wavelength. Cross-linking was observed in at least two polyester chain double bonds [47]. The polymerization with the fatty acid from soybean oil (linoleic acid) is more reactive than the reaction with the pure soybean oil by free radical [47]. The cationic polymerization with soybean oil was difficult to occur [47]. As the crosslinking increases, the ultimate strength increases and the elongation at break decreases, obtaining very good damping and shape-memory properties over a wide temperature range [46]. The fatty acid ester groups directly attached to the polymer backbone are presumed to be responsible for damping properties [46].

Damping materials have numerous applications in the aircraft, automobile, and machinery industry, being used in noise reduction as well as in the prevention of vibrational fatigue failure [46]. Shape-memory materials have applications in civil construction, mechanics and manufacturing, electronics and communications, printing and packaging, medical equipment, recreation and sports, and household items [46].

5.7 Applications of Plant Oil-Based Polyester as an Alternative for Petroleum-Based Polyester

A straight relationship of data comparing polyester obtained from different sources is still vague in the literature [2]. However, the replacement of petroleum-based polyesters by biobased polyester from plant oil has motivated the academia and industry to obtain new materials with similar or better proprieties.

For example, methyl ester of tung oil was synthesized as comonomer to replace the more frequently used styrene in UPRs because of conjugated double bonds that are reactive in cationic polymerizations [48]. Alternatively, the incorporation of a green modifier, a commercial acrylated epoxidized soybean oil, was also considered. The use of any of these modifiers allowed preparing materials with high contents of bioderived components. All the materials obtained presented good mechanical damping capacity, as well as dynamic mechanical properties comparable to those of analogous materials obtained by copolymerization with styrene. One of the formulations prepared by copolymerization with tung oil methyl ester presented shape memory behavior.

Tartaric and phthalic acid were used individually to prepare polyesters by a reaction with glycerin, but the polyesters were found to be brittle and inflexible, whereas fatty acids with glycerin showed good film-forming properties [49].

Biofoam composites from acrylated epoxidized soybean oil resin were prepared to replace the traditional unsaturated polyester foams. The plant oil-based foam showed similar compressive stress to commercial UP foam [50].

New biobased resins for use in sheetmolding compound (SMC) applications were synthesized from soybean and linseed oil [51]. The new biobased resin from soybean oil showed mechanical properties inferior to those of commercial unsaturated polyesters [52]. However, the new biobased resin from linseed oil had properties very comparable to those of the commercial unsaturated polyesters, and they showed promise as replacement petroleum-based polymers in the SMC industry [51].

As a new material, novel approaches are in constant drive to develop new biobased monomers from plant oil to discovery different polymers for diverse applications, as well as to improve the properties of known polyesters.

Acknowledgments

The authors are indebted to the financial support from FAPESP, CAPES, and CNPq.

References

- C. Okkerse, H. Van Bekkum, From fossil to green, Green Chem. 1 (1999) 107–114.
- [2] R.T. Mathers, How well can renewable resources mimic commodity monomers and polymers?, J. Polym. Sci. Part A: Polym. Chem. 50 (2012) 1–15.
- [3] M.M. Belgacem, A. Gandini, Monomers, Polymers and Composites from Renewable Resources, Elsevier, London, 2008.
- [4] A. Gandini, The irruption of polymers from renewable resources on the scene

of macromolecular science and technology, Green Chem. 13 (2011) 1061– 1083.

- [5] M.A.R. Meier, J.O. Metzger, U.S. Schubert, Plant oil renewable resources as green alternatives in polymer science, Chem. Soc. Rev. 36 (2007) 1788–1802.
- [6] S. Miao, P. Wang, Z. Su, S. Zhang, Vegetable-oil-based polymers as future polymeric biomaterials, Acta Biomater. 10 (2014) 1692–1704.
- [7] S. Warwel, F. Brüse, C. Demes, M. Kunz, M.R.G. Klaas, Polymers and surfactants on the basis of renewable resources, Chemosphere 43 (2001) 39–48.
- [8] P. Roesle, F. Stempfle, S.K. Hess, J. Zimmerer, C. Río Bártulos, B. Lepetit, A. Eckert, P.G. Kroth, S. Mecking, Synthetic polyester from algae oil, Angew. Chem. Int. Ed. 53 (2014) 6800–6804.
- [9] C. Vilela, A.F. Sousa, A.C. Fonseca, A.C. Serra, J.F.J. Coelho, C.S.R. Freire, A.J.D. Silvestre, The quest for sustainable polyesters: insights into the future, Polym. Chem. 5 (2014) 3119–3141.
- [10] S. Paul, Surface Coatings: Science and Technology, Wiley, New York, 1996.
- [11] V.C. Malshe, M.A. Sikchi. Basics of Paint Technology: For Undergraduate Students and Industrial Practitioners, AntarPrakash Centre for Yoga, India, 2004.
- [12] W. Zeno, J. Wicks, F.N. Jones, S.P. Pappas, Organic Coatings: Science and Technology, Wiley, New York, 1999.
- [13] H. Mutlu, M.A.R. Meier, Castor oil as a renewable resource for the chemical industry, Eur. J. Lipid Sci. Technol. 112 (2010) 10–30.
- [14] V. Scholz, J.N. Da Silva, Prospects and risks of the use of castor oil as a fuel, Biomass Bioenergy 32 (2008) 95–100.

- [15] M.E. Rogers, T.E. Long, Synthetic Methods in Step-Growth Polymers, Wiley, New Jersey, 2003.
- [16] N. Kolb, M. Winkler, C. Syldatk, M.A.R. Meier, Long-chain polyesters and polyamides from biochemically derived fatty acids, Eur. Polym. J. 51 (2014) 159–166.
- [17] F. Stempfle, D. Quinzler, I. Heckler, S. Mecking, Long-chain linear C19 and C23 monomers and polycondensates from unsaturated fatty acid esters, Macromolecules 44 (2011) 4159–4166.
- [18] F. Stempfle, P. Ortmann, S. Mecking, Which polyesters can mimic polyethylene?, Macromol. Rapid Commun. 34 (2013) 47–50.
- [19] J. Trzaskowski, D. Quinzler, C. Bährle, S. Mecking, Aliphatic long-chain C20 polyesters from olefin metathesis, Macromol. Rapid Commun. 32 (2011) 1352–1356.
- [20] C. Vilela, A.J.D. Silvestre, M.A.R. Meier, Plant oil-based long-chain C26 monomers and their polymers, Macromol. Chem. Phys. 213 (2012) 2220– 2227.
- [21] S. Waig Fang, P. De Caro, P.Y. Pennarun, C. Vaca-Garcia, S. Thiebaud-Roux, Synthesis and characterization of new polyesters based on renewable resources, Ind. Crops Prod. 43 (2013) 398–404.
- [22] S. Ciesielski, J. Możejko, N. Pisutpaisal. Plant oils as promising substrates for polyhydroxyalkanoates production.
 J. Cleaner Prod. (2014), http://dx.doi. org/10.1016/j.jclepro.2014.09.040
- [23] A. Soroudi, I. Jakubowicz, Recycling of bioplastics, their blends and biocomposites: a review, Eur. Polym. J. 49 (2013) 2839–2858.
- [24] R. Rai, T. Keshavarz, J.A. Roether, A.R. Boccaccini, I. Roy, Medium chain

length polyhydroxyalkanoates, promising new biomedical materials for the future, Mater. Sci. Eng. R: Rep. 72 (2011) 29–47.

- [25] P. De Waard, H. Van Der Wal, G.N. Huijberts, G. Eggink, Heteronuclear NMR analysis of unsaturated fatty acids in poly(3-hydroxyalkanoates). Study of beta-oxidation in *Pseudomonas putida*, J. Biol. Chem. 268 (1993) 315–319.
- [26] S. Chanprateep, Current trends in biodegradable polyhydroxyalkanoates, J. Biosci. Bioeng. 110 (2010) 621–632.
- [27] Z.S. Petrović, J. Milić, Y. Xu, I. Cvetković, A chemical route to high molecular weight vegetable oil-based polyhydroxyalkanoate, Macromolecules 43 (2010) 4120–4125.
- [28] M.Y. Krasko, A. Shikanov, A. Ezra, A.J. Domb, Poly(ester anhydride)s prepared by the insertion of ricinoleic acid into poly(sebacic acid), J. Polym. Sci. Part A: Polym. Chem. 41 (2003) 1059–1069.
- [29] A. Shikanov, A.J. Domb, Poly(sebacic acid-co-ricinoleic acid) biodegradable injectable in situ gelling polymer, Biomacromolecules 7 (2005) 288–296.
- [30] J.M. Raquez, M. Deléglise, M.F. Lacrampe, P. Krawczak, Thermosetting (bio)materials derived from renewable resources: a critical review, Prog. Polym. Sci. 35 (2010) 487–509.
- [31] C. Liu, W. Lei, Z. Cai, J. Chen, L. Hu, Y. Dai, Y. Zhou, Use of tung oil as a reactive toughening agent in dicyclopentadiene-terminated unsaturated polyester resins, Ind. Crops Prod. 49 (2013) 412–418.
- [32] C. Liu, J. Li, W. Lei, Y. Zhou, Development of biobased unsaturated polyester resin containing highly functionalized castor oil, Ind. Crops Prod. 52 (2014) 329–337.

- [33] E. Can, R.P. Wool, S. Küsefoğlu, Soybean- and castor-oil-based thermosetting polymers: mechanical properties, J. Appl. Polym. Sci. 102 (2006) 1497–1504.
- [34] E. Can, R.P. Wool, S. Küsefoğlu, Soybean and castor oil based monomers: synthesis and copolymerization with styrene, J. Appl. Polym. Sci. 102 (2006) 2433–2447.
- [35] H. Miyagawa, A.K. Mohanty, R. Burgueño, L.T. Drzal, M. Misra, Development of biobased unsaturated polyester containing functionalized linseed oil, Ind. Eng. Chem. Res. 45 (2006) 1014– 1018.
- [36] H. Miyagawa, A.K. Mohanty, R. Burgueño, L.T. Drzal, M. Misra, Novel biobased resins from blends of functionalized soybean oil and unsaturated polyester resin, J. Polym. Sci. Part B: Polym. Phys. 45 (2007) 698–704.
- [37] S. Ghorui, N.R. Bandyopadhyay, D. Ray, S. Sengupta, T. Kar, Use of maleated castor oil as biomodifier in unsaturated polyester resin/fly ash composites, Ind. Crops Prod. 34 (2011) 893–899.
- [38] K. Das, D. Ray, C. Banerjee, N.R. Bandyopadhyay, A.K. Mohanty, M. Misra, Novel materials from unsaturated polyester resin/styrene/tung oil blends with high impact strengths and enhanced mechanical properties, J. Appl. Polym. Sci. 119 (2011) 2174–2182.
- [39] E.U.X. Péres, F.G.D. Souza Jr., F.M. Silva, J.A. Chaker, P.A.Z. Suarez, Biopolyester from ricinoleic acid: synthesis, characterization and its use as biopolymeric matrix for magnetic nanocomposites, Ind. Crops Prod. 59 (2014) 260–267.
- [40] M.D.H. Beg, M.F. Mina, R.M. Yunus, A.K.M. Moshiul Alam, 12 – The use of

oil palm biomass (OPB) fibers as reinforcements in composites, in: O Faruk, M Sain (Eds.), Biofiber Reinforcements in Composite Materials, Woodhead Publishing, United Kingdom, 2015.

- [41] A. O'Donnell, M.A. Dweib, R.P. Wool, Natural fiber composites with plant oilbased resin, Compos. Sci. Technol. 64 (2004) 1135–1145.
- [42] S. Shinoj, R. Visvanathan, S. Panigrahi, M. Kochubabu, Oil palm fiber (OPF) and its composites: a review, Ind. Crops Prod. 33 (2011) 7–22.
- [43] O. Nabinejad, D. Sujan, M.E. Rahman, I.J. Davies, Effect of oil palm shell powder on the mechanical performance and thermal stability of polyester composites, Mater. Des. 65 (2015) 823–830.
- [44] B.L. Seal, T.C. Otero, A. Panitch, Polymeric biomaterials for tissue and organ regeneration, Mater. Sci. Eng. R: Rep. 34 (2001) 147–230.
- [45] Q. Chen, S. Liang, G.A. Thouas, Elastomeric biomaterials for tissue engineering, Prog. Polym. Sci. 38 (2013) 584–671.
- [46] V. Sharma, P.P. Kundu, Addition polymers from natural oils: a review, Prog. Polym. Sci. 31 (2006) 983–1008.
- [47] B. Hazer, S.I. Demirel, M. Borcakli, M.S. Eroglu, M. Cakmak, B. Erman, Free radical crosslinking of unsaturated bacterial polyesters obtained from soybean oily acids, Polym. Bull. 46 (2001) 389–394.
- [48] C. Meiorin, M.I. Aranguren, M.A. Mosiewicki. Polymeric networks based on tung oil: reaction and modification with green oil monomers. Eur. Polym. J. 67 (2015) 551–560 http://dx.doi.org/ 10.1016/j.eurpolymj. 2015.01.005.
- [49] M.R. Islam, M.D.H. Beg, S.S. Jamari, Development of vegetable-oil-based

polymers, J. Appl. Polym. Sci. 131 (2014) n/a-n/a.

- [50] S.P. Wu, J.F. Qiu, M.Z. Rong, M.Q. Zhang, L.Y. Zhang, Plant oil-based biofoam composites with balanced performance, Polym. Int. 58 (2009) 403– 411.
- [51] J. Lu, R.P. Wool, Novel thermosetting resins for SMC applications from linseed oil: synthesis, characterization, and properties, J. Appl. Polym. Sci. 99 (2006) 2481–2488.
- [52] J. Lu, S. Khot, R.P. Wool, New sheet molding compound resins from soybean oil. I. Synthesis and characterization, Polymer 46 (2005) 71–80.

Fana Teffera, Michael J. Forrester, and Eric W. Cochran

Department of Chemical and Biological Engineering, Iowa State University, Ames, IA, USA

Ο U T L I N E

6.1	Background				
	6.1.1	Introduction	87		
	6.1.2	Molecular Weight and Networks	88		
6.2	Metho	ods	89		
	6.2.1	Extraction Process from Natural Source	89		
	6.2.2	Purifications after Extractions	89		
	6.2.3	Modification from Oil to Polyether Monomers	90		
	6.2.4	Polymerization from Oil Source to Polyether	90		
6.3	Prope	rties of Plant Oil-Based Polyethers	91		
6.4	Applications				
6.5	Chall	enges and Trends	94		
Ref	References				

6.1 Background

6.1.1 Introduction

Plant oils are mainly triglycerides comprising a variety of different fatty acids. The fatty acid composition differs on the basis of the type of the plant and the plant's growing environment. These differences could be in chain length, composition, distribution, and location [1,2]. Historically, fossil oils have been used as the main source of polymers; however, the use of plant oils for the production of polymers has been increasing each year. Using plant oils as a source for polymers has many advantages over using petroleum as a polymer source. Their advantage over fossil-based polymers is shown by the following facts [3–9]:

- Plant oil-based polymers, such as vegetable oil-based polymers, are environmentally safer compared to the fossil oil-based alternative; this is due to the fact that plant oil-based polymers are at worst partially, and sometimes fully, biodegradable.
- Compared to petroleum-based polymer, plant oils are cheaper to extract. They also have a more environmentally friendly extraction process.



Figure 6.1 (a) Generic repeat unit of a polyether. (b) General form of triglyceride which will be processed and monomerized for the polymerization of polyether.

Plant oils are more abundant than fossil oil. This abundance – combined with the low cost of extraction, as noted previously – makes them an attractive choice for industrial use. According to Cargill's estimation, 2200 barrels of crude oil could be saved by replacing 1 million pounds of traditional polyol with soybean-oil-based polyols [2]. The current development of synthetic, enzymatic, and microbiological methods is realizing such replacements progressively [10,11].

Extraction of plant oils from these sources has been going on since distant past, but commercialization on larger scale has been following the advancement of polymer science. Plant oil seeds, such as soybean and corn, can be genetically modified to influence the extraction process. More research is needed in this area; however, research has shown that it is possible to modify the contents of the fatty acid for the oil seed with the product in consideration [2]. Significant parts of the vegetable oil production are from soybean, palm, rapeseed, sunflower, castor, tobacco, linseed oil, corn, Mesua ferrea, agremone, and canola oil [6]. Unsaturation can be measured by using iodine value (IV), which is used by measuring amount of iodine in milligram that reacts with carboncarbon double bonds in 100 g of vegetable oil [12]. Vegetable oil is a source for 80% of oil and fat production globally, of which

around 25% is soybean [10]. Refined soy oil consists of more than 99% triglycerides and a significant variety of fatty acids [2]. The chemical and physical properties of these oils significantly depend on the degree of unsaturation.

Polyethers are formed by creating ether links between monomers. Varying the amount of functional groups in the monomer will have a large impact on the properties of the polymer. Polyethers can be linear, branched, or cross-linked. Plant oil-based polyethers are derived from plant-oil based monomers. Figure 6.1 shows an example of a polyetherbased plant-oil and triglyceride.

6.1.2 Molecular Weight and Networks

Polymer material properties are highly dependent upon the molecular weight; plant oil-based polyethers can vary from oligomers to very large polymers. Typically materials that have low molecular weights tend to be much softer than materials with higher molecular weights. Also, it is much easier to produce materials that are not cross-linked at lower molecular weights. This means that at lower molecular weights it is far simpler to produce melt-processable materials. Linear polyethers are very difficult to form from triglyceride-based monomers, owing to the intrinsically multifunctional nature of triglycerides. These monomers may, on average, have only one functional group; however, it is very difficult to have a material that has an absolute value of one functional group. Due to this, making linear polymers from plant-based sources is impractical. The exception to this is if the plant oil is cleaved (e.g., through transesterification), leaving behind monofunctional fatty acids and fatty esters. Formation of cross-linked polyethers can be controlled by manipulating the context of the mixer and the nature of crosslinking agent [11]. Polyether dendrimers have an advantage over linear polyethers due to their high surface area-to-volume ratios, flexibility in functionality, and low polydispersity index [4,13].

6.2 Methods

6.2.1 Extraction Process from Natural Source

Extraction of the oil from the oil seeds is mainly achieved using mechanical or solvent-based extraction methods. The extraction method used is dependent on the size of the industry and the requirement on efficiency. The cost of the extraction process may rise proportionally with the efficiency of extracting oils from the oil seeds [2]. During mechanical extraction, squeezing the oil out of the oilseeds by using mechanical means, the protein denaturation is also achieved due to the heat generated through the extraction. This process is mainly popular with industries with capital constriction but efficiency tolerance. This method has the advantage of being more environmen-

tally friendly than solvent-based extraction. Solvent-based extraction is done by diffusing the oil out of the seed using solvents, such as hexane, as a carrier. The factors for diffusion rate are the characteristics of solvent. thickness, and size of the oil flakes. Since the extraction is diffusion dependent, the efficiency is tied with the choice of the solvent. The solvent choice is with the consideration of cost of the solvent and the efficiency; here, it will be the desired rate of oil diffusion from the oil seeds [14]. As shown by Othmer and Agarwal [15], the relationship between concentration of oil and of thickness of the flake is given as $-\frac{dC}{dt} = kF^{3.97}C^{3.5}$, where C is concentration of oil, F is the flake thickness, t is time, and k is the rate constant.

6.2.2 Purifications after Extractions

Additional extraction methods are introduced after the crude oil is extracted from the seed. The method chosen is tailored to the final product, but can include: removing phospholipids (degumming), removing free fatty acid, bleaching, and deodorizing. Phospholipids between 500 ppm and 900 ppm are not suitable for storage and transport due to the unwanted cake they form. This removal could be easily achieved by using warm water. For most applications, the fatty acid content will also need to be around 0.05% to improve the smoke point and foaming upon heating. Further bleaching is introduced to improve the appearance, especially for food-related products, and to remove more residual impurities. Deodorization is also used as needed in the final product [16,17].



Figure 6.2 Epoxidization of triglyceride by the *in situ* peracid method as described in [4].

6.2.3 Modification from Oil to Polyether Monomers

If the triglyceride does not naturally contain epoxy or hydroxyl moieties, it is necessary to functionalize the triglyceride with chemical groups – such as maleates, hydroxyl, or epoxy – through the unsaturated double bond to use the triglyceride for either polycondensation or ring opening mechanism. An example for naturally occurring epoxidized triglyceride is vernonia oil. Additional functionalities for the natural epoxy or hydroxyl triglyceride can be attached as well, i.e., vinyl functionalities [2,14,18,19] (see Figure 6.2).

6.2.4 Polymerization from Oil Source to Polyether

Natural unsaturation in vegetable oils presents a number of possibilities for monomers with minimal chemical processing. For example, vegetable oil monomers can be used to create polyethers through manipulation of the reactive sites of the monomers [20]. Triglycerides, comprising fatty acid chains with 12-18 carbon bonds and less than four double bonds, can be used to create polyethers by manipulating the double bonds, allyic carbons, and carbon alpha to the ester groups [21]. Glycolysis or amidation could also be used to convert triglycerides into monoglycerides. This converted monoglyceride could also be manipulated with various attachments to prepare suitable monomers for polymerization [5]. For example, it is possible to prepare monomer that is ready for free-radical polymerization from triglyceride using maleic anhydride after hydroxylation [2,22,23]. The thiol-ene addition is used to functionalize triglycerides without a solvent in a single step to produce bio-based polyols from a vegetable oil [24].

The popular way of producing vegetableoil-based polyethers is using cationic polymerization of epoxidized vegetable oils, which usually has higher reactivity compared to other oils [3,20]. Earlier practices of using epoxidized castor, soybean, and linseed oils to produce polyethers by photo-initiation did not have adequate mechanical properties. The desired mechanical properties are



Figure 6.3 Epoxidized methyl oleate that is derived from the triglyceride [42].

achieved by increasing the degree of crosslinking, like with epoxidized soybean oil [20,25].

Epoxidation of a triglyceride can be achieved using conventional chemical treatment, acid ion exchange resin, using enzymes, using metal catalysts and other means. Such epoxidation methods can be briefly stated as follows:

- Conventional chemical treatment includes the use of carboxylic acid with concentrated hydrogen peroxide. Although use of H₂SO₄ as a catalyst for hydrogen peroxide has been proven to be most successful, works of Jai Lian Kun et al. showed that production of cottonseed oil-based polyether polyols through ring opening reaction after epoxidation while using H₂SO₄ as acid catalysis will lead to multiple undesired side reactions. The more stable way of producing epoxidized soybean oil is to use formic acid as the catalyst [26–31].
- Acid ion exchange resin methods (AIER) are the use of peroxy acid to fill the AIER pores to lower oxirane degradation by limiting the competition with triglyceride. This will also improve the selectivity and reduce undesired side reactions in the epoxidation process [2,26,32–36].
- Enzymatic methods are more environmentally friendly and less prone to side reactions. Despite low stabil-

ity of lipase, usage of Immobilized Candida Antarctica lipase as a catalyst will improve the epoxidation reaction in temperature and enzyme load controlled environment. Lipase catalysis can be taken as an alternative of chemical treatment and they are recyclable. Warwel also showed that epoxidation of plant oils such as sunflower, soybean, linseed, and rapeseed can have high selectivity and more than 90% conversion [37–41].

Metal catalysts usually have low selectivity, is difficult to separate, and is corrosive to reactors due to the strong acids. Titanium, molybdenum, and tungsten are commonly used metal catalysts. Such metal catalysts are usually used to improve the epoxidation by increasing oxirane content, such as significantly decreasing degradation of oxirane ring by using amorphous heterogeneous Ti/SiO₂ catalysis in *tert*-butyl alcohol as shown in Campanella et al. work [26, 42, 43] (see Figure 6.3).

6.3 Properties of Plant Oil-Based Polyethers

As shown by Lligadas et al., epoxidized methyl oleate (EMO), derivative from high oleic sunflower oil, can be used to prepare polyether polyols by ring-opening cationic polymerization of EMO oligomers and partial reduction of the carboxylate groups to hydroxyl moieties with a reducing agent of lithium aluminum hydride (LiAlH₄). Partial reduction of ester groups was used to increase the variety of polyols that also have different physical characteristics at room temperature [42–44]. Preparation of polyethers from EMO is also possible by using ionic-coordinative ring opening mechanisms, as shown by Del Rio et al. This work shows the use of a limited amount of THF to copolymerize the EMO to avoid the competitions of transesterification which formed branched ester groups. These approaches will be able to produce linear polyether polyols with 6–8 kDa with limited hydroxyls [43]. Both linear and dendritic glycerol can also be used to produce biobased polyethers by using alkoxide catalysis, which different types of epoxide monomers can be copolymerized with glycidol. It was shown that hyperbranched polyols with a well-defined polyether arrangement could be formed by controlling the monomer addition and cyclization [45].

As shown by Alam et al., Pongamia oil can be used to produce a polyether product that posses good chemical resistance and physio-mechanical performance, PEFA-BMF45. They used Pongamia oil to produce N,N'-bis(2-hydroxy ethyl) Pongamia glabra oil fatty amide (HEPFA) by mixing it with diethanolamine and using HEPFA and cyclohexanedimethanol mix to produce poly(ether fatty amide) (PEFA). Then, they mixed PEFA with butylated melamine formaldehyde (BMF) (45) in xylene solvent with phosphoric acid catalyst to produce PEFA-BMF45 [46]. Recent work of their group shows polyether fatty amide, modified with tannic acid, can be used as a coating with excellent corrosion protection properties for carbon steel, with stability up to 300°C [47].

BMF modified poly(ether fatty acid amide) that shows excellent chemical resistance and thermal stability can be prepared by modifying HEPFA with cyclohexanedimethanol Pongamia oil with Diethanolamine Manawwer. Alam et al. prepared poly(ether fatty acid amide) (PEFA) by mixing amine derivative of pongamia glabra oil and cyclohexanedimethanol. Photon initiated, radical mediated thiol-ene click reactions are used for the production of cross linked polymer network with various frameworks and functional groups. Such polymerization have fast reaction rate, better regioselectivity, and low polymerizations with limited oxygen inhibition [46,48,49]. A flame retardant bio oil-based polyether was synthesized by Hoffendahl et al. This material showed excellent resistance to flame and was remarkably stable up to 320°C with most of the degradation beginning at 455°C [50] (see Figure 6.4).

Plant oil-based polyethers have properties that can range depending on the type of material that is desired. In a paper by Lu and Larock, a polyurethane that was synthesized from methoxylated soybean oil was investigated. They discovered that these are materials that are amorphous having glass transitions anywhere from 8.9°C to 33.5°C depending on the number of hydroxyl groups present in the soybean molecule. These materials can also widely vary in mechanical properties depending on how the material is synthesized. The modulus, tensile strength, and elongation before breaking are highly dependent on the amount of soybean oil present in the preparation of these polyurethanes [51,52]. In a review article, Lu et al. talked about the synthesis of another soybean oil-based polyurethane. Depending on the molecular weight of the polyether, the final polyurethane will have very different properties. For example, plant oil-based



Figure 6.4 Preparation of poly(ether fatty amide) (PEFA). Adapted from [46].

polyethers that have molecular weights above 3000 g/mol tend to produce softer flexible polyurethanes, whereas plant-based polyethers that have molecular weights in the range of 200–1200 g/mol tend to make very stiff, rigid polymers [53]. Copolymerizing the monomer by using polyhedral oligomeric silsesquioxane (POSS) compounds to modify the epoxy networks can improve the temperature stability and mechanical properties. Also, during the processing of the polymeric material, it will lower flammability, heat evolution, and viscosity [3,19].

6.4 Applications

Polyethers based on plant-based oils properties are incredibly diverse depending on many factors. This leads to a great number of potential applications especially in the production of polyurethanes, which is thus far the most substantial application for plant oil-based polyethers. It has been demonstrated that plant oil based polyurethanes have a very wide range of mechanical and thermal properties and are very stable at temperatures below 300°C [51]. Silicon/plant oil-based polyurethanes have applications as a flame retardant [1]. Plant oil-based polyethers can also be used to synthesized waterborne dispersions [51,54]. The biomedical field has large potential for the use of plant oil-based polyurethanes. Zdrahala et al. published a review article with the potential uses of polyurethanes in the biomedical fields for materials. For example, they talk about the use of polyurethanes as an implant to support an artificial organ [55]. It is possible that plant oil-based polyurethanes might have potential in this field as they are materials that have little to no toxicity and their properties are very customizable.

Polyether polyols are usually produced by the anionic ring-opening polymerization of alkylene oxides such as ethylene oxide or propylene oxide. Polyether polyols with molecular weights of 200-10,000 g/mol can be used for the formation of polyurethane, whereas rigid polyurethanes can be produced from such polyols with molecular weight of 200-1200 g/mol and flexible polyurethanes can be produced from molecular weight of above 3000 g/mol [3,36,56]. Rigid polyurethane foams are widely used in thermal insulation and packaging. Due to the uniqueness castor oil, having both unsaturated and nonconjugated hydroxyl functions, can react with polyfunctional isocyanides to form polyurethanes with a wide range of physical characteristics. It is also used to improve polyether-based foam [57-59]. Polyurethanes are used for coatings, adhesives, medical devices, and in textile industries [60]. Wu et al. showed that epoxidized rapeseed can be used as a biodegradable lubricant due to the polyether material formation on the frictional surface through tribo-polymerization [61]. As the works of Miao et al. show, polyglycerol-derived hyperbranched dendritic polyether can be used as a flexible, highly interactive drug delivery means [20,45]. Gupta et al. showed the reasonable effectiveness of polyether as an injectable drug delivery system. Their study shows that standard -PEG demonstrated ideal biocompatibility [62]. Use of polyether derivative as a lubricant was also shown in the Zhang et al. work in which they have shown the use of aqueous amine-polyoxyethylene-polyoxypropyleneether, KE-1, solution to reduce friction and wear. The antiwear property was reported improved and friction coefficient lowered for both steel-steel and steel-aluminum frictions [63]. Louie et al. showed that we can have smoother membrane surfaces by applying 1% of polyether-polyamide block copolymer (PEBAX) solution on polyamide reverse osmosis membranes. They found

that such coating will improve the fouling resistance against a surfactant, water emulsion feed or model oil for both high and low flux reverse osmosis membranes [64]. Works by Zheng et al. showed that perfluoropolyether (PFPE) can also be used as a lubricant for very thin layer of PEEK coating [65]. Alam's work shows that urethane modified plyetheramide resins (UPEtA) that is derived from linseed oil can be used as anticorrosion lubricant. They showed that this resins showed good chemical resistance in acid, alkalie, and organic solvent where UPEtA-24 outperformed other variants. The synthesized resins decomposed only 5% at 190°C and were decomposed 75% at 500°C [66].

6.5 Challenges and Trends

More work is needed to increase the effectiveness of the monomerization of fatty acid with functional groups. Considering most of the petroleum-based polyethers can be replicated using the bio-based oil sources, improving this process is important to accelerating the use of bio-based oil as a substitute. More work is needed to understand the effect of protein structures, oil content, and fatty acid type in the final product. The current trend of genetic modification of oilseeds to lower the production cost as well as the extraction efficiency needs to be strengthened more.

Modification of existing catalyst and producing new catalyst are also an important means of reducing the cost, time needed, and improving conversion. Even if improvement of catalysts is a significant solution to current challenges, developing more efficient synthesis methods without catalyst or with a limited use of catalyst will remove or reduce the dependency of monomerization on catalyst. More work is also needed in terms of developing an assessment that will help the industry understand the challenges and opportunity of the academic research. More work on production of polyether or polyether-specific monomers by thiol–ene coupling reactions with fatty acid derivatives is also needed [67,68]. Finally, these materials have exciting properties, and developing further applications is a must to make these polymers scalable.

References

- M.A. Meier, J.O. Metzger, U.S. Schubert, Plant oil renewable resources as green alternatives in polymer science, Chem. Soc. Rev. 36 (11) (2007) 1788–1802.
- [2] R. Wool, X.S. Sun, Bio-Based Polymers and Composites, Academic Press, New York, 2011.
- [3] M. Galià, L.M. de Espinosa, J.C. Ronda, G. Lligadas, V. Cádiz, Vegetable oil-based thermosetting polymers, Eur. J. Lipid Sci. Technol. 112 (1) (2010) 87–96.
- [4] N. Karak, Vegetable Oil-Based Polymers: Properties, Processing and Applications, Elsevier, Burlington, 2012.
- [5] G. Lligadas, J.C. Ronda, M. Galià, V. Cádiz, Oleic and undecylenic acids as renewable feedstocks in the synthesis of polyols and polyurethanes, Polymers 2 (4) (2010) 440–453.
- [6] B.-J.K. Ahn, S. Kraft, X.S. Sun, Chemical pathways of epoxidized and hydroxylated fatty acid methyl esters and triglycerides with phosphoric acid, J. Mater. Chem. 21 (26) (2011) 9498– 9505.

- [7] J.O. Metzger, Fats and oils as renewable feedstock for chemistry, Eur. J. Lipid Sci. Technol. 111 (9) (2009) 865–876.
- [8] F. Liu, J. Zhu, Plant-oil-based polymeric materials and their applications, Green Mater. Plant Oils (2014) 93.
- [9] R.L. Shogren, Z. Petrovic, Z. Liu, S.Z. Erhan, Biodegradation behavior of some vegetable oil-based polymers, J. Polym. Environ. 12 (3) (2004) 173–178.
- [10] J. Metzger, U. Bornscheuer, Lipids as renewable resources: current state of chemical and biotechnological conversion and diversification, Appl. Microbiol. Biotechnol. 71 (1) (2006) 13–22.
- [11] H.M. Kim, H.R. Kim, C.T. Hou, B.S. Kim, Biodegradable photo-crosslinked thin polymer networks based on vegetable oil hydroxy fatty acids, J. Am. Oil. Chem. Soc. 87 (12) (2010) 1451–1459.
- [12] Y. Xia, R.C. Larock, Vegetable oilbased polymeric materials: synthesis, properties, and applications, Green Chem. 12 (11) (2010) 1893–1909.
- [13] M.A. Carnahan, M.W. Grinstaff, Synthesis and characterization of polyetherester dendrimers from glycerol and lactic acid, J. Am. Chem. Soc. 123 (12) (2001) 2905–2906.
- [14] D. Swern, Bailey's Industrial Oil and Fat Products, Wiley, New York, 1982.
- [15] D.F. Othmer, J.C. Agarwal, Extraction Of soybeans: theory and mechanism, Chem. Eng. Prog. 51 (8) (1955) 372–378.
- [16] G. Karnofsky, The theory of solvent extraction, J. Am. Oil Chem. Soc. 26 (10) (1949) 564–569.
- [17] L. Wiedermann, Degumming, refining and bleaching soybean oil, J. Am. Oil Chem. Soc. 58 (3) (1981) 159–166.
- [18] V. Sharma, P. Kundu, Addition polymers from natural oilsa review, Prog. Polym. Sci. 31 (11) (2006) 983–1008.
- [19] G. Lligadas, J.C. Ronda, M. Galià, V. Cádiz, Bionanocomposites from renewable resources: epoxidized linseed oil-polyhedral oligomeric silsesquioxanes hybrid materials, Biomacromolecules 7 (12) (2006) 3521–3526.
- [20] S. Miao, P. Wang, Z. Su, S. Zhang, Vegetable-oil-based polymers as future polymeric biomaterials, Acta Biomater. 10 (4) (2014) 1692–1704.
- [21] L. Keshun, et al. Soybeans: Chemistry, Technology, and Utilization, Chapman and Hall, New York, 1997.
- [22] C.G. Force, F.S. Starr, Softeners, US patent 4,740,367 (Apr 26, 1988).
- [23] Y.H. Hui, A.E. Bailey, Bailey's Industrial Oil and Fat Products. 5. Industrial and Consumer Nonedible Products from Oils and Fats, Wiley, New York, 1996.
- [24] S. Caillol, M. Desroches, S. Carlotti, R. Auvergne, B. Boutevin, Synthesis of new polyurethanes from vegetable oil by thiol–ene coupling, Green Mater. 1 (1) (2013) 16–26.
- [25] S. Chakrapani, J.V. Crivello, Synthesis and photoinitiated cationic polymerization of epoxidized castor oil and its derivatives, J. Macromol. Sci. Part A: Pure Appl. Chem. 35 (4) (1998) 691–710.
- [26] T. Saurabh, M. Patnaik, S. Bhagt, V. Renge, Epoxidation of vegetable oils: a review, Int. J. Adv. Eng. Technol. 2 (4) (2011) 491–501.
- [27] S. Dinda, A.V. Patwardhan, V.V. Goud, N.C. Pradhan, Epoxidation of cottonseed oil by aqueous hydrogen peroxide catalysed by liquid inorganic acids, Bioresource Technol. 99 (9) (2008) 3737–3744.
- [28] V.V. Goud, A.V. Patwardhan, N.C. Pradhan, Studies on the epoxidation of mahua oil (madhumica indica) by hy-

drogen peroxide, Bioresource Technol. 97 (12) (2006) 1365–1371.

- [29] L.K. Jia, L.X. Gong, W.J. Ji, C.Y. Kan, Synthesis of vegetable oil based polyol with cottonseed oil and sorbitol derived from natural source, Chin. Chem. Lett. 22 (11) (2011) 1289–1292.
- [30] C. Cai, H. Dai, R. Chen, C. Su, X. Xu, S. Zhang, L. Yang, Studies on the kinetics of *in situ* epoxidation of vegetable oils, Eur. J. Lipid Sci. Technol. 110 (4) (2008) 341–346.
- [31] G.J. Piazza, A. Nuñez, T.A. Foglia, Hydrolysis of mono-and diepoxyoctadecanoates by alumina, J. Am. Oil Chem. Soc. 80 (9) (2003) 901–904.
- [32] V.V. Goud, A.V. Patwardhan, S. Dinda, N.C. Pradhan, Epoxidation of karanja (pongamia glabra) oil catalysed by acidic ion exchange resin, Eur. J. Lipid Sci. Technol. 109 (6) (2007) 575–584.
- [33] R. Mungroo, N.C. Pradhan, V.V. Goud, A.K. Dalai, Epoxidation of canola oil with hydrogen peroxide catalyzed by acidic ion exchange resin, J. Am. Oil Chem. Soc. 85 (9) (2008) 887–896.
- [34] S. Sinadinović-Fišer, M. Janković, Z.S. Petrović, Kinetics of *in situ* epoxidation of soybean oil in bulk catalyzed by ion exchange resin, J. Am. Oil Chem. Soc. 78 (7) (2001) 725–731.
- [35] S. Dinda, V. Goud, A. Patwardhan, N. Pradhan, Selective epoxidation of natural triglycerides using acidic ion exchange resin as catalyst, Asia-Pacific J. Chem. Eng. 6 (6) (2011) 870–878.
- [36] Z.S. Petrović, A. Zlatanić, C.C. Lava, S. Sinadinović-Fišer, Epoxidation of soybean oil in toluene with peroxoacetic and peroxoformic acidskinetics and side reactions, Eur. J. Lipid Sci. Technol. 104 (5) (2002) 293–299.

- [37] U. Törnvall, C. Orellana-Coca, R. Hatti-Kaul, D. Adlercreutz, Stability of immobilized candida antarctica lipase b during chemo-enzymatic epoxidation of fatty acids, Enzyme Microbiol. Technol. 40 (3) (2007) 447–451.
- [38] C. Orellana-Coca, S. Camocho, D. Adlercreutz, B. Mattiasson, R. Hatti-Kaul, Chemo-enzymatic epoxidation of linoleic acid: parameters influencing the reaction, Eur. J. Lipid Sci. Technol. 107 (12) (2005) 864–870.
- [39] U. Biermann, W. Friedt, S. Lang, W. Lühs, G. Machmüller, J.O. Metzger, M. Ruesch gen Klaas, H.J. Schaefer, M.P. Schneider, New syntheses with oils and fats as renewable raw materials for the chemical industry, Angew. Chem. Int. Ed. 39 (13) (2000) 2206–2224.
- [40] S. Sun, X. Ke, L. Cui, G. Yang, Y. Bi, F. Song, X. Xu, Enzymatic epoxidation of sapindus mukorossi seed oil by perstearic acid optimized using response surface methodology, Ind. Crops Prod. 33 (3) (2011) 676–682.
- [41] S. Warwel, et al. Complete and partial epoxidation of plant oils by lipasecatalyzed perhydrolysis, Ind. Crops Prod. 9 (2) (1999) 125–132.
- [42] A. Campanella, M.A. Baltanas, M.C. Capel-Sanchez, J.M. Campos-Martin, J.L.G. Fierro, Soybean oil epoxidation with hydrogen peroxide using an amorphous Ti/SiO₂ catalyst, Green Chem. 6 (7) (2004) 330–334.
- [43] M.T. Benaniba, N. Belhaneche-Bensemra, G. Gelbard, Kinetics of tungsten-catalyzed sunflower oil epoxidation studied by 1H NMR, Eur. J. Lipid Sci. Technol. 109 (12) (2007) 1186–1193.
- [44] S. Caillol, M. Desroches, G. Boutevin,C. Loubat, R. Auvergne, B. Boutevin,Synthesis of new polyester polyols

from epoxidized vegetable oils and biobased acids, Eur. J. Lipid Sci. Technol. 114 (12) (2012) 1447–1459.

- [45] M. Calderón, M.A. Quadir, S.K. Sharma, R. Haag, Dendritic polyglycerols for biomedical applications, Adv. Mater. 22 (2) (2010) 190–218.
- [46] M. Alam, N.M. Al-Andis, Synthesis and characterization of poly (etherfattyamide) coatings from non edible seed oil, Pigment Resin. Technol. 42 (3) (2013) 195–201.
- [47] M. Alam, N.M. Alandis, Tannic acidmodified fatty amide anticorrosive coatings from *pongamia glabra* oil, Anti-Corrosion Methods Mater. 61 (4) (2014) 232–240.
- [48] T. Yoshimura, T. Shimasaki, N. Teramoto, M. Shibata, Bio-based polymer networks by thiol-ene photopolymerizations of allyl-etherified eugenol derivatives, Eur. Polym. J. 67 (2015) 397–408.
- [49] C. Fu, Z. Zheng, Z. Yang, Y. Chen, L. Shen, A fully bio-based waterborne polyurethane dispersion from vegetable oils: from synthesis of precursors by thiol–ene reaction to study of final material, Prog. Org. Coat. 77 (1) (2014) 53–60.
- [50] J.C. Ronda, G. Lligadas, M. Galià, V. Cádiz, A renewable approach to thermosetting resins, Reactive Funct. Polym. 73 (2) (2013) 381–395.
- [51] Y. Lu, R.C. Larock, Soybean-oil-based waterborne polyurethane dispersions: effects of polyol functionality and hard segment content on properties, Biomacromolecules 9 (11) (2008) 3332–3340.
- [52] M. Chrysanthos, Novel biobased epoxy networks derived from renewable resources: structure-property relationships, PhD thesis, INSA de Lyon 2012.

- [53] M. Sacristan, J.C. Ronda, M. Galia, V. Cadiz, Silicon-containing soybeanoil-based copolymers: synthesis and properties, Biomacromolecules 10 (9) (2009) 2678–2685.
- [54] R. Gu, S. Konar, M. Sain, Preparation and characterization of sustainable polyurethane foams from soybean oils, J. Am. Oil Chem. Soc. 89 (11) (2012) 2103–2111.
- [55] R.J. Zdrahala, I.J. Zdrahala, Biomedical applications of polyurethanes: a review of past promises, present realities, and a vibrant future, J. Biomater. Appl. 14 (1) (1999) 67–90.
- [56] G. Lligadas, J.C. Ronda, M. Galià, V. Cádiz, Poly (ether urethane) networks from renewable resources as candidate biomaterials: synthesis and characterization, Biomacromolecules 8 (2) (2007) 686–692.
- [57] P.L. Nayak, Natural oil-based polymers: opportunities and challenges, J. Macromol. Sci. Part C: Polym. Rev. 40 (1) (2000) 1–21.
- [58] P.L. Nayak, S. Lenka, S.K. Panda, T. Pattnaik, Polymers from renewable resources. i. Castor oil-based interpenetrating polymer networks: thermal and mechanical properties, J. Appl. Polym. Sci. 47 (6) (1993) 1089–1096.
- [59] E. Mistri, S. Routh, D. Ray, S. Sahoo, M. Misra, Green composites from maleated castor oil and jute fibres, Ind. Crops Prod. 34 (1) (2011) 900–906.
- [60] Y. Lu, R.C. Larock, New hybrid latexes from a soybean oil-based waterborne polyurethane and acrylics via emulsion polymerization, Biomacromolecules 8 (10) (2007) 3108–3114.
- [61] X. Wu, X. Zhang, S. Yang, H. Chen, D. Wang, The study of epoxidized

rapeseed oil used as a potential biodegradable lubricant, J. Am. Oil Chem. Soc. 77 (5) (2000) 561–563.

- [62] S. Gupta, R. Tyagi, V.S. Parmar, S.K. Sharma, R. Haag, Polyether based amphiphiles for delivery of active components, Polymer 53 (15) (2012) 3053– 3078.
- [63] C. Zhang, J. Liu, C. Zhang, S. Liu, Friction reducing and anti-wear property of metallic friction pairs under lubrication of aqueous solutions with polyether added, Wear 292 (2012) 11–16.
- [64] J.S. Louie, I. Pinnau, I. Ciobanu, K.P. Ishida, A. Ng, M. Reinhard, Effects of polyether-polyamide block copolymer coating on performance and fouling of reverse osmosis membranes, J. Membr. Sci. 280 (1) (2006) 762–770.
- [65] X.Z.K. Loy, S.K. Sinha, Lubrication of polyether ether ketone (peek) surface by liquid ultrathin films for high wear durability, Wear 296 (1) (2012) 681– 692.
- [66] M. Alam, S. Ashraf, A.R. Ray, S. Ahmad, Development of anticorrosive poly (ether-urethane) amide coatings from linseed oil: a sustainable resource, J. Polym. Environ. 18 (3) (2010) 208– 215.
- [67] E.D. Rio, G. Lligadas, J. Ronda, M. Galia, M. Meier, V. Cádiz, Polyurethanes from polyols obtained by admet polymerization of a castor oil-based diene: characterization and shape memory properties, J. Polym. Sci. Part A: Polym. Chem. 49 (2) (2011) 518–525.
- [68] G. Lligadas, J.C. Ronda, M. Galià, V. Cádiz, Monomers and polymers from plant oils via click chemistry reactions, J. Polym. Sci. Part A: Polym. Chem. 51 (10) (2013) 2111–2124.

7 Plant Oil-Based Epoxy Intermediates for Polymers

Tolibjon S. Omonov and Jonathan M. Curtis

Lipid Chemistry Group (LCG), Department of Agricultural, Food, and Nutritional Science (AFNS), University of Alberta, Edmonton, AB, Canada

		Ο U T L I N E			
7.1	Introduction				
7.2	Epoxidation of Plant Oils				
7.3	Derivatives of Epoxidized Plant Oils				
	7.3.1	Epoxy Ring Opening by Alcohols	104		
	7.3.2	Epoxy Ring Opening by Water	106		
	7.3.3	Amine-to-Epoxide Reactions	106		
	7.3.4	Epoxy Ring Opening with Halogen Reagents	108		
	7.3.5	Epoxy Ring Opening by Lactic Acid	108		
	7.3.6	Hydrogenation of Epoxidized Oils	108		
	7.3.7	Acrylated Epoxidized Plant Oils	109		
	7.3.8	Carbonated Plant Oil and its Derivatives	109		
	7.3.9	Epoxy Ring-Opening Reactions by Anhydrides	110		
	7.3.10	Epoxy Ring-Opening Polymerization	111		
	7.3.11	Estolides of Epoxides	112		
7.4	.4 Biobased Epoxy Resins		113		
	7.4.1	Anhydride-Cured Epoxidized Oils	114		
	7.4.2	Resins of Acrylated-Epoxidized Plant Oils	114		
	7.4.3	Direct Polymerization of Epoxidized Plant Oils	116		
7.5	7.5 Conclusions				
Ref	References				
Fur	Further Reading				

7.1 Introduction

Epoxy resins are an important class of materials, which are used widely in electrical, automobile, and other applications because of their excellent chemical resistance, outstanding adhesion, high tensile and compressive strengths, electrical insulation properties, and superior fatigue strength [1–4]. Currently, the major components used in most epoxy-based thermoset polymer formulations originate from petroleum-based derivatives. However, resins and composites incorporating renewable resources are increasingly being used in diverse areas of manufacturing because of the heightened awareness of environmental issues. To date, these biobased resins have only represented a very small percentage of the entire plastics market, and the widespread application of biobased materials is still limited by the available materials plus their cost of production. Hence, the development of cost-effective biobased materials from abundantly available renewable sources is of great interest to prevent reliance on finite petroleum resources.

Unsaturated plant oils, such as palm oil, soybean, canola (rapeseed), camelina, flax, hemp, sunflower, safflower, cottonseed, and other oils, can be chemically modified to produce epoxidized plant oils. These epoxidized oils can, in turn, be further modified or cured into thermoset epoxy resins, in similar ways as used for their petroleum-based counterparts.

This chapter describes the materials and methods for making plant oil-based epoxides, and the use of these epoxides in making variety of the intermediate monomers, oligomers, and polymers.

7.2 Epoxidation of Plant Oils

Functionalization of the carbon-carbon double bonds (-C = C) of unsaturated plant oils and oil derivatives is a commonly used process making use of a range of reactions. Some important examples of the modification of double bonds from unsaturated oils and fats include the use of environmentally friendly processes such as acid catalyzed chemical epoxidation [5–9] by hydrogen peroxide (H₂O₂), chemo-enzymatic epoxidation [10-20], or ozonolysis reactions [21-24]. These methods can produce a wide range of oil derivatives and products. Ozonolysis of the C = C of plant oils leads to the formation of ozonides which can subsequently cleave to yield aldehydes, acids, or ketones, while the chemical and chemoenzymatic oxidation of the double bonds lead to the formation of epoxides. Despite the use of variety of catalysts for the epoxidation of unsaturated oils and oil derivatives, the acid-catalyzed epoxidation methods of unsaturates have received much attention as the most cost-effective route. The most commonly used acids in the epoxidation processes of plant oils include aliphatic (such as formic, acetic, lauric, trifluoroacetic) and aromatic (such as benzoic, *m*-chlorobenzoic) species [25]. The use of microwaves as an accelerated method for making epoxidized fatty acid esters has been reported to produce a > 90% yield of epoxides within 5 min [6].

Although there are several approaches available for the modification of double bonds of unsaturated oils and oil derivatives to make biobased epoxides, the most commercially viable method of epoxidation is the classic epoxidation reaction proposed by Prilezhaev (Prileschajew) [26]. Thus, our main focus will be put toward discussion of



Scheme 7.1 Formation of performic acid (a) and epoxidation of double bonds with this performic acid (b). Note that to reduce analytical complexities, epoxidized plant oil structure provided in Scheme 1 (b) is substituted with the monounsaturated single-chain platform such as methyl oleate. R_1 and R_2 are the mixture of hydrocarbon and ester parts of FAME; *m* and *n* are the numbers repeating hydrocarbon units in different parts of the chain; k_1 and k_2 are the performic acid and epoxide formation rates, respectively.

the aspects of the *in-situ* acid catalyzed epoxidation of double bonds of plant oils and oil derivatives using the H_2O_2 .

Epoxidation of the plant oils and oil derivatives in an industrial scale are typically carried out using peroxy (performic or peracetic) acid generated *in-situ* from carboxylic acid and H_2O_2 [27]. The epoxidation reaction takes place in two stages: first the formation of a peroxy acid and subsequently the reaction of the peroxy acid with the unsaturates [26]. As an example, Scheme 7.1 demonstrates these two stages of epoxidation reaction of unsaturated fatty acid using formic acid as a catalyst (oxygen carrier).

Reaction of the formic acid with hydrogen peroxide leads to the formation of performic acid, which is the main oxidizing product of the double bonds of the oil (Scheme 7.1a). This reaction eventually consumes a stoichiometric amount of hydrogen peroxide. The performic acid so formed then reacts with double bonds of the oil by donating an oxygen atom to form the epoxide group. At the same time, formic acid is liberated to the bulk reaction mixture (Scheme 7.1b) [28]. Due to the regeneration of the formic acid, it is possible to use low molar amount of the formic acid in the epoxidation processes. Since oil epoxides are fairly reactive they may readily undergo ring-opening reactions or participate in oligomerization processes. It was reported [29] that a high molar ratio of acid to hydrogen peroxide is deleterious to the survival of an epoxy group. The attack of epoxy groups by formic acid may lead to the formation of hydroxylated-formiated products [30,31] due to the acid catalyzed epoxy ring-opening reactions (protonation of epoxy oxygen), as shown in Scheme 7.2a.

It has been reported [31] that protons at the surface of catalysts can cause epoxy ring opening by reaction with a number of chemicals present (or formed) in the overall mixture. The epoxy ring opening can also be caused by water in the presence of a strong acid catalyst [32] to form di-hydroxy groups per epoxy site, as demonstrated in Scheme 7.2b. Note that the premature epoxy ring opening is considered at this stage to be an undesirable side reaction. In contrast, the controlled occurrence of this oxirane cleavage by nucleophiles is an important reaction route that is used in making biobased intermediates or polymers from epoxidized plant oils. Thus, it is critical to control the conditions used in manufacturing the epoxidized plant oils in order to minimize such side reactions. This is summarized in the following:



Scheme 7.2 Examples of possible epoxy ring opening: with formic acid (a) and water (b). k_3 and k_4 are the ring-opening reaction rates by carboxylic acid and water, respectively.

- 1. *In-situ* epoxidation of the plant oils needs to be carried out at a moderate temperature, usually below 70°C:
 - a. high temperature epoxidation processes lead to the excessive epoxy ring-opening reactions;
 - b. in addition, it was reported that the decomposition rate of H_2O_2 increases approximately 2.2 times for each 10°C rise in temperature in the range from 20°C to 100°C [33];
 - c. peroxy acids are readily decomposed upon heating, and can explode upon rapid heating to 80–85°C.
- 2. Use of an excess of hydrogen peroxide is necessary to ensure complete epoxidation and to compensate for the loss/ decomposition of hydrogen peroxide. Typically, 20–50% excess of H_2O_2 from the stoichiometric ratio to double bonds is sufficient.
- 3. The concentration of the carboxylic acid (homogeneous) catalyst should be significantly lower than its stoichiometric ratio to double bonds in order to avoid or minimize undesirable epoxy ring-opening reactions from occurring during epoxidation. The optimal concentration is about 0.2–0.3 moles of carboxylic acid per mole of double bonds of the unsaturated oil.

Using these methods, epoxidized plant oils can be manufactured with a range of oxirane contents, depending on the fatty acid profiles of the input oils. Usually, the conversion of double bounds of the unsaturated oils into epoxides is quantitative; however, it should be noted that small amount of epoxy ring opening may still take place.

Figure 7.1 shows the epoxidation kinetics for canola oil fatty acid methyl esters (FAME) at an epoxidation temperature of 60°C. As can be seen from Figure 7.1 (Omonov et al., unpublished work, 2015), the amount of unsaturated FAME decreases exponentially over the epoxidation time (diamond symbols), reaching a plateau region after 24 h. This indicates that complete epoxidation of FAME has been achieved, and no further change is expected as all unsaturated FAME have been converted into epoxidized derivatives. Meanwhile, the oxirane oxygen content (OOC) of the system increases in a similar way, i.e., exponentially, over epoxidation time (open-circle symbols) to approach maximum values. (Note that the term OOC is a characteristic of epoxides that describes the weight percentage of oxirane oxygen in 100 g of epoxidized product). After 24 h of epoxidation the OOC values start to decrease as a result of epoxy ring-opening reactions described in Scheme 7.2. This reaction in its turn will generate the secondary mono- or



Figure 7.1 Epoxidation kinetics of canola oil FAMEs at 60°C using formic acid and hydrogen peroxide. The ratio of H_2O_2 and formic acid is 1.5 and 0.25 moles per double bond of the canola oil, respectively. Agitation speed is ~300 RPM.

di-hydroxyl groups on the fatty acid chain. The *in-situ* formation and increase of the hydroxyl groups during the epoxidation process can be monitored by measurement of the hydroxyl value of the partially epoxidized oil, as shown in Figure 7.1 (triangle symbols). It is clear from the figure that minimizing the epoxidation time will minimize the formation of secondary hydroxy-byproducts.

7.3 Derivatives of Epoxidized Plant Oils

Plant oils mainly contain triglycerides (TAG), which are the esters of glycerol and three fatty acids. The fatty acids of the majority of plant oil TAGs consist of unsaturated $C_{16}-C_{22}$ aliphatic chains bearing double bonds (especially unsaturated C_{18} – oleic, linoleic, or linolenic) and saturated $C_{16}-C_{22}$ fatty acids (mainly stearic or palmitic). The fatty acid profiles of the major oils are presented elsewhere in the literature [34–36]. Scheme 7.3 illustrates the structure of a hypothetical TAG containing unsaturated fatty acids.



Scheme 7.3 Structure of the TAG, containing the oleic, linoleic, and linolenic fatty acids.

Plant oils with a high unsaturated fatty acid content, such as soybean, canola (rapeseed), camelina, flax, hemp, sunflower, safflower oils, have become attractive feedstocks for chemical modifications since numerous reactions, including epoxidation, could be performed at the site of their internal alkenemoieties. Epoxidation of the unsaturated TAG presented in Scheme 7.3, according to the reaction path in the Scheme 7.1b, leads to the formation of epoxidized TAG, as shown in Scheme 7.4.



Scheme 7.4 Structure of the epoxidized TAG, containing the oleic, linoleic, and linolenic fatty acids.



Scheme 7.5 Structure of the vernonia oil, containing three vernolic acids. *Adapted from* [39]

It should be noted that some naturally occurring epoxidized fatty acids have been found in certain plant oils [37]. Of these, vernonia oil extracted from the seeds of the tropical African plant *Vernonia galamensis* appears to have the best potential for commercial application. This is due to its high concentration of epoxidized fatty acid (vernolic acid, 75–80%), relatively low viscosity for its high epoxide equivalent weight (about 400 g/Eq) [37,38]. The conceptual structure of a vernonia oil TAG molecule containing three vernolic acids is given in Scheme 7.5.

Usually, the oxirane groups of epoxidized plant oils are reactive and can easily be modified using a variety of procedures to tailor-make specific monomers that possess desirable functionalities for use in making polymers. Such reactions have included hydrolysis, acrylation, ethoxylation, propoxylation, hydroxylation, hydrogenation, carbonation, oligomerization, and many others resulting in a wide range of monomer functions. Note that in the following, only epoxides containing a single fatty acyl chain are used for illustration in order to simplify the discussion. However, the same general principles also apply to TAG epoxides.

7.3.1 Epoxy Ring Opening by Alcohols

One of the most commonly used methods of modification of the epoxidized plant oils is the epoxy ring-opening reactions with alcohols. Hydroxylation (alcoholysis) of epoxidized plant oils and oil derivatives using mono-, di-, or poly-alcohols in the presence of a variety of acid catalysts is used to make biobased polyols. The concept of epoxy ring-opening reactions using monofunctional alcohols is given in Scheme 7.6. The product of epoxy ringopening reactions of epoxidized plant oils via alcoholysis is lipid derivatives containing secondary hydroxyl groups along with an alkoxy dangling chain.

A great variety of homogenous and heterogeneous catalysts have been used to catalyze epoxy ring-opening reaction, including sulfuric acid [7-9,39,40], *p*-toluenesulfonic acid [41], fluoroboric acid, activated clay [42], strongly acidic ion exchange resins [43], fluoroantimonic acid (HSbF₆) [44], *m*-chloronbenzoic [5,6], and other acids.

Similarly, a diversity of monofunctional alcohols [39,40,45–55] have been used in epoxy ring-opening reactions of the biobased epoxides. Although some authors [7] have described the possible use of C_1-C_{12} linear chain monofunctional alcohols, secondary alcohols (such as isopropanol), or cyclic monofunctional alcohols (such as cyclohexanol or benzyl alcohol) in epoxy



Scheme 7.6 The concept of epoxy ring-opening reaction by monofunctional alcohols as nucleophiles in the presence of catalyst.



Scheme 7.7 The concept of epoxy ring-opening reaction by poly-functional alcohols as nucleophiles in the presence of catalyst.

ring-opening reactions, it is preferred to use short-chain monofunctional alcohols such as methanol, ethanol, and propanol.

The process of making commercially available BiOH[®] series (Cargill, Inc.) polyols from different epoxidized plant oils with the hydroxyl number ranging between 56 and 225 mg KOH/g polyol is described in [56]. This polyol is prepared via epoxy ringopening reaction (Scheme 7.6) by methanol in the presence of fluoroboric acid as a catalyst.

Epoxy ring-opening reactions of epoxidized plant oils and oil derivatives using dialcohols have received considerable attention [7,57–61]. The use of 1,3-propanediol [60], 1,2-propanediol and ethylene glycol [60,61], or esters of ethylene glycol [7] is reported in nucleophilic ring-opening reactions in biobased epoxides in the presence of strong acids to make polyols. The concept of epoxy ring opening with poly-functional alcohols is similar to Scheme 7.2b. The resulting polyols have high hydroxyl numbers comprising a mixture of primary and secondary hydroxyl groups (Scheme 7.7).

The use of sulfuric acid [7,58,60], tetrafluoroboric acid [57,59,61], macroreticular ion exchange resin, and diethylether-trifuoride borate complexes [62] as catalysts of the alcoholysis reactions is reported at different reaction temperatures.

Alcohols including propanediol, butanediol, pentanediol, hexanediol, and polyoxyethylene have been used to make poly-hydroxylated fatty acid esters. These poly-hydroxylated fatty acid esters have further been used to make polyols from epoxidized plant oils with primary hydroxyl groups [63]. The authors [64] also used several di- and poly-alcohols in epoxy ring-opening reactions of epoxidized soybean, sunflower, and linseed oils to produce oleochemical polyols. The produced polyols had high primary and secondary hydroxyl functionality (between 370 mg KOH/g and 418 mg KOH/g) and controllable reactivity. These polyols were manufactured utilizing the hydroxylation concept described in Scheme 7.7.

Some commercial examples of this process include BASF's SOVERMOL® series of biobased polyols made from soybean oil for various applications such as solventfree UV-stable coating, flexible coating and casting, and polyurethane production. The chemical pathway of some of these biobased polyols is based on the epoxidation of vegetable oils followed by ring opening of epoxides by nucleophilic attack of alcohols, then transesterification of poly-hydroxylated oils with the same alcohol [65]. Another example is from Hobum Oleochemicals who are producing their MERGINOL® series of biobased polyester and polyether-polyester polyols for low density polyurethane applications and as raw materials for polyesters, surfactants, and lubricants. These polyols are produced via epoxidation of plant oils followed by nucleophilic ring opening by water, mono-, or di-alcohols and alcoholamines.



Scheme 7.8 The epoxy ring-opening reaction by water in the presence of catalyst.

7.3.2 Epoxy Ring Opening by Water

One of the simplest methods of epoxy ring opening is the use of the water as a nucleophile in the presence of a strong acid [32,66– 68]. This reaction yields a di-hydroxylated oil or oil derivative, containing secondary hydroxyl groups (Scheme 7.8). The use of sulphuric acid, formic acid, acetic acid, or various organometallic catalysts for this reaction has been reported. [69–74].

As an example, the Sofiproteol Group is manufacturing biobased polyols under tradename of RADIA[®]. Their polyester and polyether polyols are intended to be used for flexible and rigid PU applications respectively, while RADIA amorphous polyester polyols were developed for CASE (coatings, adhesives, sealants, and elastomers) and thermoplastic PU applications. The chemical routes for manufacturing these polyols are based on the epoxidation of oils and its derivatives by H₂O₂ with metal catalysts followed by nucleophilic ring opening of epoxides by water [75], as shown in Scheme 7.8. However, it should be noted that whilst the production of polyols with the secondary hydroxyl groups is simple and inexpensive, the demand for the use of these products is less due to the relatively low reactivity and high viscosity, compared to the other biobased polyols.

7.3.3 Amine-to-Epoxide Reactions

Monofunctional amines or alcoholamines could also directly be used as nucleophile to open epoxy ring of epoxidized plant oils. The use of monofunctional amines, such as diethylamine, *n*-propylamine, *tert*-butylamine, and isobutylamine, for the epoxide ring opening of epoxidized plant oil in the presence of zinc chloride (ZnCl₂) catalyst was reported [76]. The reaction conditions were optimized using a variety of parameters to favor epoxy ring-opening reaction for maximum aminolysis of epoxy group, without cross-linking and with little disruption of the ester linkage. The best results were achieved with diethylamine (Scheme 7.9).

These authors [76] describe that the use of epoxy ring opening by an amine is an effective way of introducing branching on the fatty acid chain of plant oils. The resulting products (defined as "aminol") possess significantly improved thermal, oxidative, and low-temperature stability compared to the equivalent unmodified oil, and can be used for antioxidant and antiwear specialty chemicals in automotive and industrial applications.

Other authors [64,77] described the formation of the mixture of biobased polyester polyols (from the amidification reaction) and



Scheme 7.9 The concept of epoxy ring-opening reaction with diethylamine. Adapted from [77].



Scheme 7.10 The concept of epoxy ring-opening reaction and amidification reaction with diethanolamine.

polyether polyols (from the oxirane ringopening reaction) from the reaction of epoxidized triacylglycerol and alcoholamines. The concept of the epoxy ring opening with diethanolamine to produce fatty amide (a) and amine (b) polyols is presented in Scheme 7.10. It was demonstrated that high temperature reactions (130–180°C) and extended reaction times (from 2 up to 7 h) favored the epoxide ring opening by diethanolamine [77].

This knowledge of making polyols from epoxidized soybean oil and alkanolamines has been realized in commercial scale by Huntsman Petrochemical LLC [78] in manufacturing JEFFADDTM B650 biobased polyols. This autocatalytic polyol is described to comprise primary and secondary hydroxyl groups (400–600 meq KOH/g), tertiary amine groups (1.5–3.5 meq/g), free from odor and acidity, water soluble and intended to use as a replacement for all, or a portion of, petroleum-derived polyols generally used in the production of polyurethane materials.

The double hydroxylation procedure is used to make biobased polyols with increased

hydroxyl numbers and moderate viscosities. First, the epoxide ring of an epoxidized plant oil is opened via the reaction shown in Scheme 7.2a to make polyols [79]. Then, the aminolysis process by hydroxyl alkylamines of the ester bond in polyols is carried out to increase the hydroxyl number and lower the viscosity [80]. This polyol is currently marketed by BioBased Technologies LLC as AGROL[®] Diamond polyol, with typical hydroxyl value of 340–370 mg KOH/g, viscosity 3300 cP, and a molecular weight of 510–560 g/mol.

The use of novel polyol synthetic routes starting from epoxidized plant oils and primary amines has been reported for coatings, adhesives, and foam applications [81]. Materials that contain mixed amino and/or hydroxyl poly functionality ("polyolamines") have been prepared by epoxy ring opening and amide formation reactions (Scheme 7.11).

The final poly-functionalized intermediate products were cross-linked with a variety of curing agents to form urethane-like polymer-ic materials [77,81,82].



Scheme 7.11 The concept of direct epoxy ring opening and amidification reactions with polyamines. Note that R is H or fatty acid adduct.

7.3.4 Epoxy Ring Opening with Halogen Reagents

Hydrohalogenation methods have been used to analyze epoxy groups via halohydrin formation reactions [83] using HCl or HBr as the halide reagent. Halogenation via ringopening reactions has been reported, using both hydrochloric (HCl) and hydrobromic (HBr) acids to prepare halogenated polyols from epoxidized plant oils (Scheme 7.12) [52,84,85]. It was anticipated that the polyurethanes made from these halogenated polyols would have enhanced fire resistant properties due to the incorporation of halogen atoms in the fatty acid chain along with the hydroxyl group [85].

The thermal stability and physical properties of these halogenated polyols were evaluated and compared with nonhalogenated polyols. Halogenated polyols exhibit higher densities and viscosities compared to their nonhalogenated counterparts. Polyurethanes incorporating halogenated polyols demonstrated higher glass transition temperatures and mechanical performance [52,84], as well as superior resistance to flammability [85].



Scheme 7.12 The concept of epoxy ring opening with hydrochloric and hydrobromic acids. Note X represents Cl– or Br–.

7.3.5 Epoxy Ring Opening by Lactic Acid

An attempt has been made to introduce two hydroxyl groups per epoxy site of the epoxidized plant oil via an epoxy ring-opening reaction with lactic acid (Scheme 7.13) [86].

A lactic acid-epoxidized soybean oil (LA-ESO) hybrid monomer has been synthesized with the 70% conversion of epoxide into dihydroxy groups. The product was claimed to be an intermediate for the production of partially biorenewable polyurethanes with a unique crosslinking structure. Polyurethane materials with the controllable, broad range of glass transition temperature (up to 96°C) were prepared from LA-ESO and MDI using different ratios of isocyanate to hydroxyl groups [86].

7.3.6 Hydrogenation of Epoxidized Oils

A modification of epoxide groups of biobased epoxides using catalytic hydrogenation in the presence of a Raney nickel catalyst has been reported [52,84,87]. Epoxy ring opening by hydrogen addition leads to the formation of hydroxylated fatty acids with only secondary hydroxyl groups (Scheme 7.14).

It was reported that the hydrogenated soy polyols $(Soy-H_2)$ demonstrated higher hydroxyl numbers (212 mg KOH/g) compared to the chlorinated (Soy-HCl), brominated (Soy-HBr), and methoxylated (Soy-Met)



Scheme 7.13 The concept of epoxy ring opening with lactic acid.



Scheme 7.14 The concept of catalytic epoxy ring-opening reaction. *Adapted from* [66,85].

polyols (below 200 mg KOH/g) [52,84]. However, the glass transition temperatures of polyurethanes from hydrogenated polyols were significantly lower (<40°C) compared to the polyurethanes from the other Soy-HCl, Soy-HBr, Soy-Met polyols (between 66°C and 88°C). These differing values of T_{a} were ascribed to differences in the crosslinking densities and in the free volumes of these substituent groups within the polyurethanes. In addition, the use of a variety of palladiumbased catalysts [88] has been reported for the catalytic hydrogenation reactions that can selectively reduce aromatic epoxides to the corresponding alcohols in high yield and under mild conditions.

7.3.7 Acrylated Epoxidized Plant Oils

Epoxidized plant oils and their derivatives can be chemically modified through simple reactions to introduce readily polymerizable groups [89,90]. One such approach is to make a reactive acrylic polymer containing both hydroxyl and acidic functional groups (Scheme 7.15).

Acrylated epoxidized plant oils were prepared using epoxidized plant oils along with acrylic acid, an amine-type catalyst and hydroquinone as a gelling inhibitor [90,91]. This primary objective of these reports was to make UV radiation curable polymer precursors. Similar strategies have been applied to the synthesis of acrylated or methacrylated plant oils using epoxidized plant oils including soybean [36], canola oil [92], linseed [93,94] castor [95], palm [96], and vernonia [97].

7.3.8 Carbonated Plant Oil and its Derivatives

Novel methods of producing carbonated plant oils have been described in the literature by several authors [98–102]. They have outlined the synthetic pathways for the modification of epoxidized oil which involves the reaction of oxirane of an epoxidized plant oil with the carbon dioxide (CO₂) to form a carbonated oil. It was claimed that they could successfully incorporate CO₂ into the fatty acids of epoxidized soybean oil in order to produce carbonated soybean oil with high yields (94% [102]), under moderate flow of CO₂ in the presence of tetrabutylammonium bromide (Scheme 7.16).

The carbonated oil produced as shown in Scheme 7.16 has been used in producing a variety of oil derivatives via cyclic carbonate ring-opening reactions with polyamines having two or more amine functionalities, under mild conditions [101– 105]. A variety of polyamines, including



Scheme 7.15 The concept of epoxy ring-opening reaction by acrylates. Adapted from [90,91].



Scheme 7.16 The concept of epoxy ring opening reaction by CO₂ to form carbonated oil.



Scheme 7.17 The concept of cyclic carbonates ring opening by polyamines to form urethane (a), amide (b) or urethane–amide compounds (c). *Adapted from [103,106]*. Note that R is hydrogen or fatty acid adduct.

1,2-ethane diamine [102,103,105], isophorone diamine [101,103], 1,6-hexanediamine [101,102,105], triethylenetetraamine and diethylenetriamine [101], tris(2-aminoethyl) amine [102], have been used in carbonate ring-opening reactions at different carbonate-to-polyamine ratios. Depending on carbonate-to-polyamine ratio the final product may contain urethane, amide bonds or the simultaneous formation of both links can be favored (Scheme 7.17) [105].

Note that these carbonate ring-opening reactions with polyamines have mainly been used to produce nonisocyanate polyurethane (NIPU) networks [102,104,105]. The effects of the amine structure and carbonate to amine ratio on biobased NIPUs structure and mechanical, physical, and swelling properties were also studied. It was reported that the stoichiometric carbonate-toamine ratio produced the polymer networks with the highest density and the highest $T_{\rm g}$ (between 35.3 °C and 40.1 °C from DMA) and hardness (between 47 and 83 shore A), and tensile strengths (between 3.80 MPa and 5.77 MPa).

7.3.9 Epoxy Ring-Opening Reactions by Anhydrides

Epoxy ring-opening reactions by anhydrides and acid anhydrides have been reported to make biobased epoxy resins using different epoxidized oils, such as epoxidized canola (ECO) [106], [107], soybean (ESO), camelina (EcamO), linseed (ELO) [106,108,109] and hemp (EHO) oils [106]. Usually, the reaction of epoxides with anhydrides involves multistep reactions, and some important simple examples are given in Scheme 7.18.

In the presence of a nucleophile, the epoxide groups can produce hydroxylated fatty



Scheme 7.18 The concept of epoxy and phthalic anhydride ring-opening reactions in the presence of nucleophilic alcohol, and possible oligomerization of components. *Adapted from* [108].

acid ethers (reaction A), while an anhydride can ring can be opened to form both monoester and carboxylic acid groups (reaction B). In these reactions, the nucleophiles or nucleophilic components could be those already existing or added or those formed via reaction (A) or (B). Hence, the hydroxyl and/ or acid groups of the components formed in (A) and (B) give rise to oligomeric structures, an example of which is indicated in reaction (C). The use of these oligomeric prepolymers as aldehyde-free internal bonding adhesives in manufacturing lignocellulosic fiberboards and particleboards has been reported [106].

7.3.10 Epoxy Ring-Opening Polymerization

A method for producing oligomeric polyether polyols through acid-catalyzed ring-opening polymerization of epoxidized methyl oleate (EMO) in the presence of $HSbF_6$ has been described [44]. A subsequent partial reduction of the ester groups resulted in the formation of primary alcohols (Scheme 7.19).

This controlled reduction of the carboxylate groups to hydroxyl moieties was carried out in tetrahydrofuran using lithium aluminum hydride (LiAlH₄). Depending on the degree of reduction, the polyols with a broad range of hydroxyl functionalities (between 94 mg KOH/g and 298 mg KOH/g) were obtained, and these range from clear liquids to white, waxy solids at room temperature. These polyols were reacted with 4,4'-methylenebis(phenyl isocyanate) to make corresponding polyurethane networks. The T_g of the polyurethanes produced varied from -15° C to 80°C (determined by DSC) depending on hydroxyl functionality of the polyols.

A similar concept involving the epoxy ring-opening oligomerization of epoxidized



Scheme 7.19 The concept of cationic epoxy ring-opening oligomerization reaction and controlled reduction of carboxylates into hydroxyl groups. *Adapted from* [45].



Scheme 7.20 The concept of epoxy ring opening by phosphoric acid and the mechanism of ether (a) and phosphate ester (b) cross-link formation. *Adapted from* [111].

plant oils has been suggested [56,110]. In this case, phosphoric acid (H₃PO₄) was used to make novel biobased pressure-sensitive adhesives (PSA) from epoxidized soybean oil (ESO). In this work, two key functions of phosphoric acid were reported: (i) acting as a general acid catalyst activating the epoxide toward nucleophilic attack by the diol (Scheme 7.20), thus generating an ether cross-linkage (reaction a); (ii) as a coreacting component in establishing phosphate ester linkages (reaction b). These reaction pathways have led to the formation of epoxidized and dihydroxylated soybean oils. Copolymerization of epoxidized and hydroxylated soybean oil in the presence of H_3PO_4 resulted in a product with the potential as a PSA. The biobased PSA showed similar peel adhesion strength properties as selected commercial PSAs.

7.3.11 Estolides of Epoxides

The estolides are a developing class of materials which can be synthesized by the condensation of a fatty acid with a second fatty acid or fatty acid alkyl ester. They have a great potential as base stocks for a wide range of lubricant applications due to their excellent cold temperature properties coupled with their good oxidative stability [111]. A variety of routes for estolide formation from fatty acids have been discussed extensively in the literature [111– 113], including the epoxy ring-opening esterification reactions. The epoxy ring opening of epoxidized plant oil derivatives (FFA, FAMEs, etc.) catalyzed in situ by the FFAs lead to the subsequent condensation of fatty acid to form the hydroxyl estolide (Scheme 7.21).



Scheme 7.21 The concept of estolide formation from epoxide and fatty acid. Adapted from [112].

A range of short-chain fatty acids, including propanoic, 2-propanoic, octanoic, and 2-ethylhexanoic acids, have been used to form hydroxylated di-esters (estolides) from propyl and octyl epoxystearate esters [114]. The best cold temperature properties (pour point of -39°C) and oxidative stability was noticed for the propyl ester of a hydroxy, 2-ethylhexyl estolide. It has been reported that epoxidation of oleic acid is used to produce dihydroxystearic acid estolides via intermolecular esterification between the hydroxyl and carboxyl groups [115]. In that work, it was reported that the synthesis of oligomers with average estolide number (the average number of fatty acid units added to a base fatty acid) of 5 was achieved within 8 h of reaction at high temperature (180°C). The use of these types of estolides in cosmetics and personal care products is described [116].

7.4 Biobased Epoxy Resins

Epoxy groups react readily with many functional groups, including amines, anhydrides, phenols, mercaptans, imidazoles, isocyanates, acids, and others, as discussed above [2,3,117,118]. The mechanisms for the polymerization of epoxy monomers to form crosslinked polymers via step polymerization, chain homo- or copolymerization, or a combination of these mechanisms were reviewed earlier [2,119]. Linear or crosslinked epoxy polymers are obtained by the reaction of epoxy monomers with curing agents (hardeners) and/or initiators. In order to obtain a linear polymer, the epoxy resin precursor and (co)curing agent should have two coreacting sites, while the formation of the branched or cross-linked polymers requires the reactive functionality of one of the (co)reacting

components to be more than two. Note that introducing monofunctional components is not desirable as their presence in the reaction medium will interrupt the growth of the polymer chain. The molar weight of linear polymers grows gradually with maximum of polydispersity of two, while the average molar mass of a crosslinked structure tends to be infinite at gelation. Thus, the amount and ratio of the reactive functional groups of (co)curing components are the key parameters to control the final polymer structure.

Although the formation of the epoxides from oil derivatives, and their curing with a variety of curing agents, have been reported in the literature, the use of epoxidized plant oils has received much attention due to the high functionally of the epoxidized triacylglycerols (TAG) compared to their epoxidized derivatives such as epoxidized fatty acid alkyl esters. For example, epoxidized canola oil has between two and seven epoxy groups per TAG molecule, while negligible amount of canola TAGs (~1 %) is expected to have zero to one oxirane group per molecule [107].

Epoxidized plant oils can be cured in similar ways as petroleum-based epoxy counterparts. Usually, the concentration of epoxy groups in the epoxy monomer used is approximately equal to the concentration of reactive groups in the hardener/initiators because, under these conditions, unwanted side reactions are minimized. A variety of curing agents or combinations of curing agents have been used to prepare biobased epoxy resins.

In Section 7.3, some possible epoxy ringopening reactions in epoxidized plant oils to make polymer precursors have been described. In contrast, the aim of this section is to give examples of the polymers manufactured directly from epoxidized plant oils. Some of the thermo-mechanical and physical properties of these epoxy resins are also addressed.

7.4.1 Anhydride-Cured Epoxidized Oils

Thermoset epoxy resins from epoxidized canola oil (ECO) were formulated with phthalic anhydride (PA) as the curing agent for different ratios of ECO to PA (1:1, 1:1.5, and 1:2 mol/mol) at curing temperatures (T_{cure}) of 155, 170, 185, and 200°C [106]. The gelation process and the viscoelastic properties of the system during curing were studied. It was found that at a fixed ECO/PA ratio the thermo-mechanical properties of the resins were not strongly dependent on the curing temperature of the resin, although elevated temperatures significantly accelerated the curing process. However, an increase in the curing agent (PA) amount significantly altered both the reaction rate and the thermomechanical properties of the final resin. Depending on the curing conditions, thermosets with a wide range of properties were produced. At low T_{cure} and low PA content, clear and transparent rubberlike flexible thermosets with the T_{a} in the range of -3.3 to -4.4 °C were produced. In contrast, at high T_{cure} and high PA contents, almost rigid, semiflexible plastics with T_{o} in the range of 37.8–40.4°C (at a) were formed. It was reported that the selective combination of the curing temperature and the molar ratios of the curing agent could be used to design the thermoset resins with selected properties. The authors of this work reported [106] the use of a variety of epoxidized plant oils, including epoxidized canola (ECO), camelina (ECamO), linseed (ELO), and soybean (ESO) in making biobased thermoset resins using anhydrides, acid anhydrides, biobased diacids, and food

acids. The influence of the types of cocuring agents and epoxy functionality of epoxides oils to the dynamic-mechanical, thermal properties of the thermoset resins have been evaluated. The applicability of these thermoset resin precursors as aldehyde-free alternatives for aldehyde-containing internal bonding adhesives (melamine, urea, or phenol formaldehyde) to manufacture lignocellulosic fiberboard or particleboard has been demonstrated.

A variety of flexible rubbery and rigid crosslinked polymers were prepared by curing ESO with anhydrides such as norbornene dicarboxylic acid anhydride, hexahydrophthalic anhydride, succinic anhydride, maleic anhydride, and phthalic anhydride in the presence of cure catalysts such as tertiary amines, imidazoles, or aluminum acetylacetonate [120]. Highly flexible and rubber-like crosslinked polymers with the T_{g} below room temperatures (between -5 and 10°C) were obtained with the use of norbornene dicarboxylic acid anhydride, hexahydrophthalic anhydride, and succinic anhydride. The use of maleic anhydride and phthalic anhydride as curing agents of ESO resulted in rigid and stiffer polymers with high flexural moduli in the range of 500–1000 MPa and T_{g} varying between 43°C and 75°C. The highest stiffness and T_{α} were observed for ESO cured with phthalic anhydride at stoichiometric ratios of epoxy and anhydride groups and 2 wt% content of aluminum acetylacetonate or benzyldimethylamine accelerators.

7.4.2 Resins of Acrylated-Epoxidized Plant Oils

The main aim from manufacturing the acrylated epoxidized plant oil is to make radiation-curable precursors of polymers.

The majority of the commercially available radiation-curable monomers and oligomers are derived from petrochemical-based materials [121] and used for applications such as in the coatings industry for the protection of wood, plastic, metal, and paper surfaces; in manufacturing printing plates, digital video disk printing, fast-drying printing ink; pressure-sensitive adhesive (PSA) manufacturing and electronic industries [122]. New environmentally friendly UV-curable acrylated-epoxidized polymer precursors from biobased sources are being actively developed and have been described [96,123,124]. Plant oil-based networks from the photooxidation, styrenization, thiol-ene coupling (TEC), acrylate coupling, and the epoxy ring-opening reactions are illustrated in the literature [125] along with the structures of some photocurable glycerol-based monomers.

Acrylated polyester prepolymers (PEPP) from RBD (refined, bleached, and distillated) and crude palm oils containing reactive diluents and photoinitiator were prepared for UV-curable wood coating applications [96]. The formulations of PEPP with reactive diluents of trimethylol propane triacrylate (TMPTA) and 1,6-hexanediol diacrylate (HDDA) were found to be most suitable for wood coating application with acceptable adhesion, scratch, solvent, and chemicalresistant properties.

Acrylated epoxidized soybean oils (AESO) have been used as a renewable reactive diluent (up to 40%) for partial replacement of commercial epoxy resins to decrease the overall cost of the thermally cured epoxy coatings [123,124]. The authors [124] have suggested that the substitution of a 10–40% of the aromatic epoxy by cycloaliphatic epoxy resin (particularly with AESO) could reduce the cost of coatings to a \$0.30–\$1.30 per pound without substantial loss in mechanical performance.

The free radical polymerization of the AESO was promoted with commercially available photoinitiators (Darocur and Irgacure series) in order to prepare photocrosslinked materials [122,126]. The kinetics of photopolymerization upon UV curing with up to 4% photoinitiators are studied in infrared spectroscopy. High gel content and conversion in the photopolymerized AESO (within 10 s of irradiation) is observed with 2% of photoinitiators [126]. The use of these UV-cured acrylated epoxidized systems in fast-drying protective coatings and the rapid manufacture of composite materials are suggested, although the mechanical performance of UV-cured materials was not addressed in these studies.

Acrylated epoxidized canola oil (AECO) and maleinized acrylated epoxidized canola oil (MAECO)-based resins were synthesized with various commercially available initiators such as Luperox LP and Luperox 26 (from Arkema), Trigonox C (from Akzo-Nobel) [127]. The effect of the type and amount of initiator, and the cure temperature on the cure and mechanical and rheological properties of resin were optimized. The T_{a} and the storage moduli of the resin cured with \tilde{b} Luperox LP were higher (70°C and 800 MPa for MAECO and 50°C and 400 MPa for AECO, respectively) compared to the resins cured with other initiators. A similar trend was observed in the mechanical performance of the biocomposites made with hemp and flax mats, while the resin impregnation rate was better for AECO-based resin.

Maleinized acrylated epoxidized soybean oils (MAESO) with varied numbers of maleic acid groups per TAG (from 1 mol to 2.5 mol) were prepared from AESO at different molar ratios of maleic anhydride [128]. The resulting monomers were then copolymerized with 33 wt% styrene to form rigid polymers. Dynamic mechanical analysis (DMA) showed the high moduli of these polymers at room temperature (1.9–2.2 GPa), which also had a T_g in the range of 100–115°C. These properties were increased by increasing the molar ratio of MA in AESO, and were found to be comparable to commercially available polyesters.

Several companies, such as Rust-Oleum (www.rustoleum.com, Varathane series clear coatings and Ultimate Poly series interior wood coatings), ECO safety products, LLC (www.ecosafetyproducts.com, Acri-Soy series of nontoxic penetrating clear concreate and wood sealer), Sherwin-Williams (www.sherwin-williams.com, ProMar and ProClassic series of low VOC soy/acrylic/ polyester paints), are deploying soybean oil acrylated epoxy products in their low VOC, environmentally friendly biobased product lines. Allnex (www.allnex.com) is proactively adding AESO to their product lines as a modifier to improve the flow and leveling of ultraviolet (UV) light or electron beam (EB) curable inks and coatings.

7.4.3 Direct Polymerization of Epoxidized Plant Oils

Epoxidized soybean oil (ESO) and epoxidized castor oil (ECasO) were synthesized and polymerized via thermally induced ringopening reaction initiated by a cationic latent thermal catalyst, *N*-benzylpyrazinium hexafluoroantimonate (BPH, 1%) [129]. The reaction is carried out at 1% of BPH load and at a temperatures of 110–150°C for 4 h. It was expected that the thermal decomposition of BPH at high temperature should yield the protic acid (HSbF₆) and catalyze the cationic polymerization of the epoxides, according to the Scheme 7.18. BPH showed an excellent thermally latent initiator role for epoxy curing that functions in the absence of coinitiators. Characterization of the thermal properties of the resulting polymers was carried out by differential scanning calorimetry (DSC), dynamic mechanical analysis (DMA), and thermal mechanical analysis (TMA). The cured ECasO/BPH resin showed a higher T_{α} (43°C), while a low crosslinking density $(0.57 \times 10^{-3} \text{ mol/cm}^3)$ and storage modulus (7.9 MPa, at 30°C) was noticed. This behavior was attributed to the greater number of intermolecular interactions in ECasO/BPH system, such as hydrogen bonding. The ESO/BPH resin had low T_g (25°C), but the crosslinking density (1.12 \times 10⁻³ mol/cm³) and storage modulus (14.7 MPa, at 30°C) were twice higher than those resins made from ECasO. This was related as a result of high epoxy numbers in ESO.

Polymerization of epoxidized soybean oil (ESO) has been carried out with maleinized soybean oil (MSO), and as well as in the presence of maleic anhydride grafted polypropylene (MA-g-PP) with 1 and 2.9% MA group (MW = 110,000 and 3900 g/ mol, respectively) [130]. In all experiments the epoxy:maleate groups ratio were kept at 1:1 to provide the best reaction conditions. The resulting polymer yield was about 94%. Thermogravimetric analysis of the resulting polymers showed that that the thermal properties of the polymers were not dependent on MA-g-PA content or molecular weight, while a dramatic increase in storage modulus of polymer with the increase of high MW MA-g-PA content was noticed. The authors of this study concluded that higher MW and therefore higher entanglement is more important in accomplishing the high mechanical properties than the crosslinking density,

which was high in polymers with low MW PP-g-MA. The use of MA-g-PP films copolymerized with ESO is described to enhance adhesion of low density polyethylene to polyesters and metal foils. These authors also reported the polymerization of ESO with maleinized polybutadiene having different maleate content [131] and with different pyridine derivatives such as 4-methylpyridine (4MP), 4-vinylpyridine (4VP), and poly(4vinylpyridine) (P4VP) [132]. The dynamic mechanical and thermal properties of the resulting polymers were evaluated, and a possible application for the new pyridine-based thermoset resin was described as its use as a gel molding resin due to its decent set of mechanical and thermal properties.

The in-situ reactive blending of ESO with phthalic anhydride in polypropylene melts (at 200°C) in the presence of aluminum acetylacetonate cure accelerator was performed using an anhydride/epoxy molar ratio of 0.75 to prevent sublimation of unreacted phthalic anhydride [120]. Interfacial adhesion of the ESO resin to the polypropylene was enhanced by adding PP-g-MA (with 5% MA content). Incorporation of 60 vol% of biobased polymers into polypropylene to afford flexible thermoplastic multiphase polyolefins has been reported [120]. A dramatic decrease of the Young modulus (from 1150 MPa to 150 MPa) and elongation at break (from 800% to 8%) were noticed in the resulting polymer blend, which could match the performance of flexible PVC-type materials.

Direct epoxy ring-opening polymerization of ESO was initiated with a different amount of boron trifluoride diethyl etherate (BF₃·OEt2) catalyst in methylene chloride at different temperatures (between 0°C and 50°C) to develop biodegradable polymers [133]. A thermally stable and white colored polymer was obtained in high yield (>99%). These biobased polymers of ESO were found to be highly crosslinked (gel fraction about 73–99%) with T_g ranging from –16 to –48°C. It was suggested that these products could be converted into hydrogels by saponification, which makes them suitable for use in personal care and health-care applications.

Various in-house made and commercially available epoxidized plant oils have been used in photoinitiated cationic polymerization to make thin films using diaryliodonium and triarylsulfonium salt photoinitiators bearing long-chain alkoxy substituents [134]. The photopolymerization rates were evaluated and found to be dependent epoxy content of the monomers. Characterization of the resulting polymers films by thermomechanical techniques revealed that they possess good adhesion and mechanical properties. Biodegradation studies (under 1 inch of soil for 6 months) of the films showed that polymer had eroded and lost their integrity due to bacterial and environmental attack. They explained these phenomena as a result of degradation of the glycerol triester units within the polymerized films.

7.5 Conclusions

Plant oils have been used as feedstocks for paints, soaps, cosmetics, lubricants, and coatings for a long time. We have seen throughout the discussion in this chapter that there are enormous possibilities for the use of epoxidized plant oils and oil derivatives in manufacturing intermediate chemicals, monomers, and polymers.

However, the direct polymerization of biobased epoxides has received less attention in making polymeric end-products due to their relatively poor mechanical performance compared to conventional petrochemical-based counterparts. This poor mechanical behavior of biobased epoxides has been associated with the low reactivity of epoxy groups together with a tendency for intramolecular bonding, leading to a low degree of cross-linking. However, it should be noted that the plant oil-based polymers are potentially biodegradable via hydrolytic cleavage of glycerol ester bonds present in the triglyceride oils [27].

Nevertheless, currently, plant oil-based epoxides are being used to replace a portion of petroleum based epoxides in manufacturing polymers and composites, thus preserving acceptable mechanical performance while increasing the renewable content of the final product. Although there are many problems that exist and manufacturing cost-effective drop-in replacements for petroleum-based epoxides remains challenging, progress is being made in this area and their further adoption is likely.

By designing and synthesizing new (co) curing agents in combination with the use of novel polymerization methods, it is likely that better as well as unique epoxidized plant oil-based polymers will be manufactured in the future to expand the application and use of this class of materials.

References

- C.D. Craver, C.E. Carraher, Applied Polymer Science: 21st Century, Elsevier Science Ltd, Oxford, 2000.
- [2] J.-P. Pascault, R.J.J. Williams, Epoxy Polymers: New Materials and Innovations, Wiley-VCH, Germany, 2010.
- [3] J.-P. Pascault, H. Sautereau, J. Verdu, R.J.J. Williams, Thermosetting Polymers, Marcel Dekker, New York, 2002.

- [4] E.M. Petrie, Epoxy Adhesive Formulations, McGraw-Hill, New York, 2006.
- [5] F. Fringuelli, R. Germani, F. Pizzo, G. Savelli, Epoxidation reaction with Mchloroperoxybenzoic acid in water, Tetrahedron Lett. 30 (1989) 1427–1428.
- [6] M.S.F.L.K. Jie, C. Yankit, The use of a microwave-oven in the chemical transformation of long-chain fatty-acid esters, Lipids 23 (1988) 367–369.
- [7] H. Kluth, A. Meffert, 1985, Polyurethane prepolymers based on oleochemical polyols. US 4,508,853 A.
- [8] H. Kluth, B. Gruber, A. Meffert, W. Huebner, 1988. Polyurethane prepolymers based on oleochemical polyols, their production and use, US 4,742,087 A.
- [9] A. Meffert, H. Kluth, 1989. Process for the preparation of modified triglycerides. US 4,886,893 A.
- [10] I. Hilker, D. Bothe, J. Pruss, H.J. Warnecke, Chemo-enzymatic epoxidation of unsaturated plant oils, Chem. Eng. Sci. 56 (2001) 427–432.
- [11] M.R.G. Klaas, S. Warwel, Chemoenzymatic epoxidation of unsaturated fatty acid esters and plant oils, J. Am. Oil Chem. Soc. 73 (1996) 1453–1457.
- [12] C. Orellana-Coca, S. Camocho, D. Adlercreutz, B. Mattiasson, R. Hatti-Kaul, Chemo-enzymatic epoxidation of linoleic acid: parameters influencing the reaction, Eur. J. Lipid Sci. Technol. 107 (2005) 864–870.
- [13] C. Orellana-Coca, U. Tornvall, D. Adlercreutz, B. Mattiasson, R. Hatti-Kaul, Chemo-enzymatic epoxidation of oleic acid and methyl oleate in solventfree medium, Biocatal. Biotransform. 23 (2005) 431–437.
- [14] G.J. Piazza, T.A. Foglia, A. Nunez, Preparation of fatty epoxy alcohols

using oat seed peroxygenase in nonaqueous media, J. Am. Oil Chem. Soc. 76 (1999) 551–555.

- [15] G.J. Piazza, T.A. Foglia, A. Nunez, Epoxidation of fatty acids with membranesupported peroxygenase, Biotechnol. Lett. 22 (2000) 217–221.
- [16] G.J. Piazza, T.A. Foglia, A. Nunez, Optimizing reaction parameters for the enzymatic synthesis of epoxidized oleic acid with oat seed peroxygenase, J. Am. Oil Chem. Soc. 78 (2001) 589–592.
- [17] G.J. Piazza, A. Nunez, T.A. Foglia, Enzymatic conversion of unsaturated fatty acids to epoxides using peroxygenase from oat seeds, Abstr. Papers Am. Chem. Soc. 224 (2002) U219–U220.
- [18] G.J. Piazza, A. Nunez, T.A. Foglia, Epoxidation of fatty acids, fatty methyl esters, and alkenes by immobilized oat seed peroxygenase, J. Mol. Catal. B-Enzymatic 21 (2003) 143–151.
- [19] R.D.S. Schneider, L.R.S. Lara, T.B. Bitencourt, M.D. Nascimento, M.R.D. Nunes, Chemo-enzymatic epoxidation of sunflower oil methyl esters, J. Brazilian Chem. Soc. 20 (2009) 1473–1477.
- [20] S. Warwel, M.R.G. Klaas, Chemoenzymatic epoxidation of unsaturated carboxylic acids, J. Mol. Catal. B-Enzymatic 1 (1995) 29–35.
- [21] C. Fitchett, N. Laughton, C. Chappell, M. Khan, V. Tverezovskiy, J. Tomkinson, P. Fowler, 2005. Oil ozonolysis. US 20,050,010,069 A1.
- [22] P. Fitton, E.H. Pryde, J.C. Cowan, Preparation of malonaldehyde acetals by ozonolysis of polyunsaturated fatty esters, J. Am. Oil Chem. Soc. 42 (1965) 14.
- [23] T.S. Omonov, E. Kharraz, J.M. Curtis, Ozonolysis of canola oil: a study of

product yields and ozonolysis kinetics in different solvent systems, J. Am. Oil Chem. Soc. 88 (2010) 689–705.

- [24] E.H. Pryde, C.M. Thierfelder, J.C. Cowan, Alcohols from ozonolysis products of unsaturated fatty esters, J. Am. Oil Chem. Soc. 53 (1976) 90–93.
- [25] F.D. Gunstone, Harwood, L. John, Dijkstra, J. Albert, The Lipid Handbook with CD-ROM, CRC Press, Boca Raton, FL, 2007.
- [26] N. Prileschajew, Oxydation ungesättigter Verbindungen mittels organischer Superoxyde, Berichte deutschen Chem. Gesellschaft 42 (1909) 4811–4815.
- [27] H. Dodiuk, S.H. Goodman, Handbook of Thermoset Plastics, Elsevier, Oxford, UK, 2014.
- [28] C.S. Cai, H.G. Dai, R.S. Chen, C.X. Su, X.Y. Xu, S. Zhang, L.T. Yang, Studies on the kinetics of in situ epoxidation of vegetable oils, Eur. J. Lipid Sci. Technol. 110 (2008) 341–346.
- [29] R.J. Gall, F.P. Greenspan, 1956. In situ epoxidation of oleic acid using formic acid. US 2,774,774 A.
- [30] M.C. Brasil, A.E. Gerbase, M.A. De luca, J.R. Gregorio, Organic-inorganic hybrid films based on hydroxylated soybean oils, J. Am. Oil Chem. Soc. 84 (2007) 289–295.
- [31] J. Salimon, B.M. Abdullah, R.M. Yusop, N. Salih, Synthesis, reactivity and application studies for different biolubricants, Chem. Central J. (2014) 8.
- [32] M. Smith, Organic Chemistry: An Acid - Base Approach, CRC Press, Boca Raton, FL, 2010.
- [33] G. Strukul, Catalytic Oxidations with Hydrogen Peroxide as Oxidant, Springer, Berlin, Germany, 1993.

- [34] M.N. Belgacem, A. Gandini, Monomers, Polymers and Composites from Renewable Resources, Elsevier, Amsterdam, The Netherlands, 2008.
- [35] F.S. Guner, Y. Yagci, A.T. Erciyes, Polymers from triglyceride oils, Prog. Polym. Sci. 31 (2006) 633–670.
- [36] S.N. Khot, J.J. Lascala, E. Can, S.S. Morye, G.I. Williams, G.R. Palmese, S.H. Kusefoglu, R.P. Wool, Development and application of triglyceridebased polymers and composites, J. Appl. Polym. Sci. 82 (2001) 703–723.
- [37] F.R. Earle, Epoxy oils from plant seeds, J. Am. Oil Chem. Soc. 47 (1970) A81.
- [38] L.W. Barrett, L.H. Sperling, C.J. Murphy, Naturally functionalized triglyceride oils in interpenetrating polymer networks, J. Am. Oil Chem. Soc. 70 (1993) 523–534.
- [39] H.S. Hwang, S.Z. Erhan, Modification of epoxidized soybean oil for lubricant formulations with improved oxidative stability and low pour point, J. Am. Oil Chem. Soc. 78 (2001) 1179–1184.
- [40] H.S. Hwang, A. Adhvaryu, S.Z. Erhan, Preparation and properties of lubricant basestocks from epoxidized soybean oil and 2-ethylhexanol, J. Am. Oil Chem. Soc. 80 (2003) 811–815.
- [41] A. Shani, Functionalization at the double-bond region of jojoba oil. 3. Hydroxylic derivatives, Ind. Eng. Chem. Prod. Res. Dev. 22 (1983) 121–123.
- [42] D.W. Chasar, M.J. Hughes, 2005. Method of making oleochemical oilbased polyols. US 6,891,053 B2.
- [43] P.S. Lathi, B. Mattiasson, Green approach for the preparation of biodegradable lubricant base stock from epoxidized vegetable oil, Appl. Catal. B-Environ. 69 (2007) 207–212.

- [44] G. Lligadas, J.C. Ronda, M. Galia, U. Biermann, J.O. Metzger, Synthesis and characterization of polyurethanes from epoxidized methyl oleate based polyether polyols as renewable resources, J. Polym. Sci. Part a-Polym. Chem. 44 (2006) 634–645.
- [45] B.M. Abdullah, J. Salimon, Epoxidation of vegetable oils and fatty acids: catalysts, methods and advantages, J. Appl. Sci. 10 (2010) 1545–1553.
- [46] M. Ionescu, Chemistry and Technology of Polyols for Polyurethanes, Rapra Technology Ltd, Shrewsbury, 2005.
- [47] M. Ionescu, Z.S. Petrovic, X.M. Wan, Ethoxylated soybean polyols for polyurethanes, J. Polym. Environ. 18 (2007) 1–7.
- [48] Z. Petrovic, I. Javni, A. Guo, W. Zhang, 2002. Method of making natural oilbased polyols and polyurethanes therefrom. US 6,433,121 B1.
- [49] Z. Petrovic, A. Guo, I. Javni, 2000a. Process for the preparation of vegetable oil-based polyols and electroninsulating casting compounds created from vegetable oil-based polyols. Google Patents.
- [50] Z. Petrovic, I. Javni, A. Zlatanic, A. Guo, 2006. Modified vegetable oilbased polyols. US 20,060,041,157 A1.
- [51] Z.S. Petrovic, Polyurethanes from vegetable oils, Polym. Rev. 48 (2008) 109– 155.
- [52] Z.S. Petrovic, A. Guo, W. Zhang, Structure and properties of polyurethanes based on halogenated and nonhalogenated soy-polyols, J. Polym. Sci. Part a-Polym. Chem. 38 (2000) 4062–4069.
- [53] Z.S. Petrovic, A. Guo, I. Javni, I. Cvetkovic, D.P. Hong, Polyurethane networks from polyols obtained by hydroformylation of soybean oil, Polym. Int. 57 (2008) 275–281.

- [54] A. Zlatanic, Z.S. Petrovic, K. Dusek, Structure and properties of trioleinbased polyurethane networks, Biomacromolecules 3 (2002) 1048–1056.
- [55] A. Zlatanic, C. Lava, W. Zhang, Z.S. Petrovic, Effect of structure on properties of polyols and polyurethanes based on different vegetable oils, J. Polym. Sci. Part B-Polym. Phys. 42 (2004) 809–819.
- [56] P.B. Smith, Richard A. Gross, Biobased Monomers, Polymers, and Materials, Oxford University Press, Indianapolis, 2012.
- [57] T.W. Abraham, D.G. Dai, G.R. De, J. Malsam, 2008. Viscoelastic polyurethane foams comprising oligomeric natural oil polyols. WO 2,008,063,594 A1.
- [58] M.Y. Cheong, T.L. Ooi, S. Ahmad, W.M.Z.W. Yunus, D. Kuang, Synthesis and characterization of palm-based resin for UV coating, J. Appl. Polym. Sci. 111 (2009) 2353–2361.
- [59] S.W. Choi, D.W. Seo, Y.D. Lim, Y.G. Jeong, M.S.I. Mollah, H. Park, T.W. Hong, W.G. Kim, Synthesis and properties of multihydroxy soybean oil from soybean oil and polymeric methylene-diphenyl-4,4'-diisocyanate/ multihydroxy soybean oil polyurethane adhesive to wood, J. Appl. Polym. Sci. 121 (2011) 764–769.
- [60] J. Curtis, G. Liu, T. Omonov, E. Kharraz, 2013, Polyol synthesis from fatty acids and oils. US 2013/0274494 A1.
- [61] H.H. Dai, L.T. Yang, B. Lin, C.S. Wang, G. Shi, Synthesis and characterization of the different soy-based polyols by ring opening of epoxidized soybean oil with methanol, 1,2-ethanediol and 1,2-propanediol, J. Am. Oil Chem. Soc. 86 (2009) 261–267.

- [62] H.A. Hassan, T.N.M.T. Ismail, M.N. Sattar, S.S. Hoong, T.L. Ooi, S. Ahmad, K.D.P. Palam, M.Y. Cheong, 2008. Process to produce polyols. US 20,080,293,913 A1.
- [63] H. Cramail, A. Boyer, E. Cloutet, R. Bakhiyi, C. Alfos, 2013, Novel method for preparing polyols and products obtained. US 20,130,005,937 A1.
- [64] N. Ramírez-De-Arellano-Aburto, A. Cohen-Barki, J. Cruz-Gómez, 2003. Process for the production of oleochemical polyols. US 6,548,609 B2.
- [65] M. Desroches, M. Escouvois, R. Auvergne, S. Caillol, B. Boutevin, From vegetable oils to polyurethanes: synthetic routes to polyols and main industrial products, Polym. Rev. 52 (2012) 38–79.
- [66] S. Ahmad, S.M. Ashraf, E. Sharmin, F. Zafar, A. Hasnat, Studies on ambient cured polyurethane modified epoxy coatings synthesized from a sustainable resource, Prog. Cryst. Growth Characterization Mater. 45 (2002) 83–88.
- [67] R.E. Harry-O'kuru, C.J. Carriere, Synthesis, rheological characterization, and constitutive modeling of polyhydroxy triglycerides derived from milkweed oil, J. Agric. Food Chem. 50 (2002) 3214–3221.
- [68] R.E. Harry-O'kuru, R.A. Holser, T.P. Abbott, D. Weisleder, Synthesis and characteristics of polyhydroxy triglycerides from milkweed oil, Ind. Crops Prod. 15 (2002) 51–58.
- [69] A. Adhvaryu, S.Z. Erhan, J.M. Perez, Tribological studies of thermally and chemically modified vegetable oils for use as environmentally friendly lubricants, Wear 257 (2004) 359–367.
- [70] A. Adhvaryu, Z. Liu, S.Z. Erhan, Synthesis of novel alkoxylated

triacylglycerols and their lubricant base oil properties, Ind. Crops Prod. 21 (2005) 113–119.

- [71] S.Z. Erhan, Industrial Uses of Vegetable Oils, AOCS Press, Illinois, 2005.
- [72] S.Z. Erhan, B.K. Sharma, Z.S. Liu, A. Adhvaryu, Lubricant base stock potential of chemically modified vegetable oils, J. Agric. Food Chem. 56 (2008) 8919–8925.
- [73] B.K. Sharma, A. Adhvaryu, Z.S. Liu, S.Z. Erhan, Chemical modification of vegetable oils for lubricant applications, J. Am. Oil Chem. Soc. 83 (2006) 129–136.
- [74] B.K. Sharma, Z.S. Liu, A. Adhvaryu, S.Z. Erhan, One-pot synthesis of chemically modified vegetable oils, J. Agric. Food Chem. 56 (2008) 3049–3056.
- [75] L. Bernard, D. Charlemagne, M. Huet, P. Metivier, 2000, Method for expoxidizing or dihydroxilating unsaturated fatty acids. WO 2,000,044,704 A1.
- [76] A. Biswas, A. Adhvaryu, S.H. Gordon, S.Z. Erhan, J.L. Willett, Synthesis of diethylamine-functionalized soybean oil, J. Agric. Food Chem. 53 (2005) 9485–9490.
- [77] C.S. Lee, T.L. Ooi, C.H. Chuah, S. Ahmad, Synthesis of palm oil-based diethanolamides, J. Am. Oil Chem. Soc. 84 (2007) 945–952.
- [78] H. Zhao, R. Herrington, F. Rodriguez, 2011. Natural oil based autocatalytic polyols. US 20,110,118,432 A1.
- [79] D. Casper, T. Newbold, 2006, Methods of preparing hydroxy functional vegetable oils. US 20,060,041,156 A1.
- [80] N. Luo, T. Newbold, 2008. Method for vegetable oil derived polyols and polyurethanes made therefrom. US 20,080,262,259 A1.
- [81] T.C. Waidner, V.M. Granquist, 2007. Novel polyol-polyamine synthesized

from vegetable oils. US 20,070,155,934 A1.

- [82] C.S. Lee, T.L. Ooi, C.H. Chuah, S. Ahmad, Rigid polyurethane foam production from palm oil-based epoxidized diethanolamides, J. Am. Oil Chem. Soc. 84 (2007) 1161–1167.
- [83] G. Maerker, Determination of oxirane content of derivatives of fats, J. Am. Oil Chem. Soc. 42 (1965) 329.
- [84] A. Guo, Y.J. Cho, Z.S. Petrovic, Structure and properties of halogenated and nonhalogenated soy-based polyols, J. Polym. Sci. Part a-Polym. Chem. 38 (2000) 3900–3910.
- [85] F. Scholnic, E.J. Saggese, A.N. Wrigley, G.R. Riser, Urethane foams from animal fats 5. Flame resistant foams from hypohalogenated glycerides, J. Am. Oil Chem. Soc. 47 (1970) 180–182.
- [86] S.D. Miao, S.P. Zhang, Z.G. Su, P. Wang, A novel vegetable oil-lactate hybrid monomer for synthesis of high-T-g polyurethanes, J. Polym. Sci. Part a-Polym. Chem. 48 (2010) 243–250.
- [87] Z.S. Petrovic, L.T. Yang, A. Zlatanic, W. Zhang, I. Javni, Network structure and properties of polyurethanes from soybean oil, J. Appl. Polym. Sci. 105 (2007) 2717–2727.
- [88] S.V. Ley, A.J.P. Stewart-Liddon, D. Pears, R.H. Perni, K. Treacher, Hydrogenation of aromatic ketones, aldehydes, and epoxides with hydrogen and Pd(0)EnCat (TM) 30NP, Beilstein J. Org. Chem. 2 (2006).
- [89] U. Biermann, W. Friedt, S. Lang, W. Luhs, G. Machmuller, J.O. Metzger, M.R. Klaas, H.J. Schafer, M.P. Schneider, New syntheses with oils and fats as renewable raw materials for the chemical industry, Angew. Chem. Int. Ed. 39 (2000) 2206–2224.

- [90] R. Wool, S. Kusefoglu, G. Palmese, S. Khot, R. Zhao, 2000. High modulus polymers and composites from plant oils. US 6,121,398 A.
- [91] F. Habib, M. Bajpai, Synthesis and characterization of acrylated epoxidized soybean oil for UV cured coatings, Chem. Chem. Technol. 5 (3) (2011) 317–327.
- [92] A. Campanella, M. Fahimian, R.P. Wool, J. Raghavan, Synthesis and rheology of chemically modified canola oil, J. Biobased Mater. Bioenergy 3 (2009) 91–99.
- [93] E. Sharmin, S.M. Ashraf, S. Ahmad, Epoxidation hydroxylation acrylation and urethanation of Linum usitatissimum seed oil and its derivatives, Eur. J. Lipid Sci. Technol. 109 (2007) 134–146.
- [94] G. Wuzella, A.R. Mahendran, U. Muller, A. Kandelbauer, A. Teischinger, Photocrosslinking of an acrylated epoxidized linseed oil: kinetics and its application for optimized wood coatings, J. Polym. Environ. 20 (2012) 1063–1074.
- [95] J.G. Homan, X.H. Yu, T.J. Connor, S.L. Cooper, Castor-oil based UVcurable polyurethane-acrylate interpenetrating networks, J. Appl. Polym. Sci. 43 (1991) 2249–2257.
- [96] M.A. Ali, L.T. Ooi, A. Salmiah, U.S. Ishiaku, Z.A.M. Ishak, New polyester acrylate resins from palm oil for wood coating application, J. Appl. Polym. Sci. 79 (2001) 2156–2163.
- [97] V. Kolot, S. Grinberg, Vernonia oilbased acrylate and methacrylate polymers and interpenetrating polymer networks with epoxy resins, J. Appl. Polym. Sci. 91 (2004) 3835–3843.
- [98] T. Endo, Effective fixation of carbon dioxide and its application to polymer synthesis, Abstracts Papers Am. Chem. Soc. 227 (2004) U1076–U11076.

- [99] M. Jalilian, H. Yeganeh, M.N. Haghighi, Synthesis and properties of polyurethane networks derived from new soybean oil-based polyol and a bulky blocked polyisocyanate, Polym. Int. 57 (2008) 1385–1394.
- [100] Z.R. Li, Y.H. Zhao, S.R. Yan, X.K. Wang, M.Q. Kang, J.W. Wang, H.W. Xiang, Catalytic synthesis of carbonated soybean oil, Catal. Lett. 123 (2008) 246–251.
- [101] P.G. Parzuchowski, M. Jurczyk-Kowalska, J. Ryszkowska, G. Rokicki, Epoxy resin modified with soybean oil containing cyclic carbonate groups, J. Appl. Polym. Sci. 102 (2006) 2904– 2914.
- [102] G.L. Wilkes, S. Sohn, B. Tamami, 2006. Nonisocyanate polyurethane materials, and their preparation from epoxidized soybean oils and related epoxidized vegetable oils, incorporation of carbon dioxide into soybean oil, and carbonation of vegetable oils. US 7,045,577 B2.
- [103] M. Bahr, R. Mulhaupt, Linseed and soybean oil-based polyurethanes prepared via the non-isocyanate route and catalytic carbon dioxide conversion, Green Chem. 14 (2012) 483–489.
- [104] J. Guan, Y.H. Song, Y. Lin, X.Z. Yin, M. Zuo, Y.H. Zhao, X.L. Tao, Q. Zheng, Progress in study of nonisocyanate polyurethane, Ind. Eng. Chem. Res. 50 (2011) 6517–6527.
- [105] I. Javni, D.P. Hong, Z.S. Petrovic, Soy-based polyurethanes by nonisocyanate route, J. Appl. Polym. Sci. 108 (2008) 3867–3875.
- [106] T.S. Omonov, J.M. Curtis, 2012.
 Thermoset biobased resins and aldehyde-free biocomposites. US PPA 61,727,357. November 16, 2012.

- [107] T.S. Omonov, J.M. Curtis, Biobased epoxy resin from canola oil, J. Appl. Polym. Sci. 131 (8) (2014) 40142.
- [108] N. Boquillon, Use of an epoxidized oil-based resin as matrix in vegetable fibers-reinforced composites, J. Appl. Polym. Sci. 101 (2006) 4037–4043.
- [109] A.R. Mahendran, G. Wuzella, A. Kandelbauer, N. Aust, Thermal cure kinetics of epoxidized linseed oil with anhydride hardener, J. Thermal Anal. Calorimetry 107 (2012) 989–998.
- [110] B.J.K. Ahn, S. Kraft, X.S. Sun, Chemical pathways of epoxidized and hydroxylated fatty acid methyl esters and triglycerides with phosphoric acid, J. Mater. Chem. 21 (2011) 9498–9505.
- [111] T.A. Isbell, Chemistry and physical properties of estolides, Grasas Y Aceites 62 (2011) 8–20.
- [112] R. Awang, A.N. Azizan, S. Ahmad, 2009, Method to produce polyhydroxy carboxylic acid esters of polyethylene glycol. US 20,090,192,329 A1.
- [113] S.C. Cermak, T.A. Isbell, Synthesis of estolides from oleic and saturated fatty acids, J. Am. Oil Chem. Soc. 78 (2001) 557–565.
- [114] B.R. Moser, B.K. Sharma, K.M. Doll, S.Z. Erhan, Diesters from oleic acid: synthesis, low temperature properties, and oxidation stability, J. Am. Oil Chem. Soc. 84 (2007) 675–680.
- [115] R. Awang, A.N. Azizan, S. Ahmad, W.M.Z.W. Yunus, Characterization of estolides from dihydroxystearic acid, J. Oil Palm Res. 19 (2007) 350–355.
- [116] R. Awang, A.N. Azizan, S. Ahmad, W.M.Z.W. Yunus, N. Ahmad, Palmbased estolide as an ingredient in shampoo: a preliminary study, J. Oil Palm Res. 19 (2007) 416–420.

- [117] N. Karak, Vegetable Oil-Based Polymers. Properties, Processing and Applications, Woodhead Publishing Limited, Cambridge, UK, 2012.
- [118] R.P. Wool, X.S. Sun, Bio-Based Polymers and Composites, Burlington, MA, Academic Press, 2005.
- [119] A. Bhattacharya, J.W. Rawlins, P. Ray, Polymer Grafting and Crosslinking, John Wiley and Sons, New York, 2008.
- [120] J. Rosch, R. Mulhaupt, Polymers from renewable resources: polyester resins and blends based upon anhydridecured epoxidized soybean oil, Polym. Bull. 31 (1993) 679–685.
- [121] S.P. Pappas, Radiation Curing: Science and Technology, Springer, Berlin, Germany, 1992.
- [122] C. Decker, Kinetic study and new applications of UV radiation curing, Macromol. Rapid Commun. 23 (2002) 1067–1093.
- [123] Z. Chen, 2014, Plant oil-based UVcurable coating. US 8,815,969 B2.
- [124] R. Raghavachar, G. Sarnecki, J. Baghdachi, J. Massingill, Cationic, thermally cured coatings using epoxidized soybean oil, J. Coat. Technol. 72 (2000) 125–133.
- [125] L. Fertier, H. Koleilat, M. Stemmelen, O. Giani, C. Joly-Duhamel, V. Lapinte, J.J. Robin, The use of renewable feedstock in UV-curable materials: a new age for polymers and green chemistry, Prog. Polym. Sci. 38 (2013) 932–962.
- [126] K. Pelletier, N. Belgacem, A. Gandini, Acrylated vegetable oils as photocrosslinkable materials, J. Appl. Polym. Sci. 99 (2006) 3218–3221.
- [127] M. Fahimian, D. Adhikari, J. Raghavan, R. Wool, Biocomposites from

canola oil based resins and hemp and flax fibers. 13th European Conference on Composite Materials 2008 Stockhol, Sweden. ECCM13.

- [128] J. Lu, S. Khot, R.P. Wool, New sheet molding compound resins from soybean oil. I. Synthesis and characterization, Polymer 46 (2005) 71–80.
- [129] S.J. Park, F.L. Jin, J.R. Lee, Synthesis and thermal properties of epoxidized vegetable oil, Macromol. Rapid Commun. 25 (2004) 724–727.
- [130] C. Ozturk, S.H. Kusefoglu, Polymerization of epoxidized soybean oil with maleinized soybean oil and maleic anhydride grafted polypropylene mixtures, J. Appl. Polym. Sci. 118 (2010) 3311–3317.
- [131] C. Ozturk, S.H. Kusefoglu, Polymerization of epoxidized soybean oil with maleinized polybutadiene, J. Appl. Polym. Sci. 120 (2011) 116–123.

- [132] C. Ozturk, S.H. Kusefoglu, New polymers from epoxidized soybean oil with pyridine derivatives, J. Appl. Polym. Sci. 121 (2011) 2976–2984.
- [133] Z.S. Liu, S.Z. Erhan, Ring-opening polymerization of epoxidized soybean oil, J. Am. Oil Chem. Soc. 87 (2010) 437–444.
- [134] J.V. Crivello, R. Narayan, Epoxidized triglycerides as renewable monomers in photoinitiated cationic polymerization, Chem. Mater. 4 (1992) 692–699.

Further Reading

Available: www.hobum.de. [Accessed March 2 2015].

Page left intentionally blank

8 Enzyme-Assisted Synthesis of Plant Oil-Based Polymers

Yi-Shu Tai and Kechun Zhang

Department of Chemical Engineering and Materials Science, University of Minnesota, Minneapolis, MN, USA

Ο U T L I N E

8.1	Introduction 1					
8.2	Enzyme-Assisted Synthesis of Plant Oil-Based Polyesters					
	8.2.1	Lipase-Catalyzed Synthesis of Functional Polyesters	129			
	8.2.2	Lipase-Catalyzed Polycondensation of Unsaturated or Epoxidized α , ω -Dicarboxylic Acids and Diols	132			
	8.2.3	Lipase-Catalyzed Polycondensation of Hydroxy Fatty Acids and Derivatives	135			
8.3	Enzyme-Assisted Plant Oil Transformations					
	8.3.1	Sugar Fatty Acid Esters	138			
	8.3.2	Transesterification	139			
	8.3.3	Enzyme-Catalyzed Epoxidation	141			
	8.3.4	Fatty Acids Hydroxylation	143			
8.4	Concl	usion	144			
Refe	References 14					

8.1 Introduction

Plant oil is an attractive renewable resource for polymer synthesis based on its availability and relatively low price. From the perspective of green manufacturing, we should consider not only feedstocks but also the synthesis process. Traditional chemical catalysis for polymer synthesis usually requires high temperature and toxic catalysts. The harsh reaction conditions could result in lack of selectivity. Chemical polymerization becomes more challenging when synthesizing complex or well-defined polymers. Enzymes can be an alternative to the process. Enzymes work under milder reaction conditions and are generally highly chemo-, regio-, enantio-, stero-, and choroselective [1,2]. Therefore, enzymatic processes are cleaner (no metal used and less byproducts), consuming less energy, having better structural control, and easier for operation (protection-deprotection not required). Currently, lipases are the predominantly used enzyme in polymer syntheses [3]. Lipases do not require any cofactor and thus can be applied as cell-free biocatalysts. In this chapter, several examples are discussed on enzymecatalyzed polymerization. Furthermore, biotransformations of plant oil are discussed in the second part of the chapter to demonstrate the diverse application of enzymes. The functionality of important enzymes discussed in this chapter is summarized in Table 8.1.

Table 8.1 Representative Reactions Catalyzed by Enzymes for Plant Oil Polymerization and Transformation

Enzymes	Reactions	Reactants	Products
Lipase CA	Polycondensation	DVS, glycerol, and unsaturated fatty acids (C18:2, C18:3)	As shown in Figure 8.1
Lipase CA	Polycondensation	DVS, glycerol, and epoxidized fatty acids (C18:1, C18:2, C18:3)	As shown in Figure 8.2
N435	Epoxidation	Oleic acid	As shown in Figure 8.3
N435	Polycondensation	Diols (C3, C4), unsaturated/ epoxidized α , ω -DMDA (C18, C20, C26)	As shown in Figure 8.4
N435	Polycondensation	Diols (C3, C6, C16), unsaturated/epoxidized α , ω -diacid (C18, C22)	As shown in Figure 8.5
Lipase RM	Polycondensation	Pentaerythritol, ricinoleaic acid	As shown in Figure 8.6
Lipase PC	Polycondensation	Ricinoleaic acid or methyl ricinoleaic acid	As shown in Figure 8.7
N435	ROP	Ambrettolide epoxide	As shown in Figure 8.8
N435	Polycondensation	Isopropyl aleuriteate	As shown in Figure 8.8
N435	Polycondensation	<i>cis</i> -9,10-epoxy-18- hydroxyoctadecanoic acid	As shown in Figure 8.9
Lipase CA	Esterification	Sucrose, palmitic acid	As shown in Figure 8.10
Lipase RD	Transesterification	Tripalmitin, ethanol, oleic acid	As shown in Figure 8.11
P450 (CYP52)	<i>w</i> -Hydroxylation	Palmitic acid	16-hydroxy palmitic acid
P450 (CYP102)	<i>ω</i> -2 Hydroxylation	Arachidonic acid	As shown in Figure 8.14b
FAO	Oxidation	<i>ω</i> -Hydroxy fatty acid	Fatty acid semialdehyde
FALDH	Oxidation	Fatty acid semialdehyde	α, <i>ω</i> -Diacid
Hydratase	Hydroxylation	Linoleic acid	As shown in Figure 8.14c

Abbreviations: CA, Candida Antarctica. N435, Novozym 435. P450, Cytochromes P450 monooxygenase. DVS, divinyl sebacate. DMDA, α , ω -dicarboxylic acid methyl esters. RM, Rhizomucor miehei. PC, Pseudomonas cepacia. ROP, ring opening polymerization. RD, Rhizopus delemar. FAO, fatty alcohol oxidase. FALDH, fatty aldehyde dehydrogenase.

8.2 Enzyme-Assisted Synthesis of Plant Oil-Based Polyesters

8.2.1 Lipase-Catalyzed Synthesis of Functional Polyesters

8.2.1.1 Biodegradable and Cross-Linkable Polyesters

Cross-linkable polyesters with an unsaturated fatty acid moiety in the side chain have been produced from renewable plant oil by lipase-catalyzed synthesis. These polyesters were synthesized by the polymerization of divinyl sebacate and glycerol in the presence of higher unsaturated acids, which are plant-oil derivatives (Figure 8.1) [4,5]. To identify enzymes that are active for the polymerization, lipases from Candida antarctica, Mucor miehei, and Pseudomonas cepacia were used as catalysts for the polymerization of equimolar ratio of divinyl sebacate 1 and glycerol 2 in the presence of linoleic acid (3a) (Figure 8.1) in bulk at 60°C for 24 h. All the three candidates could catalyze the regioselective polymerization of 1 and 2



Figure 8.1 Lipase-catalyzed synthesis of crosslinkable polyesters.

[6]. Among these enzymes, *Candida antarc*tica lipase had the highest catalytic activity for the production of 4 in a 58% yield; however, the other two lipases showed no or very low activity toward the synthesis of this oily polymer. The estimated molecular weight of 4 (Figure 8.1) determined by size-exclusion chromatography (SEC) was 4700. Therefore, Candida antarctica lipase was used to optimize the process parameters on both linoleic acid (3a) and linolenic acid (3b) (Figure 8.1). Compared with the optimal concentration, when an excessive amount of enzyme was used, the yield decreased because of the more frequent hydrolysis of the vinyl ester. With a smaller amount of enzyme, higher molecular weight was observed but the unsaturated acid content and the yield were lower. The yield and molecular weight were improved if the polymerization was carried out in a small amount of toluene: nonetheless, the acid content was lower compared with that of the bulk polymerization. The effect of temperature on the polymerization reaction has been studied. Both the yield and molecular weight increased when temperature increased up to 75°C, while the acid content of the polymer remained almost the same. Reducing pressure usually improves the molecular weight in the enzymatic synthesis of polyesters since the removal of the leaving group (vinyl alcohol) drives the equilibrium to the production of polymer [7]. Compared with the polymer produced under atmospheric pressure, the yield and molecular weight increased when the polymerization was carried out under reduced pressure at 2700 Pa. The polyester produced under reduced pressure using linoleic acid or linolenic acid was subjected to hardening by cobalt naphthenate or thermal treatment to give transparent and gloss films. The pencil-scratch hardness [8] of the thermally treated films from linoleic acid and linolenic acid after 24 h was 5 and 2 B,

respectively, which were superior to that produced by cobalt naphthenate oxidation. The biodegradability of the cross-linked film was evaluated by biochemical oxygen demand (BOD) in an activated sludge. Results showed that the film could be degraded gradually to 45% after 42 days, which demonstrated the good biodegradability of this cross-linked film.

8.2.1.2 Epoxide-Containing Polyesters

Plant oil-based unsaturated fatty acids such as oleic, linoleic, and linolenic acids were used as starting substrate for the enzymatic synthesis of epoxide-containing polyesters. Theses polyesters containing an epoxide moiety in the side chain were synthesized using lipase catalysis via two routes (Figure 8.2) [9]. In both routes, Candida antarctica lipase was used to catalyze the polymerization and the epoxidation of unsaturated groups in fatty acids. In route A, polyester 4 possessing an unsaturated group in the side chain was first synthesized by the lipasecatalyzed polymerization of divinyl sebacate 1 and glycerol 2 in the presence of unsaturated fatty acids 3 (Figure 8.2). The reaction was catalyzed by Candida antarctica lipase at 60°C and was carried out under reduced pressure (2700 Pa) to shift the equilibrium to polymer production. The unsaturated fatty



Figure 8.2 Two routes for lipase-catalyzed synthesis of epoxide-containing polyesters.

acid groups in polyester 4 were then enzymatically epoxidized using hydrogen peroxide as oxidizing agent at room temperature to generate the target epoxide-containing polyesters 6 (Figure 8.2). To synthesize the epoxidecontaining polyesters 6 via the other route B, the epoxidation of unsaturated fatty acids 3 was first catalyzed by lipase using hydrogen peroxide as the oxidizing agent, followed by the lipase-catalyzed polymerization of divinyl sebacate 1 and glycerol 2 in the presence of the epoxidized fatty acids 5 under reduced pressure (Figure 8.2). The epoxidized ratio of **5** and **6** was similar, indicating that the epoxy groups remained intact during polymerization. NMR and IR spectra were used to confirm the polymer structure. High epoxidized ratios (>75%) were achieved by both routes. The epoxide-containing polyesters produced via route B were subjected to curing by thermal treatment at 150°C for 2 h to give transparent and glossy films. The pencil-scratch hardness of the cured films increased with the increasing epoxidation content. Compared with the films produced from the polyesters possessing unsaturated acid moieties in the side chain mentioned in Section 8.2.1.1, the hardness of the epoxide-containing films was improved. The hardest cured film reached pencil-scratch hardness 3H was achieved using linolenic acid as the starting substrate. The cured film prepared from linolenic acid showed good biodegradability which was tested by BOD in an activated sludge and the biodegradation reached more than 50% after 50 day.

8.2.1.3 Oleic Acid-Based Polyesters

Besides using plant oil-derived fatty acids to introduce functional side-chain moieties, fatty acids can also be polymerized to produce polymers with lipids in the backbone structure. Linear and cross-linked poly(oleic acid) was synthesized via a chemo-enzymatic route (Figure 8.3). The fatty acid derived monomer was produced via epoxidation of oleic acid. Instead of achieving the epoxidation chemically by reacting unsaturated fatty acids with peroxy acids, enzyme-catalyzed epoxidation [10] provides a better alternative with fewer side reactions and can be performed under milder conditions (enzymatic epoxidation will be further discussed in Section 8.3.3). Hydrogen peroxide was used as the oxidizing reagent to epoxidize unsaturated fatty acids catalyzed by lipase. *Candida antarctica* lipase B has been broadly used as catalysts in polymerization and epoxidation reactions



Figure 8.3 Chemo-enzymatic synthesis of crosslinkable oleic acid–base polyesters.
because of its high thermal stability, activity, as well as its high regio-, chemo-, and enantio-selectivity [1]. Therefore, Novozym 435 consisting of immobilized Candida antarctica lipase B on a macroporous poly(methyl methacrylate-co-butyl methacrylate) resin was used as the catalyst for epoxidation of oleic acid 1 into oleic acid epoxide 2 (Figure 8.3). Since the produced epoxide was not thermally stable, the reaction was performed at 40°C and oleic acid was completely converted within 5 h based on HPLC tests. Oleic acid epoxides produced were recovered by solvent evaporation and were polymerized by heating up to different temperatures at 90, 120, or 150°C for 6 h. IR spectra of the produced polymers suggested that better conversion yield and faster reaction rate could be achieved with higher reaction temperature. However, polymers produced at 150°C solidified within several minutes and could not be dissolved in common organic solvents like chloroform. On the other hand, polymers produced at 120°C had lower molecular weight but stayed as a viscous liquid, which made them feasible for downstream processing. NMR spectra showed the existence of ether bond from ring opening of the epoxy moiety, but polyesters were dominant which accounted for 90% in the polymer structure. Hydroxyl groups in the produced polyesters **3** provide the opportunity of synthesizing more cross-linked polymers 4 (Figure 8.3). Transparent films were produced by mixing poly(oleic acid) with diisocyanate compound such as 4,4'-methylenebis(phenyl isocyanate). The hardness of the cross-linked films could be adjusted by the amount of diisocyanate added. Thermal gravimetric analysis (TGA) was applied and three decomposition peak temperatures (260-350, 350-395, and 395–475°C) were identified which were corresponding to the decomposition of urethane

bonds, the ester bonds, and the gasification of remaining, respectively.

8.2.2 Lipase-Catalyzed Polycondensation of Unsaturated or Epoxidized α, ω-Dicarboxylic Acids and Diols

Symmetrically unsaturated and expoxidized α, ω -dicarboxylic acid methyl esters $(C_{18}, C_{20}, and C_{26})$ were copolymerized with 1,3-propanediol or 1,4-butanediol catalyzed by Novozym 435 (Figure 8.4) [11]. Unsaturated fatty acid methyl esters serving as the starting materials were obtained by tranesterification of vegetable oil with methanol; for example, methyl oleate and methyl erucate were derived from high oleic sunflower oil and high erucic rapeseed oil, respectively. First, Grubbs catalysts [12] were applied to metathetically cleave these methyl esters to yield 1-decene and methyl 9-decenoate 1a or 1-decne and methyl 13-tetradecenoate 1c (Figure 8.4a). It is worth noting that methyl 10-undecenoate **1b** is available as a product from castor oil. Subsequently, Grubbs catalysts were used to perform the metathetical dimerization of 1a, 1b, and 1c, resulting in long-chain, symmetrically unsaturated α, ω -dicarboxylic acid dimethyl esters (2a-2c) (Figure 8.4b). Catalyzed by Novozym 435, these diesters (2a-2c) were converted to epoxidized diesters (3a-3c) by hydrogen peroxide in methyl acetate with yields of more than 97% (Figure 8.4b). The diesters (2a–2c) and their epoxidized derivatives (3a-3c) were then polymerized with 1,3-propanediol or 1,4-butanediol in diphenyl ether using Novozym 435 as catalysts (Figure 8.4c). Temperature effect of the lipase-catalyzed



Figure 8.4 Novozym 435-catalyzed copolymerization of unsaturated and epoxidized α , ω -dicarboxylic acids and diols.

polycondensation was examined and the results showed higher yield (62-79%) and higher molecular weights (1900-7860 g/mol) could be achieved with higher reaction temperature $(30-70^{\circ}\text{C})$. The molecular weights increased with increasing reaction time up to 4 h. Determined by SEC, molecular weights of the unsaturated polyesters from 1,3-propanediol (4a-4c) ranged between 1950 g/mol and 2250 g/mol and marginally higher molar masses with the corresponding epoxidized polyesters (5a-5c) were 2450–3300 g/mol (Figure 8.4c). The polyesters had melting temperatures between 47 and 75°C determined by differential scanning

calorimetry (DSC). The polyesters prepared from 1,4-butanediol had higher molecular weights than those of the corresponding polymers from 1,3-propanediol. Molecular weights were within the range of 7900–9900 and 9000–11,600 g/mol for the unsaturated polyesters (**6a–6c**) and the expoxidized homologues (**7a–7c**), respectively (Figure 8.4c). These polyesters had melting temperatures between 55°C and 74°C. Compared with the unsaturated polyesters, epoxidized polyesters homologues produced via lipase-catalyzed polycondensation did not give higher melting temperature. All the double bonds or epoxy groups remained intact during the polymerizations. With these functional groups, the produced polyesters were capable of being further derivatized or crosslinked.

Alternatively, α, ω -dicarboxylic acid monomers could be prepared from fatty acids via whole-cell biocatalysis using Candida tropicalis ATCC20962 or related engineered strains. This biotransformation could be realized by endogenous enzymes when β-oxidation pathway was blocked. Cytochrome P450 monooxygenase (CYP52) first catalyzed ω -hydroxylation (more discussions about P450 monooxygenase can be found in Section 8.3.4). The ω -hydroxy fatty acid was further transformed to the corresponding fatty aldehyde catalyzed by fatty alcohol oxidase and to the target compound, α , ω -dicarboxylic acid, via fatty aldehyde dehydrogenase. For example, three α, ω -dicarboxylic fatty acids, 1,18-cis-9-octadecenedioic, 1,22-cis-9-docosenedioic, and 1,18-cis-9,10-epoxyoctadecanedioic acids (2a-2b, 3c), were synthesized from renewable fatty acids, oleic, erucic, epoxy stearic acids (1a-1c), respectively (Figure 8.5) via whole-cell biocatalysis catalyzed by C. tropicalis ATCC20962 [13]. The biotransformations were conducted in 30 mL conversion media in shake flasks supplementing with 20 g/L fatty acids. The production dicarboxylic acids and the consumption of fed fatty acids were monitored by liquid chromatography (LC)-mass spectrometry (MS). The production titers of target dicarboxylic acids reached 17.3 (77.6 mol%), 14.2 (64.8 mol%), and 19.1 (86.2 mol%) g/L after 48 h for oleic acid and 72 hours for erucic and epoxy stearic acids, respectively. The low solubility of longerchain fatty acids might cause their slower conversion rates. Biotransformation of oleic acid to 1.18-cis-9-octadecenedioic acid was further scaled up in a 3-L fermentor with



Figure 8.5 Whole-cell biocatalyst applied for α, ω -dicarboxylic acid generation followed copolymerization with diols catalyzed by immobilized lipase.

better pH, glucose feeding, and dissolved oxygen controls, resulting in 31 g/L of the dicarboxylic acid after 60 h with higher productivities up to 0.5 g/L/h. These α, ω -dicarboxylic acids were then copolymerized with diols (1,3-propanediol, 1,8-octanediol, and 1,16-hexadecanediol) to produce functional polyesters using immobilized Candida antarctica lipase B (Novozym 435) as catalysts (Figure 8.5). For example, molecular weight of polyester produced by in-bulk polycondensations of 1,18-cis-9-octadecenedioic with 1,8-octanediol increased from 15,000 to 37,000, and then to 44,000 g/mol at 2, 24, and 48 h, respectively. Overall, polyesters with molecular weights from 25,000 g/mol to 57,000 g/mol were produced with different combinations of diols and α, ω diacids. Although higher molecular-weight

polyester was produced when the reaction was performed in diphenyl ether, in-bulk polymerization was preferred due to difficult separation of diphenyl ether and low efficiency of the overall process. Similar to the previous example, molecular weights of polyesters from the same diacids increased with increasing diol molecular weights. Unsaturated and epoxy moieties on these diacids remained intact during polymerization. These polyesters had lower melting points (23–40°C) than their corresponding saturated polyesters due to the disruption of crystallization by functional groups.

8.2.3 Lipase-Catalyzed Polycondensation of Hydroxy Fatty Acids and Derivatives

8.2.3.1 Poly(Ricinoleaic Acid) and its Derivatives

Ricinoleaic acid (RA), a hydroxyl fatty acid formally named as 12-hydroxy-9-cisoctadecenoic acid, is a main component of the seed oil of castor (Ricinus comminus). RA is a bifunctional molecule containing a secondary hydroxyl group and a double bond. Poly(hydroxyl acids) such as poly(RA) and their derivatives are biodegradable and can be used as environmentally friendly lubricant materials because of their low melting point and high viscosity. Poly(RA) was prepared via lipase-catalyzed polycondensation under mild temperature to avoid problems of discoloration, odour, dehydration, and high energy costs associated with high-temperature chemical processes. Poly(RA) 1 acyl donors and polyols (trimethylolpropane (TMP), pentaerythritol 2 (PE), and dimer diol(DD)) acyl acceptors could then be used to synthe-



Figure 8.6 Branched star polyesters synthesis catalyzed by lipase.

size branched star polyesters such as pentaerythrotol-poly(ricinoleic acid) polyester 3 in bulk by lipase-catalyzed polyconsensation [14] (Figure 8.6). Poly(RA) was first prepared using immobilized Candida antarctica lipase B as the catalyst. Lipase-catalyzed polymerization of RA proceeded slowly since RA has a high viscosity and only contains a secondary hydroxyl group for esterification. Poly(RA) with an average MW of 724 and 1040 was obtained from technical grade RA in bulk polymerization at 70°C for 7 and 10 days, respectively. The propagation step in the polymerization of RA was rate-limiting, following first-order kinectics in respect of monomer concentration. In the next step, the enzymatically synthesized Poly(RA) was mixed with polyols (TME, PE, DD) for the synthesis of polyol-poly(RA) polyesters catalyzed by immobilized lipases from Candida antarctica or Rhizomucor miehei. The synthesis was carried out at 70°C for 14 days and the reaction rates catalyzed by both lipases were similar. All three polyols were successfully used as acyl donors. The highest molecular weight product was synthesized from PE and poly(RA) with an average molecular weight of 4850 according to ¹H-NMR. Gel permeation chromatography indicated that the product mixture was composed of 82.7% of the polyol tetraester and 17.3% noncondensed linear poly(RA). 78% of the hydroxyl groups of PE were esterified and the average acyl groups per chain were 5.4. The produced polyesters had melting points below 7.5°C. Their high viscosity and viscosity index (155) enable their applications as environmentally friendly lubricant materials.

In another example, poly(RA) was prepared by polymerization of RA **1a** or methyl RA **1b** via the catalysis of lipases from various organisms [15] (Figure 8.7). First, five different lipases from *Candida antarctica*, *Porcine pancreas*, *Candida rugosa*, *Pseudomonas* sp., *Pseudomonas cepacia* were screened for their activities on the polymerization of RA. Among these enzymes, immobilized lipase from *Pseudomonas cepacia* showed the best activity, achieving the



Figure 8.7 Polycondensation of ricinoleic acid and methyl recinoleate catalyzed by lipase.

production poly(RA) with an average MW of 5600 at 60°C after 7 days. Thus, further optimization of the process was based on this enzyme. Temperature effect was studied on the range of 60-100°C and highest MW of poly(RA) was obtained at 80°C. For the polymerization of methyl RA catalyzed by 50% (w/w) immobilized Pseudomonas cepacia lipase, placing 4 Å molecular sieves MS4A in the vapor phase significantly improved the MW of produced poly(RA) from 2100 to 73,200 since the absorbed methanol by MS4A shifted the equilibrium toward polymer formation. This result is intriguing since polycondensation activities of lipases are generally low via esterification of secondary hydroxyl groups; however, this process can be optimized by shifting reaction equilibrium. Moreover, increasing the lipase concentration to 150% led to the highest molecular weight of 98,000 and the produced poly(RA) was a viscous liquid at room temperature with a low glass transition temperature of -74.8°C. BOD biodegradability of this poly(RA) by activated sludge reached over 60% after 28 days. It showed no crystallinity and was readily cured by dicumyl peroxide at 170°C for 30 min (Figure 8.7). The cross-linked poly(RA) was insoluble in chloroform with a hardness of 50 A based on measurement of a durometer A.

8.2.3.2 Poly(Ambrettolide Epoxide) and Poly(Aleuritic Acid)

Ambrettolide, 17-oxacycloheptadec-6-en-1-one, used widely in perfume industry, is a compound found in the oil of musk mallow (*Abelmoschus moschatus*) seeds. Aleuritic acid ((9R,10S)-9,10,16-trihydroxyhexadecanoic acid) is a derivative of ambrettolide which contains three hydroxyl groups, two secondary and one primary (on ω -position). To prepare monomers, ambrettolide was epoxidized by meta-chloroperoxybenzoic acid to produce ambrettolide epoxide and isopropyl aleuriteate was synthesized by the esterification of aleuritic acid and isopropanol in the presence of sulfuric acid. These two transformations are both one-step with high conversion yields (>87%). These monomers were then used to synthesize poly(ambrettolide epoxide) and poly(aleuritic acid) by the catalysis of immobilized Candida antarctica lipase B (Novozym 435) [16] (Figure 8.8). Ring-opening polymerization of ambrettolide epoxide 1 was catalyzed by Novozym 435 (10 mg/mmol lactone) carried out in toluene at 90°C for 18 h; molecular weight of the resulting poly(ambrettolide epoxide) 2 was 9700 g/mol with a polydiversity index (PDI) of 1.9 determined by GPC analysis (Figure 8.8). According to the NMR spectra, the epoxide group was not affected during the



Figure 8.8 Lipase-catalyzed syntheses of poly (ambrettolide epoxide) and poly(aleuritic acid).

polymerization. Polymerization of isopropyl aleuriteate 3 to poly(aleuritic acid) 4 was catalyzed by Novozym 435 in toluene/2,4dimethyl-3-pentanol solvent mixture at 90°C and 55 kPa for 21 h (Figure 8.8). The released isopropanol from the reaction was removed because of reduced pressure to drive the equilibrium toward polymer. Upon precipitation in pentane, poly(aleuritic acid) was isolated in moderate yield (43%) with molecular weight of 5600 g/mol and PDI of 3.2. It is worth noting that this lipasecatalyed polymerization is highly selective for primary ω -hydroxyl group with no detectable esterification at the other two secondary hydroxyl moieties. However, when the polymerization of isopropyl aleuriteate was carried out chemically using Ti(IV)(n- $BuO)_4$ at high temperature (200°C), a yellow and hardly soluble product was obtained with ester bonds formed at both primary and secondary hydroxyl groups. Furthermore, copolymerization of isopropyl aleuriteate and ε -caprolactone (ε -CL) were applied to synthesize polyesters with tunable hydrophilicity and biocompatibility. Polymerizations were catalyzed by Novozym 435 in toluene at 90°C under different molar ratios of ε-CL resulting in poly(E-CL-co-isopropyl aleuriteate) with molecular weights ranging in 10,400-27,200 g/mol with good yields (71-78%). Selective esterification on the primary hydroxyl groups was again observed and the composition of these copolymers agreed consistently with the feeding ratio. In a postmodification reaction of the copolymers, the secondary hydroxyl groups in the polymer backbone were easily reacted with hexyl isocyanate, suggesting the potential use of poly(e-CL-co-isopropyl aleuriteate) as a next-generation biomaterial. Prepolymers based on ϵ -CL aleuritic acid may play an important role in thermoplastic elastomoers

because not only their hydrophilicity and biodegradation are tunable, but also bioreactive molecules can be easily attached.

8.2.3.3 Polyesters from the Monomer *cis*-9,10-Epoxy-18-Hydroxyoctadecanoic Acid

Polymers with reactive functional groups such as epoxides can play an important role in the development of biomedical or commodity materials. A naturally occurring fatty acid, cis-9,10-epoxy-18-hydroxyoctadecanoic acid, can serve as a "green" starting material for the synthesis of functionalized polymers. Polyesters found in higher plants are usually composed of complex monomers and differ between species, which makes them challenging for further applications. Fortunately, cis-9,10-epoxy-18-hydroxyoctadecanoic acid is an exception, which is a main constituent of dry outer bark (100 g/kg) in Betula verrucosa. Alkali hydrolysis can be easily applied on the outer bark to produce this epoxy acid with a high yield. For polymerization of epoxy-containing monomers, lipasecatalyzed polycondensation was attractive due to its mild reaction condition. Therefore, Novozym 435-catalyzed polymerization of cis-9,10-epoxy-18-hydroxyoctadecanoic acid 1 to its corresponding functional polyester 2 was investigated (Figure 8.9) [11].

Four solvents, acetonitrile, dioxane, toluene, and monomer melt (bulk), were tested. The highest molecular weight (20,000 g/mol) was obtained after 68 h using toluene; however, bulk polymerizations resulted in a close molecular weight (16,000 g/mol) in a much shorter time (5 h). The H-NMR spectra of the produced polyesters showed that the *cis*-epoxy group was intact during polymerization.

8.3 Enzyme-Assisted Plant Oil Transformations

8.3.1 Sugar Fatty Acid Esters

Sugar fatty acid esters such as polyoxyethylene sorbitan esters (Tween[®]) are extensively used as nonionic biosurfactants and emulsifiers. They have been used in food, cosmetic, and pharmaceutical industries since they are antimicrobial, nontoxic, and biodegradable. Moreover, the functional characteristics can be tuned by the constitutive sugar and fatty acid moieties. Although both carbohydrates and fatty acids are derived from renewable resource, almost all the current commercial sugar esters are produced via chemical syntheses. Due to the safety concerns with chemical synthesis,



Figure 8.9 Lipase-catalyzed polymerization of cis-9,10-epoxy-18-hydroxyoctadecanoic acid.

enzyme-catalyzed processes seem a better alternative for the production of food-grade sugar esters. Compared to the chemical processes, enzymatic processes have advantages such as low energy input, excellent regioselectivity, simple operational procedures, and simple product purification [17]. The lipasecatalyzed syntheses of sugar esters have been recently reviewed [17,18]. It is noteworthy that several monosaccharides such as D-glucose, D-galactose, and D-fructose can be converted regioselectively into 6-O-acyl esters using immobilized Candida Antarctica lipase B (Novozym 435) as the catalyst in aprotic solvents such as tetrafuran or dioxan. Several plant oil-derived fatty acids caprylic, capric, myristic, lauric, palmitic, stearic, and erucic acids have been successfully applied as acyl donors [19]. Chemical versus enzymatic catalysis for the synthesis sucrose fatty acids has been studied (Figure 8.10) [20]. Chemical process normally requires high temperature (>100°C) and reduced pressure in the presence of a basic or metallic catalyst. High-temperature reactions not only consume more energy but also result in different degrees and positions of esterification and various undesirable byproducts (route A in Figure 8.10). In the opposite, enzymatic synthesis provides a low-energy (usually <60°C), highly chemo-, regio-, and enantioselective, and environmentally friendly alternative. The product is typically a monoester and easy for downstream separation (route B, Figure 8.10).

8.3.2 Transesterification

Transesterification can be used to convert natural triglycerides found in plant oils into various fatty acid esters monomers with a



Figure 8.10 Comparison between chemical (a) and enzymatical (b) syntheses of sugar fatty acid esters.

wide range of functional characteristics. For example, as demonstrated in previous examples, many fatty acid methyl esters can be used as better monomers in polycondensation since methanol is a better leaving group than water. Transesterification reactions were catalyzed by alkaline in most commercial processes such as sodium hydroxide (NaOH) and potassium hydroxide (KOH). These chemicals are economically feasible and can achieve fast rate and high conversion yield. However, highly pure triglycerides are required since free fatty acids (FFA) in the feedstock are converted to soap and presence of water further increases soap formation. Multistep purification is necessary since the end products are a mixture of glycerol, tri-, di-, monoglycerides, alcohol, catalyst, and so on. For enzymatic transesterification, no soap formation due to FFA and FFA can be converted into according esters in one step. This suggests that low-cost feedstocks with high FFA and water content such as waste cooking oil or lard can be directly used for the enzymatic process. Reaction condition for biocatalyst transesterification is milder (reducing energy consumption) and downstream processing is not required. Overall, enzymatic transesterification is an attractive alternative to chemical transesterification in terms of the reduced energy cost, environmental impact, feedstock sustainability, and downstream processing [21]. The high enzyme cost is the main challenge for commercialization, which can be overcome by immobilization. Although the conversion yields of most enzymatic transesterifications (<90%) are lower than chemical processes (nearly 99%), some biocatalyst processes showed great yields and productivity. For example, it has been reported that rapeseed oil can be converted to its corresponding methyl esters by the catalysis of immobilized Candida antartica lipase with a conversion yield of 95% and productivity of 2.6 g methyl esters/g enzyme/h. Recently, synthesis of fatty acid methyl esters by enzymatic transesterification has been extensively reviewed [21,22]. Other than methanol, various alcohols have also been used as acyl acceptors in transesterification reactions. For example, trimethylolpropane esters were synthesized by lipase-catalyzed transesterification with rapeseed oil methyl esters [23]. Total conversion of trimethylolpropane esters could reach 98% using Candida rugosa lipase as the catalyst at a reduced pressure of 2.0 kPa, 42°C, and 15% added water. The maximal conversion to the tri-ester reached 64% in 24 h when the reaction temperature was slightly increased to 47°C. The produced trimethylopropane esters of rapeseed oil can be used as hydraulic fluids with superior properties such as good cold and oxidation stability, and excellent frication and wear resistance.

Moreover, due to the high regioselectivity of some lipases, they have been applied to tailor natural lipids by transesterification to improve nutritional values or material properties. The most well-known example is the conversion of palm mid fraction and shea stearin into cocoa butter equivalents (CBE). Cocoa butter is mainly composed of oleic, palmitic, stearic acids in the form of 1,3-disaturated-2-oleyl-glyceride. Unilever [24] and Fuji Oil [25] have developed the process using 1,3-specific lipase to convert palm oil and stearic acids into CBE. The 1,3-specific lipase catalyzes the transesterfication that replaces palmitic acid with stearic acid at the 1- and 3-position to produce CBE. The produced CBE is crystalline with a melting temperature around 32-35°C, which is completely compatible with natural cocoa butter. Similar strategies have been applied for the production of structured triglycerides



Figure 8.11 Convert tripalmitin to Betapol using 1,3-specific lipase.

(sTAG) such as Betapol. Betapol is a sTAG used for infany nutrition containing oleic acid at the 1- and 3-position and plamitic acid at the 2-position (1,3-oleoyl-2-palmitoylglycerol). A two-step lipase-catalyzed process was developed to convert tripalmitin 1 into Betapol 3 (Figure 8.11) [26]. First, Rhizopus delemar lipase immobilized on polypropylene (EP-100) was used for alcoholysis of tripalmitin to produce the corresponding 2-monopalmitin. The 2-monopalmitin 2 was then isolated with >95% purity and then further esterified with oleic acid with the same enzyme in *n*-hexane to produce Betapol 3 (Figure 8.11). The conversion yield reached 70% in 5 h and the produced Betapol contained 92% palmitic acid in the β position. More studies related to enzyme-catalyzed reactions have been discussed in [27].

8.3.3 Enzyme-Catalyzed Epoxidation

As shown in Sections 8.2.1.2 and 8.2.1.3, epoxide functional groups are versatile moi-

eties that not only can be used for crosslinking but also can alter functional characteristics of polymers. Currently, epoxide groups were mainly generated chemically using peroxicarboxylic acids as the oxidant. The highly acidic condition associated with in situ production of peroxy acid results in hazardous handling and formation of undesired byproducts. On the other hand, enzymecatalyzed epoxidation can be carried out in a much milder condition with high regioselectiviry. Lipase-catalyzed epoxidation of simple olefins was first reported in 1990 [28]. Five different lipases were screened for the epoxidation reaction and Candida Antarctica lipase showed the best activity. Lipase was first catalyzed by the production of peroxycarboxylic acid from the parent carboxylic acid using hydrogen peroxide as an oxidant. The produced peroxy acids were then used to expoxidiz the alkenes (Figure 8.12). This enzymatic transformation has then been applied to different substrates. Unsaturated fatty acids having a carboxyl group can be used for percarboxylic acid generation and carbon double bond(s) can be epoxidized. Therefore, if hydrogen peroxide is used to treat an unsaturated fatty acid in the presence of Novozym 435, the unsaturated acid





Figure 8.12 Lipase-catalyzed epoxidation of alkenes using hydrogen peroxide as oxidant. is epoxidized (hence, this is called a chemoenzymatic process) [29]. Taking oleic acid as an example, the mechanism of the reaction is that some unstaturated fatty acids 1 are first converted to peroxy acids 2 by hydrogen peroxide catalyzed by the lipase, and then the peroxy acids epoxidize the unsaturated moieties chemically without other catalyst to produce the expoxidized fatty acids 3 (Figure 8.13). It has been demonstrated that unsaturated fatty acids with varying chain length can be epoxidized with high yields (72–95%) and superior selectivity (>98%). Either terminal or internal unsaturated groups can be enzymatically epoxidized. For fatty acid containing hydroxyl group such as ricinoleic acid, the hydroxyl group was involved in the epoxidation reaction. The enzymatic epoxidation process has also been applied to meadowfoam (Limnanthes alba) fatty acids to produce a mixture of 5,6-epoxyeicosanoic, 13,14-epoxydocosanoic, 5,6-epoxydocosanoic, and 5,6-13,14-diepoxydocosanoic acids with 98% yield [30]. These highly pure epoxy products provide great starting materials for the subsequent production of 6-hydroxy-δ-lactone. Besides free fatty ac-



Figure 8.13 Mechanism of lipase-catalyzed epoxidation of oleic acid.

ids, unsaturated triglycerides can be epoxidized enzymatically. If triglycerides were treated with hydrogen peroxide directly in the presence of lipase, the undesired byproducts such as epoxidized mono or diglycerides were found in the product mixture. Adding a small amount of free fatty acids (5 mol%) at the beginning of the reaction was found to completely inhibit the production of monoand diglycerides [31]. The product mixture consists of only epoxidized triglycerides and epoxidized free acids since any free hydroxyl group produced was immediately reesterified by the excess free fatty acid. Since transesterification occurred in the reaction, free fatty acids with the same composition as the triglycerides were used to maintain fatty acid distribution of the triglycerides. It has been demonstrated that rapeseed, sunflower, soybean, linseed, and castor oils could be epoxidized enzymatically with conversion yields of 88-96% and a selectivity of >92% [31]. Besides complete epoxidation, controlled partial epoxidation by optimizing $H_2O_2/C = C$ -bonds ratio could be applied to accurately predict the oxirane oxygen contents and prepare synthetic substitutes that mimic natural epoxy oils such as vernonia oil [32]. Moreover, the immobilized lipase can be easily recovered and can maintain activity after multiple runs.

Recently, enzymatic epoxidation of *Sapindus mukoross*i seed oil (SMSO) was studied. SMSO composed of oleic, linoleic, linolenic, and eicosanoic acid was epoxidized using hydrogen peroxide as oxidant and stearic acid as active oxygen carrier catalyzed by Novozym 435. The effect caused by varying amount of stearic acid added was first studied. Response surface methodology (RSM) was then applied to investigate and optimize reaction parameters on the epoxy oxygen group content (EOC). Reaction temperature and enzyme load were found to be the most important parameters. Under the optimal conditions of 50°C, 7 hours, 2% enzyme load, and 4:1 molar ratio of $H_0O_0/C = C$ -bonds, the EOC could reach 4.6%. Chemoenzymatic epoxidation of soybean oil has also been optimized [33]. Oleic acid was used for peroxy acid production. The optimization was performed step by step with one parameter at a time by the order of catalyst concentration, solvent concentration, $H_2O_2/C = C$ -bonds ratio, initial oleic acid concentration, and reaction temperature. The conversion yield after optimization reached over 90% after 24 hours under the condition at 50°C, 4.0 wt% Novozym 435, 2:1 molar ratio of $H_2O_2/C = C$ -bonds, in 110 wt% toluene and 8.0 wt% of oleic acid. More works about this chemoenzymatic epoxidation catalyzed by lipase can be found in these reviews [34–36]. Besides lipase, peroxygenase from oat (Avena sativa) seeds was found to be able to catalyze the conversion of double bonds to epoxide groups in the presence of an oxidant [37]. Among the oxidants tested, tert-butyl hydroperoxide showed the best activity. Oleic acid with a cis double bond was preferred compared to its trans analog, elaidic acid. This selectivity toward double bond cis configuration was not observed with lipase-catalyzed epoxidation.

8.3.4 Fatty Acids Hydroxylation

Hydroxy fatty acids are important chemical derivatives from plant oil due to the extra hydroxyl functional group(s). For example, the secondary hydroxyl group in ricinoleic acid (RA) has been transformed through various reactions such as esterification (as discussed in Section 8.2.3.1) dehydration, hydrogenation, sulfonation to produce versatile platform chemicals. Hence, RA can be used for the synthesis of a vast amount of polymers such as polyamides, polyurethanes, and polyesters that are widely used in lubricant, cosmetic, and candle industries [38]. Moreover, some hydroxyl fatty acids have antibiotic, antitumor, or antiinflammation activities, suggesting their applications in pharmaceutical industry. For example, 9-hydroxy-10,12-octadecadienoic acid found in plants is a ligand for peroxisome proliferator-activated receptor- γ [39] and thus provides antitumor activity. Another polyhydroxyl fatty acid, 7,10-dihydroxy-8E-octadecenoic acid, produced by Pseudomonas aeruginosa can reduce surface tension and inhibit growth of the pathogenic microorganism Candida albicans [40]. Since chemical hydroxylation cannot provide the required regio- and steroselectivity for most applications, enzymecatalyzed hydroxylation of fatty acid is preferred. Several enzyme families were found to catalyze fatty acid hydroxylation reactions by different mechanisms. Therefore, enzymes catalysts provide synthetic routes to regiospecific but diverse hydroxyl fatty acids. Recently, these fatty acid hydroxylation enzymes were extensively reviewed [41]; therefore, we will focus on two important enzyme families, P450s and hydratases, to demonstrate the versatilities of different enzyme families.

Cytochrome P450 monooxygenases (one example is mentioned in Section 8.2.2) are one of the largest and oldest enzyme superfamilies and are reported in all kinds of organisms [42]. They are divided into families starting with *CYP* followed by an Arabic number. P450s incorporate an oxygen atom from molecular oxygen into an organic substrate using cofactors NADH or NADPH as reducing equivalents. Fatty acid-hydroxylating P450s can be categorized into terminal (α -hydroxylases and ω -hydroxylases) and subterminal fatty acid hydroxlases. For example, some enzymes belonging to CYP52 family from *Candida* species [43] show specifically terminal hydroxylation activity toward C₁₂ to C_{18} fatty acids. The ω -terminal hydroxylation followed by fatty alcohol oxidases and dehydrogenases can convert fatty acid into its corresponding α, ω -dicarboxylic acid. For example, palmitic acid 1 can be converted into 16-hydroxy palmitic acid 2 by CYP52, as shown in Figure 8.14a. Another P450 family CYP102 isolated from Bacillus species is self-sufficient and has higher turnover rates since they are fusion proteins with FAD- and FMN-containing reductase. CYP102 family also belongs to ω -hydroxylases and *CYP*102A1 is found to mainly (80%) catalyze the hydroxylation of arachidonic acid 3 at the ω -2 to produce enantiomerically pure (R)-18-hydroxyeicosatetraenoic acid 4 (Figure 8.14b) [44].

Hydratases add the hydroxyl group at the unsaturated functional group. These enzymes catalyze the regio-, steroselective irreversible reaction by adding water to the carbon-carbon cis-double bond of unsaturated fatty acids. Hydratase activity was first observed in 1962 [45]: an isolated Pseudomonas species hydrated oleic acid at the unsaturated 9,10-position to produce 10-hydroxystearic acid. The overall yield was only 14%. Hydration reactions of fatty acid on the unsaturated group have then been observed in many bacteria species such as Nocardia, Rhodococcus, Enterococcus, Pediococcus, Corynebacterium [46–49]. Mycobacterium fortuitum was reported to produce 10-hydroxystearic acid from oleic acid with an 80% yield [50]. Stenotrophomonas nitritireducens showed the highest



Figure 8.14 Comparison between different fatty acid-hydroxylation enzymes.

activity using linoleic acid as the substrate, but the hydroxylation is regioselective at the 9-, 10-position. The strain could convert linoleic acid **5** into 10-hydroxy-12(Z)-octadecenoic acid **6** (Figure 8.14c) with a yield of 67.5% [51].

8.4 Conclusion

This chapter provides an overview of enzyme-assisted processes applied in plant oil-based polymer industry. Enzymes are versatile and promising alternatives to chemical catalysts due to some characteristics. For example, they are able to catalyze reactions at mild conditions with excellent regio-, chemo-, and steroselectivity [1], and thus reduce energy consumption and the use of protection–deprotection reagents. This is especially important for polymerizing multifunctional or temperature-sensitive monomers and for structural control. Besides polymer synthesis catalyzed by isolated enzymes, whole-cell biocatalysts can be a useful tool for transforming plant oil into a variety of functional monomers by epoxidation, oxidation, and transesterification. Furthermore, biotechnology techniques can be applied to optimize enzyme efficiency or to engineer production of nonnatural chemicals derived from plant oil.

References

- S. Kobayashi, Recent developments in lipase-catalyzed synthesis of polyesters, Macromol. Rapid Commun. 30 (2009) 237–266.
- [2] H. Uyama, S. Kobayashi, Enzymatic synthesis of polyesters via polycondensation, in: S. Kobayashi, H. Ritter, D. Kaplan (Eds.), Enzyme-Catalyzed Synthesis of Polymers, Springer-Verlag Berlin, Berlin, 2006.
- [3] R.D. Schmid, R. Verger, Lipases: interfacial enzymes with attractive applications, Angew. Chem. Int. Ed. 37 (1998) 1609–1633.
- [4] T. Tsujimoto, H. Uyama, S. Kobayashi, Enzymatic synthesis of cross-linkable polyesters from renewable resources, Biomacromolecules 2 (2001) 29–31.
- [5] T. Tsujimoto, H. Uyama, S. Kobayashi, Enzymatic synthesis and curing of biodegradable crosslinkable polyesters, Macromol. Biosci. 2 (2002) 329–335.
- [6] H. Uyama, K. Inada, S. Kobayashi, Regioselectivity control in lipase-catalyzed polymerization of divinyl sebacate and triols, Macromol. Biosci. 1 (2001) 40–44.

- [7] Y.Y. Linko, J. Seppala, Producing high molecular weight biodegradable polyesters, Chemtech 26 (1996) 25–31.
- [8] P.R. Guevin, State-of-the-art instruments to measure coating hardness, J. Coat. Technol. 67 (1995) 61–65.
- [9] H. Uyama, M. Kuwabara, T. Tsujimoto, S. Kobayashi, Enzymatic synthesis and curing of biodegradable epoxidecontaining polyesters from renewable resources, Biomacromolecules 4 (2003) 211–215.
- [10] F. Björkling, H. Frykman, S.E. Godtfredsen, O. Kirk, Lipase catalyzed synthesis of peroxycarboxylic acids and lipase mediated oxidations, Tetrahedron 48 (1992) 4587–4592.
- [11] S. Warwel, C. Demes, G. Steinke, Polyesters by lipase-catalyzed polycondensation of unsaturated and epoxidized long-chain alpha,omega-dicarboxylic acid methyl esters with diols, J. Polym. Sci. Part a-Polym. Chem. 39 (2001) 1601–1609.
- [12] P. Schwab, M.B. France, J.W. Ziller, R.H. Grubbs, A series of well-defined metathesis catalysts: synthesis of [RuCl₂(=CHR')(PR₃)₂] and its reactions, Angew. Chem. Int. Ed. Eng. 34 (1995) 2039–2041.
- [13] Y.X. Yang, W.H. Lu, X.Y. Zhang, W.C. Xie, M.M. Cai, R.A. Gross, Two-step biocatalytic route to biobased functional polyesters from omega-carboxy fatty acids and diols, Biomacromolecules 11 (2010) 259–268.
- [14] A.R. Kelly, D.G. Hayes, Lipasecatalyzed synthesis of polyhydric alcohol-poly(ricinoleic acid) ester star polymers, J. Appl. Polym. Sci. 101 (2006) 1646–1656.
- [15] H. Ebata, K. Toshima, S. Matsumura, Lipase-catalyzed synthesis and curing

of high-molecular-weight polyricinoleate, Macromol. Biosci. 7 (2007) 798–803.

- [16] M.A.J. Veld, A.R.A. Palmans, E.W. Meijer, Selective polymerization of functional monomers with novozym 435, J. Polym. Sci. Part a-Polym. Chem. 45 (2007) 5968–5978.
- [17] A.M. Gumel, M.S.M. Annuar, T. Heidelberg, Y. Chisti, Lipase mediated synthesis of sugar fatty acid esters, Process Biochem. 46 (2011) 2079–2090.
- [18] T. Kobayashi, Lipase-catalyzed syntheses of sugar esters in non-aqueous media, Biotechnol. Lett. 33 (2011) 1911–1919.
- [19] M. Schneider, B. Haase, G. Machmueller, 1996. Preparation of mono:acylated mono-, di- or oligosaccharide: by reacting the primary hydroxyl group with acyl donors in the presence of lipase. DE19626943-A1.
- [20] M. Ferrer, M.A. Cruces, F.J. Plou, E. Pastor, G. Fuentes, M. Bernabé, J.L. Parra, A. Ballesteros, Chemical versus enzymatic catalysis for the regioselective synthesis of sucrose esters of fatty acids, in: F.V.M.S.M. Avelino Corma, G.F. José Luis (Eds.), Studies in Surface Science and Catalysis, Elsevier, Amsterdam, 2000.
- [21] K.R. Jegannathan, S. Abang, D. Poncelet, E.S. Chan, P. Ravindra, Production of biodiesel using immobilized lipase: a critical review, Crit. Rev. Biotechnol. 28 (2008) 253–264.
- [22] A.E. Ghaly, D. Dave, M.S. Brooks, S. Budge, Production of biodiesel by enzymatic transesterification: review, Am. J. Biochem. Biotechnol. 6 (2010) 54–76.
- [23] E. Uosukainen, Y.Y. Linko, M. Lamsa, T. Tervakangas, P. Linko, Transesteri-

fication of trimethylolpropane and rapeseed oil methyl ester to environmentally acceptable lubricants, J. Am. Oil Chem. Soc. 75 (1998) 1557–1563.

- [24] M.H. Coleman, A.R. Macrae, Rearrangement of fatty acid esters in fat reaction reactants. Germany patent application DE2705608, 1977.
- [25] T. Matsuo, N. Sawamura, Y. Hashimoto, W. Hashida, 1981. Method for enzymatic interesterification of lipid and enzyme used therein. EP0035883.
- [26] U. Schmid, U.T. Bornscheuer, M.M. Soumanou, G.P. Mcneill, R.D. Schmid, Highly selective synthesis of 1,3-oleoyl-2-palmitoylglycerol by lipase catalysis, Biotechnol. Bioeng. 64 (1999) 678–684.
- [27] J.O. Metzger, U. Bornscheuer, Lipids as renewable resources: current state of chemical and biotechnological conversion and diversification, Appl. Microbiol. Biotechnol. 71 (2006) 13–22.
- [28] F. Bjorkling, S.E. Godtfredsen, O. Kirk, Lipase-mediated formation of peroxycarboxylic acids used in catalytic epoxidation of alkenes, J. Chem. Soc. Chem. Commun. (1990) 1301–1303.
- [29] S. Warwel, M.R.G. Klaas, Chemoenzymatic epoxidation of unsaturated carboxylic acids, J. Mol. Catal. B-Enzymatic 1 (1995) 29–35.
- [30] H.B. Frykman, T.A. Isbell, Synthesis of 6-hydroxy delta-lactones and 5,6-dihydroxy eicosanoic/docosanoic acids from meadowfoam fatty acids via a lipase-mediated self-epoxidation, J. Am. Oil Chem. Soc. 74 (1997) 719–722.
- [31] M.R.G. Klaas, S. Warwel, Chemoenzymatic epoxidation of unsaturated fatty acid esters and plant oils, J. Am. Oil Chem. Soc. 73 (1996) 1453–1457.

- [32] M.R. Klaas, S. Warwel, Complete and partial epoxidation of plant oils by lipase-catalyzed perhydrolysis, Ind. Crops Prod. 9 (1999) 125–132.
- [33] T. Vlcek, Z.S. Petrovic, Optimization of the chemoenzymatic epoxidation of soybean oil, J. Am. Oil Chem. Soc. 73 (2006) 247–252.
- [34] A. Kockritz, A. Martin, Oxidation of unsaturated fatty acid derivatives and vegetable oils, Eur. J. Lipid Sci. Technol. 110 (2008) 812–824.
- [35] D.S. Rolf, C.M. Steffen, Biocatalysts for the epoxidation and hydroxylation of fatty acids and fatty alcohols, Handbook of Industrial Biocatalysis, CRC Press, Boca Raton, FL, (2005).
- [36] S. Warwel, M.R.G. Klaas, New Peroxycarboxylic Acids by Lipase Catalysis: Preparation and Oxidation Properties. Peroxide Chemistry, Wiley-VCH Verlag GmbH & Co. KGaA, Weinheim, Germany, 2005.
- [37] G.J. Piazza, T.A. Foglia, A. Nunez, Epoxidation of fatty acids with membrane-supported peroxygenase, Biotechnol. Lett. 22 (2000) 217–221.
- [38] H. Mutlu, M.A.R. Meier, Castor oil as a renewable resource for the chemical industry, Eur. J. Lipid Sci. Technol. 112 (2010) 10–30.
- [39] T. Itoh, L. Fairall, K. Amin, Y. Inaba, A. Szanto, B.L. Balint, L. Nagy, K. Yamamoto, J.W.R. Schwabe, Structural basis for the activation of PPAR gamma by oxidized fatty acids, Nat. Struct. Mol. Biol. 15 (2008) 924–931.
- [40] T.M. Kuo, A.C. Lanser, Factors influencing the production of a novel compound, 7, 10-dihydroxy-8(E)-octadecenoic acid, by Pseudomonas aeruginosa PR3 (NRRL b-18602) in batch cultures, Curr. Microbiol. 47 (2003) 186–191.

- [41] K.-R. Kim, D.-K. Oh, Production of hydroxy fatty acids by microbial fatty acid-hydroxylation enzymes, Biotechnol. Adv. 31 (2013) 1473–1485.
- [42] D.W. Nebert, F.J. Gonzalez, P450 genes-sturcture, evolution, and regulation, Annu. Rev. Biochem. 56 (1987) 945–993.
- [43] W. Seghezzi, C. Meili, R. Ruffiner, R. Kuenzi, D. Sanglard, A. Fiechter, Identification and characterization of additional members of the cytochrome-P450 multigene family *CYP52* of *Candida tropicalis*, DNA Cell Biol. 11 (1992) 767–780.
- [44] J.H. Capdevila, S.Z. Wei, C. Helvig, J.R. Falck, Y. Belosludtsev, G. Truan, S.E. Grahamlorence, J.A. Peterson, The highly stereoselective oxidation of polyunsaturated fatty acids by cytochrome P450BM-3, J. Biol. Chem. 271 (1996) 22663–22671.
- [45] L.L. Wallen, R.G. Benedict, R.W. Jackson, The microbiological production of 10-Hydroxystearic acid from oleic acid, Arch. Biochem. Biophys. 99 (1962) 249–253.
- [46] W. Blank, H. Takayanagi, T. Kido, F. Meussdoerffer, N. Esaki, K. Soda, Transformation of oleic-acid and its esters by sarcina-lutea, Agric. Biol. Chem. 55 (1991) 2651–2652.
- [47] J.A. Hudson, B. Morvan, K.N. Joblin, Hydration of linoleic acid by bacteria isolated from ruminants, Fems. Microbiol. Lett. 169 (1998) 277–282.
- [48] S. Koritala, L. Hosie, C.T. Hou, C.W. Hesseltine, M.O. Bagby, Microbial conversion of oleic acid to 10-hydroxystearic acid, Appl. Microbiol. Biotechnol. 32 (1989) 299–304.
- [49] C.W. Seo, Y. Yamada, N. Takada, H. Okada, Hydration of squalene and

oleic-acid by corynebacterium SP S-401, Agric. Biol. Chem. 45 (1981) 2025–2030.

- [50] S.H. Elsharkawy, W. Yang, L. Dostal, J.P.N. Rosazza, Microbial oxidation of oleic-acid, Appl. Environ. Microbiol. 58 (1992) 2116–2122.
- [51] I.S. Yu, S.J. Yeom, H.J. Kim, J.K. Lee, Y.H. Kim, D.K. Oh, Substrate specificity of Stenotrophomonas nitritireducens in the hydroxylation of unsaturated fatty acid, Appl. Microbiol. Biotechnol. 78 (2008) 157–163.

Samy A. Madbouly^{1,2}, Chaoqun Zhang³, and Jing Zhang⁴

¹Department of Chemistry, Faculty of Science, Cairo University, Orman-Giza, Egypt ²Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA ³Department of Mechanical Engineering, Texas A&M University, College Station, TX, USA ⁴Department of Corrosion Prevention and Polymer Materials, College of Materials Science and Engineering, Jiangsu University of Science and Technology, Zhenjiang, China

		Ο U T L I N E	
9.1	Introduction		149
9.2	Plant Oil-Based/Natural Filler Composites		152
9.3	Plant	Oil-Based/Carbon Nanotubes Composites	154
9.4	Plant	Oil-Based PU/Graphene Oxide Composites	155
	9.4.1	Surface Modification of Graphene Oxide	156
	9.4.2	Synthesis and Characterization of Bio-Based PU/GO Nanocomposites	158
9.5	Plant	Oil-Based/Clay and Silica Nanocomposites	159
References			

9.1 Introduction

Plant oil-based polymer nanocomposites can be defined as a mixture of a sustainable polymer matrix and nanosized particles, such as fibers, platelets, or tubes. The plant oilbased polymer matrix can be an amorphous thermoplastic material or a crosslinked threedimensional polymer network. The plant oil-based polymer matrix holds or binds the fillers together and protects them from damage by distributing any stress through the whole specimen. Recently, plant oil-based polymer composites have received considerable attention due to their potential to significantly improve and enhance the properties of neat polymer matrix. Incorporation of small amounts of filler leads to an improvement in material properties, such as modulus, strength, heat resistance, flame retardancy, and lowered gas permeability [1-14]. Furthermore, addition of nanofillers to the plant oil-based polymer matrix could also yield novel functions, such as electrical [15–18], magnetic [19–25], and optical functions [26–29], as well as biofunctionality [30–33]. In principle, the improvement of material properties and creation of novel functions has been linked to the interfacial interaction between the polymer matrix and fillers as well as the formation of a network of interconnected filler particles. The interconnected particles can conduct heat and electrical current [15–25].



Figure 9.1 Surface/volume (S/V) ratios for varying filler geometries, *r* is the radius, *l* is the length, and *t* is the thickness of filler. Taken with permission from [37].

Development and tailoring of polymer composites offer the possibility to promote their use in automotive, aerospace, building, electrical, optoelectronic, and biomedical applications [34–36]. The novel functions and properties enhancement of polymer composites can also be controlled by micro or nanostructural parameters such as dimension, shape, distribution, volume fraction, alignment, and packing arrangement of fillers. Anisotropic properties could be obtained from fiber filled composites, except for the very short, randomly distributed fibers, whereas physical properties of polymers with randomly distributed particles are isotropic. Furthermore, composites with nanometersized fillers had different properties compared to those filled with macrosized fillers. Some of the properties of nanocomposites, such as increased tensile strength, may be achieved by using higher macrosized filler concentration at the expense of increased weight and decreased gloss. Other properties of nanocomposites such as optical clarity or improved barrier properties could not be achieved by high concentration of macrosized fillers. The filler concentration required for substantial improvement in the overall material properties is called critical filler volume fraction. The fillers could be classified either by their geometry or by their size. Three different categories of filler materials could be found, namely particles (e.g., silica, metal, POSS, and other organic and inorganic particles), layered materials (e.g., graphite and layered silicate), and fibrous materials (e.g., nanofibers and single- (SWCNTs) and multiwalled nanotubes (MWCNTs)).

The surface area/volume ratio of the fillers is a very important morphological factor normally used to understand and control the structure-property relationship of nanocomposites [37]. As shown in Figure 9.1, the change in particle diameter, layer thickness, or fibrous material diameter from micrometer to nanometer changes the surface area/ volume ratio by three orders of magnitude. According to this scale, there is a distinct size dependence of the material properties. In addition, the properties of the composite are normally controlled by the properties of the interface or interphase when the interfacial area drastically increased. Major challenges in design and fundamental understanding of polymer composites are related to the complexity of the composite structure, dispersibility of fillers, and the relationship between dispersion and optimal properties.



Figure 9.2 Schematic diagram of the hard and soft domains of PU structure. The TEM shows microphase separation morphology for the hard domains (*dark particles*) and soft domains (*bright matrix*). Adapted with permission from [41].

Uniform dispersion of nanoparticles and nanotubes against their agglomeration due to van der Waals bonding is the first step in the processing of nanocomposites [38-40]. Exfoliation of clays and graphitic layers are essential. Several strategies have been studied to achieve well-dispersed fillers in polymer matrix, including melt processing, solvent casting method (often with surface functionalization and/or sonication pretreatment), and in situ polymerization. Melt processing by itself often led to limited filler dispersion in the polymer matrix. Blending polymer and fillers in solvent or in situ polymerization resulted in a better dispersion. It is well established that the modification of the filler surface by grafting of macromolecules onto its surface is preferable. In this case, the filler is highly compatible to the polymer matrix. This is often desirable since it provides the best possible adhesion and allows for optimal transfer of stress from the matrix to the fiber.

Polyurethanes (PUs), typically synthesized through addition polymerization of polyisocyanates and polyols are among the most versatile polymeric materials and have been widely used as foams, coatings, and sealants. PU is normally comprised of alternating soft polyol and hard polyurethane–urea segments. These two segments are microphase separated into hard and soft domains [41]. The microphase separation is responsible for the excellent elastomeric properties of PU. Figure 9.2 depicts schematically the structure of PU elastomers (linear structure) as multiblock copolymers with hard (urethane) and soft (polyol) segments. The TEM shows how these two segments can segregate in different domains having a nanoscale morphology. The very small dark particles in the micrograph are the hard domains, and the bright matrix is the soft domain of the PU [41]. Currently, most of the polyisocyanates and polyols for PU production are derived from petroleum. Increasing concerns about the unsustainability of chemicals from petroleum resources and environmental problems caused by fossil fuels, have led to considerable efforts to develop materials based on renewable resources, such as natural oils, starch, cellulose, and lignin. For example, vegetable oil based PUs were successfully developed and their structureproperty relationships have been extensively investigated.

In this chapter, the substantial enhancement in mechanical, thermal, and morphological properties of plant oil-based polymers (mainly PU) will be highlighted exemplarily for different types of surface modified and functionalized nanofillers including, natural fillers (e.g., cellulose and lignin), graphene, carbon nanotubes (CNTs), and silica.

9.2 Plant Oil-Based/Natural Filler Composites

Natural fillers, such as lignin and cellulosic fibers have good mechanical properties, low density, and they are available with huge amount with low cost from renewable resources. In addition, the surface of natural fillers has hydroxyl groups that allow chemical modification to enhance the interaction and improve the interface with the polymer matrix.

Lignin, one of the world's most abundant natural polymers, is found in all vascular plants, mostly between the cells and in the cell walls. Lignin is relatively hydrophobic and aromatic in nature, and is composed of up to three different phenyl propane monomers, namely; p-coumaryl alcohol, coniferyl alcohol, and sinapyl alcohol (see Figure 9.3a). These three lignols monomers are incorporated into a complex, ill-defined structure as shown in Figure 9.3b. Although pure lignin can be easily obtained with huge quantities as by-product of wood pulping or ethanol production from biomass, the industrial applications of lignin is very limited due to its complex chemical structure and complete insolubility in common organic solvents. Chemical incorporation of lignin to plant oil-based polymers is not an easy process because of the complex structure of lignin. Generally, the utilization of lignin during the polymerization process of plant oil monomers can be established through chemical modification and functionalization of lignin. The significant number of hydroxyl groups on the lignin backbone provides an excellent opportunity for chemical modification to tailor lignin properties to a certain application.

Soybean-based PU biofoam was investigated with different concentrations of lignin as a reinforcement filler by Luo et al. [43]. The PU/lignin biofoams were synthesized via self-rising method using water as a blowing agent. The mechanical properties of the PU biofoam composites increased significantly with increasing lignin concentration up to 10 wt%. In addition, the glass transition temperature, foam density, thermal stability, and storage modulus increased systematically with increasing lignin content. The authors provided a simple synthesis procedure to prepare a novel biofoams and explored the potential of using lignin as a biofiller in PU applications.

A series of natural fibers such as flax. cellulose, pulp, and hemp with different contents up to 50 wt% were used as reinforcement fillers for acrylated epoxidized soybean oil via resin vacuum infusion process or vacuum-assisted resin transfer molding [44]. Recycled paper, an example for a cheap cellulose fiber was found to have a significant improvement on the flow behavior and mechanical properties of the neat biobased polymer matrix. The bio-based composites were cured at room temperature and flexural modulus of the obtained composites increased significantly to a range between 1.5 GPa and 6 GPa depending on the nature of the used fiber. In addition, the authors claimed that the low cost acrylated epoxidized soybean-based natural fiber composites have good mechanical properties suitable for applications in housing construction materials, furniture, and automotive parts.

Bio-based maleate castor oil and lignin composites have been synthesized through free radical polymerization. The composites chemical structure and morphology were



Figure 9.3 The three phenyl propane monomers in lignin (a) and the proposed chemical structure of lignin (b) [42].

characterized using Fourier transform infrared spectroscope (FTIR) and scanning electron microscope (SEM), respectively [45]. In addition, the mechanical properties of the composites showed a considerable improvement in the tensile properties by adding lignin.

Recently, polyol based on lignin and oleic acid have been synthesized with no organic solvent or catalyst [46]. The chemical structure of this new lignin–fatty acid-based polyol was characterized using ¹H NMR and FTIR analyses [46]. The lignin–fatty acid polyol was then used to synthesize biobased PU via two steps procedure. The thermomechanical properties of this lignin–fatty acid-based PU were investigated using DSC, TGA, and DMA. This study depicted an important output for the valorization of lignin and they may be an optimal alternative to conventional PUs.

The effect of lignin and cellulose fibers obtained from sugarcane straw on the structure and thermal behavior of castor oil-based PU has been reported by Mileo et al. [47]. The milled cellulose and lignin were obtained from acid hydrolysis of sugarcane straw. The bio-based composites were prepared with different concentrations of lignin and cellulose fillers. A significant increase in the stiffness was observed by adding 40 and 30 wt% of lignin and cellulose fibers, respectively. At higher concentration of the bio-based fillers, the stiffness of the composites decreased greatly. In addition, the thermal stability of the composites improved by adding the bio-based fillers.

9.3 Plant Oil-Based/Carbon Nanotubes Composites

Polymer composites with CNTs as filler are highly conductive materials with significantly reduced in the electric resistance. The carbon compounds could conduct electricity in the plane of each covalently bonded sheet due to the delocalization of outer electrons to form π -cloud. This resulted in a lower bulk electrical conductivity for carbon than for most metals. Generally at atmospheric pressure, carbon took the form of graphite in which each atom can be bonded trigonally to three others in a plane composed of fused hexagonal rings, similar to aromatic hydrocarbons. The resulting network is twodimensional, and the resulting flat sheets were stacked and loosely bonded through weak van der Waals forces (see Figure 9.4a). This gave graphite its softness and its cleaving properties (the sheets slip easily off one another). CNTs could be considered as the result of folding graphite layers into carbon cylinders and might be composed of a single shell-SWCNTs (Figure 9.4b) or of several shells-MWCNTs (Figure 9.4c) [48-50].

CNTs with their impressive mechanical properties are ideal material. Reinforcement bio-based nanocomposites from acrylated epoxidized soy oil (AESO) and CNTs with different concentrations has been reported by McAninch et al. [51]. Styrene and methyl methacrylate were used as dispersing agents for CNTs in AESO polymer matrix. It has been found that the MMA significantly improved the dispensability of CNTs in polymer matrix than styrene. The processing of composites was carried out using mechanical shear mixing method with considerable improvement in the mechanical properties. The homogenous distribution of CNTs was investigated by SEM and optical microscopy. The morphology of the composites revealed some aggregation of CNTs in the polymer matrix.

Liang et al. [52] developed bio-based PU foams from soybean oil and MWCNTs. The composites were prepared with different MW-CNTs concentrations of 0.5–1.0 wt%. The



Figure 9.4 Schematic diagrams for graphene (a), single-wall carbon nanotube (b), and multiwall carbon nanotube (c). (a) Reprinted with permission from [59], (b) reprinted with permission from [66], and (c) reprinted with permission from [67].

mechanical properties of the soy-based PU nanocomposites increased significantly with adding MWCNTs. The compressive, flexural, and tensile properties of the composites foams improved to about 30% by adding MWCNTs. In addition the maximum improvement in the compressive and flexural properties of the composites were obtained at 0.5 wt% MW-CNTs, while the tensile stress reached a maximum value at 1 wt% MWCNTs.

9.4 Plant Oil-Based PU/Graphene Oxide Composites

Graphene oxide (GO) can be made from chemical exfoliation by reactions that have been known for 150 years. Recently, interest in this old material has resurged, because GO is a promising solution-processable precursor for the bulk production of graphene. There are abundant oxygenated functional groups on GO's basal plane and edges, such as carboxyl, hydroxyl, and epoxy. Moreover, GO is a two-dimensional sheet with feature sizes at two abruptly different length scales. The apparent thickness of GO is approximately 1 nm, while the lateral dimensions can range from several nanometers to hundreds of micrometers. The di-isocynate can react easily with the oxygenated functional groups attached to GO to produce exfoliated PU nanocomposites with enhanced mechanical, physical, and thermal properties. Nguyen et al. prepared a cast nanocomposite film from a mixture of thermoplastic polyurethane (TPU) solution and functionalized graphene sheet (FGS) suspended in methyl ethyl ketone. The FGS efficiently reinforced the TPU matrix particularly in the temperature region above the soft segment melt [53]. Cai et al. reported the substantial improvement in the stiffness, toughness, and thermal stability of linear PU resulting from the incorporation of GO [54]. Kim et al. reported exfoliated GO reinforced TPU nanocomposites with improved gas barrier and electrical conductivity [55]. Lee et al. prepared waterborne polyurethane (WPU) nanocomposites with FGS by in situ method; the conductivity and modulus of nanocomposites improved, while the thermal stability and tensile strength of nanocomposites deteriorated [56]. Wang et al. prepared GO sheets reinforced linear PU composites by in situ polymerization. The authors found that, the nanocomposites displayed high electrical conductivity and good thermal stability [57]. Chen and Lu reported simultaneous improvement in strength and toughness while maintaining the good ductility of PU elastomers by adding FGS [58]. Little work has been reported for bio-based PU nanocomposites. In this section, the effects of surface modified GO on the mechanical, thermal, and morphological properties of bio-based PU will be considered. The thermosetting PU will be synthesized from epoxidized soybean oil and castor oil fatty acid.

9.4.1 Surface Modification of Graphene Oxide

Generally, graphene oxide can be synthesized via a modified pressurized oxidation method. Briefly, graphite, KMnO₄, H₂SO₄ (98%), Teflon reactor, and stainless steel autoclave were completely cooled, respectively, in a refrigerator at 0–4°C before use. The cooled graphite (2 g) and KMnO₄ (10 g) were put into the reactor, and then, H_2SO_4 (100 mL) was added to the reagent mixture. The reactor was then moved into an ice water bath and kept mechanical stir at 800 rpm for 1 h. After that, the reactor put into the stainless steel autoclave and tightly covered at 80°C for 2 h. The obtained mud was diluted with a large amount of deionized water. With mechanical stirring, H_2O_2 (30%) was dripped into the suspension until the slurry tuned golden yellow. The suspension was washed with hot HCl and deionized water until the pH reached 7.0 to obtain the GO.

Figure 9.5 shows the Raman spectrum, which reflects structural changes occurring in graphite and GO. Highly ordered graphite has a couple of Raman-active bands visible in the spectra, the in-phase vibration of the graphic lattice (G band at 1565 cm⁻¹), and the weak disorder band caused by the graphite edges (D band at 1343 cm⁻¹). Both the G and the D band undergo significant changes upon oxidation and exfoliation of graphite as GO contains a certain fraction of sp³ carbons. In the case of GO, the D band shifts to 1355 cm⁻¹, and becomes broader with higher relative intensity compared with that of the G band, which indicates the



Figure 9.5 (a) Pressurized oxidation method and (b) modified pressurized oxidation method [59].



Figure 9.6 Raman spectra of graphite and GO obtained from the modified pressurized oxidation method [59].

higher disorder in GO. The broader G band also indicated the deconstruction of highly ordered graphite. Moreover, the blue shift of G band to 1595 cm⁻¹ is mainly due to the isolated double bonds of GO resonates at higher frequencies than that in graphite [59,60] (Figure 9.6).

AFM was performed to observe the morphology of GO. It can be seen from the AFM profile that GO is fairly small (around 1 μ m) and ultrathin. The thickness of GO is

0.8–1.2 nm (see Figure 9.7), which is typical for a one-atom-thick GO nanolayer. Such a small thickness indicates that the specific surface area of GO must be very large, which is important for polymer/layered compound nanocomposites [61]. The layers are a bit thicker than individual graphene due to the presence of oxygenated functional groups such as hydroxyl, carboxyl, and epoxy, which disrupt the original conjugation and introduce lattice defects to form folds and distortions



Figure 9.7 AFM profile for GO obtained from the modified pressurized oxidation method [59].

on the GO layers. Such disruption and lattice defects reduce the mechanical strength of the GO layers.

9.4.2 Synthesis and Characterization of Bio-Based PU/GO Nanocomposites

The obtained GO was dried at 60°C under vacuum overnight. The dried GO (0.1 g) was exfoliated in 10 g DMF using ultrasonication with a power of 70 W for 0.5 h at room temperature. Calculated amount of the obtained GO/DMF mixture was mixed with 1 g biobased polyols at room temperature, and then 0.31 g IPDI and one to two drops DBTDL catalyst were added to the mixture and kept stirring at 70°C for 2 h. Afterward, the solution was poured into a Teflon mold to produce 100-mm length and 50-mm width film with approximately 0.5-mm thickness, which was cut into specific dimensions for characterization. The elementary steps of PU/GO synthesis are illustrated in Figure 9.8.

The thermal stability of the PU/GO nanocomposites was investigated using TGA. Figure 9.9 demonstrates a typical TGA measurement for PU/GO nanocomposites at 20°C/ min heating rate under a nitrogen atmosphere. Clearly, all samples are thermally stable at temperatures up to 290°C. The onset of the thermal degradation process of all PU/GO nanocomposites is almost identical and slightly lower than that of pure PU. It is also clear from this figure that the soft segments of PU start to degrade first at 290-370°C and the hard segments degrades later on at a temperature higher than 370°C. The nanocomposites have more thermal stability at temperature higher than 400°C. In addition, the PU/GO nanocomposites with $GO \ge 0.4$ wt% have about 5 wt% inert residues, while the pure PU and PU/GO with 0.2 wt% GO have no any inert residue remaining. In conclusion, incorporation of GO does not significantly enhance the thermal stability of PU under a nitrogen atmosphere [59].

The fracture surface of the PU/GO nanocomposites with different GO contents was investigated by SEM. As shown in Figure 9.10, the fracture surface of the neat PU film after tensile testing is relatively smooth, and exhibits stiffness. While the fracture surfaces of the PU/GO films after tensile testing become rough, and exhibit toughness until the



Figure 9.8 Elementary steps for the synthesis of PU/GO nanocomposite [59].



Figure 9.9 TGA measurement for PU/GO nanocomposites at 20°C/min heating rate under a nitrogen atmosphere [59].

concentration of GO reaches 0.4%. Moreover, most of the GO nanosheets were well dispersed and embedded into the PU matrix with the concentration of GO no more than 0.4%. With the addition of more GO, it began to aggregate, and the fracture-surface images of the films exhibit a stack of sheets (bright stripe), just like that of the GO membranes, which provided a barrier to cure by physically blocking functional group diffusion [59] (Figure 9.10).

9.5 Plant Oil-Based/Clay and Silica Nanocomposites

Epoxidized soybean oil (ESO) and nanoclay thermoset as reinforcement filler has been reported by Liu et al. [62]. The mechanical properties increased with adding nanoclay up to 8 wt% followed by a decrease in the mechanical properties with higher filler contents. The obtained results were related to the aggregation of the nanoclay at a high concentration.



Figure 9.10 SEM images of (a) 0.2 wt% GO nanocomposite, (b) 0.4 wt% GO nanocomposite, and (c) 0.6 wt% GO nanocomposite [59].



Figure 9.11 TEM and X-ray diffraction for castor oil-based PU/nanoclay composites with different concentrations of nanoclay. Reprinted with permission of Elsevier [64].

The glass relaxation process of these nanocomposites measured by DMA (peak maximum of tan δ) shifted to higher temperature from 11.8°C to 20.7°C with increasing the nanoclay concentration up to 5 wt% [62].

Organically modified montmorillonite with a wide range of concentration up to 15 wt% has been used as reinforcement filler for cationically polymerization of ESO [63]. The ESO/nanoclay composites have high storage modulus and glass transition temperature with high clay concentration. This improvement in thermomechanical properties of the nanocomposites was related to the copolymerization of the surface modified nanoclay with the ESO.

Nanocomposites from castor oil-based PU and a quaternary ammonium modified montmorillonite was reported by Kaushik et al. [64]. The nanocomposites were prepared with different modified clay contents from 0 wt% to 5 wt%. The X-ray diffraction technique was used to evaluate the exfoliation behavior of the nanoclay in the castor oil-based PU matrix. In addition, the TEM was also used to evaluate the morphology

of the nanocomposites. Delaminated stacks of silicate and individual platelets as well as clay aggregation were observed in the TEM morphology. Figure 9.11 summarized both the TEM and X-ray data for the castor oil-based PU/nanoclay composites [64]. The mechanical properties (Young's modulus, tensile strength, and elongation at break) of the nanocomposites increased with increasing the content of modified clay. In addition, the thermal stability, barrier properties to the diffusion of vapor and liquid water of castor oil-based PU improved significantly by adding the modified nanoclay.

Aqueous castor oil-based PU and surface modified silica nanocomposites have been synthesized with different contents of silica through a sol–gel process [65]. Figure 9.12 shows the TEM morphologies of the diluted aqueous dispersions. The gray and black particles are the PU and silica nanoparticles, respectively. The black silica aggregated to large particles with increasing silica concentration. The authors also found that the silica particles are mainly embedded in



Figure 9.12 TEM images for PU/Si aqueous dispersion with (a) 0 wt% Si, (b) 0.5 wt% Si, (c) 1 wt% Si, and (c) 2 wt% Si. Reprinted with permission from Elsevier [65].

the PU particles (a core-shell structure). Both mechanical properties and thermal stability of solid films obtained from solvent cast of the aqueous dispersions increased significantly with increasing the nanosilica content. The crosslink density of the nanocomposites was also calculated from the storage modulus plateau of the DMA data at 40°C above the T_g based on rubber elasticity theory. The value of the crosslink density increased from 90 mol/m³ for pure castor oil-based PU to

766 mol/m³ for the PU/silica composite with 2 wt% silica. However, on the other hand the T_g of the nanocomposites slightly decreased with increasing silica contents.

References

 E. Bekyarova, E.T. Thostenson, A. Yu, H. Kim, J. Gao, J. Tang, H.T. Hahn, T.W. Chou, M.E. Itkis, R.C. Haddon, Multiscale carbon nanotube-carbon fiber reinforcement for advanced epoxy composites, Langmuir 23 (7) (2007) 3970–3974.

- [2] C.M. Huang, K.H. Wei, Pseudo-singlecrystalline self-assembled structure formed from hydrophilic CdSe and hydrophobic an nanoparticles in the polystyrene and poly(4-vinylpyridine) blocks, respectively, of a polystyrene-b-poly(4vinylpyridine) diblock copolymer, Macromolecules 41 (19) (2008) 6876–6879.
- [3] A. Satake, Y. Miyajima, Y. Kobuke, Porphyrin – Carbon nanotube composites formed by noncovalent polymer wrapping, Chem. Mater. 17 (4) (2005) 716–724.
- [4] H.M. Song, Y.J. Kim, J.H. Park, Threedimensional hierarchically organized magnetic nanoparticle polymer composites: achievement of monodispersity and enhanced tensile strength, J. Phys. Chem. C 112 (14) (2008) 5397–5404.
- [5] P. Xu, X.J. Han, C. Wang, D.H. Zhou, Z.S. Lv, A.H. Wen, X.H. Wang, B. Zhang, Synthesis of electromagnetic functionalized nickel/polypyrrole core/ shell composites, J. Phys. Chem. B 112 (34) (2008) 10443–10448.
- [6] E.G. Bellomo, T.J. Deming, Monoliths of aligned silica-polypeptide hexagonal platelets, J. Am. Chem. Soc. 128 (7) (2006) 2276–2279.
- [7] W.F. Chen, J.S. Wu, P.L. Kuo, Poly(oxyalkylene)diamine-functionalized carbon nanotube/perfluorosulfonated polymer composites: synthesis, water state, and conductivity, Chem. Mater. 20 (18) (2008) 5756–5767.
- [8] S.B. Kharchenko, J.F. Douglas, J. Obrzut, E.A. Grulke, K.B. Migler, Flowinduced properties of nanotube-filled polymer materials, Nat. Mater. 3 (8) (2004) 564–568.

- [9] Y.J. Li, H. Shimizu, High-shear processing induced homogenous dispersion of pristine multiwalled carbon nanotubes in a thermoplastic elastomer, Polymer 48 (8) (2007) 2203–2207.
- [10] S. Pegel, P. Potschke, G. Petzold, I. Alig, S.M. Dudkin, D. Lellinger, Dispersion, agglomeration, and network formation of multiwalled carbon nanotubes in polycarbonate melts, Polymer 49 (4) (2008) 974–984.
- [11] D.W. Schaefer, R.S. Justice, How nano are nanocomposites?, Macromolecules 40 (24) (2007) 8501–8517.
- [12] H.Z. Sun, H. Zhang, J. Ju, J.H. Zhang, G. Qian, C.L. Wang, B. Yang, Z.Y. Wang, One-step synthesis of high-quality gradient CdHgTe nanocrystals: a prerequisite to prepare CdHgTe-polymer bulk composites with intense near-infrared photoluminescence, Chem. Mater. 20 (21) (2008) 6764–6769.
- [13] T.D. Fornes, D.R. Paul, Structure and properties of nanocomposites based on nylon-11 and-12 compared with those based on nylon-6, Macromolecules 37 (20) (2004) 7698–7709.
- [14] J.E. Mark, Some novel polymeric nanocomposites, Acc. Chem. Res. 39 (12) (2006) 881–888.
- [15] Q. Chen, L. Zhao, C. Li, G.Q. Shi, Electrochemical fabrication of a memory device based on conducting polymer nanocomposites, J. Phys. Chem. C 111 (49) (2007) 18392–18396.
- [16] R. Gangopadhyay, A. De, Conducting polymer nanocomposites: a brief overview, Chem. Mater. 12 (7) (2000) 2064–12064.
- [17] N. Liu, M.M. Shi, X.W. Pan, W.M. Qiu, J.H. Zhu, H.P. He, H.Z. Chen, M. Wang, Photoinduced electron transfer and enhancement of photoconductivity in silicon nanoparticles/perylene diimide

composites in a polymer matrix, J. Phys. Chem. C 112 (40) (2008) 15865–15869.

- [18] G.H. Lu, H.W. Tang, Y.P. Qu, L.G. Li, X.N. Yang, Enhanced electrical conductivity of highly crystalline polythiophene/insulating-polymer composite, Macromolecules 40 (18) (2007) 6579–6584.
- [19] H.J. Ding, X.M. Liu, M.X. Wan, S.Y. Fu, Electromagnetic functionalized cage-like polyaniline composite nanostructures, J. Phys. Chem. B 112 (31) (2008) 9289–9294.
- [20] F.F. Fang, J.H. Kim, H.J. Choi, Synthesis of core-shell structured PS/Fe₃O₄ microbeads and their magnetorheology, Polymer 50 (10) (2009) 2290–2293.
- [21] F.G. Guo, Q.Y. Zhang, B.L. Zhang, H.P. Zhang, L. Zhang, Preparation and characterization of magnetic composite microspheres using a free radical polymerization system consisting of DPE, Polymer 50 (8) (2009) 1887–1894.
- [22] N.S. Kommareddi, M. Tata, V.T. John, G.L. McPherson, M.F. Herman, Y.S. Lee, C.J. O'Connor, J.A. Akkara, D.L. Kaplan, Synthesis of superparamagnetic polymer-ferrite composites using surfactant microstructures, Chem. Mater. 8 (3) (1996) 801–809.
- [23] G.Y. Liu, H. Wang, X.L. Yang, Synthesis of pH-sensitive hollow polymer microspheres with movable magnetic core, Polymer 50 (12) (2009) 2578–2586.
- [24] Y.L. Yang, M.C. Gupta, Novel carbon nanotube-polystyrene foam composites for electromagnetic interference shielding, Nano. Lett. 5 (11) (2005) 2131–2134.
- [25] P.Xu,X.J.Han,C.Wang,H.T.Zhao,J.Y. Wang, X.H. Wang, B. Zhang, Synthesis of electromagnetic functionalized barium ferrite nanoparticles embedded in polypyrrole, J. Phys. Chem. B 112 (10) (2008) 2775–2781.

- [26] L.L. Beecroft, C.K. Ober, Nanocomposite materials for optical applications, Chem. Mater. 9 (6) (1997) 1302–1317.
- [27] L. Dennany, E.J. O'Reilly, P.C. Innis, G.G. Wallace, R.J. Forster, Solid state photochemistry of novel composites containing luminescent metal centers and poly(2-methoxyaniline-5-sulfonic acid), J. Phys. Chem. B 113 (21) (2009) 7443–7448.
- [28] R.Ravindranath, P.K. Ajikumar, N.B.M. Hanafiah, W. Knoll, S. Valiyaveettil, Synthesis and characterization of luminescent conjugated polymer-silica composite spheres, Chem. Mater. 18 (5) (2006) 1213–1218.
- [29] R. Sainz, W.R. Small, N.A. Young, C. Valles, A.M. Benito, W.K. Maser, M.I.H. Panhuis, Synthesis and properties of optically active polyaniline carbon nanotube composites, Macromolecules 39 (21) (2006) 7324–7332.
- [30] E. Chiellini, P. Cinelli, V.I. Hieva, M. Martera, Biodegradable thermoplastic composites based on polyvinyl alcohol and algae, Biomacromolecules 9 (3) (2008) 1007–1013.
- [31] U. Costantino, V. Bugatti, G. Gorrasi, F. Montanari, M. Nocchetti, L. Tammaro, V. Vittoria, New polymeric composites based on poly(epsilon-caprolactone) and layered double hydroxides containing antimicrobial species, ACS Appl. Mater. Inter. 1 (3) (2009) 668–677.
- [32] M.C. Urbina, S. Zinoveva, T. Miller, C.M. Sabliov, W.T. Monroe, C.S.S.R. Kumar, Investigation of magnetic nanoparticlepolymer composites for multiplecontrolled drug delivery, J. Phys. Chem. C 112 (30) (2008) 11102–11108.
- [33] D.H. Zhang, M.A. Kandadai, J. Cech, S. Roth, S.A. Curran, Poly(L-lactide) (PLLA)/multiwalled carbon nanotube (MWCNT) composite: characterization

and biocompatibility evaluation, J. Phys. Chem. B 110 (26) (2006) 12910–12915.

- [34] A. Kelly, Composites in context, Compos. Sci. Technol. 23 (3) (1985) 171–199.
- [35] A. Kelly, C.H. Zweben, Comprehensive Composite Materials, Elsevier, Amsterdam, New York, (2000).
- [36] E.T. Thostenson, C.Y. Li, T.W. Chou, Nanocomposites in context, Compos. Sci. Technol. 65 (3–4) (2005) 491–516.
- [37] F. Hussain, M. Hojjati, M. Okamoto, R.E. Gorga, Review article: polymermatrix nanocomposites, processing, manufacturing, and application: an overview, J. Compos. Mater. 40 (17) (2006) 1511–1575.
- [38] S. Iyer, D.A. Schiraldi, Role of specific interactions and solubility in the reinforcement of bisphenol A polymers with polyhedral oligomeric silsesquioxanes, Macromolecules 40 (14) (2007) 4942–4952.
- [39] A.A. Koval'chuk, V.G. Shevchenko, A.N. Shchegolikhin, P.M. Nedorezova, A.N. Klyamkina, A.M. Aladyshev, Effect of carbon nanotube functionalization on the structural and mechanical properties of polypropylene/MWCNT composites, Macromolecules 41 (20) (2008) 7536–7542.
- [40] H.Z. Sun, H. Zhang, J.H. Zhang, Y. Ning, T.J. Yao, X. Bao, C.L. Wang, M.J. Li, B. Yang, Effect of electrostatic interactions on the photophysical properties of the composites of CdTe nanocrystals and carbazole-containing polymers, J. Phys. Chem. C 112 (7) (2008) 2317–2324.
- [41] S.A. Madbouly, J.U. Otaigbe, A.K. Nanda, D.A. Wicks, Rheological behavior of POSS/polyurethane-urea nanocomposite films prepared by ho-

mogeneous solution polymerization in aqueous dispersions, Macromolecules 40 (14) (2007) 4982–4991.

- [42] http://en.wikipedia.org/wiki/Lignin.
- [43] X. Luo, A. Mohanty, M. Misra, Lignin as a reactive reinforcing filler for waterblown rigid biofoam composites from soy oil-based polyurethane, Polym. Int. 54 (2005) 829–836.
- [44] A. O'Donnell, M.A. Dweib, R.P. Wool, Natural fiber composites with plant oilbased resin, Compos. Sci. Technol. 9 (2004) 1135–1145.
- [45] H.J. Wang, H.Y. Xiao, F.Q. Sun, J.H. Zhang, Fabrication and properties of lignin/castor oil-based composites, Adv. Mater. Res. 311–313 (2011) 1535–1538.
- [46] S. Laurichesse, C. Huillet, L. Avérous, Original polyols based on organosolv lignin and fatty acids: new bio-based building blocks for segmented polyurethane synthesis, Green Chem. 16 (2014) 3958–3970.
- [47] P. Câmara Mileo, M.F. De Oliveira, S.M. Luz, G.J.M. Rocha, A.R. Gonçalves, Evaluation of castor oil polyurethane reinforced with lignin and cellulose from sugarcane straw, Adv. Mater. Res. 123–125 (2010) 1143–1146.
- [48] P.J.F. Harris, Carbon Nanotube Science: Synthesis, Properties and Applications, Revised and updated ed., Cambridge University Press, Cambridge, UK, New York, (2009) p. x, 301 p.
- [49] S.A. Madbouly, A. Lendlein, Shapememory polymer composites, Shape-Memory Polym. 226 (2010) 41–95.
- [50] S. Reich, C. Thomsen, J. Maultzsch, Carbon Nanotubes: Basic Concepts and Physical Properties, Wiley-VCH, Weinheim, Cambridge, (2004) p. ix, 215 p.
- [51] McAninch, I.M., Wool, R.P. Carbon nanotube composites from soy-based

resins. The 10th Annual Green Chemistry and Engineering Conference (June 26–30, 2006).

- [52] K.W. Liang, S.Q. Shi, Soy-based polyurethane foam reinforced with carbon nanotubes, Key Eng. Mater. 419–420 (2009) 477–480.
- [53] D.A. Nguyen, Y.R. Lee, A.V. Raghu, H.M. Jeong, C.M. Shin, B.K. Kim, Morphological and physical properties of a thermoplastic polyurethane reinforced with functionalized graphene sheet, Polym. Int. 58 (4) (2009) 412–417.
- [54] D.Y. Cai, J. Jin, K. Yusoh, R. Rafiq, M. Song, High performance polyurethane/ functionalized graphene nanocomposites with improved mechanical and thermal properties, Compos. Sci. Technol. 72 (6) (2012) 702–707.
- [55] H. Kim, Y. Miura, C.W. Macosko, Graphene/polyurethane nanocomposites for improved gas barrier and electrical conductivity, Chem. Mater. 22 (11) (2010) 3441–3450.
- [56] Y.R. Lee, A.V. Raghu, H.M. Jeong, B.K. Kim, Properties of waterborne polyurethane/functionalized graphene sheet nanocomposites prepared by an *in situ* method, Macromol. Chem. Phys. 210 (15) (2009) 1247–1254.
- [57] X. Wang, Y.A. Hu, L. Song, H.Y. Yang, W.Y. Xing, H.D. Lu, *In situ* polymerization of graphene nanosheets and polyurethane with enhanced mechanical and thermal properties, J. Mater. Chem. 21 (12) (2011) 4222–4227.
- [58] Z.X. Chen, H.B. Lu, Constructing sacrificial bonds and hidden lengths for ductile graphene/polyurethane elastomers with improved strength and toughness, J. Mater. Chem. 22 (25) (2012) 12479–12490.

- [59] J. Zhang, C.Q. Zhang, S.A. Madbouly, *In situ* polymerization of bio-based thermosetting polyurethane/graphene oxide nanocomposites, J. Appl. Polym. Sci. 132 (13) (2015) 41751.
- [60] K.N. Kudin, B. Ozbas, H.C. Schniepp, R.K. Prud'homme, I.A. Aksay, R. Car, Raman spectra of graphite oxide and functionalized graphene sheets, Nano. Lett. 8 (1) (2008) 36–41.
- [61] U. Khan, P. May, A. O'Neill, J.N. Coleman, Development of stiff, strong, yet tough composites by the addition of solvent exfoliated graphene to polyurethane, Carbon 48 (14) (2010) 4035–4041.
- [62] Z. Liu, S.Z. Erhan, J. Xu, Preparation, characterization and mechanical of epoxidized soybean oil/clay nanocomposites, Polymer 46 (2005) 10119–10127.
- [63] H. Uyama, M. Kuwabara, T. Tsujimoto, M. Nakano, A. Usuki, S. Kobayashi, Green nanocomposites from renewable resources: plant oil-clay hybrid materials, Chem. Mater. 15 (2003) 2492–2494.
- [64] A. Kaushik, D. Ahuja, V. Salwani, Synthesis and characterization of organically modified clay/castor oil based chain extended polyurethane nanocomposites, Compos. Part A 42 (2011) 1534–1541.
- [65] Y. Xia, R.C. Larock, Preparation and properties of aqueous castor oil-based polyurethane–silica nanocomposite dispersions through a sol–gel process, Macromol. Rapid Commun. 32 (2011) 1331–1337.
- [66] M.L. Cohen, Mater. Sci. Eng. C 15 (2001) 1.
- [67] T. Guo, P. Nikolaev, A.G. Rinzler, D. Tomanek, D.T. Colbert, R.E. Smalley, J. Phys. Chem. 99 (1995) 10694.

Page left intentionally blank

10 Fiber Reinforced Plant Oil-Based Composites

Kunwei Liu¹, Chaoqun Zhang², and Samy A. Madbouly^{3,4}

¹Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA ²Department of Mechanical Engineering, Texas A&M University, College Station, TX, USA ³Department of Chemistry, Faculty of Science, Cairo University, Orman-Giza, Egypt ⁴Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA

OUTLINE	
10.1 Introduction	167
10.2 Fiber-Reinforced Plant Oil-Based Vinyl Polymer Composites	168
10.3 Fiber-Reinforced Plant Oil-Based Epoxy Composites	176
10.4 Fiber-Reinforced Plant Oil-Based Polyurethane Composites	179
10.5 Fiber-Reinforced Plant Oil-Based Foam Composites	184
10.6 Conclusion	186
References	187

10.1 Introduction

Due to the continuously increasing price of petroleum resources and an increase in environmental awareness, researchers are actively developing polymeric materials based on biorenewable resources to replace the traditional petroleum-based plastics. Thermoplastics such as polystyrene, polyethylene, polystyrene, poly(vinyl chloride) and thermosets such as epoxies and polyesters are used everywhere in our daily life. These plastics are used in applications including packaging, textiles, coatings, automobile components, biomedical devices, and household items. Even though plastics are versatile and essential materials in the modern society, one should not neglect the fact that petroleum resources are limited. It is proposed that all petroleum resources will be exhausted within the next 100 years [1]. As a result, it is important to continuously put effort in finding environmentalfriendly starting materials and methods to synthesis plastics.

Plant oils have been widely used as the building blocks of polymers because they are relatively cheap, available in large quantity, and environmentally friendly. Plant oils are biorenewable and can be obtained from a variety of plants, soybean, castor, tung tree, and palm tree. Plant oils are excellent biorenewable materials because the triglyceride molecule consists of many reactive functional groups, such as double bonds and hydroxyl groups. Soybean oil and linseed oil contain relatively high amount of unsaturation content compared to other plant oils; thus, these two types
of plant oils are used most often to synthesize polymers. Higher unsaturation content in plant oils will lead to higher number of reactive crosslink sites, resulting in polymers with better mechanical properties. Castor oil is also a popular starting material for producing polymers such as polyurethane because the castor oil itself contains hydroxyl groups that can be used to react with isocyanates.

Fiber reinforcements have been widely used in plastic industry to produce composites that have higher modulus and high strength compared to the neat plastic alone. Fibers such as glass fiber and carbon fiber are strong materials because they have relatively small diameter: thus, there are less defects in fiber compared to bulk materials. Plastics such as epoxies act as the matrix that consolidates every single fiber into a solid entity. A variety of fibers have been incorporated into the plant oil-based thermosets, including the traditional synthetic fibers such as glass fibers and carbon fibers as well as biobased fibers. The use of biobased fibers can not only enhance the mechanical properties of the plant oilbased matrix, but also increase the biorenewable content in the final products. Due to the lower cost of some biorenewable fibers, the incorporation of the biorenewable fibers can also decrease the cost. Biorenewable fibers can be divided into two categories, namely, plant-based fibers and animal-based fibers. Plant-based fibers include flax, hemp, kenaf, jute, sisal, banana, cottons, etc. Animal fibers such as chicken feathers, wool, and silk have also been used as reinforcements. However, biorenewable fibers have several drawbacks compared to the synthetic fibers. Biorenewable fibers generally possess lower mechanical properties and lower thermal stabilities when compared to the synthetic fibers. In addition, their hydrophilicity makes them prone to water damage. Since polymer matrixes are hydrophobic, the hydrophilic biorenewable fibers may not interact well with the matrix, producing composites with low interfacial adhesion between the fiber and the matrix. In this chapter, the experiment details of composites prepared using different plant oilbased matrixes and fibers are discussed. Plant oil-based vinyl thermoset, epoxies, and polyurethane have been widely used as the matrix for composites.

10.2 Fiber-Reinforced Plant Oil-Based Vinyl Polymer Composites

Akesson et al. prepared a series of structural composites based on acrylated epoxidized soybean oil (AESO) and various natural fiber mats [2]. The AESO was cured using 1,1-di-(tert-butylperoxy)-cyclohexane free radical initiator, and the concentration of the initiator was kept at 1 wt%. The composite was made by first spraying resin on the fiber mats, followed by compression molding under elevated temperature and pressure. Three different types of natural fiber mats were used in this study, including air laid flax mat, coarse hemp mat, air laid flax mat-PET hybrid. In this study, the tensile properties of the composites with 40-70 wt% air laid flax fiber mat were recorded in order to investigate the effect of fiber mat percentage. As expected, the mechanical properties increased with increasing amount of fiber mat in the composite. The composites with 40%flax mat have a tensile strength and a Young's modulus around 50 MPa and 5.0 GPA, respectively. When the fiber mat content increased to 70 wt%, the tensile strength of the composite increased to 78 MPa to 9.7 GPa, respectively. Dynamical mechanical analysis (DMA) showed that the storage modulus



Figure 10.1 (a) Storage modulus of air laid flax composites at different fiber contents. (b) Dynamical mechanical results for composites with varying fiber content [2].

also increased with fiber mat content (see Figure 10.1a), and this can be interpreted as a result of good interfacial adhesion between the fiber and the matrix. The glass transition temperatures obtained from the peak of tan δ curve also increased with increasing fiber mat content as shown in Figure 10.1b, and this increase was also attributed to the good adhesion between fiber and the polymer matrix. High interfacial adhesion will decrease the mobility of the polymer chains in the interface, resulting in an increase in glass transition temperature. The impact resistance was the composites made from AESO and air laid flax mat was measured using a Charpy impact strength tester, and the results showed that the composites with higher fiber content possess higher impact resistance. The author also compared the impact resistance of the natural fiber filled AESO composites with those of glass fiber reinforced polyester, and the test results indicated that the glass fiber reinforced polyester has much higher impact resistance than the biocomposites composed of AESO and flax mat reinforcement.

Besides AESO, other versions of modified soybean oil were also used to produce composite materials. Adekunle et al. produced composites with methacrylated soybean oil (MSO), methacrylic anhydride modified soybean oil (MMSO), and acetic anhydride modified soybean oil (AMSO) [3]. MSO was synthesized using a similar route to the synthesis route of AESO, except that methacrylic acid instead of acrylic acid was used to ring open the epoxy ring on epoxidized soybean oil (ESO). MMSO and AMSO were produced by reacting methacrylic anhydride and acetic anhydride with ESO, respectively. Two types of flax fibers were used in this study: airlaid flax mat and woven fabric. Six combinations of vegetable oils were utilized as matrix: MSO, MMSO, AMSO, and the addition of 30 wt% ST to each type of vegetable oil. Adding styrene to the soybean oil increased the flexural strength and moduli of the composites considerably because styrene can produce composites with longer crosslinks. Composites made from MMSO have higher strength and modulus when compared to those made from MSO because MMSO contains more reactive double bond in the structure. Composites with AMSO possess lower flexural properties than MSO. On the other hand, composites made from MMSO showed relatively low impact strength compared with composites made with AMSO and MSO, and this is attributed to the fact that the

high crosslink density produce from MMSO makes the polymer chain less flexible.

The effects of fabric architectures on the properties of the MESO-based composites were also investigated by Adekunle et al. [4]. It was found that flax mats with plainweave architecture can produce composites with higher flexural properties and impact strength compared to twill-weave architecture. Hybrid composites with Lyocell fiber mat as the inner ply were also prepared in this paper. The hybridization with Lyocell fiber can reduce the water absorption dramatically because the Lyocell fiber itself has lower water absorption than other natural fibers. In the other paper published by the same author, the effect of surface weight (distance between adjacent roving wefts and warps) of the woven jute fabric was also investigated [5]. It was found that increasing the surface weight results in a decrease in mechanical properties. Because the weight percentage of fiber reinforcement in this study was controlled at a fixed percentage, increasing the surface weight results in a reduce number of plies in composites.

Khot et al. also produced flax composites, hemp composites, and glass/flax composites using copolymer of AESO, ST, and divinylbenzene (DVB) as resin [6]. The resin injection molding process was utilized to produce glass fiber reinforced composites. The glass fiber reinforced AESO composites containing 35 wt% of glass fiber possessed a tensile modulus of 5.2 GPa and tensile strength of 463 MPa. When the fiber loading increased to 50%, composites with tensile modulus as high as 24.8 GPa can be observed.

Glass/flax hybrid composites were also produced to obtain composite materials with a wide range of properties [7]. It was found that the mechanical properties of the glass/ flax hybrid composites are highly dependent on the glass/flax ratio and the arrangement of fibers (see Figure 10.2) in the composite. As expected, the flexural modulus, flexural strength, and impact resistance decreased with increasing amount of flax fiber in the hybrid composite. Symmetric arrangement can produce composites with higher mechanical properties when compared to unsymmetrical arrangement at the same glass/ flax ratio. Impact test showed interesting results. For unsymmetrical composites, if the flax face was impacted, the composites can absorb higher energy when compared to the situation when the glass face was impacted. In flexural testing of unsymmetrical arrangement, when the load is on the glass face, the composites showed lower flexural modulus and higher flexural strength than if the load is on the flax face. This study showed that for hybrid composites, the ratio between different fibers as well as the construction of the laminates will have a significant impact on the properties of composites.



Figure 10.2 Different modes of arrangement of fibers in the composite: symmetic composites (a); unsymmetric composites, loading on the glass fiber face (b); unsymmetric composites, loading on the flax fiber face (c) [7].



Figure 10.3 Schematic of vacuum-assisted resin transfer molding technique [8].

The vacuum-assisted resin transfer molding technique (see Figure 10.3) was also utilized to produce composite panel out of AESO-based resin [8]. Several kinds of natural fiber mats, including flax, cellulose, pulp, hemp, and recycled corrugated cardboard, were used as reinforcements. Among these reinforcements, recycled cardboard paper was found out to be the best reinforcement. With 55.2 wt% of the recycled cardboard paper, the storage modulus of the AESO resin was improved more than five times.

Hong et al. used keratin fiber to reinforce AESO-based and soybean oil pentaerythritol glyceride maleates (SOPERMA) resin [9]. Keratin fiber was obtained from chicken feather, and keratin fiber (KF) was made into mat. Composites containing 5–30 wt% of KF mat were put into the resin using vacuum-assisted resin transfer molding. The storage modulus was significantly increased with increasing amount of KF. For example, when 30% of KF was introduced into the AESO composites, the storage modulus increased by 58.8% when compared to neat resin. The glass transition temperature remains unaffected. However, when KF fiber was added to SOPERMA resin, the increase in modulus is not very pronounced. The maximum improvement in storage modulus is only 9.6% as shown in Table 10.1. On the other hand, the fracture toughness can also be increased by adding KF, and the increase in fracture toughness is more pronounced in AESO resin. In addition, the dielectric constants can be increased by adding more KF reinforcement.

Pyrolyzed chicken feather fibers (PFF) were also incorporated into vegetable oilbased resin to produce environmental friendly composite. A mixture of AESO and methacrylated lauric acid (MLAU) was used as the matrix. MLAU acts as a reactive diluent for AESO, and the ratio of AESO and MLAU was kept at 1:1 in this study. The storage modulus, tensile modulus, tensile strength, and fracture energy were found to be increased with increasing amount of PCFF; however, there is not a big difference in the reinforcement effect between untreated chicken feather and pyrolyzed chicken feather.

Bacterial cellulose (BC) was used as a binder for AESO-based resin and sisal fibers [10].

KF Content (wt%)	Max. Load N	Fracture Toughness <i>K</i> _{ic} (MPa m ^{1/2})	Fracture Energy G _{ic} (KJ/m²)
0	97.6	1.458	1.420
5	97.1	1.455 (-0.2%)	1.610 (+13.4%)
10	103.0	1.524 (+4.5%)	1.759 (+23.9%)
20	113.0	1.672 (+14.7%)	1.820 (+28.2%)
30 (KF mat)	120.1	1.768 (+21.3%)	1.927 (+35.7%)
30 (hybrid mat)	130.7	1.931 (+32.4%)	1.945 (+37.0%)

 Table 10.1
 Fracture Properties of AESO Composites Prepared by Vacuum-Assisted Resin Transfer

 Molding [9]

BC is a special type of cellulose produced by reacting *Gluconacetobacter xylinus* with cellulose, yielding a cellulose product with degree of polymerization around 6500. It was found that the BC can increase the mechanical properties of the sisal fiber. The BCmodified composites increased the tensile modulus and strength of neat AESO thermoset by 1300% and 600%, respectively. The reinforcement effect of BC-modified sisal fiber is much greater than the neat sisal fiber. The decrease in amplitude of tan δ was also observed after incorporating BC modified sisal, indicating that the fiber–matrix adhesion has been increased.

The effect of surface treatment on ramie fiber on the interfacial adhesion between the modified fiber and AESO was investigated [11]. Three different surface treatments were studied: (i) acetylation using acetic anhydride; (ii) silane treatment using triethoxyvinylsilane; and (iii) peroxide treatment using dicumyl peroxide. It was found that using surface treatment to modify or cover the OH groups on ramie fiber can lower the hydrophilicity of ramie fiber, and this can increase the interaction between the reinforcement and the matrix. In addition, the interfacial shear strength (IFSS) of AESO/ramie composites was increased significantly. Silane treatment yielded composites with highest interfacial adhesion between the fibers and the AEOS matrix. After the silane treatment, the IFSS increased around 75%. The author also investigated the effect of acetylation time on the IFSS of the final composites, and the result indicated that the IFSS of the composites increased with acetylation time. TGA data indicated that acetylation decreased the thermal stability of the ramie fiber, and this is ascribed to the etching and the disruption of the fiber structure. On the other hand, silane and peroxide treatments increased the thermal stability of the ramie fiber, which is due to the removal of hemicellulose and impurity by silane and peroxide treatments.

Green composites from maleated castor oil (MACO) and jute fibers were also studied [12]. The maleated castor oil was produced by adding maleic anhydride to castor oil in 3:1 molar ratio, followed by heating the mixture up at 125°C for 4 h. Due to the high viscosity of MACO, 20 wt% of tung oil was added to a reactive diluent. This resin system was polymerized using a free radical initiator. Nonwoven jute felts were used as reinforcement. In this study, the mechanical properties MACO/jute composites were compared to those of unsaturated polyester (UPE)/jute composites. The flexural moduli were very similar for these two composites; however, the flexural strength of MACO/jute composites was much lower than the flexural strength of UPE/jute composites. The UPE jute composites possess a flexural strength of around 35 MPa, while the MACO/jute biocomposites only have a flexural strength of 15 MPa, which can be ascribed to the higher hydrophobicity of the MACO resin compared to that of UPE. Because natural fibers are hydrophilic, the interfacial adhesion between the reinforcement and the matrix can be decreased if resin with higher hydrophobicity is used. The breaking strain of UPE/jute was about 60% lower. On the other hand, the impact strength of MACO/jute composites was 42% higher than that of UPE jute composites. The high impact strength of MACO/ jute composite may be due to the presence of flexible fatty acid structure.

Vegetable oil thermoset synthesized by the ring-opening metathesis polymerization (ROMP) method was also utilized to produce composite materials. Henna et al. prepared glass fiber reinforced composites using a copolymer of Dilulin (a type of linseed oil modified with cyclopentadiene) and dicyclopentadiene (DCPD) as the matrix and the short glass fiber mat as the reinforcement. Second-generation Grubb's catalyst was utilized to initiate the polymerization process. An increase in DCPD content resulted in a dramatic increase in glass transition temperature, tensile strength, and Young's modulus because DCPD is a rigid crosslinker in the crosslink network. Thermoset containing 30 wt% of DCPD exhibits a tensile strength and Young's modulus of 0.5 and 0.91 MPa, respectively, while thermoset with 70% of DCPD possesses a tensile strength closed to 30 MPa and a Young's modulus around 525 MPa. Higher glass fiber content increased the tensile properties of the composites. Composites fabricated using 70% DCPD as resin and 56 wt% glass fiber can

reach a tensile strength of around 170 MPa and a Young's modulus of around 1.5 GPa. The amount of glass fiber incorporated into the composites did not affect the glass transition temperature. In addition, SEM images indicated that there is a weak interfacial adhesion between this type of vegetable oil-based matrix and glass. The author also found that the modulus values of the obtained composites are far below those predicted by the Halpin–Tsai and Cox model, indicating that a weak adhesion exists between the fiber and the matrix.

The copolymer of Dilulin and DCPD showed weak adhesion with the fiber reinforcement; hence, Cui et al. investigated the possibility of using silane coupling agent to increase the fiber-matrix adhesion [13]. Norbornenylethyldimethylchlorosilane (MCS) and Norbornenylethyltrichlorosilane (TCS) were used because they contain norbornene groups that can participate in the ROMP reaction. MSC contains only one hydrolyzable group, while TCS contains three hydrolyzable groups. The silane is first hydrolyzed in the presence of water and then undergoes condensation reaction with hydroxyl groups on the surface of glass fiber to form strong siloxane bonds. The results indicated that grafting of silane onto the glass fiber can improve the interfacial adhesion between the matrix and the fiber, the storage modulus, interfacial shear strength, and interlaminar shear strength of the final composites have been improved. MCS can provide a better improvement of the mechanical properties compared to TCS because MCS has a higher covalent bond density as shown in Figure 10.4. Cui et al. also used the pultrusion process to fabricate glass fiber reinforced composites using Dilulin/DCPD as the matrix along with MCS as the silane coupling agent [14]. Flexural testing showed



Figure 10.4 SEM images of fracture surfaces for composites made with different types of fibers [13]. (a) Untreated, (b) heat-cleaned, (c) MCS-treated, (d) TCS-treated.

that the flexural strength and modulus can be increased by increasing silane concentration; however, if the concentration of silane coupling agent was above 3%, the increase in interfacial bonding became less pronounced.

A series of composites using conjugated vegetable oil as the matrix have been studied [15-22]. Most vegetable oils do not contain conjugated double bonds; as a result, the reactivity of the unmodified vegetable oils is very low, resulting in materials with low mechanical properties. The conjugation of vegetable oil can be achieved by using a rhodium-based catalyst [23]. Lu et al. produced glass fiber mat reinforced composites based on regular and conjugated corn oil. ST and divinylbenzene (DVB) were used as comonomers in order to increase the mechanical properties of the corn oil-based thermoset [15]. The polymerization was initiated by a cationic polymerization initiator called

boron trifluoride diethyl etherate (BFE). The tensile modulus of the conjugated corn oilbased composites increased from 4.7 MPa to 933 MPa, and the tensile strength increased from 2.0 MPa to 12 MPa when 45 wt% of glass fiber was incorporated. Composites using conjugated corn oil as the matrix showed higher mechanical properties than composites made from regular corn oil, which is ascribed to the increase in reactivity of corn oil after conjugation. An increase in DVB content can increase the crosslink density of the thermoset, resulting in a significant improvement in mechanical properties and thermal stability of the composites. Even though the mechanical properties have been improved significantly by adding glass fiber and DVB, the composites only showed a maximum tensile strength of around 12 MPa, which is relatively low compared to the commercially available polymer composites. Conjugated

soybean oil and conjugated low saturation soybean oil were used by Lu et al. to produce composites reinforced by glass fiber mat [16]. ST and DVB were also used as comonomers. The conclusions are generally the same as those in Ref. [15]. However, the composites produced using soybean oil have higher mechanical properties than those produced from soybean oil because soybean oil contains more carbon–carbon doubles than corn oil.

Pfister et al. developed biocomposites using free radically polymerized conjugated linseed oil resin as the matrix and wheat straw (WS) as reinforcement. DVB and *n*-butyl methacrylate (BMA) were used as copolymer, and 5% *t*-butyl peroxide (TBPO) was used as a free radical initiator. Wheat straw acts as an effective reinforcement for the vegetable oil-based matrix. The modulus and tensile strength increased systematically with increasing amount of WS when the content of WS was in the range of 50–80 wt%. The mechanical properties of the composites with 90 wt% of WS do not follow the above trend due to improper wetting. Since agriculture fibers are generally not thermally stable, adding more WS resulted in a decrease in faster thermal degradation. When the length of the WS was reduced, the Young's modulus of the composite was decreased, while the tensile strength did not exhibit a clear tread. In addition, increasing the size of the WS fibers also resulted in significant increase in water absorption. The effect of compression molding pressure on the final mechanical properties was also investigated. When the molding pressure increased from 400 to 600 psi, the Young's modulus, tensile modulus, and thermal stability increased slightly, while the water absorption was reduced considerably. It was also found that incorporating maleic anhydride can increase the fibermatrix interaction, thus increasing the thermal and mechanical properties. In addition, similar results were found when using switchgrass and corn stover (see Figure 10.5) as the reinforcement [19,22]. The same natural fibers were also incorporated into cationically cured



Figure 10.5 Microscope images of 0.5, 1, and 2 mm CS at 8× magnification [19].

vegetable oil thermoset, and the results were slightly different than the composites made with free radically cured resin. When the fiber content was controlled at 75 wt%, the Young's modulus increased significantly; however, unlike the composites made with free radically cured resin, the tensile strength of these cationically cured composites was inferior to the tensile strength of the pure resins. The decrease in tensile strength is probably due to the poor interfacial interaction between the fiber and the matrix. Among corn stover, wheat straw, and switchgrass, the wheat straw can provide the highest improvement in Young's modulus.

10.3 Fiber-Reinforced Plant Oil-Based Epoxy Composites

Chandrashekhara et al. reported composites produced by reinforcing partially biobased epoxy by glass fiber; the pultrusion process was used to fabricate composites [24]. Epoxidized allyl soyate (EAS) was blended with Epon 9500, a petroleum-based epoxy resin designed for the pultrusion process. The EAS was synthesized by first performing transesterification on soybean oil suing methyl alcohol and allyl alcohol to produce fatty acid methyl ester and fatty acid ally ester. Then, the fatty acid esters were epoxidized to produce soyate epoxy resin. 10% to 30% of EAS was blended with EPON 9500. Composites with 63% volume fraction of glass fiber were produced using a pultrusion machine. It was found that incorporation of 10-20 wt% of epoxidized allyl soyate into epoxy yielded a material with a toughness that is much higher than pure Epon 9500 resin. The epoxy with 10% EAS was found to have higher tensile strength and tensile strain than that of pure Epon 9500. The

flexural strength of the composites decreased systematically with increasing amount of EAS in the resin. The composites containing 10% EAS in the resin have a flexural modulus about 100 MPa lower than that of composites made using 100% Epon 9500 as the resin. In addition, the incorporation of EAS can improve the impact damage resistance of the composites. It was also found that the average pulling force needed in pultrusion decreased with increasing amount of EAS in the resin, indicating that adding EAS can make the petroleum-based resin more suitable for the pultrusion process.

Boquillon et al. used epoxidized oil-based thermoset reinforced with hemp fibers to produce biorenewable composites [25]. The vegetable oil-based thermosetting resin contains epoxidized linseed oil with methyl terahydrophthalic anhydride as a hardener. The flexural modulus and flexural strength of the composites increased systematically when the hemp fiber volume fraction was in the range of 0–30%. Composites with more than 30% volume fraction of fibers possessed flexural properties lower than that of composites with less than 30% volume fraction of hemp fibers. A good level of interfacial adhesion was also observed under SEM. Composites with different lengths of hemp fibers were also produced. The flexural modulus increased with increasing fiber length. The flexural strength also increased with increasing fiber length, except when the fiber length is below 2 mm. Composites reinforced by hemp fibers with a length of 1 mm possessed lower flexural strength than the neat resin, indicating that the critical length for this composite system is between 1 mm and 2 mm. The DMA test showed that the introduction of hemp fibers significantly increased the storage modulus in rubbery region. On the other hand, the glass transition temperature of the composites decreased with increasing volume fraction of hemp fiber reinforcement. The decrease in glass transition temperature is probably due to the reaction between the anhydride hardener and the hydroxyl groups on the hemp fiber; as a result, less amount of anhydride was reacted with the epoxide group, producing materials with lower glass transition temperature.

Liu et al. fabricated composites using a blend of ESO and petroleum-based epoxy as resin using solid freeform fabrication (SFF) methods. SFF is a rapid prototyping technique that produces products without molds. This concept is very similar to 3D printing. In this study, ESO was blended with Epon 828 resin using a weight ratio of 1:0.3. Four different fibers were used as reinforcement, namely, E-glass fiber, carbon fiber, Franklin fiber, and wollastonite. It was found that the glass fiber and carbon fiber produced composites with better mechanical properties compared to those reinforced with mineral fibers. Combinations of fibers were also used to produce hybrid composite, and the result indicated that the hybrid composites have higher strength and modulus than composites made with a single type of fiber. Three different amine-curing agents were used, including diethylenetriamine (DETA), triehylenetetraamine (TETA), and polyethylenimine. It was found that polyethylenimine yielded composites with lowest properties, while TETA performed the best.

Composites from epoxidized soybean oil and flax fiber were studied [26]. ESO was blended with 1,1,1-tris(*p*-hydroxyphenyl) ethane triglycidyl ether (THPE-GE) in different ratios. The flexural modulus and flexural strength increased when the THPE-GE amount increased in the resin. The flax fibers were chopped into short fibers. In general, the mechanical properties of a composite will be improved by adding more volume fraction of fiber; however, in this study, the mechanical properties of the composites reached its maximum at 10 wt% of fiber. At fiber content above 10 wt%, the mechanical properties of the composites decreased. The decrease in flexural properties at high fiber loading is ascribed to the increase in fiber-to-fiber interaction and problems in fiber dispersion. In addition, it was found that the longer the fiber, the higher the mechanical properties. The other study performed by the same author focused on modification of flax fibers to produce biocomposites with enhanced properties [27]. Two methods of chemical modifications were investigated. The first method is to coat the flax fiber with a milk protein gel to increase the water resistance of the fiber. The second method involved acrylation in the presence of excess canola oil with Rhizopus oryzae lipase as the catalyst. Both these two modification methods can increase the water resistance of the composites, and lipid acrylation produced composites with lowest water uptake. When the flax fibers are coated with protein, methanol groups on the protein can react with hydroxyl groups on the flax fibers to form covalent bonds, which are more water resistant. On the other hand, the increase in water resistance of flax fibers after lipid acrylation may be due to the hydrophobicity of the fatty acid chain and high efficiency of acrylation catalyzed by enzyme. Lipid acrylation can also increase the tensile properties of the composites, which can be ascribed to the increase in interfacial adhesion between the fiber and the matrix. The increase in interfacial adhesion can also be due to the compatibility between the fatty acid chain on the lipid acrylated flax fibers and the ESO.

Besides using epoxidized vegetable oil and natural fibers to produce green materials,

attempts have been made to use biobased anhydride to cure the epoxidized vegetable oil [28]. Takahashi et al. used a terpene-based acid anhydride (TPAn) to cure ESO. In addition, maleinated linseed oil (LOAn), hyxahydrophthalic anhydride (HPAn), and thermal latent cationic polymerization catalyst (CPI) were also used for comparison. Among these four curing agents, ESO-TPAn resin can produce materials with highest glass transition temperature and highest tensile properties. Lyocell fibers were used to reinforce the ESO-TPAn resin (with epoxy/anhydride ratio of 1:1). The tensile strength of the composites increased with increasing Lyocell fiber content. After 75 wt% of Lyocell fiber was added, composites with a tensile strength of 65 MPa and a tensile modulus of 2.3 GPa could be produced, and the mechanical properties of the composites were three times higher than those of neat ESO-TPAn resin.

Crivello et al. prepared glass fiber reinforcement composites using UV-cured epoxidized vegetable oils as the matrix [29]. ESO or epoxidized linseed oil (ELO) was blended with a petroleum-based epoxide. Onium salt cationic photoinitiator was added to the resin to allow for UV polymerization. With the photoinitiator, the gelation time of the resin can be as short as 7 s. It was also found that the sizing on the glass fiber increased the gelation time of the resin, indicating that the sizing agents retarded the photopolymerization process, but the retardation effect was not strong enough to completely inhibit polymerization. On the other hand, if the size agents were removed, the modulus of the composites decreased significantly, indicating that removing sizing agent decreased the fibermatrix adhesion. This UV-cured epoxidized vegetable oil-based has almost infinite shelf life when it is not exposed to light, and this resin system does not require refrigeration.

Manthey et al. prepared composites based on epoxidized hemp oil (EHO) [30]. Unlike soybean oil, which is the most commonly used vegetable oil in polymer science research, hemp oil is not used in food production. Using hemp oil to produce polymeric materials can eliminate ethical issues. In this study. EHO or ESO is mixed with a petroleum-based epoxy resin, and the properties of EHO-based thermoset and ESO-based thermoset were compared. The impact resistance of the petroleum epoxy can be increased by adding EHO or ESO. The incorporation of fatty acid chain increased the flexibility of the crosslink network, resulting in an increase in impact resistance. Adding ESO to petroleum-based epoxy resulted in a more pronounced increase in impact resistance compared to adding EHO. This is due to the lower number of crosslinking sites of ESO when compared to EHO and the petroleum-based epoxy. On the other hand, adding EHO or ESO to the petroleum-based epoxy resulted in a decrease in ILSS due to the increase in interfacial strength. The flexural testing data indicated that EHObased thermoset possessed higher flexural strength than the ESO-based thermoset. The same trend was observed for jute fiber mat-reinforced composite samples. Interestingly, the flexural strength of the biocomposites was lower than the flexural strength of the neat resin, and this may be a result of low fiber-matrix adhesion. Water absorption data showed that EHO-based resin and the composite have marginally lower water absorption than ESO-based resin and composites. The author concluded that the EHO can compete with the commercially produced ESO in terms of mechanical properties and water-absorption characteristics when using as the matrix material for jute fibers.

10.4 Fiber-Reinforced Plant Oil-Based Polyurethane Composites

Polyurethane (PU) composites were produced using soybean oil-derived polyol reinforced by glass fiber [31]. Soybean phosphate ester polyol (SEPEP) was combined with polymeric 4,4'-disphenylmethane dissocynate (pMDI) to produce PU. Increasing the glass fiber content increases the storage modulus and the glass transition temperature of the composite. Flexural testing and tensile testing both indicate that the strength and modulus of the PU can be increased when more glass fibers were added. Moreover, the impact strength was enhanced when more glass fiber was added, indicating that the incorporation of glass fiber introduces a new energy dissipation mechanism. SEM image showed that a good distribution of glass fiber, good wetting, and good interfacial adhesion can be observed in the composites with soybean oil-based PU as the matrix material.

Husic et al. compared the properties of the resins and the glass-reinforced composites prepared using soybean oil-based PU or petroleum-based PU as the matrix [32]. The soybean oil-based polyol used in this study has a hydroxyl number of 204 mg KOH/g, while the hydroxyl number of the petroleum-based polyol (polypropylene oxidebased triol) is 650 mg KOH/g. The soybean oil-based PU composites possessed similar flexural strength, tensile strength, and ILSS when compared to the composites prepared by the petroleum-based PU, even though the petroleum-based showed marginally better strength as shown in Figure 10.6. The flexural modulus and Young's modulus were lower for the soybean oil-based PU composite due to lower crosslinking density and the presence of dangling fatty acid chains. For the same reason, the glass transition temperatures of the soybean oil-based PU and composite were lower than those of petroleumbased PU and composite.

The fracture toughness of castor oil-derived PU/natural fibers composites was extensively studied by Silva et al. [33]. Sisal short fibers, coconut short fibers, and woven sisal fabric were used in their original states or after chemical treatment with sodium hydroxide. The alkaline treatment using sodium hydroxide decreased the fracture toughness of the sisal fiber composites due to the increase in interfacial adhesion between the fiber and the matrix. The increase in interfacial adhesion reduced the chance of debonding and fiber pull-out, which is



Figure 10.6 Flexural, tensile and interlaminar shear strength (ILSS), modulus of the soy and Jeffol composites [32].

the main energy absorption mechanism. In contrast, treating coconut fiber using sodium hydroxide was beneficial to the coconut fiber composites. The increase in fracture toughness could be ascribed to the fibrillation process after alkaline treatment. The fibrillation process promoted an easy pathway for crack propagation and increased the friction between split fibers under loading. The fracture toughness of the sisal fabric composites was higher than those composites containing sisal short fiber and coconut short fibers. The high fracture toughness is due to the contraction of yarns in the interlacement points of fabric and the friction and shear among the fibers. Even though alkaline treatment can enhance the fiber-matrix interfacial adhesion, resulting in the improvement of mechanical strength; however, at the same time a strong interfacial adhesion can decrease the fracture toughness, damping, and fatigue resistance [34].

Milanese et al. compared the flexural properties of the sisal/castor oil-based PU composites to sisal/phenolic composites [35]. Sisal fibers were used in its woven form. The effects of thermal treatment at 60°C for 72 h on the final mechanical properties of the composites were also studied. The thermal treatment can increase the adhesion between fiber and matrix and reduce the water absorption of the sisal fibers [36]. 44 wt% of sisal fibers was added to the sisal/PU resin and 33 wt% of sisal fibers was added to the sisal/phenolic resin. The addition of sisal fibers can increase the flexural strength of both the resin system, but the phenolic resin possessed much higher mechanical properties compared to the biobased PU. After thermal treatment of the sisal fibers, the flexural strength of the sisal/PU composites increased from 1.6 MPa to 3.7 MPa, while the flexural strength of the sisal/phenolic composites increased from 10.7 MPa to 11.2 MPa. The flexural modulus also increased if sisal fibers were thermally treated.

Polyester nonwoven fabric was used to reinforce castor oil-derived PU [37]. A significant increase in mechanical properties was observed after adding the polyester nonwoven fabric to the castor oil-derived PU. Two different isocyantes were used to react with castor oil-based polyol to produce PU, namely, toluene-2,4-diisocyanate (TDI) and hexamethylene diisocyanate (HMDI). The tensile strength increased about four times and seven times for PU using TDI and HMDI as isocyanate, respectively. Moreover, the tensile strength increased about nine times and seven times for PU made using TDI and HDMI, respectively. The increase in tensile properties indicates the good interfacial adhesion between the matrix and the fiber. The HMDI-based PU showed better chemical resistance compared to the TDI-based PU. The water resistance is very similar for both types of PU.

Composites produced from castor oilderived PU and alfa stems fiber were studied [38]. The PU was prepared from directly using castor oil and HDI. The fibers were produced from using potassium hydroxide to extract the cellulose in the alfa plant. The composites were prepared by adding the alfa stem fibers to the PU prepolymer in acetone. The glass transition temperature of the composites increased as the percentage of fiber increased. The increase in glass transition temperature is due to the interaction between the fiber and the matrix due to hydrogen bonding and/or covalent bonding. The low thermal stability of the natural fibers made the thermal stability of the overall composites decrease with increasing fiber loading. The Young's modulus increased with increasing fiber loading up to 20 wt% fiber. The modulus of the composites started to increase with fiber loading above 20 wt% due to low fiber-matrix adhesion. The composites presented in this chapter possessed low mechanical properties even with the alfa stem fiber reinforcement. The maximum Young's modulus and tensile strength of this composite system were 110 and 3.17 MPa, respectively. Since the author reacted the castor oil directly with HDI without any comonomers or other chemical modifications, the mechanical properties of the PU should be low due to the low number of hydroxyl groups in castor oil.

Manjula et al. used silk fibers as the reinforcement for a chain-extended castor oilderived PU to produce biocomposites [39]. The chain-extended PU was synthesized by reacting castor oil with diisocyante, followed by adding glutaric acid as a chain extender as shown in Figure 10.7. Five to ten percent of silk fiber was added to produce composites. The purpose of adding chain extender is to increase the hydrogen bonding in the polymer. The addition of 10% fibers can increase the tensile strength and the tensile modulus of the composites around 100%. The surface hardness can also be increased with the incorporation of silk fibers. DMA test confirmed that the glass transition temperature of the composites increased with higher fiber loading. The increase in mechanical properties and glass transition temperature is due to the good interfacial adhesion between the matrix and the reinforcement.

PU composites prepared using palm oilderived polyol and oil palm trunk (OPT) fiber were studied [40]. The PU was synthesized by reacting palm oil-oleic acid and glycerol to produce a polyol, and then the polyol is reacted with 4,4-diphenylmethane diisocyante (MDI) to produce PU matrix. Ten to forty percent of polycaprolactone (PCL) was also added as a compactabilizer of the fiber



Figure 10.7 (a) Chain extended PU and (b) hydrogen bonding in chain extended PU [39].

and the PU matrix. The tensile strength and tensile modulus of the composites increased with increasing amount of PCL. SEM micrographs indicated that an improvement of interfacial adhesion was observed when more PCL was added. The addition of PCL can also increase the thermal stability of the composites. Moreover, PCL can also increase the glass transition temperature of the composites.

Rubber seed oil (RSO) was also used to produce biobased PU and composites [41]. Rubber seed oil contains about 77% of unsaturated fatty acids. RSO was saponified into monoglyceride by reacting RSO with glycerol at higher temperature in the presence of calcium oxide catalyst. Afterward, TDI was added to produce PU. Biocomposites were prepared by adding 10-30% of unidirectional long sisal fibers to the matrix. As expected, the tensile and flexural strength and moduli were enhanced with the addition of sisal fibers. The tensile strength of the neat PU polymer was 4.8 MPa. After 15% of reinforced was added, the tensile strength increased remarkably to 57.4 MPa. However, when the fiber content was at 30 wt%, the flexural modulus and flexural strength of the composites at this fiber loading were lower than the composites with 25 wt% fiber. The decrease of flexural properties at high fiber loading is due to fiber agglomeration and the formation of microvoids. The thermal stability of the composites was lower than that of the neat RSO-derived PU but higher than that of neat sisal fibers. In addition, the effects of water absorption on the mechanical properties of the composites were also investigated. Since natural fibers are hydrophilic, when more sisal fibers were added, the water absorption of the composites increased. A light loss in tensile strength and flexural strength was observed. On the other hand, enhancement in flexural modulus was observed after water immersion. The maximum tensile elongation was increased after water immersion. The increase in elongation is due to the fact that water molecules act as a plasticizer agent in the composites. Water molecule can fill the cavities and the cracks within the composites, leading to an increase in the flexural modulus [42].

The influences of alkaline treatment on short random banana fibers and the length of fibers on the properties of banana fibers/ castor oil-derived PU were investigated [43]. After surface treatment, the tensile strength of the fiber was higher than those of the untreated banana fibers. The untreated banana fibers possess a tensile strength and tensile modulus of 162 MPa and 8.5 GPa, respectively. After alkaline treatment, the tensile strength and modulus increased to 269 MPa; however, the Young's modulus decreased to 7.1 GPa. The maximum elongation of the fiber doubled after alkaline treatment. The increase in tensile strength of the fibers after treatment is due to the removal of lignin and hemicelluloses, which allows the rearrangement of fibrils along the direction of tensile load. The IFSS increased dramatically after alkaline treatment. When the length of the embedded fiber increased, the IFSS was also improved. The tensile strength and Young's modulus also increased when treated fiber and longer fiber were used. The DMA test also showed the glass transition temperature shifted to higher temperature with the addition of banana fibers. The enhancements in mechanical properties and glass transition temperature are due to higher interfacial adhesion between the fibers and the matrix.

Merlini et al. developed electric conductive composites for pressure-sensitive applications by using polyaniline-coated coconut fibers as the reinforcement and castor oilderived PU as the matrix [44]. The polyaniline coated coconut fibers (CF-PANI) were prepared by *in situ* oxidative polymerization. The reaction was carried out using hydrochloric acid as a solution and iron (III) chloride hexahydrate (FeCl₃) or ammonium persulfate (APS) as the oxidant. The composites were prepared by using compression molding. The electrical conductivity of the coconut fibers increased from 3.13×10^{-10} to 1.5×10^{-1} and 1.9×10^{-2} when using FeCl₃ and APS as the oxidant, respectively. The polyanilinecoated fibers have lower conductivity compared to polyaniline itself. The PU composites containing 20–25 wt% of CF-PANI with FeCl₃ as an oxidant showed a variation in electrical conductivity when a compress load was applied as shown in Figure 10.8. This property can be used to develop conducting polymer composites for pressure sensor applications.



Figure 10.8 Electrical resistivity as a function of compressive loading–unloading stress cycles applied to PU/CF-PANI.FeCl₃ composites with CF-PANI.FeCl₃ contents of (a) 20 wt% and (b) 25 wt% [44].

10.5 Fiber-Reinforced Plant Oil-Based Foam Composites

Wu et al. prepared biobased foam composite using AESO as matrix and short sisal fibers as reinforcement [45]. This biobased foam composite can find their applications in car interior and acoustic insulation applications. The sisal fibers were chemically treated using alkali or silane coupling agent. AESO with 20 wt% of ST was used as the foam matrix. The foam was prepared by using a water-blown method with NaHCO₂ as the blowing agent. The addition of sisal fibers can increase the compressive modulus and the compressive strength of the composites. The compressive properties of these composites were also compared with a commercially available petroleum-based unsaturated polyester foam, and the SEM image showed the sisal fibers were embedded in the cell walls of the foam matrix, which resulted in good mechanical properties. Both alkali treatment and silane treatment led to increase in interfacial adhesion. The author also conducted a biodegradation test by burying the composites under soil. After 3 months, 12-20% of weight loss was observed depending on the fiber loading. Higher sisal fiber loading led to acceleration of the biodegradation rate. This study indicated that AESO-based foam reinforced with sisal fibers can replace the petroleum-based unsaturated polyester foam in terms of mechanical properties. Moreover, the biobased foam possesses biodegradability while the unsaturated polyester does not.

Zhu et al. fabricated soybean oil-derived rigid polyurethane foams composites reinforced by cellulose microfibers and nanoclays [46]. The foam was prepared using a water-blown method. SEM and X-ray study revealed that the addition of microfibers and nanoclays can modify the cellular structure of the foams. The average cell size was decreased and the cell size distribution was narrowed by the addition of microfibers and nanoclays as shown in Figure 10.9. The incorporation of cellulose nanofibers made the composite more thermally stable. The addition of cellulose nanofibers and nanoclays also increased the mechanical properties of the foam.

Aranguren et al. prepared microfoams based on castor oil-derived PU and vegetable fibers (see Figure 10.10) such as pine wood fibers and hemp fibers [47]. PU foams reinforced with hemp fibers showed better mechanical properties due to its larger aspect ratio. For pine wood fibers, the flexural properties decreased as the length of the fiber increased. The glass transition temperature was also increased by the addition of fibers. In addition, this PU foam contains no catalyst; as a result, these materials will suffer aging due to the secondary reaction of the unsaturation presented in the castor oilbased polyol. Since aging is the crosslinking reaction, the modulus and strength can be improved after aging.

Gu et al. investigated the effects of wood fiber and microclay on the properties of soybean oil-derived PU foams [48]. It was found that the NCO index greatly influenced the mechanical and thermal properties of the PU foams. Different NCO indexes will produce polyurethane with different rigidity. The author of this chapter produced three different types of PU foams by altering the NCO index: flexible foams, semirigid foams, and rigid foams. Both the wood fiber and microclay can increase the compressive strength of the semirigid to rigid foams. For semirigid and rigid foams, wood fibers can produce PU foams with higher compressive properties when compared to the microclay reinforced



Figure 10.9 SEM micrographs (parallel to the blowing direction of foam): (a) neat foam; (b) foam with microfibers; (c) foam with Cloisite Na+; (d) foam with Cloisite 93A; (e) foam with Cloisite 30B [46].



Figure 10.10 Images of reinforcing fibers (a) pine wood fibers, (b) hemp fibers [47].

foams. The tensile strength was degraded for the rigid foams with the addition of the wood fiber or microclay due to the formation of rigid polyurea, which deteriorates the tensile strength. For the flexible foam, the addition of wood fiber or microclay decreased both the tensile and compressive properties. The author also proposed that wood fibers and microclay exhibited different reinforcement mechanisms. Wood fiber tends to form chemical bonds during foaming, while microclay tends to form physical insertion.

10.6 Conclusion

In recent years, many plant oil-based polymers with a wide range of properties have been developed. Vegetable oils are attractive starting materials for preparing polymers because they are relatively inexpensive, commercially available, and most of them are not toxic to humans. Vegetable oils also contain many active functional groups, such as double bonds and hydroxyl groups, which can be chemically modified to produce polymers with desired properties. By using different chemical modification methods and different polymerization methods, plant oil-based vinyl polymers, plant oil-based epoxies, and plant oil-based polyurethane and polyurethane foams, both synthetic fibers and natural fibers have been incorporated into different kinds of plant oil-based polymer to produce composites. Glass fibers are the most commonly used reinforcement in the area of polymer science. Synthetic fibers such as glass fibers have high mechanical properties compared to natural fibers; however, synthetic fibers are relatively expensive and not biorenewable. On the other hand, natural fibers such as hemp fibers, jute fibers, sisal fibers, and cellulose fibers are relatively cheap and biorenewable, but they have drawbacks such as low mechanical properties and being hydrophilic. The addition of fiber reinforcement can generally increase the strength and modulus of the neat matrix. Depending on the level of interfacial adhesion, an increase in glass transition can also be observed in some cases.

When plant oil-based vinyl polymers were used as matrix, acrylate version of

soybean oil, such as acrylated epoxidized soybean oil, is most commonly used as the plant oil component of the matrix due to its high availability, high mechanical properties, and high reaction rate during free radical polymerization. Due to the high viscosity of AESO, reactive diluent such as styrene is often added to AESO to make it more processable. Conjugated vegetable oils can be obtained by using Rhodium-based catalyst, and the conjugated version of vegetable oil has been reinforced by glass fibers and many other natural fibers to produce composites using free radical polymerization or cationic polymerization method. Ring-opening metathesis polymerization (ROMP) has been used to produce matrix with good mechanical properties. When using plant oil-based epoxy resin as the matrix, the epoxidized vegetable oil is always blended with petroleum-based epoxy resin to produce materials with acceptable mechanical properties. Normally, the percentage of the plant oil-based component is below 40%.

Surface modifications on the fibers can also influence the properties of the composites significantly. Alkaline treatment and silane treatment are the most important ones. Novel modification methods such as protein coating and peroxide treatment have also been developed. Surface modifications on fibers can increase the interfacial adhesion between the fibers and the matrix, resulting in composites with higher mechanical and thermal properties. Besides the surface modifications, the length, architecture, and the order of the fiber reinforcements also have great influence on the properties of the composites.

Almost all of the studies in vegetable oilbased composites involve the use of petroleum-based comonomers, so the composites matrix is not completely biobased. In the future, composites with increasing biorenewable contents and higher mechanical properties should be produced.

References

- C.K. Williams, M.A. Hillmyer, Polymers from renewable resources: a perspective for a special issue of polymer reviews, Polym. Rev. 48 (1) (2008) 1–10.
- [2] D. Akesson, M. Skrifvars, P. Walkenstrom, Preparation of thermoset composites from natural fibres and acrylate modified soybean oil resins, J. Appl. Polym. Sci. 114 (4) (2009) 2502–2508.
- [3] K. Adekunle, D. Akesson, M. Skrifvars, Biobased composites prepared by compression molding with a novel thermoset resin from soybean oil and a natural-fiber reinforcement, J. Appl. Polym. Sci. 116 (3) (2010) 1759–1765.
- [4] K. Adekunle, S.W. Cho, C. Patzelt, T. Blomfeldt, M. Skrifvars, Impact and flexural properties of flax fabrics and Lyocell fiber-reinforced bio-based thermoset, J. Reinf. Plast. Comp. 30 (8) (2011) 685–697.
- [5] K. Adekunle, C. Patzelt, A. Kalantar, M. Skrifvars, Mechanical and viscoelastic properties of soybean oil thermoset reinforced with jute fabrics and carded Lyocell fiber, J. Appl. Polym. Sci. 122 (5) (2011) 2855–2863.
- [6] S.N. Khot, J.J. Lascala, E. Can, S.S. Morye, G.I. Williams, G.R. Palmese, et al. Development and application of triglyceride-based polymers and composites, J. Appl. Polym. Sci. 82 (3) (2001) 703–723.
- [7] S.S. Morye, R.P. Wool, Mechanical properties of glass/flax hybrid composites based on a novel modified soybean

oil matrix material, Polym. Compos. 26 (4) (2005) 407–416.

- [8] A. O'donnell, M. Dweib, R. Wool, Natural fiber composites with plant oilbased resin, Compos. Sci. Technol. 64 (9) (2004) 1135–1145.
- [9] C.K. Hong, R.P. Wool, Development of a bio-based composite material from soybean oil and keratin fibers, J. Appl. Polym. Sci. 95 (6) (2005) 1524–1538.
- [10] K.Y. Lee, K.K.C. Ho, K. Schlufter, A. Bismarck, Hierarchical composites reinforced with robust short sisal fibre preforms utilising bacterial cellulose as binder, Compos. Sci. Technol. 72 (13) (2012) 1479–1486.
- [11] T.S. Lee, H.Y. Choi, H.N. Choi, K.Y. Lee, S.H. Kim, S.G. Lee, et al. Effect of surface treatment of ramie fiber on the interfacial adhesion of ramie/acetylated epoxidized soybean oil (AESO) green composite, J. Adhes. Sci. Technol. 27 (12) (2013) 1335–1347.
- [12] E. Mistri, S. Routh, D. Ray, S. Sahoo, M. Misra, Green composites from maleated castor oil and jute fibres, Ind. Crop Prod. 34 (1) (2011) 900–906.
- [13] H.Y. Cui, M.R. Kessler, Glass fiber reinforced ROMP-based bio-renewable polymers: enhancement of the interface with silane coupling agents, Compos. Sci. Technol. 72 (11) (2012) 1264–1272.
- [14] H.Y. Cui, M.R. Kessler, Pultruded glass fiber/bio-based polymer: interface tailoring with silane coupling agent, Compos. Part A-Appl. Sci. 65 (2014) 83–90.
- [15] Y.S. Lu, R.C. Larock, Corn oil-based composites reinforced with continuous glass fibers: fabrication and properties, J. Appl. Polym. Sci. 102 (4) (2006) 3345–3353.
- [16] Y.S. Lu, R.C. Larock, Fabrication, morphology and properties of soybean

oil-based composites reinforced with continuous glass fibers, Macromol. Mater. Eng. 292 (10–11) (2007) 1085–1094.

- [17] R.L. Quirino, R.C. Larock, Synthesis and properties of soy hull-reinforced biocomposites from conjugated soybean oil, J. Appl. Polym. Sci. 112 (4) (2009) 2033–2043.
- [18] D.P. Pfister, R.C. Larock, Green composites from a conjugated linseed oilbased resin and wheat straw, Compos. Part a-Appl. Sci. 41 (9) (2010) 1279– 1288.
- [19] D.P. Pfister, R.C. Larock, Thermophysical properties of conjugated soybean oil/corn stover biocomposites, Bioresource Technol. 101 (15) (2010) 6200–6206.
- [20] D.P. Pfister, R.C. Larock, Cationically cured natural oil-based green composites: effect of the natural oil and the agricultural fiber, J. Appl. Polym. Sci. 123 (3) (2012) 1392–1400.
- [21] R.L. Quirino, J. Woodford, R.C. Larock, Soybean and linseed oil-based composites reinforced with wood flour and wood fibers, J. Appl. Polym. Sci. 124 (2) (2012) 1520–1528.
- [22] D.P. Pfister, R.C. Larock, Green composites using switchgrass as a reinforcement for a conjugated linseed oilbased resin, J. Appl. Polym. Sci. 127 (3) (2013) 1921–1928.
- [23] R.C. Larock, X.Y. Dong, S. Chung, C.K. Reddy, L.E. Ehlers, Preparation of conjugated soybean oil and other natural oils and fatty acids by homogeneous transition metal catalysis, J. Am. Oil Chem. Soc. 78 (5) (2001) 447–453.
- [24] K. Chandrashekhara, S. Sundararaman, V. Flanigan, S. Kapila, Affordable composites using renewable materials, Mater. Sci. Eng. a-Struct. 412 (1–2) (2005) 2–6.

- [25] N. Boquillon, Use of an epoxidized oil-based resin as matrix in vegetable fibers-reinforced composites, J. Appl. Polym. Sci. 101 (6) (2006) 4037–4043.
- [26] Z.S. Liu, S.Z. Erhan, D.E. Akin, F.E. Barton, "Green" composites from renewable resources: preparation of epoxidized soybean oil and flax fiber composites, J. Agric. Food Chem. 54 (6) (2006) 2134–2137.
- [27] Z.S. Liu, S.Z. Erhan, D.E. Akin, F.E. Barton, C. Onwulata, T.A. McKeon, Modified flax fibers reinforced soybased composites: mechanical properties and water absorption behavior, Compos. Interface 15 (2-3) (2008) 207–220.
- [28] T. Takahashi, K.I. Hirayarna, N. Teramoto, M. Shibata, Biocomposites composed of epoxidized soybean oil cured with terpene-based acid anhydride and cellulose fibers, J. Appl. Polym. Sci. 108 (3) (2008) 1596–1602.
- [29] J.V. Crivello, R. Narayan, S.S. Sternstein, Fabrication and mechanical characterization of glass fiber reinforced UV-cured composites from epoxidized vegetable oils, J. Appl. Polym. Sci. 64 (11) (1997) 2073–2087.
- [30] N.W. Manthey, F. Cardona, G. Francucci, T. Aravinthan, Thermo-mechanical properties of epoxidized hemp oil-based bioresins and biocomposites, J. Reinf. Plast. Comp. 32 (19) (2013) 1444–1456.
- [31] J.P.L. Dwan'isa, A.K. Mohanty, M. Misra, L.T. Drzal, M. Kazemizadeh, Biobased polyurethane and its composite with glass fiber, J. Mater. Sci. 39 (6) (2004) 2081–2087.
- [32] S. Husic, I. Javni, Z.S. Petrovic, Thermal and mechanical properties of glass reinforced soy-based polyurethane composites, Compos. Sci. Technol. 65 (1) (2005) 19–25.

- [33] R.V. Silva, D. Spinelli, W.W. Bose, S.C. Neto, G.O. Chierice, J.R. Tarpani, Fracture toughness of natural fibers/ castor oil polyurethane composites, Compos. Sci. Technol. 66 (10) (2006) 1328–1335.
- [34] D. Ray, B.K. Sarkar, N.R. Bose, Impact fatigue behaviour of vinylester resin matrix composites reinforced with alkali treated jute fibres, Compos. Part A: Appl. Sci. Manuf. 33 (2) (2002) 233–241.
- [35] A.C. Milanese, M.O.H. Cioffi, H.J.C. Voorwald, Flexural behavior of sisal/ castor oil-based polyurethane and sisal/ phenolic composites, Mater. Res. 15 (2) (2012) 191–197.
- [36] Y. Li, Y.W. Mai, L. Ye, Sisal fibre and its composites: a review of recent developments, Compos. Sci. Technol. 60 (11) (2000) 2037–2055.
- [37] M.N.S. Kumar, K.S. Manjula, Siddaramaiah. Castor oil-based polyurethane– polyester nonwoven fabric composites: mechanical properties, chemical resistance, and water sorption behavior at different temperatures, J. Appl. Polym. Sci. 105 (6) (2007) 3153–3161.
- [38] E. Maafi, L. Tighzert, F. Malek, Elaboration and characterization of composites of castor oil-based polyurethane and fibers from alfa stems, J. Appl. Polym. Sci. 118 (2) (2010) 902–909.
- [39] K.S. Manjula, M.N.S. Kumar, B.G. Soare, P. Picciani, Siddaramaiah. Biobased chain extended polyurethane and its composites with silk fiber, Polym. Eng. Sci. 50 (4) (2010) 851–856.
- [40] Z. Yaakob, A.M. Min, M.M. Hilmi, H. Zaman, M. Khairul, S. Kamarudin, Effect of compactabilization of polymer on the properties of polyurethane-palm fiber composites, J. Polym. Eng. 29 (8–9) (2009) 503–520.

- [41] I.O. Bakare, F.E. Okieimen, C. Pavithran, H.P.S.A. Khalil, M. Brahmakumar, Mechanical and thermal properties of sisal fiber-reinforced rubber seed oil-based polyurethane composites, Mater. Des. 31 (9) (2010) 4274–4280.
- [42] H. Dhakal, Z. Zhang, M. Richardson, Effect of water absorption on the mechanical properties of hemp fibre reinforced unsaturated polyester composites, Compos. Sci. Technol. 67 (7) (2007) 1674–1683.
- [43] C. Merlini, V. Soldi, G.M.O. Barra, Influence of fiber surface treatment and length on physico-chemical properties of short random banana fiber-reinforced castor oil polyurethane composites, Polym. Test 30 (8) (2011) 833–840.
- [44] C. Merlini, G.M.O. Barra, D.P. Schmitz, S.D.A.S. Ramoa, A. Silveira, T.M. Araujo, et al. Polyaniline-coated coconut fibers: structure, properties and their use as conductive additives in matrix of polyurethane derived from castor oil, Polym. Test 38 (2014) 18–25.
- [45] S.P. Wu, J.F. Qiu, M.Z. Rong, M.Q. Zhang, L.Y. Zhang, Plant oil-based biofoam composites with balanced performance, Polym. Int. 58 (4) (2009) 403–411.
- [46] M. Zhu, S. Bandyopadhyay-Ghosh, M. Khazabi, H. Cai, C. Correa, M. Sain, Reinforcement of soy polyol-based rigid polyurethane foams by cellulose microfibers and nanoclays, J. Appl. Polym. Sci. 124 (6) (2012) 4702–4710.
- [47] M.I. Aranguren, I. Racz, N.E. Marcovich, Microfoams based on castor oil polyurethanes and vegetable fibers, J. Appl. Polym. Sci. 105 (5) (2007) 2791–2800.
- [48] R.J. Gu, M.M. Sain, Effects of wood fiber and microclay on the performance of soy based polyurethane foams, J. Polym. Environ. 21 (1) (2013) 30–38.

Page left intentionally blank

11 Application of Plant Oil-Based Products in Structural Health Monitoring

Simon Laflamme^{1,2} and Samy A. Madbouly^{3,4}

¹Department of Civil, Construction, and Environmental Engineering, Iowa State University, Ames, IA, USA ²Department of Electrical and Computer Engineering, Iowa State University, Ames, IA, USA ³Department of Materials Science and Engineering, Iowa State University, Ames, IA, USA ⁴Department of Chemistry, Faculty of Science, Cairo University, Orman-Giza, Egypt

Ο U Τ L Ι Ν Ε	
11.1 Introduction	191
11.2 Background	194
11.2.1 Materials	194
11.2.2 Sensor Fabrication	194
11.2.3 Sensing Principle	196
11.3 Materials Properties	197
11.3.1 Dispersion of Nanofiller	197
11.3.2 Dielectric Properties	198
11.3.3 Thermogravimetric Analysis (TGA)	199
11.4 Laboratory Experiments	
11.4.1 Experimental Setup	199
11.4.2 Results and Discussion	201
11.5 Conclusion	203
References	205

11.1 Introduction

Structural health monitoring (SHM) is the automated process of damage diagnosis, localization, and prognosis, which leads to structural condition assessment. Such automation is typically provided by a combination of sensors, data acquisition systems, and signal processing algorithms, and is meant to replace visual inspections, which are ineffective and judgement-dependent, and nondestructive evaluation techniques [1,2], which are expensive and require highly trained inspectors [3]. There is a particular challenge in SHM associated with condition assessment of large-scale structures, including energy systems (e.g., wind turbine blades), aerospace structures (e.g., aircrafts), and civil infrastructures (e.g., bridges), due to the lack of scalability of existing sensing solutions. Off-the-shelf sensors, for instance accelerometers and strain gauges, are expensive to deploy over large geometries and/or do not provide a clear link between the signal and the condition of a structure.

With the recent technological developments in conducting polymers [4], it is now possible to fabricate and deploy skin-type sensors. Such a sensing strategy has the potential to cover very large surfaces and detect local damages over a large surface by mimicking the biological function of skin. The field of SHM terms this technology as large area electronic (LAE).

Popular applications in SHM include the utilization of carbon nanotube (CNT) nanocomposites to create resistive strain sensors [5,6]. CNTs are typically used due to their strength and super-elastic attribute [7], but their utilization results in high fabrication costs and difficult scalability. Capacitancebased strain sensors have been proposed for various applications including strain [8,9], pressure [10], triaxial force [11], and humidity [12,13] sensing.

The authors have also developed an LAEbased sensor, termed soft elastomeric capacitor (SEC). The SEC is a strain gauge, and its measurement principle based on a measurable change in capacitance provoked by a change in its geometry. The SEC was originally constituted from a styrene–ethylene/butylene–styrene (SEBS) matrix filled with titania (TiO₂) particles to form the dielectric, sandwiched between compliant electrodes fabricated from the same



Figure 11.1 A single SEC.

SEBS, but filled with CB to provide conductivity. The fabrication method of this sensor is described in [14]. Figure 11.1 is a picture of a typical SEC measuring $75 \times 75 \text{ mm}^2$ (3 × 3 in²). The sensors' quasistatic behavior [15], dynamic behavior [16], and nanocomposite optimization [17] have been studied.

The SEC discussed previously showed promising electrical and mechanical behavior, and its fabrication process and inexpensive nanocomposite enable scalability for large-scale deployment. However, like the vast majority of LAE, the SEC is also fabricated using a petroleum-based polymer [18]. The volatility of the natural oil markets and the increasing cost of petroleum have led to a push to reduce the dependence on petroleum products. Together with an increase in environmental awareness, this has promoted the use of alternative, biorenewable, environmentally friendly products, such as biomass. The growing interest in replacing petroleumbased products by inexpensive, renewable, natural materials is important for sustainable development into the future and will have a

significant impact on the polymer industry and the environment, in particular, for any large-scale deployments.

The authors have studied the opportunity to fabricate a biobased LAE instead of petroleum-based. Plant oils represent one of the most promising renewable resources that can be used to create and develop a new class of bioplastics with tailored multifunctional properties due to their unique chemical structure. These natural oils exhibit a triglyceride structure linked with different fatty acids that can be functionalized/modified with more readily polymerizable groups to synthesize biobased copolymer thermosets with outstanding mechanical properties. These copolymer thermosets also offer sustainability, low cost, and are environmentally friendly, biocompatible, and potentially biodegradable. The plant oil can be also chemically converted to polyols to synthesis plant oil-based polyurethane (PU). With the exception of castor oil, nearly all vegetable oils do not naturally have the hydroxyl groups necessary to produce PU, but the carboncarbon double bonds and the ester functionality present in triglycerides allow for the introduction of such groups. A majority of the vegetable oil-based polyols produced take advantage of the carbon-carbon double bonds to incorporate hydroxyl groups. The plant oil-based PU materials can be flexible and rigid, soft elastomers, or ductile and hard plastics, depending on the different types and concentrations of plant oil-based polyols. Plant oil-based polyurethanes have been studied to a much lesser extent in the literature than petroleum-based PUDs.

Vegetable oil-based waterborne polyurethane has recently emerged as a new branch of PU chemistry in an effort to reduce negative impacts on the environment and minimize fabrication costs. This branch has been rapidly growing, driven by the versatility and environmental friendliness of these polyurethanes. Integrating biobased composites with the concept of the SEC is a pioneer application of plant oil-based products in structural health monitoring. It is motivated by the improvement of the sensor's sustainability at both the material and fabrication process levels.

In this chapter, we discuss a biobased LAE for SHM applications presented in [19], developed by the authors as a possible replacement to the petroleum-based SEC aforementioned. The sensor is a biobased version of the SEC. Its dielectric is constituted from a castor oil-based polyurethane (PU) mixed with TiO₂ particles. The choice of castor oil is motivated by the presence of hydroxyl groups necessary to produce polyurethane dispersions (PUDs). In addition, waterborne castor oil-based PUD is used in order to partly eliminate organic solvents in the SEC's fabrication process [20-22]. This significantly reduces the amount toxic volatile organic compounds present in conventional PUs as well as hazardous air pollutants [21]. Unlike solvent-based PUDs, the aqueous PUDs can be applied with high solids content because their viscosity does not depend on the molecular weight of polyurethane. It is important to make the distinction that at this current state of research, only the dielectric is a biobased product. The compliant electrodes are still constituted from an SEBS-carbon black mix because the dispersion of carbon black within a castor-oil substrate requires further investigation.

The rest of this chapter is organized as follows. In the next section, the background of the sensor is described, which includes a description of the materials used, sensor fabrication process, and sensing principle. It is followed by a study of the sensor on a materials perspective, with a discussion on the dispersion of the filler and decay of the dielectric. Subsequently, the sensor's linearity and theoretical gauge factor are experimentally verified. The last section concludes this chapter.

11.2 Background

This section provides the background on the SEC. First, materials used in the fabrication process are discussed. Second, the fabrication process of the sensor is described. Last, the electromechanical model is derived.

11.2.1 Materials

For the dielectric, castor oil, isophorone diisocyanate (IPDI), dimethylol propionic acid (DMPA), and dibutyltin dilaurate (DBTDL) were obtained from Aldrich Chemical Company (Milwaukee, WI). Methyl ethyl ketone (MEK) and triethylamine (TEA) were purchased from Fisher Scientific Company (Fair Lawn, NJ). Titania-type rutile R320 was acquired from Sachtleben Chemie GmbH (Germany). The nanofiller was selected to increase the dielectric permittivity and durability of the nanocomposite [23].

For the electrodes, SEBS was acquired from VTC Elastoteknik AB, Sweden. SEBS is a block copolymer widely used for medical applications because of its purity, softness, elasticity, and strength [24]. CB-type Printex XE-2B (2% ash content and 500 ppm sieve residue 45 μ m) was obtained from Orion Engineered Carbons (Kingswood, TX). It is characterized by a high structure (minimum oil absorption 380 cc/100 g), which facilitates higher conductivity [25]. All materials were used as received without further purification or analysis.

11.2.2 Sensor Fabrication

Details about the synthesis of castor oilbased PUD can be found in our recent publication [22]. Briefly, the castor oil-based PUD are synthesized by a reaction of IPDI, castor oil, and DMPA as internal surfactant. The DMPA incorporates carboxylic functionality in the prepolymer backbone. Tertiary amine (e.g., TEA) is then used to neutralize the carboxylic groups and produce ionic centers to stabilize the polymer particles in water. Scheme 11.1 shows the elementary steps for the synthesis of castor oil-based PUD and its TEM morphology. The PUD exhibited dispersed nano-PU particles with a diameter of approximately 30 nm.

Castor oil-based PU is then mixed with TiO_2 , which both serve as a dielectric of the capacitor. TiO_2 , an environmentally friendly and readily available filler [26], is added and dispersed in the castor oil to increase the permittivity of the PU [27]. The preparation of SEC is finalized by sandwiching the dielectric with two conductive electrodes fabricated from a carbon black and SEBS mix. Figure 11.2 illustrates the composition of a biobased SEC.

The sensor's fabrication process, as shown in Figure 11.3, follows a solution casting method:

- 1. TiO_2 nanoparticles are dissolved using methanol solvent before they are added to the castor oil PU at various volume percentages (5%, 10%, and 15%) and dispersed via sonication using an ultrasonic tip.
- 2. The resulting homogenous solution is drop casted onto a glass plate and dried at room temperature for about 3 days to allow evaporation of water.



Scheme 11.1 Elementary steps for the synthesis of anionic castor oil-based PUD and its TEM nanostructure morphology.



Figure 11.2 A single biobased SEC [19].

- 3. A 15 ml of SEBS/toluene solution is added to 0.79 g of carbon black to create the compliant electrodes. A sonication bath (Branson CPXH 2800) is used to disperse the carbon black particles.
- 4. The carbon black-SEBS solution is sprayed on both surfaces of the dried polymer to form the sensor.



Figure 11.3 Sensor fabrication [19].

5. Two conductive copper tapes are attached to the electrodes during the drying process to create mechanical connections for connecting to the data acquisition system (DAQ).

11.2.3 Sensing Principle

The sensor is assumed to behave as a nonlossy capacitor when sampled at relatively low frequencies (≤ 1 kHz). The capacitance of the sensor is written as

$$C = \frac{e_0 e_r A}{h} \tag{11.1}$$

where A = w.l. is the surface area of the electrodes of width w and length l, h is the thickness of the dielectric, $e_0 = 8.854$ pF/m is the vacuum permittivity, and e_r is the dimensionless relative permittivity of the composite. Assuming small deformation, the derivative of Equation (11.1) leads to the following expression:

$$\Delta C = \left(\frac{\Delta l}{l} + \frac{\Delta w}{w} + \frac{\Delta h}{h}\right)C \qquad (11.2)$$

or

$$\frac{\Delta C}{C} = \varepsilon_x + \varepsilon_y - \varepsilon_z \qquad (11.3)$$

where ε is the strain in the principal axes illustrated in Figure 11.4. The SEC is designed to be adhered to the surface of the monitored structure in the *x*-*y* plane using an epoxy. A surface deployment methodology is described in [28]. Using Hooke's law specialized for in-plane stress, one obtains an expression for ε_z :

$$\varepsilon_{z} = -\frac{v}{1-v}(\varepsilon_{x} + \varepsilon_{y}) \qquad (11.4)$$

where *v* is the Poisson's ratio of the sensor. Substituting Equation (11.4) into Equation (11.3) gives and expression for the gauge factor λ :

$$\frac{\Delta C}{C} = \lambda(\varepsilon_x + \varepsilon_y) = \frac{1}{1 - \nu}(\varepsilon_x + \varepsilon_y) \quad (11.5)$$



Figure 11.4 Sensing principle [19].

Equation (11.5) can be rearranged using Equation (11.1) to obtain an expression for the sensitivity of the sensor *S*:

$$S = \frac{\Delta C}{(\varepsilon_x + \varepsilon_y)} = \frac{\lambda e_0 e_r A}{h}$$
(11.6)

It is, therefore, possible to increase the sensor's sensitivity Equation (11.6) by altering the geometry or the dielectric permittivity. A formulation for uniaxial strain applications can be obtained by specializing Equation (11.5). Assuming that the monitored material (e.g., fiberglass, steel, concrete, aluminum) has a significantly higher stiffness than the nanocomposite itself, the strain along the *y*-axis is written as $\varepsilon_{y} = -v_{m}\varepsilon_{x}$, where v_{m} is the Poisson's ratio of the monitored material. Here, Equation (11.5) becomes

$$\frac{\Delta C}{C} = \lambda \varepsilon_x = \frac{1 - v_{\rm m}}{1 - v} \varepsilon_x \qquad (11.7)$$

11.3 Materials Properties

In this section, key mechanical properties of the nanocomposite are investigated. First, the dispersion of the nanofiller is examined. A homogenous dispersion is critical to ensure linearity of the sensor as a function of strain. Second, the dielectric properties are discussed. Such properties define the sensitivity of the sensor *S*. Last, a thermogravimetric analysis (TGA) is conducted to determine the thermal stability of the composite.

11.3.1 Dispersion of Nanofiller

The quality of dispersion was investigated by conducting a morphological inspection of the castor oil-based PU/TiO, nanocomposites using scanning electron microscopy (SEM). Three samples with different TiO₂ contents were studied. The test methodology consisted of fracturing the samples in liquid nitrogen, fixing them on the SEM holders, and sputtering with gold. The SEM equipment was a field-emission scanning electron microscope (FE-SEM, FEI Quanta 250) operating at 10 kV under high vacuum. Micrographs were taken 3 days after fabrication. Figure 11.5 shows typical SEM micrographs for PU/TiO₂ nanocomposites with 5, 10, and 15 vol% TiO₂ to polymer contents. Results show that the TiO_2 is dispersed well along the horizontal axis, but appears to have settled in the vertical axis. Figure 11.6 shows a blowup on the region where TiO_2 has settled. TiO_2 dispersed well in the PU matrix with an average particle size as small as 200 nm in this region. Given the utilization of the sensor in an in-plane mode, uniform dispersion is only required along the horizontal axis, yet preferable among the entire volume. The section on laboratory experiments will confirm the



Figure 11.5 SEM micrographs for PU/TiO₂ nanocomposites for different compositions [19].



Figure 11.6 SEM micrographs for PU/TiO₂ nanocomposites for different compositions, blow up on dispersed TiO₂ particles [19].

homogeneous in-plane dispersion of the particles by demonstrating the linearity of the sensor and verifying the theoretical gauge factor experimentally. However, the nonuniform settlement along the sensor's thickness needs further investigation. In particular, other dispersion techniques need to be investigated, including the utilization of surfactants and coated TiO_2 .

11.3.2 Dielectric Properties

Experience with the fabrication of castor oil-based SECs led to believe in a possible decay in the materials' dielectric over time. Such decay has never been observed with SEBS-based SECs. The authors have studied the change in the relative permittivity over time by recording the value of e_r over 3 weeks for three different TiO₂ concentration levels (5, 10, and 15 vol%). The methodology consisted of measuring the capacitance of each sample using an Agilent 4263B LCR Meter and back-calculating e_r from Equation (11.1). Three samples we used per TiO_2 concentration level. The average value of the relative permittivity for each set of specimens is plotted in Figure 11.7 (no recording was taken for the 9th and 10th days). The decay of the dielectric value is evident, and stabilizes after approximately 16 days. Possible causes include a slow evaporation of the soluble materials and a reorganization of the nanofiller. All experiments presented later in this chapter were performed on specimens older than 21 days to ensure a stable relative permittivity.



Figure 11.7 Average value of relative permittivity for each set over time [19].

11.3.3 Thermogravimetric Analysis (TGA)

The thermal stability of the PU/TiO, nanocomposite at different TiO₂ concentration levels was investigated using TGA. The equipment consisted of a TA Instruments TGA model 50. Tests were conducted over a temperature range of 30-600°C at 20°C/min under nitrogen atmosphere. Figure 11.8 shows a typical TGA measurement for PU/TiO, nanocomposites for different TiO₂ concentration levels. Results show that the pure PU sample (0 vol% TiO₂) was thermally stable at a temperature range up to 250°C. Approximately 10 wt% of the sample was degraded at approximately 250-300°C due to the evaporation of soluble materials and unreacted oil fragments. With increasing the temperature up to 600°C, two fast degradation processes were observed. The first process occurred over the temperature range of 300-380°C. Approximately 60 wt% of the sample was lost due to the degradation of the polymer backbone. The second process occurred over the temperature range 400-600°C. It was possibly caused by further decomposition



Figure 11.8 TGA measurements for PU/TiO₂ composites at 20°C/min heating rate under nitrogen atmosphere [19].

of the PU fragments. The multiple thermal degradation processes of biobased polymers from vegetable oils are very common in the literature. The addition of TiO_2 significantly increased the thermal stability of the nano-composite, evidenced by a shift to higher temperature ranges. Results from this section showed that the TiO_2 dispersed well along the horizontal axis of the castor oil-based PU, and that it increased both its dielectric permittivity and thermal stability.

11.4 Laboratory Experiments

Laboratory experiments were conducted to verify the sensor linearity and compare its performance with the previous pretroleumbased version of the SEC. First, the experiment setup is described. Second, the results are presented and discussed.

11.4.1 Experimental Setup

The sensor linearity was verified on two different configurations: in a free-standing mode, and on a bending beam. Tests were conducted over the range 0-6% strain. The gauge factor was experimentally computed from the bending beam test and compared against the solution from Equation (11.7).

For the free-standing configuration, three SEC with 5, 10, and 15 vol% TiO_2 concentration levels were clamped separately into an Instron table-top mechanical testing machine (model 5569). Each specimen was subjected to six tensile strain cycles (from 1% to 6% strain with 1% strain increment), with each strain plateau attained at a loading rate of 5 mm/min. This range of strain was governed by the allowable elongation of the testing equipment for the sample sizes, and is

well beyond the failure point of typical structural materials. Data from the SECs were acquired using an off-the-shelf capacitance data acquisition system (ACAM PCap01), and sampled at 95 Hz. Figure 11.9 shows the laboratory setup for the free-standing sensor test. Remark that Equation (11.7) cannot be directly verified in a free-standing configuration because of the nonuniform distribution of strain within the dielectric.

For the bending beam test, the SEC specimens were installed onto the bottom surface of a simply supported aluminum beam of dimensions $910 \times 200 \times 6.5 \text{ mm}^3$ ($36 \times 8 \times 0.25 \text{ in}^3$) subjected to a four-point load setup as shown in Figure 11.10. Each sensor was



Figure 11.9 (a) Tensile test laboratory setup; (b) schematic of the clamped SEC on the Instron machine [19].



Figure 11.10 (a) Bending plate test laboratory setup; (b) schematic from side and under the plate (connection sensors-DAQ only shown for two sensors for clarity) [19].

deployed within the constant moment region using a thin layer of an off-the-shelf epoxy (JB Kwik) after sanding the plate surface and applying a primer. Step loads (approximately 20, 40, 60, and 80 lb.) were applied at 1/3 and 2/3 of the beam length using a hand operated hydraulic test system (Enerpac). Data from the SECs were acquired using an ACAM PCap01. Data from the RSGs were acquired using n Hewlett-Packard 3852 DAQ.

11.4.2 Results and Discussion

Figures 11.11 and 11.12 show the results from the free-standing tests. Figure 11.11

compares time series of strain input and capacitance measurements. Time axes of strain and capacitance were manually aligned given the inaccurate synchronization of both DAQs. The comparison of time series measurements shows an agreement between the experimental values of strain and measured capacitance. There is an upward slope in the capacitance measurements that becomes significant at high levels of strain. This slope is attributed to the viscoelastic behavior of the nanocomposite.

Figure 11.12 is a plot of the normalized change in capacitance versus strain. Normalized measurements provide a better comparison given that Equation (11.7) does not hold in a free-standing configuration. Data are



Figure 11.11 Strain history of SEC versus Instron RSG (free-standing test). (a) 5% TiO_2 content; (b) 10% TiO_2 content; (c) 15% TiO_2 content [19].



Figure 11.12 Verification of linearity for free-standing biobased SECs. (a) $5\% \text{ TiO}_2$ content; (b) $10\% \text{ TiO}_2$ content; (c) $15\% \text{ TiO}_2$ content [19].

fitted linearly using a least-squares estimator. Results show that the sensor remains linear over the range 0–6% strain.

Figures 11.13 and 11.14 show the results from the bending beam test. Comparison of time series data (Figure 11.13) also shows agreement between strain load and measured capacitance. A noticeable feature in the signals is a drift of the signal after each step load. It is hypothesized that this drift is caused by the epoxy interface, which results in a slow relaxation process of the dielectric following a step load. The experimental gauge factor is obtained by plotting $\Delta C/C$ versus strain and taking the slope of the linear fit (Figure 11.14). Assuming a value of $v_m = 0.33$ for aluminum and v = 0.45-0.5 for thermosets, the theoretical gauge factor of the castor oil-based SEC is $1.34 < \lambda < 1.49$ (Equation (11.7)). Table 11.1 summarizes

Table 11.1 Strain Gauge Factors of SECInstalled on an Aluminum Plate

	5%	10%	15%
Gauge factor λ	1.415	1.376	1.463



Figure 11.13 Strain history of SEC versus RSG (bending plate test). (a) 5% TiO_2 content; (b) 10% TiO_2 content; (c) 15% TiO_2 content [19].

the values obtained in Figure 11.14. The experimental gauge factors are all comprised within 1.34 and 1.49. Cross-specimens fluctuation can be explained by slight imperfections in the dispersion of the TiO_2 , and changes in environmental conditions (temperature and humidity).

11.5 Conclusion

In this chapter, a pioneer application of plant oil-based polymer for structural health monitoring was presented. The proposed application is an SEC, constituted from a dielectric made of castor oil-based PU filled with titanium dioxide, and conductive plates made of SEBS filled with carbon black. The application was motivated by the inexpensive and biorenewable characteristics of the castor oil, therefore reducing the environmental footprint of LAEs. The proposed sensor could be deployed in a large network configuration to cover large-scale surfaces, enabling monitoring of local strain over global areas.

The sensor showed to reach stability after approximately 16 days from fabrication.


Figure 11.14 Verification of gauge factor for biobased SEC (bending plate test). (a) $5\% \text{ TiO}_2$ content; (b) $10\% \text{ TiO}_2$ content; (c) $15\% \text{ TiO}_2$ content [19].

Scanning electron microscopy tests showed that all concentrations of TiO_2 (5%, 10%, and 15%) are dispersed well in the horizontal axis, but appear to have settled in the vertical axis. The TGA showed good physical and chemical properties.

Static load tests were conducted on freestanding specimens and on specimens adhered to an aluminum plate subjected to bending. Results from free-standing specimens showed that the sensor remained linear in the range 0–6% strain, and bending tests verified the theoretical gauge factor experimentally. These results confirmed the good dispersion of the particles within the castor oil PU and that the SEC can be used as a strain sensor.

The proposed castor oil-based SEC constitutes a promising sensor for monitoring of mesoscale surfaces. It demonstrates the utilization of biobased polymers in the fabrication of sensors, which can result in important environmental benefits.

Acknowledgment

This work was supported by grant 13-02 from the Iowa Energy Center and grant 1001062565 from the Iowa Alliance for Wind Innovation; their support is gratefully acknowledged.

References

- [1] J. Blitz, S. Geoff, Ultrasonic Methods of Non-Destructive Testing vol. 2, Springer, Berlin, Germany, 1996.
- [2] C.U. Grosse, Acoustic Emission Testing, Springer, Heidelberg, 2008.
- [3] K.L. Rens, T.J. Wipf, F.W. Klaiber, Review of nondestructive evaluation techniques of civil infrastructure, J. Performance Constructed Facilitie 11 (4) (1997) 152–160.
- [4] R. Gangopadhyay, A. De, Conducting polymer nanocomposites: a brief overview, Chem. Mater. 12 (3) (2000) 608– 622.
- [5] L. Gao, et al. Damage monitoring in fiber-reinforced composites under fatigue loading using carbon nanotube networks, Philos. Mag. 90 (31–32) (2010) 4085–4099.
- [6] K.J. Loh, et al. Carbon nanotube sensing skins for spatial strain and impact damage identification, J. Nondestructive Eval. 28 (1) (2009) 9–25.
- [7] I. Kang, et al. A carbon nanotube strain sensor for structural health monitoring, Smart Mater. Struct. 15 (3) (2006) 737.
- [8] K.I. Arshak, D. McDonagh, M.A. Durcan, Development of new capacitive strain sensors based on thick film polymer and cermet technologies, Sens. Actuators A: Phys. 79 (2) (2000) 102–114.
- [9] M. Suster, et al. A high-performance MEMS capacitive strain sensing system, Microelectromech. Syst. J. 15 (5) (2006) 1069–1077.
- [10] D.J. Lipomi, et al. Skin-like pressure and strain sensors based on transparent elastic films of carbon nanotubes, Nat. Nanotechnol. 6 (12) (2011) 788–792.
- [11] J.A. Dobrzynska, M.A.M. Gijs, Polymer-based flexible capacitive sensor

for three-axial force measurements, J. Micromech. Microeng. 23 (1) (2013) 015009.

- [12] P.M. Harrey, et al. Capacitive-type humidity sensors fabricated using the offset lithographic printing process, Sens. Actuators B: Chem. 87 (2) (2002) 226–232.
- [13] H.P. Hong, et al. A highly fast capacitivetype humidity sensor using percolating carbon nanotube films as a porous electrode material, Proc. IEEE Sens. (2012) 1–4.
- [14] S. Laflamme, M. Kollosche, J.J. Connor, G. Kofod, Robust flexible capacitive surface sensor for structural health monitoring applications, J. Eng. Mech. 139 (7) (2012) 879–885.
- [15] S. Laflamme, H.S. Saleem, B.K. Vasan, R.L. Geiger, D. Chen, M.R. Kessler, K. Rajan, Soft elastomeric capacitor network for strain sensing over large surfaces, IEEE/ASME Trans. Mechatron. 18 (6) (2013) 1647–1654.
- [16] S. Laflamme, F. Ubertini, H. Saleem, A. D'Alessandro, A. Downey, H. Ceylan, A.L. Materazzi, Dynamic characterization of a soft elastomeric capacitor for structural health monitoring, J. Struct. Eng. (2014), doi: 10.1061/(ASCE) ST.1943-541X.0001151.
- [17] H. Saleem, M. Thunga, M. Kollosche, M.R. Kessler, S. Laflamme, Interfacial treatment effects on behavior of soft nano-composites for highly stretchable dielectrics, Polymer 55 (17) (2014) 4531–4537.
- [18] Y. Xia, R.C. Larock, Vegetable oilbased polymeric materials: synthesis, properties, and applications, Green Chem. 12 (11) (2010) 1893–1909.
- [19] S. Kharroub, S. Laflamme, S. Madbouly, F. Ubertini, Bio-based soft elastomeric capacitor for structural health

monitoring applications, Struct. Health Monitoring 14 (2) (2014) 158–167.

- [20] Y. Lu, et al. Preparation and properties of starch thermoplastics modified with waterborne polyurethane from renewable resources, Polymer 46 (23) (2005) 9863–9870.
- [21] Y. Lu, C.R. Larock, Soybean-oilbased waterborne polyurethane dispersions: effects of polyol functionality and hard segment content on properties, Biomacromolecules 9 (11) (2008) 3332–3340.
- [22] S.A. Madbouly, X. Ying, R.M. Kessler, Rheological behavior of environmentally friendly castor oil-based waterborne polyurethane dispersions, Macromolecules 46 (11) (2013) 4606– 4616.
- [23] H. Stoyanov, M. Kollosche, S. Risse, D.N. McCarthy, G. Kofod, Elastic block copolymer nanocomposites with controlled interfacial interactions for artificial muscles with direct volt-

age control, Soft Matter 7 (1) (2011) 194–202.

- [24] R. Yoda, Elastomers for biomedical applications, J. Biomater. Sci., Polym. Ed. 9 (6) (1998) 561–626.
- [25] J.C. Huang, Carbon black filled conducting polymers and polymer blends, Adv. Polym. Technol. 21 (4) (2002) 299–313.
- [26] D. Yang, et al. A high-performance dielectric elastomer consisting of biobased polyester elastomer and titanium dioxide powder, J. Appl. Phys. 114 (15) (2013) 154104.
- [27] M. Kollosche, H. Stoyanov, S. Laflamme, G. Kofod, Strongly enhanced sensitivity in elastic capacitive strain sensors, J. Mater. Chem. 21 (23) (2011) 8292–8294.
- [28] W. Zeng, W. Sun, N. Bowler, S. Laflamme, Peel resistance of adhesive joints with elastomer-carbon black composite as surface sensing membranes, Int. J. Adhes. Adhes. 58 (2015) 28–33.

12 The Future of Essential Oils as a Pest Biocontrol Method

Olivier Sparagano¹, James Pritchard², and David George^{3,4}

¹Vice-Chancellor Office, Coventry University, Coventry, UK ²Royal Veterinary College, Hatfield, Hertfordshire, UK ³Northumbria University, Newcastle upon Tyne, UK

⁴Head of Entomology, Stockbridge Technology Centre, North Yorkshire, UK

	OUTLINE	
12.1 Introduction		207
12.2 Range of Use		208
12.3 Mechanisms		208
12.4 Future Research		208
12.5 Conclusion		209
References		209
Further Reading		211

12.1 Introduction

The use of synthetic pesticides has come under significant scrutiny in the last decade, with continued reliance on "conventional" insecticides and acaricides threatened by current or impending legislation being implemented at a range of geographical scales. In the EU numerous plant protection products are at risk of withdrawal under Regulation (EC) No. 1107/2009, the Water Framework Directive, and the Sustainable Use Directive. The combined impact of such legislation, coupled with reviews of substances previously approved, will be the loss of many plant protection products from the sector [1], and a need for more considered use of those that remain.

Even where synthetic pesticides remain in use, their effectiveness may be compromised by development of pest resistance. Resistance has, for example, been observed in lice [2], mites [3,4], and ticks [5] of medical/veterinary significance, and is now widespread among agricultural pests. The peach potato aphid (*Myzus persicae*) is perhaps the most well studied example, being considered one of the world's most strongly resistant pests [6].

To develop alternative methods of pest control research has turned to plant essential oils, many of which display complex and highly pesticidal chemistries [7–9]. Essential oils are generally considered "lower risk" than many synthetics, having short persistence times and typically, though not universally, displaying low vertebrate toxicity [7]. The high volatility of many of these products also lends itself well to the use of essential oils as pest repellents, including against major disease vectors such as mosquitoes [10,11], for example, recently demonstrated the repellent effect of several essential oils to *Aedes aegypti* mosquitoes.

12.2 Range of Use

The potential of essential oils, and/or their chemical components, is beginning to be realized in virtually every sector in which pests pose a threat to health and/or production. In agriculture, the commercial development of botanically based products (along with other "biopesticides") now far exceeds that of synthetic pesticides, with a CAGR of 16% (Markets and Markets, 2014) [12]. Commercial products are also becoming more widely available for use against parasites of companion animals, supported by a wealth of research over the last 20 years (e.g., [13–16]). Evidence supports that they are similarly effective against an array of other pests of veterinary, as well as medical, significance [7,17]. Thymol, the primary chemical component of thyme essential oil, has been found to kill Varroa mites [18,19] as is now widely used in apiculture, as are several essential oilbased feed supplements (e.g., Vita GoldTM and GreenTM).

12.3 Mechanisms

Within the majority of sectors essential oils are primarily researched and developed as pesticides for direct topical or spray application. Their use as mosquito repellents in the medical sector is an exception, with the repellent potential of essential oils also recognized against other veterinary/medical ectoparasites [20–22]. Use or potential of essential oils as pest attractants is rarer, though had been documented (Kraus et al., 1994) [23].

The mode of action of essential oils on arthropods is not always clear, and for many products has yet to be fully studied. In addition to attacking the arthropod cuticle directly, essential oils have been generally described as having toxic, antifeedant, and antioviposition properties, also exerting developmental effects on pests (Koul et al., 2008) [24]. Those studies that have considered biochemical responses have shown essential oils to express their effects through a variety of routes, primarily acting upon GABA, tyramine and octopamine receptors/ synapses, and the inhibition of AChE [7].

A potentially attractive attribute of at least some essential oils is that a single product may target pests in a multitude of ways, thus reducing the possibility of pests developing resistance to them. Thymol, for example, may block octopamine receptors and also act upon the GABA-gated chloride channel [25]. However, essential oils may not be immune to pest resistance [7], especially as arthropods possess a diverse array of detoxification pathways, as well as other resistance "strategies."

12.4 Future Research

Essential oils are often relatively volatile compounds and remain on treated surfaces for only short periods. Though rapid dissipation could be considered beneficial in some circumstances (e.g., when looking to release biological control agents shortly after treatment), in others prolonged residual activity is likely to be preferred. To this end, new research is working on the encapsulation of essential oils to maintain longer lasting exposure times. Recently, Kraus et al. [23] demonstrated that encapsulation on a chitosan-benzoic acid gel increased the half-life and antifungal properties of thyme essential oils. Similarly, Koul et al. [24] showed that zein/casein complex nanoparticles delivered a more controlled release of thymol and eugenol over 24 h. For details of further methods, and comparison between them. De Oliveira et al. [25] recently reviewed different carrier systems used for botanical compounds such as azadirachtin, rotenone, carvacrol, thymol, eugenol, and curcumin.

Issues with such an approach include maintaining sufficient efficacy of the bioactive compounds during mass production, whilst keeping particle chemically stable, though overcoming such problems may produce further benefits in addition to prolonged efficacy. Controlling the reactivity of particles through encapsulation could theoretically limit their toxicity to nontarget organisms, such as beneficial arthropods, livestock, humans, and crop plants. Though generally considered "safer" alternatives to synthetics, some essential oils may nevertheless produce cytoplasmic, cytotoxic, and phytotoxic effects [8,26], so minimizing nontarget exposure should still be considered important.

It appears that essential oils extracted from plants can produce a vast number of compounds, though not all of them being active against arthropod pests, diluting sometimes the effect of the more potent compounds. The chirality of some molecules also influences the level of bioactivity and should be investigated further to understand better how to produce the right essential oil molecules in mass.

12.5 Conclusion

Essential oils are already being used successfully against a broad range of arthropod pests and, through research and development, it is likely that more essential oil-based products will penetrate pest control markets in multiple sectors in the near future. Essential oils display many favorable characteristics as pesticides and repellents, with shortcomings that are currently being addressed through R&D. Future research should be encouraged to better clarify modes of action, molecular activity, and further means to prolong efficacy, whilst also reducing production costs. As with any new pest management product, assessing the potential for development of pest resistance to essential oils would also be beneficial, as an inability of pests to overcome essential oil chemistries should not be automatically assumed.

Overall, the future of essential oils in pest management seems relatively assured. The available evidence supports that in many sectors these, and other biopesticides (such as microbials), offer a potential alternative to conventional pest control. At worst they are a viable, and potential vital pest management solution wherein synthetic pesticides fail or are unavailable.

References

- Anderson's Report, Healthy Harvest: The impact of losing plant protection products on UK food production. NFU, UK. 16p, 2014.
- [2] L. Ellse, F. Burden, R. Wall, Pyrethroid tolerance in the chewing louse *Bovicola* (*Werneckiella*) ocellatus, Vet. Parasitol. 188 (2012) 134–139.
- [3] F. Beugnet, C. Chauve, M. Gauthey, L. Beert, Resistance of the red poultry

mite to pyrethroid in France, Vet. Rec. 140 (1997) 577–579.

- [4] M. Marangi, M.A. Cafiero, G. Capelli, A. Camarda, O. Sparagano, A. Giangaspero, Evaluation of the poultry red mite (*Dermanyssus gallinae*, Acarina: Dermanyssidae) susceptibility to some acaricides in field populations from Italy. Exper. Appl. Acarol., 48 (1–2) 2009 11–18.
- [5] L.D. Foil, P. Coleman, M. Eisler, et al. Factors that influence the prevalence of acaricide resistance and tick-borne diseases, Vet. Parasitol. 125 (2004) 163– 181.
- [6] C. Bass, A.M. Puinean, C.T. Zimmer, I. Denholm, L.M. Field, S.P. Foster, O. Gutbrod, R. Nauen, R. Slater, M.S. Williamson, The evolution of insecticide resistance in the peach potato aphid, Myzus persicae, Insect Biochem. Mol. Biol. 51 (2014) 41–51.
- [7] D.R. George, R.D. Finn, K.M. Graham, O.A.E. Sparagano, Present and future potential of plant derived products to control arthropods of veterinary and medical significance, Parasites Vectors 7 (2014) 28–39.
- [8] M.B. Isman, Plant essential oils for pest and disease management, Crop Protect. 19 (2000) 603–608.
- [9] M.B. Isman, Botanical insecticides, deterrents, and repellents in modern agriculture and an increasingly regulated world, Annu. Rev. Entomol. 51 (2006) 45–66.
- [10] W. Boonyuan, J.P. Grieco, M.J. Bangs, A. Prabaripai, S. Tantakom, T. Chareonviriyaphap, Excito-repellency of essential oils against an *Aedes aegypti* (L.) field population in Thailand, J. Vector Ecol. 39 (2014) 112–122.
- [11] L.S. Nerio, J. Olivero-Verbel, E. Stashenko, Reppellent activity of essential

oils: a review, Bioresource Technol. 101 (2010) 372–378.

- [12] Markets and Markets, AGI 2716: Biopesticides Market by Active Ingredient (Microbials & Biorationals), by Types (Bioinsecticides, Biofungicides, Bionematicides & Bioherbicides), by Application, by Formulation, by Crop Type & by Geography – Global Trends & Forecasts to 2019. Markets and Markets, Dallas, TX, USA, 2014.
- [13] F. Macchioni, S. Perrucci, F. Cecchi, P.L. Cioni, et al. Acaricidal activity of aqueous extracts of camomile flowers, *Matricaria chamomilla* against the mite *Psoroptes cuniculi*, Med. Vet. Entomol. 18 (2004) 205–207.
- [14] B.R. Mignon, B.S. Losson, Efficacy of a phyto-aromatic gel against auricular mange in rabbits and carnivores, Vet Rec. 138 (1996) 329–332.
- [15] S. Perruccci, G. Macchioni, P.L. Cioni, et al. Structure/activity relationsip of some natural monoterpenes as acaricides against Psoroptes cuniculi, J. Natural Prod. 58 (1995) 1261–1264.
- [16] L. Ellse, R. Wall, The use of essential oils in veterinary ectoparasite control: a review, Med. Vet. Entomol. 28 (2014) 233–243.
- [17] N.W. Calderone, W.T. Wilson, M. Spivak, Plant extracts used for control of the parasitic mites *Varroa jacobsoni* (Acari: Varroidae) and *Acarapis woodi* (Acari: Tarsonemidae) in colonies of *Apis mellifera* (Hymenoptera: Apidae), J. Econ. Entomol. 90 (1997) 1080–1086.
- [18] N.D. Rice, R. Whittington, H.A. Higo, Comparison of release mechanisms for botanical oils to control Varroa destructor (Acari: Varroidae) and Acarapis woodi (Acari: Tarsonemidae) in colonies of honey bees (Hymenoptera:

Apidae), J. Econ. Entomol. 95 (2002) 221–226.

- [19] D.R. George, O.A.E. Sparagano, G. Port, E. Okello, R.S. Shiel, J.H. Guy, Repellence of plant essential oils to *Dermanyssus gallinae* and toxicity to the nontarget invertebrate *Tenebrio molitor*, Vet. Parasitol. 162 (2009) 129–134.
- [20] W. Lwande, A.J. Ndakala, A. Hassanali, et al. *Gynandropis gynandra* essential oil and its constituents as tick (*Rhipicephalus appendiculatus*) repellents, Phytochem. 50 (1999) 401–405.
- [21] E. Magi, T. Jarbis, I. Miller, Effects of different plant products against pig mange mites, Acta Vet. Brunensis 75 (2006) 283–287.
- [22] R.S. Rattan, Mechanism of action of insecticidal secondary metabolites of plant origin, Crop Prot. 29 (2010) 913–920.
- [23] B. Kraus, N. Koeniger, S. Fuchs, Screening of substances for their effect on *Varroa jacobsoni* – attractiveness, repellency, toxicity and masking effects of etheral oils, J. Apic. Res. 33 (1994) 34–43.
- [24] O. Koul, S. Walia, G.S. Dhaliwal, Essential oils as green pesticides: potential

and constraints, Biopestic. Int. 4 (2008) 63–84.

- [25] J.L. De Oliveira, E.V. Ramos Campos, M. Bakshi, et al. Application of nanotechnology for the encapsulation of botanical insecticides for sustainable agriculture: prospects and promises, Biotechnol. Adv. 32 (2014) 1550–1561.
- [26] F. Bakkali, S. Averbeck, D. Averbeck, M. Idaomar, Biological effects of essential oils: a review, Food Chem. Toxicol. 46 (2008) 446–475.

Further Reading

- D.R. George, J.H. Guy, S. Arkle, D. Harrington, C. De Luna, E.J. Okello, R.S. Shiel, G. Port, O.A.E. Sparagano, Use of plant-derived products to control arthropods of veterinary importance: a review, Ann. N. Y. Acad. Sci. 1149 (2008) 23–26.
- S. Ghayempour, S.M. Mortazavi, Antibacterial activity of peppermint fragrance micro-nanocapsules prepared with a new electrospraying method, J. Essential Oil Res. 26 (2014) 492–498.

Page left intentionally blank

Α

Acetic anhydride modified soybean oil (AMSO), 169 Acid ion exchange resin methods (AIER), 91 Acrylated epoxidized soybean oil (AESO), 57-59, 115, 116, 154, 168, 169, 171, 172, 184 Aedes aegypti, 207 AESO. See Acrylated epoxidized soybean oil (AESO) AIER. See Acid ion exchange resin methods (AIER) Air laid flax composites, 169 Alcohols, 13, 22, 105, 139 aliphatic, 79 epoxidation, 22 hydroformylation, 23 ozonolysis, 23 poly-functional, 105 ring opening by, 44, 104, 105 transesterification, 23 Alcoholysis reactions, 22, 45, 79, 104, 105, 140 Ambrettolide epoxide, 136, 137 Amidation, 44 of alcohols, 23 polyol methods, 42, 44 of triglycerides, 90 Amide diols, 44, 45 Amine-to-epoxide reactions, 106 Ammonium persulfate (APS), 182 AMSO. See Acetic anhydride modified soybean oil (AMSO) APS. See Ammonium persulfate (APS) Attapulgite nanocomposites, 67 Azadiriachta indica juss, 44

В

Bacterial cellulose (BC), 171 Bending beam test, 200 Betapol, 139–141 *Betula verrucosa*, 138 Biobased epoxy resins, 113 Biochemical oxygen demand (BOD), 129, 130, 136 Biodegradable polyesters, 77. *See also* Polyesters Biofoam, 152 BiOH[®] series, 105 BOD. *See* Biochemical oxygen demand (BOD) *n*-Butyl methacrylate (BMA), 175 *t*-Butyl peroxide (TBPO), 175

С

Camelina sativa, 40 Candida albicans, 143 Candida antarctica, 28, 91, 129-131, 134-136, 141 Candida rugosa, 136, 139 Canola oil, 109, 113, 177 epoxidation kinetics of, 102, 103, 114 extraction of, 88 fatty acid methyl esters (FAME), 102, 103 Capillary supercritical fluid chromatography (SFC), 10 Carbonated plant oils and derivatives, 109, 110 Carbon nanotubes (CNTs), 31, 192 SEM images of, 32 spray pyrolysis setup, 31 Cationic polymerization catalyst (CPI), 177 CBE. See Cocoa butter equivalents (CBE) CCC. See Counter current chromatography (CCC) CF-PANI. See Polyaniline coated coconut fibers (CF-PANI) π -Cloud, 154 CNTs. See Carbon nanotubes (CNTs) Coatings, 46, 47, 56, 167 aliphatic isocyanates, use in, 38 antimicrobial, 48 cured epoxy, 115 fast-drying protective, 115 polymer, 59 polyurethane, 38 thermoplastic polyolefin, 66 waterborne dispersions, 48 water-repellent, 60 Cobalt naphthenate oxidation, 129 Cocoa butter equivalents (CBE), 139 Counter current chromatography (CCC), 10 Cox model, 173 CPI. See Cationic polymerization catalyst (CPI) Cross-linked polymers, 78, 79, 92, 113, 131

Cyclic carbonates, 61, 110 bifunctional, 66 bio-based, 68 1,4-butane diamine (BDA), 62 1,2-ethane diamine (EDA), 62 isophorone diamine (IPDA), 62 Cytochrome P450 monooxygenases, 143

D

DAQ. See Data acquisition system (DAO) Data acquisition system (DAQ), 196 DBTDL. See Dibutyltin dilaurate (DBTDL) DCPD. See Dicyclopentadiene (DCPD) Deoxyxylulose phosphate (DOX) pathway, 4 DETA. See Diethylenetriamine (DETA) Dibutyltin dilaurate (DBTDL), 194 Dicyclopentadiene (DCPD), 173 Dielectric permittivity, 194 Diels-Alder reaction, 81 Diethylenetriamine (DETA), 177 Differential scanning calorimetry (DSC), 111, 116, 132, 154 Diglycerides, 2 Diisocyanates, structure of, 39, 56, 57 Dimethylallyl pyrophosphate (DMAPP), 4 Dimethylol propionic acid (DMPA), 48, 58, 59, 194, 195 4,4-Diphenylmethane diisocyante (MDI), 181 Divinylbenzene (DVB), 174 DMA. See Dynamic mechanical analysis (DMA) DMAPP. See Dimethylallyl pyrophosphate (DMAPP) DMPA. See Dimethylol propionic acid (DMPA) DSC. See Differential scanning calorimetry DVB. See Divinylbenzene (DVB) Dynamic mechanical analysis (DMA), 115, 116, 168 test, 176

Ε

EAS. *See* Epoxidized allyl soyate (EAS) EHO. *See* Epoxidized hemp oil (EHO) Elaeis guineensis, 44 ELO. See Epoxidized linseed oil (ELO) EML. See Epoxidized methyl linseedate (EML) EMO. See Epoxidized methyl oleate (EMO) Enzymes, 11, 91, 127, 129, 134 fatty acid-hydroxylation, 144 identification of, 129 for plant oil polymerization, 128 processes, 127 EOC. See Epoxy oxygen group content (EOC) Epoxidation, 22, 28, 44, 58, 91, 101, 141 of carbon double bonds by hydroperoxide, 28 chemo-enzymatic, 28 enzyme-catalyzed, 141 of fatty acids by lipase, 28 high temperature, 102 in-situ, 102 kinetics, 103 of plant oils, 100 polyol methods, 42 of triglyceride, 91 of unsaturated vegetable oils, 61 Epoxidized allyl soyate (EAS), 176 Epoxidized α, ω -dicarboxylic acids, 132 Epoxidized canola (ECO), 110 Epoxidized hemp oil (EHO), 178 Epoxidized linseed oil (ELO), 44, 178 Epoxidized methyl linseedate (EML), 79 Epoxidized methyl oleate (EMO), 91, 111 Epoxidized oils, 57, 62, 114 anhydride cured, 114 hydrogenation of, 108 Epoxidized soybean oil (ESO), 44, 57, 111, 114, 159, 169, 178 in-situ reactive blending of, 117 polymerization of, 116 reaction of, 30 Epoxy oxygen group content (EOC), 142 Epoxy resins, 100 biobased, 110, 113 thermoset, 100, 114 Epoxy ring opening, 101, 102 by alcohols, 104, 105 hydroxylation reaction, 104 by anhydrides, 110, 111 by halogen reagents, 108 by lactic acid, 108 polymerization of, 111 by water, 106, 107 ESO. See Epoxidized soybean oil (ESO) Ethylene glycol, 105

F

FAME. See Fatty acid methyl ester (FAME) Fatty acid methyl ester (FAME), 74 Fatty acids, 21 based amide polyols, 45 biosynthesis of, 6 derived from acetyl CoA, 7 derived from malonyl-Coenzyme A (malonyl CoA), 7 epoxidized, 41 hydroxylated, 41, 135, 143 lipase-catalyzed polycondensation of, 135 post-extraction processing of, 13 structure of, 75 transesterification of, 27 unsaturated, 141 vegetable oils, composition in, 75, 76 Fatty alcohols, 22, 23. See also Alcohols Fatty amides, 21, 22 Fatty amines, 21, 22 Fatty nitriles, 21, 22 FE-SEM. See Field-emission scanning electron microscope (FE-SEM) FFA. See Free fatty acids (FFA) FGS. See Functionalized graphene sheet (FGS) Fibers, 168 biobased, 168 biorenewable, 168 carbon, 168 glass, 168 pine wood, 184 reinforcements, 168 synthetic, 168 Field-emission scanning electron microscope (FE-SEM), 197 Flame retardant polyols, 45 Flax fibers, 169 architectural effects of, 170 arrangement modes of, 170 types of, 169 Flexural testing, 173, 179 Foams, 46, 57, 93, 152, 154, 184 flexible, 47 polyester, 82 rigid, 47 Fourier transform infrared spectroscope (FTIR), 152 Free fatty acids (FFA), 139 FTIR. See Fourier transform infrared spectroscope (FTIR) Functionalized graphene sheet (FGS),

G

Gauge factor, 193, 196, 202, 204 Gel permeation chromatography, 135

155

Gluconacetobacter xylinus, 171 Glycerides, 3, 6, 7 Glycerols, 62 alkylated, 5 application of, 24 difunctional cyclic carbonate, synthesis of, 64 hydroxyurethanes, synthesis of, 63 phenoxycarbonyloxymethyl ethylene carbonate, synthesis of, 64 transformation of, 24 Grains, production of, 3 Graphene oxide composites, 155 surface modification of, 156 modified pressurized oxidation method, 156 pressurized oxidation method, 156 raman spectrum of, 157 Grubbs catalysts, 132

Η

Halpin-Tsai model, 173 Helianthus annuus L., 40 Hexamethylene diisocyanate (HMDI), 180 HMDI. See Hexamethylene diisocyanate (HMDI) Hooke's law, 196 HPAn. See Hyxahydrophthalic anhydride (HPAn) Hydratases, 143, 144 Hydroformylation, 23, 42 Hydrohalogenation methods, 108 of epoxidized oils, 108 Hydroxyl estolide, 112 Hydroxyurethane, 61. See also Polyhydroxyurethanes (PHUs) Hyxahydrophthalic anhydride (HPAn), 177

IFSS. See Interfacial shear strength (IFSS) ILSS. See Interlaminar shear strength (ILSS) Interfacial shear strength (IFSS), 172 Interlaminar shear strength (ILSS), 179 IPDI. See Isophorone diisocyanate (IPDI) IPP. See Isopentyl diphosphate (IPP) Isocyanates, 38 aromatic, 38 hexamethylene diisocyanate (HDI), 38 isophorone diisocyanate (IPDI), 38 methylene diphenyl diisocyanate (MDI), 38 pentamethylene diisocyanate (PDI), 38 toluene diisocyanate (TDI), 38

Isopentyl diphosphate (IPP), 4, 6 mavelonate (MVA) pathway, 4 methylerythritol (MEP) pathway, 4 Isophorone diisocyanate (IPDI), 38, 58, 194 Isoprenoid-derived oils, 1, 3

J

Jatropha curcas L., 41 Jute composites, 172

K

Keratin fiber (KF), 171 KF. *See* Keratin fiber (KF)

L

Layered graphite, 155 Lesquerolic acid, 40 Lewis acid catalysts, 81 Lignin, 151, 152, 182 composites, 152 different concentrations of, 152 and fatty acid polyol, 154 phenyl propane monomers in, 152, 153 *Limnanthes alba*, 141 Lipases, 127, 129, 135, 136, 140, 141 Lipid acrylation, 177 Liquid chromatography (LC)–mass spectrometry (MS), 134 Lyocell fiber, 170, 177

Μ

MACO. See Maleated castor oil (MACO) Macroreticular ion exchange resin, 105 MAECO. See Maleinized acrylated epoxidized canola oil (MAECO) MAESO. See Maleinized acrylated epoxidized soybean oils (MAESO) Maleated castor oil (MACO), 79, 80, 172 Maleinized acrylated epoxidized canola oil (MAECO), 115 Maleinized acrylated epoxidized soybean oils (MAESO), 115 Maleinized soybean oil (MSO), 116, 169 Mavelonate (MVA) pathway, 4 MCS. See Norbornenylehyldimethylchlorosilane (MCS) MEK. See Methyl ethyl ketone (MEK) MERGINOL® series, 105 Mesua ferrea L., 41 Methacrylated lauric acid (MLAU), 171 Methacrylated soybean oil (MSO), 169 Methacrylic anhydride modified soybean oil (MMSO), 169

Methoxylated soybean oil polyols (MSOLs), 58 Methylerythritol (MEP) pathway, 4. See also Deoxyxylulose phosphate (DOX) pathway Methyl ethyl ketone (MEK), 194 4-Methylpyridine, 116 MLAU. See Methacrylated lauric acid (MLAU) MMSO. See Methacrylic anhydride modified soybean oil (MMSO) Monoglycerides, 2, 25, 90, 139 Montmorillonite, 160 MSO. See Maleinized soybean oil (MSO); Methacrylated soybean oil (MSO) MSOLs. See Methoxylated soybean oil polyols (MSOLs) Mucor miehei, 129 Myzus persicae, 207

Ν

Nanocomposites, 150, 194 durability, 194 melt processing of, 150 Nanofiller, 149, 151, 197, 198 dispersion of, 197 NIPU. *See* Nonisocyanate polyurethanes (NIPU) Nonisocyanate polyurethanes (NIPU), 59, 110 Norbornenylethyldimethylchlorosilane (MCS), 173 Norbornenylethyltrichlorosilane (TCS), 173 Novozym 435, 132

0

Octopamine receptors, 208 Oil-producing crops, production, 3 Oleic acid, 7, 113, 131, 144, 154 based polyesters, 131 epoxidation of, 131 lipase-catalyzed epoxidation of, 142 OOC. *See* Oxirane oxygen content (OOC) Organosiliconic polymers, 66 Oxirane oxygen content (OOC), 102 Oxiranes, ring opening of, 44 Ozonolysis, 23, 42, 44, 100

Ρ

Palm oil, 2, 12, 20, 31, 40, 47, 115, 140 extraction of, 12 fractionation of, 12 refining of, 12 SEM images of CNT, 32 *Passiflora edulis* Sims *f. flavicarpa* Degener, 41 PCL. See Polycaprolactone (PCL) PEFA. See Poly(ether fatty amide) (PEFA) Performic acid formation of, 101 ring-opening by, 28, 101 Perhydrolysis, 29 Pest control, 207, 209 conventional, 209 Pest resistance, 207-209 Petroleum-based polymers, 55, 83 Petroleum resources, 167 PFF. See Pyrolyzed chicken feather fibers (PFF) PHAs. See Polyhydroxyalkanoates (PHAs) Phospholipids, 2, 12, 89 Phosphorylated polyols, 45 Photopolymerization, 115, 117, 178 PHUs. See Polyhydroxyurethanes (PHUs) Plant biosynthesis, 1 Plant oil, 1, 8, 127, 167, 193 acid value, 1 alkene moieties, presence of, 8 based carbon nanotube composites, 154 based clay and silica nanocomposites, 159 based epoxy composites, fiber-reinforced, 176 based foam composites, fiber-reinforced, 184, 185 based graphene oxide composites, 155 based polyurethane composites, fiber-reinforced, 179 based vinyl polymer composites, fiber-reinforced, 168 biochemistry of, 3 chemical characterization, 10 chemical reactivity of, 6-9 chemistry of, 3 conversion into polyols, 58 epoxidation of, 29, 100 acrylated epoxidized, 109 acrylated resins of, 114 derivatives of, 103 direct polymerization of, 116 extraction and processing of, 11 fatty acids, 1, 40, 41 glycerides, 3 hydroxyl value, 1 iodine values of, 41 isoprenoid derived hydrocarbons, 1 large-scale extraction and processing, 12 major sources, 11 modified polyesters, 73 advantages of, 73 applications of, 73

Plant oil (cont.) fatty acid process, 75 industrial applications, use in, 74 monoglyceride process, 75 olefin/iodine value, 1 perhydrolysis of, 29 peroxide value, 1 small scale extraction and processing, 11-12 terpenoids, 3 transesterification reaction of, 74 transformations, 138 enzyme assisted, 138 Plant oil-based derivatives, 21 fatty acids, 21 fatty amines, 21 Plant oil-based filler composites, 152 cellulosic fibers, 152 lignin fibers, 152 Plant oil-based polymer nanocomposites, 149 anisotropic properties of, 150 development and tailoring of, 150 structure-property relationship of, 150 Plant oil-based polymers, 74, 87 advantages of, 87 applications of, 82 molecular weight, 88 oil extraction methods, 89 modification of monomers, 90 from natural source, 89 polymerization from oil source, 90 polyamides, 74 polyesters, 74 polyethers, 74 polyolefins, 74 properties of, 91 Plasticizers, 29, 30 Poisson's ratio, 196 Polyaniline coated coconut fibers (CF-PANI), 182 Polycaprolactone (PCL), 181 Polyesters, 74, 129 biodegradable, 129 cross-linkable, 129 epoxide containing, 130 functional, 129 lipase-catalyzed synthesis, 129 linear, 76 oleic acid containing, 131 plant oil-based, 129 biodegradable, 129 cross-link, 129 enzyme-assisted synthesis, 129 functional, 129 polycondensation reaction, 74 Polyetheramides, 45 Poly(ether fatty amide) (PEFA), 92, 93 Polyethers, 88 applications of, 93 polyols, 93

Polyhedral oligomeric silsesquioxane (POSS) compounds, 92 Polyhydroxyalkanoates (PHAs), 77 Polyhydroxyurethanes (PHUs), 56, 59,68 alternative systems, 67 bio-based, 61 and composites, 66 4,9-dioxa-1,12-dodecanediamine (DODDA), 60 petroleum-based, 60 polysiloxane hybrid material synthesis, 67 synthesis from limonene, 66 p-xylylenediamine (p-XDA), 60 Polymeric 4,4 -disphenylmethane dissocynate (pMDI), 179 Polymerization, 127 chemical, 127 enzyme catalyzed, 127 Polymers, 127 biomaterials, 81 industry, 192 science, 73 synthesis, 127, 167 Polyolamines, 107 Polyols, 42 bio-based, 57 structure of, 57 Poly(sebacic acid-co-ricinoleic acid), 78 Polyurethane dispersions (PUDs), 193 Polyurethanes (PU), 38, 45, 68, 151 alternative routes for making, 45 applications of, 46 from bio-based polyols, 57 chemical structure of, 56 foams, 46 microphase separation of, 151 nanocomposites, 158 fracture surface of, 158 synthesis of, 158 TGA measurement for, 159 thermal stability of, 158 petroleum-based, 56 plant oil-based, 38 urethane reaction of, 38 POSS. See Polyhedral oligomeric silsesquioxane (POSS) compounds Poylisoprenoids, 4 Pressure-sensitive adhesives (PSA), 111, 114 Prileschajew reaction, 28 1,2-Propanediol, 105 1,3-Propanediol, 105 Prosopis juliflora, 41 PSA. See Pressure-sensitive adhesives (PSA) Pseudomonas cepacia, 129 PU. See Polyurethanes (PU)

PUDs. *See* Polyurethane dispersions (PUDs) Pyrolyzed chicken feather fibers (PFF), 171

R

RA. See Ricinoleaic acid Raman spectrum, 156, 157 Response surface methodology (RSM), 142 Rhizopus delemar, 139, 140 Rhizopus oryzae, 177 Ricinoleaic acid (RA), 7, 19, 40, 42, 58, 81, 135, 136, 141 Ring-opening metathesis polymerization (ROMP), 173 ROMP. See Ring-opening metathesis polymerization (ROMP) RSM. See Response surface methodology RSO. See Rubber seed oil (RSO) Rubber seed oil (RSO), 182

S

Sapindus mukorossi seed oil (SMSO), 142 Scanning electron microscope (SEM), 152, 197 SEM. See Scanning electron microscope (SEM) Sensing principle, 193, 196, 197 Sensor fabrication, 193, 194, 196 SEPEP. See Soybean phosphate ester polyol (SEPEP) SFC. See Capillary supercritical fluid chromatography (SFC) Shape-memory polymers, 48 SHM. See Structural health monitoring (SHM) SMSO. See Sapindus mukorossi seed oil Solid freeform fabrication (SFF), 177 Sonication, 150, 158, 194 SOPERMA. See Soybean oil pentaerythritol glyceride maleates (SOPERMA) SOVERMOL® series, 105 Soybean oil, 13, 20, 28, 31, 40, 61, 82, 91, 152, 167, 169, 171, 174, 176, 178 based cationic polyurethane coatings, 48 based polyols, 47, 48 based polyurethanes, 58, 92 carbonated, 109 cationic polymerization of, 82 chemical structure of, 58 with diamines, 46 epoxidized, 29, 30, 45 lactic acid-epoxidized, 108 methacrylated, 169 methoxylated, 92

INDEX

structures of maleates alcoholyzed triglycerides, 58, 80 Soybean oil pentaerythritol glyceride maleates (SOPERMA), 171 Soybean phosphate ester polyol (SEPEP), 179 *Stenotrophomonas nitritireducens*, 144 Structural health monitoring (SHM), 191, 192 Sugar fatty acid esters, 138 Sulfuric acid, 104, 105, 136 Synthetic pesticides, 207–209

Τ

TBAB. See Tetrabutylammonium bromide (TBAB) TCS. See Norbornenylethyltrichlorosilane (TCS) TDI. See Toluene-2,4-diisocyanate (TDI) TEA. See Triethylamine (TEA) Tensile test laboratory setup, 200 Terpene-based acid anhydride (TPAn), 177 Terpenes, 5, 65 dimethylallyl pyrophosphate (DMAPP), 4 isopentyl diphosphate (IPP), 4 isoprene-derived, 65 use of, 65 Terpenoids, 5 biosynthesis of, 3 Tertiary amines, 21, 107, 194 TETA. See Triehylenetetraamine (TETA) Tetrabutylammonium bromide (TBAB), 62 Tetrafluoroboric acid, 105 TGA. See Thermogravimetric analysis (TGA) Thermal mechanical analysis (TMA), 116

Thermogravimetric analysis (TGA), 131, 154, 158, 159, 172, 197, 199 Thermoplastic polyesters, 76. See also Polyesters aliphatic groups, 76 aromatic groups, 76 Thermoplastic polyurethane (TPU), 155 Thymol, 5, 208 TMA. See Thermal mechanical analysis (TMA) Toluene-2,4-diisocyanate (TDI), 180 TPAn. See Terpene-based acid anhydride (TPAn) TPU. See Thermoplastic polyurethane (TPU) Transesterification, 23, 41, 48, 64, 74, 91, 105, 139, 176 fatty acid esters of, 27 lipase-catalyzed, 139 triglycerides of, 26 vegetable oils of, 28 of waste oils rapeseed waste frying oil, 27 Triehylenetetraamine (TETA), 177 Triethylamine (TEA), 194 Triglycerides, 2, 61, 103 epoxidization of, 90 epoxidized methyl oleate, 91 structure of, 40 transesterification of, 26 Tripalmitin, 140, 141 Tung oil, 19, 30, 81, 82, 172

U

Unsaturated α,ω-dicarboxylic acids, 132 lipase-catalyzed polycondensation of, 132 Unsaturated polyester resin (UPR), 78

V

Vacuum-assisted resin transfer molding technique, 171, 172 Van der Waals bonding, 150 Varroa mites, 208 Vegetable oils, 11, 19, 55 based polyols, 43 carbon nanotubes (CNTs), 31 conjugates, 30 degree of unsaturation, 20 epoxidation of, 28 epoxy derivatives, 28, 29 ester derivatives, 25 fatty acids composition in, 20 hydrolysis of, 21 sources of, 40 transesterification of, 28 viscosity of, 26 world production of, 20 Vernolic acid, 7, 40, 104 Vernonia galamensis, 104 Vernonia oil, 19, 40, 90, 104, 141 4-Vinylpyridine, 116 VOCs. See Volatile organic compounds (VOCs) Volatile organic compounds (VOCs), 47, 48, 58, 116

W

Waterborne polyurethane (WPU), 155 Waterborne, polyurethane dispersions (PUDs), 48, 58, 193–195 Wheat straw (WS), 175 WPU. *See* Waterborne polyurethane (WPU)

Υ

Young's modulus, 117, 173

Page left intentionally blank

A Volume in the Plastics Design Library Series Bio-Based Plant Oil Polymers and Composites

Edited by Samy A. Madbouly, Chaoqun Zhang, Michael R. Kessler

The state of the art in the processing, characterization, and application of plant oil-based polymeric and composite materials, enabling engineers and scientists to meet growing demands for sustainable materials and products.

Plant oil is one of the most attractive options as a substitute for nonrenewable resources in polymers and composites, and is producing materials with very promising thermomechanical properties compared to traditional, petroleum-based polymers. *Plant Oil-Based Polymers* and *Composites* helps engineers and materials scientists to take advantage of the latest research conducted in this rapidly advancing field – enabling them to develop and commercialize their own products quickly and more successfully.

In addition to critical processing and characterization information, the book assists engineers in deciding whether or not they should use a plant oil-based polymer over a petroleum-based polymer, discussing sustainability concerns, biodegradability, associated costs, and recommended applications.

The book details the advancements in the development of polymeric materials and composites from plant oils, as well as providing a critical review of current applications in various fields, including packaging, biomedical, and automotive applications. This also includes the latest progress in developing multifunctional bio-based polymers – such as by increasing thermal conductivity or adding antibacterial properties.

ABOUT THE EDITORS

Samy A. Madbouly received his PhD degree from the Department of Organic and Polymeric Materials, Tokyo Institute of Technology, Japan. He has served as a professor at Cairo University, Faculty of Science, Chemistry Department, and at the Department of Materials Science and Engineering, Iowa State University. Dr Madbouly has a broad expertise in biopolymers, polymer blends, nanocomposites, shape-memory polymers, processing and phase behavior of polymeric materials.

Chaoqun Zhang received his PhD degree in Materials Science and Engineering in 2014 from Iowa State University, USA. Following his PhD, he served as a postdoctoral research associate working on developing biopolymers and nanocomposites from natural resources at the Center for Sustainable Polymers at the University of Minnesota from 2014 to 2015. Now, he is working at Texas A&M University on bioinspired materials and antiflammable nanocoatings via layer-by-layer assembly.

Michael R. Kessler received his PhD degree from the University of Illinois at Urbana-Champaign in 2002. He is currently Professor and Director of the School of Mechanical and Materials Engineering at Washington State University, where he is also the site director the Center for Bioplastics and Biocomposites, a National Science Foundation Industry/University Cooperative Research Center that focuses on developing high-value bio-based products from agricultural and forestry feedstocks.

RELATED TITLES

Biopolymers: Processing and Products, Niaounakis, 2014, 9780323266987 Biopolymers: Applications and Trends, Niaounakis, 2015, 9780323353991 Vegetable Oil-Based Polymers, Karak, 2012, 9780857097101



